Strategies to improve nitrogen use efficiency and yields in UK forage crops

Nicola Mary Capstaff



University of East Anglia

A thesis submitted for the degree of Doctor of Philosophy September 2019

This copy of the thesis has been supplied on condition that anyone who consults it is understood to recognise that its copyright rests with the author and that use of any information derived there from must be in accordance with current UK Copyright law. In addition, any quotation or extract must include full attribution. This page is intentionally left blank blank pages are used throughout to allow for formatting in the printed version of this document

ABSTRACT

Grasses and lucerne are the main crops of choice for dried forage crop growers in the UK, who use vegetative forage tissue to formulate bulk animal feeds. To maximise yield, the forage crop industry uses substantial fertiliser applications, however not efficiently, which is directly related to limited research into forages. Efficiency must be increased throughout the industry; therefore this thesis aims to investigate several strategies to increase yield and efficiency of UK forage crop production.

Firstly, a range of nitrogen status marker genes for use in forage crops were validated in the laboratory. Testing of field samples showed a link between soil conditions and future yields. This provides knowledge platforms that can be used by growers to ensure adequate, but not excessive, fertiliser use.

Soil nitrate availability to forage crops was tested with a new method using soil columns in conjunction with nitrate-selective sensors. This method provided extensive data of the soil nitrate profile in columns, and showed to be superior to techniques in current published literature. The data collected with the use of soil columns and nitrate-selective sensors was then used to investigate management practices, including how defoliation and intercropping can affect soil profiles.

Next, the potential of the biostimulant fulvic acid to increase vegetative yield in forage crops was investigated. It was found that the treatment with fulvic acid resulted in vegetative yield increases in numerous lucerne cultivars, across a range of growth conditions. Furthermore, it was observed that nodulation and microbial growth was also affected by fulvic acid treatment. To assess the mode of the action that determines biomass increases transcriptome analysis was undertaken, which suggested fulvic acid may be a viable biostimulant for lucerne, providing yield increases without significantly higher inputs.

Taken together, this research provides a great resource of information to aid growers in producing high quality forage more efficiently.

iii

"I know I've got a degree. Why does that mean I have to spend my life with intellectuals? I've got a life-saving certificate but I don't spend my evenings diving for a rubber brick with my pyjamas on." - Victoria Wood Oxford Dictionary of Quotations by Subject (ISBN 0199567069)

TABLE OF CONTENTS

Abstract	iii
Table of contents	. v
Acknowledgementsx	iii
List of figures	×٧
List of tables	хx
List of common abbreviationsxx	ĸii
Chapter 1: Introduction	. 1
1.1 The importance of forage crops	. 2
1.1.1 Temperate crops and cultivars	. 2
1.1.2 Relevance and cultivation in the UK	.5
1.1.3 Nutritional content and yield	.9
Digestibility	.9
Carbohydrates	.9
Protein1	10
Lipids1	10
Trace Elements1	1
Biomass1	1
Current research in forage crop production1	12
1.2 Forage drying in the UK – British Association of Green Crop Driers (BAGCD)1	14
1.2.1 Methodology conception1	14
1.2.2 Present production of dried forage1	٤4
1.2.3 The main difficulties associated with the forage crop industry is nitrogen use efficiency1	16
1.2.4 Grassland research2	21
1.3 Strategies to improve nitrogen use efficiency in forage crops	22
1.3.1 Project aims2	22
1.3.2 Thesis overview, contents, and hypotheses2	22
Chapter 2: Materials and methods2	22
Chapter 3: Marker genes in Festulolium can be used to assess crop and soil N status, t aid with grower decisions to improve future yield2	to 23
Chapter 4: Soil sensors can detect N profile changes under forage crops with different management regimes	t 23
Chapter 5: Fulvic acid increases vegetative growth in the forage legume <i>Medicago sativa</i> , and is associated with influencing microbial activity	24
Chapter 6: Transcriptome analysis shows preferential enrichment of nodulation regulation and signalling-related genes in <i>Medicago sativa</i> following fulvic acid application	24
•••	

Chapter 7: General discussion	. 25
1.3.3 Contributions to thesis	. 25
Chapter 2: Materials and methods	. 27
To Note	. 28
2.1 N status marker genes	. 29
2.1.1 Arabidopsis growth conditions	. 29
2.1.2 RNA extraction	. 29
2.1.3 cDNA synthesis	. 30
2.1.4 Candidate gene identification and primer design	. 30
2.1.5 Quantitative RT-PCR	.31
2.1.6 Candidate gene primer design for forages	. 34
2.1.7 Forage crop growth conditions	. 37
2.1.8 Candidate gene testing for forages with RT-PCR	. 38
2.1.9 Gel electrophoresis and sequencing	. 38
2.1.10 Primer efficiency testing	.41
2.1.11 Larger forage screen of developed N status marker primers	.44
2.1.12 Developed N status marker primers forage crop qRT-PCR	.44
2.1.13 2018 field sampling of developed N status marker primers	.45
2.1.14 Soil extraction with KCl	.48
2.1.15 Spectrophotometric determination of nitrate in soil extractions	.48
2.1.16 Protein analysis using Kjeldahl method	.49
2.1.17 Chlorophyll analysis using Soxhlet extraction	. 50
2.1.18 Further qPCR of developed N status marker primers in Festulolium	.51
2.1.19 2019 field sampling of developed N status marker primers	.51
2.1.20 Modelling N status marker gene expression	. 53
2.2 Nitrate availability monitoring using soil sensors	. 55
2.2.1 Nitrate-selective sensor construction	. 55
2.2.2 Nitrate ion-selective sensor calibration	.56
2.2.3 Sensor running and data analysis	. 57
2.2.4 Column experiments of <i>Lolium perenne</i> monocropping	. 58
2.2.5 Column experiments of Lolium perenne and Medicago sativa intercropping	. 59
2.3 Fulvic acid testing of forage crops	.61
2.3.1 Fulvic acid materials	.61
2.3.2 Initial test of FAs on forage crop vegetative biomass	.61
2.3.3 Soil pH measurement	. 62
2.3.4 FA Chemical analysis standard solutions	. 62
2.3.5 FA Inductively Coupled Plasma Spectrometry	.63
2.3.6 FA Thin Layer Chromatography	.65

2.3.7 FA Liquid Chromatography Mass Spectrometry	. 65
2.3.8 FA Gas Chromatography Mass Spectrometry	.66
2.3.9 FA control solutions	. 68
2.3.10 FA Nuclear Magnetic Resonance	. 70
2.3.11 Glasshouse FA vegetative growth and nodulation assays	. 70
2.3.12 Plate FA vegetative growth and nodulation assays	.72
2.3.13 Sinorhizobium colony forming unit counts	.72
2.3.14 Soil Microbial Phospholipid Fatty Acids Extraction	.73
2.3.15 Analysis of FAME profiles	. 75
2.3.16 Soil microbial DNA extraction	.76
2.3.17 Soil DNA qPCR	.77
2.3.18 Field trials with VFA	. 79
2.4 Fulvic acid RNA-sequencing of Medicago sativa	. 80
2.4.1 RNAseq of VFA and VC experimental design	. 80
2.4.2 RNAseq plant material and RNA extraction	. 82
2.4.3 RNA clean-up and quality control	. 82
2.4.4 RNA Sequencing	. 83
2.4.5 Read alignments	. 83
2.4.6 Differential gene expression analysis	. 84
2.4.7 Functional annotation	. 85
2.4.8 GO term enrichment testing	. 85
2.4.9 Validation of RNAseq by qRT-PCR	.86
Chapter 3: Marker genes in <i>Festulolium</i> can be used to assess crop and soil N status, to ai with grower decisions to improve future yield	id . 89
3.1 Introduction	.90
3.1.1 Soil N status affects plant NUE and yield	. 90
3.1.2 There is a wide variety of genes associated with NUE	.91
Nitrate transporters (NRTs)	.92
Nitrogen assimilation (GS and NADH-GOGAT)	.93
Nitrogen reduction (NR and NIR)	.94
Protein storage (RBCS and VSPs)	.95
Vacuole protein pathway (<i>TIPs</i>)	.97
3.1.3 Plants marker genes for a variety of parameters	.98
3.1.4 Current testing for soil N lacks spatial and temporal detail	.99
3.1.5 Aims of this chapter: Testing N status marker genes in forage crops1	101
3.1.6 Materials and Methods1	101
3.2 Results1	102

3.2.1 Nitrogen status candidate genes have different expression patterns in <i>A. thaliana</i> in relation to NO ₃ ⁻ supply
3.2.2 Model species sequence alignments can be used to investigate forage crops 104
3.2.3 N status genes also showed different gene expression patterns in grasses in relation to NO_3^- supply107
3.2.4 Glasshouse testing of N status marker genes in vegetative tissue could be used to assess field samples using a Gaussian model approach
3.2.5 N status marker genes can be comparable to current conventional testing, providing information of both crop and soil status119
3.3 Discussion
3.3.1 Testing N status candidate genes provides more information of how NO ₃ - environment can affect NUE associated gene expression
3.3.2 Fields are variable for both soil N and crop yield123
3.3.3 Refining N status marker gene suite with model provides promising developments for grass research
Chapter 4: Soil sensors can detect N profile changes under forage crops with different management regimes
4.1 Introduction
4.1.1 Nitrogen fertiliser production for crops is demanding
4.1.2 Studies have estimated N changes in grassland and forage cropland
4.1.3 Current testing methods of soil N are complicated to use
4.1.4 Ion-selective nitrate sensors may be deployed in trials to assess soil concentrations
4.1.5 Aims of this chapter: Measuring soil NO ₃ ⁻ changes with sensors in forage crops in response to fertiliser application, defoliation, and intercropping
4.1.6 Materials and Methods141
4.2 Results
4.2.1 Nitrate soil profiles of Lolium perenne columns vary with management practices 142
4.2.1 Soil sensors detect a nitrate 'burst' in the mid column range following defoliation of Lolium perenne
4.2.3 Nitrate soil profiles were assessed when <i>Lolium perenne</i> columns were intercropped with <i>Medicago sativa</i> 149
4.2.4 No nitrate 'burst' was detected following defoliation of <i>Lolium perenne</i> when intercropped with <i>Medicago sativa</i> 153
4.3 Discussion
4.3.1 Soil columns can be used to produce standardised measurements of nitrate profiles
4.3.2 Soil sensors match conventional measurements, but with higher definition and lower interference
4.3.3 The detected nitrate 'burst' following defoliation of <i>Lolium perenne</i> , but not when cultivated with <i>Medicago sativa</i> 160

Chapter 5: Fulvic acid increases vegetative growth in the forage legume <i>Medicago sativa</i> , and is associated with influencing microbial activity	3
5.1 Introduction	4
5.1.1 Using biostimulants to aid crop yields16	4
5.1.2 Humic substances are potential biostimulants, and include fulvic acid	5
5.1.3 Studies have investigated humic and fulvic acids effects on plant growth	6
5.1.4 Limited evidence for the mode of action of humic substances	7
5.1.5 Aims of this chapter: Assessing the potential biostimulant fulvic acid in forage crop 16	s 8
5.1.6 Materials and Methods16	8
5.2 Results	9
5.2.1 A screen of two fulvic acid applications found a vegetative yield increase in forage legumes, but not in grasses	9
5.2.2 Analysis of fulvic acids found varying chemical composition17	3
5.2.3 Fulvic acid increase growth of <i>Medicago sativa</i> and is specific to its composition and not only a nutritional effect	7
5.2.4 Fulvic acid application caused an increased number of pink nodules in <i>Medicago</i> sativa	0
5.2.5 Vegetative growth phenotypes with Fulvic acid application is dependent on Sinorhizobium presence	3
5.2.5 Sinorhizobium growth is affected by fulvic acids18	6
5.2.6 Soil microbial population structures are affected by fulvic acid18	8
5.2.7 Quantification of DNA from soil substrate confirms fulvic acid increases Sinorhizobium numbers	1
5.2.8 Fulvic acid vegetative growth effect on <i>Medicago sativa</i> was recorded in in independent field trials, with no change to nutritional content	3
5.3 Discussion	6
5.3.1 <i>Medicago sativa</i> vegetative yield can be increased by adding fulvic acid into crop management practices19	6
5.3.2 The vegetative growth increase is associated with microbial changes including <i>Sinorhizobium</i> populations	6
5.3.3 Fulvic acids are complex chemicals and require more work to define their mode of action, including field applications19	8
Chapter 6: Transcriptome analysis shows preferential enrichment of nodulation regulation and signalling-related genes in <i>Medicago sativa</i> following fulvic acid application	1
6.1 Introduction	2
6.1.1 Using RNA-seq to identify transcriptional changes in plants	2
6.1.2 The effect of Fulvic acid on <i>Medicago sativa</i> transcriptome in both shoot and root tissue	3
6.1.3 <i>Rhizobium</i> /legume nodulation initiation	4

6.1.4 Aims of this chapter: RNA-seq analysis in <i>Medicago sativa</i> to investigate c transcription following fulvic acid application	hanges in 206
6.1.5 Materials and Methods	
6.2 Results	
6.2.1 VFA induces differential expression of transcripts in <i>Medicago sativa</i> after treatment	⁻ 3 days of 207
6.2.2 Significant transcriptional changes occured in roots following VFA treatme	ent 212
6.2.3 Differentially expressed transcripts have many homologues related to leg model species, particularly <i>Medicago truncatula</i>	ume 214
6.2.4 Biological processes associated with VFA treatment include oxidation-red gene regulation, metabolism, transport and defence	uction <i>,</i> 216
6.2.5 Molecular functions associated with VFA treatment include catalytic activ binding	ity and 218
6.2.6 Metabolism was significantly enriched by VFA treatment, particularly N m	etabolism 220
6.2.7 Enriched DE transcripts in VFA roots are associated with well-known nodu regulatory and signalling genes	ılation 223
6.2.8 qRT-PCR confirmed a subset of DE transcripts which are associated with n	odulation 227
6.3 Discussion	229
6.3.1 De novo transcriptome assembly of Medicago sativa was successful	229
6.3.2 Fulvic acid treatment causes increased transcription for metabolic process upregulated transporter activity predominantly in the root tissue	ses with 229
6.3.3 Fulvic acid treatment causes enriched upregulation of nodulation regulaton nodule-associated signalling-related genes	ory and 230
Chapter 7: General Discussion	233
7.1 Summary of results with implications	234
7.1.1 N status marker genes have use across forage crop industry	236
7.1.2 Studying the field soil N availability provides more detail for growers to puper precision agriculture techniques	ırsue 238
7.1.3 New management practices including fulvic acid application could be adva in forage crops	antageous 240
7.2 Open Questions	242
7.2.1 Improving tools for tackling nitrogen use efficiency need to be accessible growers	to 242
7.2.2 Management practices require thorough testing to demonstrate their efference as grassland research is not standardised	ectiveness 243
7.2.3 Fertiliser use must be curtailed ahead of resource depletion	244
7.3 Concluding remarks	246
Appendix A: Chapter 3 supplemental figures	248
Appendix B: Chapter 4 supplemental figures	

272
282
of 288
289
292
301
305
305
310
314
318
322

This page is intentionally left blank

ACKNOWLEDGEMENTS

Throughout writing this thesis I've looked back on how I have reached this point. My family have been an endless source of encouragement. Mam, Dad, and Lindsey - thank you for reassuring me it was worth it. Niamh Mary, you have been an infinite motivation; I hope I support you in what you do in the future as well as everyone has for me.

I owe much to my undergraduate mentors. Alison Baker had immeasurable patience in teaching me about opportunities I didn't even know existed - I realise now so much of what I have achieved began in her tutorials. My research supervisors, Yoselin Benetiz-Alfonso, Marc Knight, Julian Hibberd, and Stefan Kepinski, each provided me with knowledge across a breadth of science, and their laboratory members taught me considerably more. The Gatsby Charitable Foundation has provided amazing training from Undergrad to present, and I thank all the advisors for their time and advice over the last six years.

Undertaking a PhD at John Innes Centre has provided me with skills and opportunities alongside many inspiring colleagues. Thank you to my supervisor, Tony Miller, for your never-ending support both scientifically and pastorally. Thanks too to Claire Domoney for supervisory assistance, along with Steph Bornemann for guidance.

This project has been funded with gratitude by BBSRC as an iCASE studentship with the British Association of Green Crop Driers as collaborators. Thanks to all the BAGCD members, particularly the farms where trials were conducted. Elizabeth Harding provided support and expert information which helped me to really understand the industry and applicability of my research.

The Sanders/Miller Laboratory has had many iterations in my time, and I thank each member for their support; Tom, Millie and Marco have been particularly supportive throughout my time, thank you for the solidarity and the laughs. Supervising the amazing Freddie Morrison taught me so much, thanks for your hard work.

I have been lucky to work on many projects with such great colleagues across NBI. My heart is full when I think of the achievements I have been part of, and to the friends I have made along the way. Countless thanks to the love, support and care from the inhabitants and associated strays of Bury Street and Park Lane. I am excited for what the future has in store for all of us.

And to Adam, wow. Thank you for the unconditional love and support, including the hugs I never asked for and the alcoholic refreshments I insisted on. You have given me so much, I am truly grateful.

LIST OF FIGURES

Figure 1.1: Fodder crop production in the top forage producing European countries4
Figure 1.2: Temporary grassland area in the UK7
Figure 1.3: Utilised agricultural area (UAA) of UK crop types in 2018
Figure 1.4: Targets for improved forage crops
Figure 1.5: N fertiliser efficiency of all crop types in the highest forage producing European countries
Figure 1.6: Nitrate vulnerable zones (NVZs) in England
Figure 2.1: Fields sampled at Blankney Estates Ltd. in May 2018.
Figure 2.2: Fields sampled at Blankney Estates Ltd. in April 2019
Figure 2.3: RNAseq experimental design for comparing transcriptional differences in shoots and roots of <i>Medicago sativa</i> cv. Daisy
Figure 3.1: Gene expression patterns of candidate N status marker genes in <i>A. thaliana</i> vegetative tissue after three weeks of growth
Figure 3.2: Phylogeny of temperate forage and related model species for <i>Nitrite</i> <i>Reductase</i> (<i>NIR</i>)
Figure 3.3: Pipeline for candidate gene primer design for forages crops
Figure 3.4: Gene expression patterns of N status marker genes in grasses <i>Lolium perenne</i> and <i>Festulolium</i> hybrid vegetative tissue at three weeks
Figure 3.5: Gene expression patterns of N status marker genes in <i>Festulolium</i> vegetative tissue at three weeks, with yield photographs
Figure 3.6: Normalised expression data 'mean.delta.ct.norm' of 2018 and 2019 glasshouse grown <i>Festulolium</i>
Figure 3.7: Gaussian process models of 2018 and 2019 glasshouse grown Festulolium. 115
Figure 3.8: Analysis of individual gene models for field grown Festulolium
Figure 3.9: Analysis of combined gene models for field grown Festulolium
Figure 3.10: Extractable soil NO ₃ ⁻ and <i>Festulolium</i> yield for fields from 2018 and 2019, with comparison to combined gene models generated with glasshouse data
Figure 4.1: NO ₃ -selective sensors (Probe) correlated to conventional soil nitrate extraction analysis methods (Assay)
Figure 4.2: Field trial example of NO ₃ -selective sensors in plots treated with different potassium nitrate concentrations
Figure 4.3: Schematic of column set-up for NO ₃ -selective sensor experiments
Figure 4.4: Lolium perenne monocrop column experiment NO ₃ -selective sensor data. 145

Figure 4.5: Monocrop column experiment NO ₃ ⁻ -selective sensor data for 'Monocrop 2' and 'Monocrop 3'
Figure 4.6: Repeated measurements ANOVA for NO ₃ -selective sensor data for 'Monocrop 2' and 'Monocrop 3'
Figure 4.7: Lolium perenne and Medicago sativa intercrop column experiment NO ₃ - selective sensor data
Figure 4.8: Intercrop column experiment NO ₃ ⁻ -selective sensor data for 'Intercrop 2' and 'Intercrop 3'
Figure 4.9: Repeated measurements ANOVAs for NO ₃ -selective sensor data for 'Intercrop 2' and 'Intercrop 3'
Figure 4.10: Column experiment NO ₃ -selective sensor data for 'Monocrop 3' and 'Intercrop 3'
Figure 5.1: Vegetative tissue of forage crops following one of two fulvic acid treatments compared to a control
Figure 5.2: Vegetative biomass of forage crops following one of two fulvic acid treatments relative to a control
Figure 5.3: Analysis of two fulvic acids (MFA and VFA) using ¹³ C-NMR for carbon compounds in evaporated samples
Figure 5.4: <i>Medicago sativa</i> cv. Daisy seedlings following treatment with fulvic acids or controls.
Figure 5.5: Vegetative biomass of <i>Medicago sativa</i> cultivars and <i>Lolium perenne</i> following treatment with fulvic acids or controls
Figure 5.6: <i>Medicago sativa</i> cv. Daisy nodules following treatment with fulvic acids (MFA or VFA) or controls
Figure 5.7: Nodulation counts of two <i>Medicago sativa</i> cultivars following treatment with fulvic acids or controls
Figure 5.8: Vegetative and nodule phenotypes of <i>Medicago sativa</i> cv. Daisy following treatment with fulvic acids or controls, with or without inoculation of <i>Sinorhizobium meliloti</i>
Figure 5.9: Growth effects of fulvic acid on the growth of <i>Sinorhizobium meliloti</i> in TY medium, compared to controls
Figure 5.10: Two-dimensional principal component analysis (PCA) of fatty acid methyl esters (FAMEs) relative contents from soil treated with fulvic acid or controls
Figure 5.11: Quantification of <i>Sinorhizobium meliloti</i> DNA using <i>nodC</i> gene from soil treated with fulvic acid or controls
Figure 5.12: Vegetative biomass of <i>Medicago sativa</i> cultivars in independent field trials following treatment with a fulvic acid or controls194
Figure 5.13: Vegetative tissue of <i>Medicago sativa</i> of first cut following fulvic acid treatment compared to a control, from field trial plots

Figure 6.1: Multi-dimensional scaling (MDS) plots to compare all transcripts in experimental replicates clustered for treatment and timepoints
Figure 6.2: Differentially expressed transcripts in <i>Medicago sativa</i> shoot and root tissue with treatments of either fulvic acid (VFA, orange) or the control (VC, yellow)
Figure 6.3: Volcano plot of DE transcripts as log Fold Change (logFC) between 0 day and 3 day for VFA treatment RNA samples
Figure 6.4: BLAST sequencing results for DE transcripts of VFA treatment from <i>de novo</i> RNAseq analysis, showing results for root tissue only
Figure 6.5: 'Biological Process' GO term hits for individual DE transcripts from VFA treatment of <i>M. sativa</i> roots
Figure 6.6: Direct Acyclic Graphs (DAGs) of GOSlim 'Molecular Function' associated with DE transcripts from VFA treatment of <i>Medicago sativa</i> roots
Figure 6.7: Quantitative reverse transcription PCR (qRT-PCR) of a subset of differentially expressed (DE) transcripts to confirm RNA-seq in <i>Medicago sativa</i>
Figure 7.1: Results compared to targets for improved forage crops in Figure 1.4
Figure A1: Arabidopsis eFP Browser [128-130] images for expression candidate N status genes
Figure A2: Forage crop RT-PCR screen of developed N status marker gene primers electrophoresis gel photos
Figure A3: Gene expression patterns of N status marker genes in <i>Medicago sativa</i> , cv. Daisy and Luzelle, vegetative tissue at three weeks
Figure A4: Normalised expression data 'mean.delta.ct.norm' of 2018 glasshouse grown Festulolium. 256
Figure A5: Gaussian model of 2018 glasshouse grown Festulolium
Figure A6: Analysis of individual gene models for 2018 field grown Festulolium
Figure A7: Analysis of combined gene models for 2018 field grown Festulolium
Figure A8: Normalised expression data 'mean.delta.ct.norm' of 2019 glasshouse grown Festulolium
Figure A9: Gaussian model of 2019 glasshouse grown Festulolium
Figure A10: Analysis of individual gene models for 2019 field grown Festulolium
Figure A11: Analysis of combined gene models for 2019 field grown Festulolium
Figure A12: Extractable soil NO ₃ ⁻ for <i>Festulolium</i> fields from 2018 and 2019, correlated to model predicted soil NO ₃ ⁻ from probability distribution from combined gene models generated with glasshouse data
Figure A13: <i>Festulolium</i> yield for fields from 2018 and 2019, with comparison to vegetative chlorophyll and total protein
Figure B1: Schematic of nitrate-selective sensor construction
Figure B2: Photograph of soil column set-up for nitrate-selective sensor experiments. 268

Figure B3: Lolium perenne monocrop column experiment NO ₃ -selective sensor data for 'No crop' and 'Monocrop 1'
Figure B4: Lolium perenne and Medicago sativa intercrop column experiment NO ₃ - selective sensor data for 'No crop' and 'Intercrop 1'
Figure C1: Weed counts and soil pH of pots with forage crops following one of two fulvic acid treatments relative to a control
Figure C2: Gas chromatogram spectra of (a) citric acid standard and b) MFA, with c)-d) NIST Atomic Spectra database 1A v14 matches for citric acid, 4TMS derivative and Trimethyl TMS derivative
Figure C3: Gas chromatogram spectra of (a) PEG-400 standard and b) VFA, with c) NIST Atomic Spectra database 1A v14 matches for poly(ethylene glycol) (heptaethylene glycol)
Figure C4: Vegetative and nodule phenotypes of <i>Medicago sativa</i> cv. Luzelle following treatment with fulvic acids or elemental controls, with or without inoculation of <i>Sinorhizobium meliloti</i>
Figure C5: Vegetative and nodule phenotypes of <i>Medicago sativa</i> cv. Gea following treatment with fulvic acids or controls, with or without inoculation of <i>Sinorhizobium meliloti</i>
Figure C6: CFU counting of <i>Sinorhizobium meliloti</i> for growth effects of fulvic acid in TY medium, compared to elemental controls
Figure C7: Fatty acid methyl esters (FAMEs) relative contents from soil treated with fulvic acid or controls
Figure C8: Fatty acid methyl esters (FAMEs) biomarker assignments to microbial popoulation relative contents from soil treated with fulvic acid or controls
Figure C9: Field plots for fulvic acid trials performed in 2017 and 2018
Figure D1: Gel from Bioanalyzer 2100 run of RNAseq samples
Figure D2: Differential transcript expression using voom/Limma method [184] between 0 day (green) and 3 day (red) for VFA treatment RNA samples (see 2.4.1) using Degust [183] as in Section 2.4.6
Figure E1: Typical nutritional value of different processed forage grass, dried grass (blue) [64] and two examples of silage grass (grey) [829; 830].
Figure E2: Assorted crop drying advertisements from 1970s [831; 832], showing the huge gain of interest in the agricultural techniques first developed from 1925 [62]
Figure E3: UK and ROI forage growers and driers belonging to the British Association of Green Crop Driers (BAGCD) in (A) 1975 [832], (B) 1980s - 2000s [4; 834; 398], and (C) 2018 [8; 398]
Figure E4: Total European Union output for all crops in 2016 from Eurostat data [841]. 298
Figure E5: Extent of forest of the UK and surrounding Western European Countries from 1990 - 2015, data from from the "The Global Forest Resources Assessment 2015: Desk reference" [843]

Figure E6: Global population since 1700 to the present day; adapted from "World population to 2300" and "World Populations Prospects: The 2017 Revision" [852; 853].	2
Figure E7: The vital exemption for BAGCD members in the RB209, outlined by the Crown© [414]	4
Figure E8: Nitrate vulnerable zones (NVZs) map of the UK for 2017, directly reproduced from the "Review of Nitrate Vulnerable Zone designations for implementation in 2017: Environment Agency report and recommendations to DEFRA: Report Number HOEV151604/R" [82]	17
Figure E9: Trend in UK air pollutant emissions in million tonnes including Nitrogen Oxide (NO and NO ₂), Ammonia (NH ₃) and Non-Methane Volatile Organic Compounds (NMVOCs)	s 19
Figure E10: 'Ecosystem services' and their link to 'Constituents of Well-being', directly reproduced from the "Ecosystems and Human Well-being: Synthesis" [101]	.2
Figure E11: 'Expenditure of UK biodiversity', data from the "UK Biodiversity Indicators 2018" [889]	.7
Figure E12: Global livestock feed rations, adapted from "GLEAM 2.0 - Assessment of greenhouse gas emissions and mitigation potential" [893]	.0

LIST OF TABLES

Table 1.1: Forage crops cultivated in the UK by the British Association of Green Crops Driers (BAGCD). 15
Table 1.2: The multiple definitions surrounding nitrogen use efficiency (NUE). 17
Table 2.1: Primers used in study for Arabidopsis thaliana. 33
Table 2.2: Primers used in study for Medicago sativa testing based on legume consensus. 35
Table 2.3: Primers used in study for Lolium perenne testing based on grass consensus or published primers. 36
Table 2.4: Forage grass species used in this study. 39
Table 2.5: Forage legume species used in this study. 40
Table 2.6: Primers used in study for legumes based on Medicago sativa testing
Table 2.7: Primers used in study for grasses based on Lolium perenne testing. 43
Table 2.8: VFA detectable content in mg/L for individual elements by ICP-OES and ICP-MS if 0.5 % solution.
Table 2.9: MFA detectable content in mg/L for individual elements by ICP-OES and ICP- MS for 0.5 % solution. 64
Table 2.10: VFA control solution (VC) components for 1 % application as in 2.3.1. 69
Table 2.11: MFA control solution (MC) components for 0.5 % application as in 2.3.1 69
Table 2.12: Primers used in study for soil DNA extractions for testing Sinorhizobium meliloti activity. 78
Table 2.13: Primers used in study for Medicago sativa qRT-PCR confirmation of RNAseq analysis. 87
Table 4.1: Column set-up for the Lolium perenne monocropping experiment described in Section 2.2.4.
Table 4.2: Column set-up for the Lolium perenne and Medicago sativa cv. Daisy intercropping experiment described in Section 2.2.5.
Table 6.1: Pseudoalignment summary of RNA-seq samples with <i>de novo</i> transcriptome assembly.
Table 6.2: Fold enrichment of top 20 'Biological Process' for upregulated DE transcripts from VFA treatment of Medicago sativa roots.
Table 6.3: Fold enrichment of top 20 'Molecular Function' for upregulated DE transcripts from VFA treatment of Medicago sativa roots. 222
Table 6.4a/b/c: Enriched DE transcripts in VFA roots which are putatively classed as highly preferential nodulation regulatory genes and nodule-associated signalling-related genes as in Roux et al., (2014) [705]

Table A1: R Script outlined for Gaussian Process (GP) fitting model of N status marker	
252	
283	
285	
؛ 286	

Table E1: Forage crops currently cultivated by BAGCD members across the UK, or thosenot cultivated but popular for other forage crop growers.290

LIST OF COMMON ABBREVIATIONS

	British Association of Green Crop
BAGCD	Driers
BLAST	Basic local alignment search tool
CFU	Colony forming unit
DE	Differentially expressed
	Department for Environment, Food
DEFRA	and Rural Affairs
dH ₂ O	Distilled water
DM	Dry matter
EIN	Early initiating nodule
FA	Fulvic acid
FAME	Fatty acid methyl ester
GO	Gene ontology
HS	Humic substances
ICP	Ion-coupled plasma
LV	Lytic vacuole
MC	MFA control
MDS	Multi-dimensional scaling
MFA	MPXA FA
MS	Mass spectrometry
MSA	Multiple sequence alignment
Ν	Nitrogen
N ₂ O	Nitrous oxide
NA	No application
NH4 ⁺	Ammonium
NMR	Nuclear magnetic resonance
NO	Nitric oxide
NO ₃ -	Nitrate
NUE	Nitrogen use efficiency
NVZ	Nitrate vulnerable zone
OD	Optical density
PCA	Principle component analysis
PLFA	Phospholipid fatty acid
PM	Pink nodule
PSV	Protein storage vacuole
PUFA	Polyunsaturated fatty acids
	Quantitative reverse transcription
qRT-PCR	PCR
RNA-seq	RNA-sequencing
RT	Room temperature
UAA	Utilised agricultural area
VC	VFA control
VFA	Vitalink FA
WN	White nodule

Chapter 1: Introduction

1.1 THE IMPORTANCE OF FORAGE CROPS

Forage crops are grasses, legumes, root crops, and trees, grown globally for animal feed. Their cultivation provides the necessary bulky feed for the animal product industry for human consumption. In this section the relevance of these crops in Europe and the UK will be discussed.

1.1.1 Temperate crops and cultivars

Forage grasslands represent 26 % of global land area, and 70 % of agricultural land [1]. These grasslands are used to feed livestock with forage crops available for tropical and temperate climates. In the UK, temperate forage crops are cultivated, and these are usually grasses (*Poaceae*) or herbaceous legumes (*Fabaceae*). In temperate climates like the UK, the main forage grasses include ryegrass (*Lolium* spp.), fescue (*Festuca* spp.), bentgrass (*Agrostis* spp.), and orchard grass (*Dactylis* spp.), or hybrids such as *Festulolium* [2; 3]. The most commonly cultivated legumes are medics (*Medicago* spp.), trefoil (*Lotus corniculatus*), clover (*Trifolium* spp.), and vetches (*Vicia* spp.). The globally important legume lucerne or alfalfa (*Medicago* sativa) is of particular prominence in UK forage production, even termed the 'wonder protein plant' [4], and is the most widely cultivated crop along with *Lolium*, *Festuca*, and *Festulolium* grasses. Some *Brassicaceae* and *Amaranthaceae* forage species can also be cultivated for their vegetative and root tissues.

The vast array of forages, as well as the multiple cultivars available within each species or hybrid family, means a crop can be chosen for the specific environment of cultivation. These can be cultivated as stand-alone crops, or as a mixed or intercropped system. Forage crops will either be grown as part of a permanent pasture, with little to no management influence, or as low to high intensity cultivated temporary grassland, sometimes in rotation with commodity crops. A general term used by Eurostat (the statistical directorate-general of the European Commission) is "fodder crops", which describes forage crops grown using cultivation methods in utilised agricultural area (UAA), and so not as a permanent pasture. Figure

2

1.1 below shows the UAA of the top forage producing countries using the fodder crops metric [5]; 1.1a shows the percentage of UAA land in these countries as fodder cropland; 1.1b shows this fodder cropland as area in hectares.



a) Arable land percentage of utilised agricultural land in Europe

<u>Figure 1.1:</u> Fodder crop production in the top forage producing European countries.

Source of data is Eurostat [5] with online data codes: ef_lac_main and Eurostat calculations. Data is for 2016 and shows top forage producing European countries for a) fodder crop arable land percentage of utilised agricultural area, and b) the total fodder area in hectares of forage crops. Note that Italy is based on predicted data, whereas all other countries are accurate measurements.

1.1.2 Relevance and cultivation in the UK

Wild forage pastures began at 9000 - 7000 BC, coinciding with the domestication of animals across North Africa, Europe, and Asia [6]. Domestication of forage species occurred from ~ 1000 BC. During the 13th - 19th centuries developments in agricultural practice and haylage production led to more forage crop cultivation. In the 19th – 20th centuries there was the rise of agricultural intensification, which was particularly prominent throughout Europe and the UK [7; 6]. Now forage crops are widespread across the UK, as shown in the temporary grassland production map of Figure 1.2. This figure shows forage production occurs across climates and soil type, including sandy soils and marginal land. This is possible as species and cultivars can be chosen to match the local climate. Livestock can be fed directly with fresh forage crops, or with processed, partially dried or pre-digested feeds, including hay, haylage, and silage. Dried feed in the UK is predominantly produced by the British Association of Green Crop Driers (BAGCD); BAGCD produce ~ 90 % of UK dried feeds [8]. The location of BAGCD members is shown in Figure 1.2. These feeds are termed bulky feeds and are produced from grass and legume cropping, most predominantly using *L. perenne* (ryegrass) and *M. sativa* (lucerne), or a mix of these and similar grass/legume mixes. Animals can also be fed concentrates, which are generally cereals, oilseeds and legume seeds. Additionally, sugar-rich high energy feedstuffs are added to concentrates, such as molasses and fats. Using concentrates as feeds means the energy required to produce animal products is greatly increased compared to the energy input required for animal products produced from bulky feeds, with sometimes over double dry matter required per-capita yr⁻¹ [9]. Moreover, these concentrate feeds for animals are in direct competition with materials for foodstuffs for human consumptions [10]. As animal production is heavily resource intensive, and concentrate food diverts land from direct food production for humans, using forage crops as bulky feed instead of cereal-based concentrate feeds is necessary to lowering feed vs. food competition. Making forage crops more efficient in their production will greatly help this aim.

Cultivation of forages is widespread across the UK, and globally there has been a rise in many countries specific breeding programmes for forage crop species [11-15]. There are many UK and European forage breeders, especially since there has been a rise in the use of forages as energy and cover crops, and these include DSV United Kingdom Ltd., LS Plant Breeding, Nickerson, Germinal Holdings Ltd., and Limagrain UK Ltd. Designated breeding programmes are possible due to the economic importance of forage crops. This is shown in Figure 1.3 where the UAA production percentage of crops is divided by type, with temporary grass being 19 % [16], showing how intense cultivation is hugely important to UK agriculture. As forage crop cultivation is intrinsic to intensive agriculture, forage crop cultivation is essential for supporting a growing human population.

Throughout the decades, there have been periodic developments which have intensified forage cultivation and the regulation of the industry has undergone many changes. My review entitled 'UK dried forage production: a review of industry changes and assessment of prospects for both policy and science' can be found in Appendix E. This review details the recent changes in UK forage crop cultivation, particularly those in the dried forage industry relating to BAGCD members. Most BAGCD forage crops are grown in temporary grassland, which are shown in Figure 1.2 as area of UAA. This production is high intensity compared to permanent grasslands, with high nitrogen (N) fertiliser applications and intense cutting regimes.



Figure 1.2: Temporary grassland area in the UK.

Map shows grassland area in hectares produced in the UK as temporary grassland, reproduced and adapted from ADAS NNFCC project 08-004 report [17]. The key shows the area in hectares of grassland per unit square, with yellow depicting less than 200 ha and dark blue as more than 1400 ha. Red triangles show the sites of British Association of Green Crop Driers (BAGCD) farms.



Figure 1.3: Utilised agricultural area (UAA) of UK crop types in 2018.

The percentages of UK crop types are adapted from the DEFRA Farming Statistics 2019 [16]. Forage crops include the 'Temporary grass' in dark green at 19 %, but also includes a proportion of 'Other arable crops' in dark blue, and forage legumes are designated as this crop types.

1.1.3 Nutritional content and yield

Not only are there numerous species cultivated in the UK, but their nutritional content and growth can vary significantly. The value of a forage crop depends on 1) its nutritional content, including the content of complex molecules that can change the nutrient availability, and 2) the ability to rapidly increase the vegetative biomass. Herein will be described the factors effecting nutritional content and yield in forage crops; a more detailed review can be found in Capstaff and Miller (2018) [18].

Digestibility

The nutritional status of a forage crop depends upon the relative concentrations of carbohydrates, proteins and lipids. This reflects the digestibility (D-value) of a crop, and this along with trace elements, vitamins, and minerals, provides the nutritional value available for the animal on consumption. The metabolisable energy is measured in MJ / kg dry matter (DM) [19]. The following paragraphs discuss the nutritional content of forage crops in the context of digestibility.

Carbohydrates

In forage crop vegetative tissue, carbohydrates can be 50 - 80 % of DM. Carbohydrates will primarily be insoluble structural polysaccharides including cellulose and hemicellulose, or in the storage forms of starch and water-soluble sugars. The different ratios of these carbohydrates within a forage crop will alter its downstream digestibility. This is especially true if cell wall structures constrain digestion through limiting cell wall penetration by the animal's microbial populations within the gut [20]. As such, molecules like lignin, a polyphenolic compound within forage and not a carbohydrate, can bind the structural carbohydrates cellulose and hemicellulose, having a huge impact on forage digestibility [21]. Forages with decreased lignin concentration during growth will increase the digestible DM of the feed. This is true for *L*. perenne, where high soluble sugar content alongside low lignin content provides a high D-value [22].

<u>Protein</u>

Forage proteins in bulky feeds determine the N availability to animals. The majority of this protein, like with all land plants, will be in the form of Ribulose-1,5-bisphosphate carboxylase/oxygenase (RuBisCO), although the relative amounts do vary between species and cultivars [23]. Other major N-containing compounds include N-storage proteins, nucleic acids, and nitrate [23]. Forage crops with especially high protein contents are the forage legumes, especially M. sativa, T. pratense, and T. repens [22]. Lignin can negatively affect the digestibility of protein, as well as carbohydrates, by binding and preventing hydrolysis and the subsequent breakdown of the molecules. Some micronutrients like condensed tannins or proanthocyanidins can affect protein digestibility, by inhibiting degradation through binding. This binding can be advantageous for feed, as binding reduces rapid protein degradation which reduces bloat [24; 25]. There is a delicate balance between N availability through protein degradation and reducing bloat for animal productivity [26]. Grasses contain little or no proanthocyanidins content, but many legumes have very high levels of up to 18 % DM [27; 26].

<u>Lipids</u>

Lipid content is the final factor effecting D-value in forage crops. They are mostly found as polyunsaturated fatty acids (PUFAs); PUFAs can be in the range of 10 - 30 g kg⁻¹ [28] and the most abundant kinds is a-linolenic acid at ~ 62 % total lipids [29]. The PUFAs of linolenic and palmitic acid are also present in high numbers [30]. These lipids are important in both the feed composition, but also the quality of the final animal product. It has been shown that forage diets with low PUFA levels when compared to cereal concentrate diets will produce leaner meat [31; 32]. In addition, forage has been shown to produce milk with lower PUFA content and higher levels of trans-fatty acids [33; 34]. Therefore, studies have been carried out to profile PUFAs across forage species [35]. Differences in PUFA content can be seen between cultivars, species, harvest period, and environment [36; 29]. The grasses *L. perenne, F. pratensis*, and

10

Festulolium hybrids vary in their PUFA content at the beginning and end of their growth season, as well as between individual cuttings [37].

Trace Elements

Trace elements and minerals in forage crops are important for animal health. It is important that the nutritional balance of feeds is optimal for the animal, to ensure healthy processes like immunity without need for unnecessary antibiotic use. Zinc is important for immunity and, although supplements can be used to aid animal health, such additions can cause wasteful excretion [38]. In contrast, the accumulation of toxic minerals must also be avoided in forages. The required balance in trace element nutritional content is best shown with selenium as an example, where low levels are beneficial for animals and high concentrations cause toxicity [39]. Trace elements can also make the forage unpalatable for animals, and must also be considered when improving the trace element content of forage crops.

Biomass

Along with studies aligned to animal nutrition, the most important area of forage crop research is in rapid biomass production, particularly for height and vegetative biomass [40]. This is because crops are either cut or grazed directly, with high frequency. For dried forage producers such as the BAGCD, crops are cut throughout the spring/summer growing season on average 6 – 8 weeks (or 4 – 7 cuts depending on local climate, weather conditions and growth). Although many different plant species can be grown for forage, grasses are most desirable as they have a shoot meristem which promptly responds to cutting by increasing vegetative growth. In many forage grasses, the correlation of aboveground cutting or grazing has been correlated to increased shoot growth alongside increased root exudation [41-47]. These exudates cause a flush of carbon-containing compounds into the surrounding soils, which microbes can use to mobilise nutrients, which in turn will sustain the aboveground regrowth of the plant. Therefore, maintaining both nutrient and water supply for regrowth postcutting is important in forage crop biomass production [48], and N

availability to the crop is especially important [45; 49; 50]. Strategies to improve biomass production in forage crops will be discussed in Chapter 5.

Current research in forage crop production

Despite Figures 1.1, 1.2, and 1.3 showing forage wide cultivation and importance as crops, including specific breeding initiatives for species, forage crops have been relatively forgotten in modern research when compared to cereals, vegetables and fruits [18], and their potential remains largely unexplored [51]. Techniques in genetic transformation of forage crops have been developed [52], which have made the study of individual genes possible for their effect on nutritional content. Figure 1.4 below shows some of the main targets for improvements in forage crops. Individual advances have been reviewed elsewhere [13; 53-58; 18; 59; 60], and will discussed further in Chapter 3.



Figure 1.4: Targets for improved forage crops.

Adapted from Capstaff and Miller (2018) [18], showing the main targets for improvement in forage crop research.

1.2 FORAGE DRYING IN THE UK – BRITISH ASSOCIATION OF GREEN CROP DRIERS (BAGCD)

1.2.1 Methodology conception

With the increase in agricultural intensification at the beginning of the 20th Century [7; 6; 18], new technologies were developed. In the 1920s drying technology gained attention [61; 62], and in the 1930s dried forage was bulk produced in the UK [63]. The process of high temperature drying established, enabled the production of pelleted feed from forage which lasted longer than normal fresh fodder, haylage or silage, whilst preserving nutritional content [64]. High temperature drying was particularly prominent post-War, where lowering waste from fresh forages was paramount to stockpiling dried feed efforts [65]. From the 1950s to 1970s, new drying technologies continued to develop, and many UK growers purchased rotary machines. These rotary machines use high temperature and rotation to dry grass and legumes for manufacture as pellets and are used by BAGCD sites to this day. Further information on the method development across Europe and the UK can be found in Appendix E.

1.2.2 Present production of dried forage

Present production locations of dried forage are shown in Figure 1.2, with sites represented as red triangles on UK map. BAGCD is an association of numerous farms based in Cambridgeshire, Devon, Essex, Hertfordshire, Lincolnshire, Perthshire and Yorkshire. In total they farm approximately 7500 hectares of land through high intensity, temporary grasslands across all these sites, with only one farm as permanent pasture. Crop rotations of forage crops are cultivated with break crops of wheat, maize or oilseed. As before, cuts are taken throughout the spring/summer growing season every 6 – 8 weeks. The crops grown by BAGCD members can be found in Table 1.1; crops which are currently cultivated are shown, alongside those recently cultivated or of interest to members.
<u>Table 1.1:</u> Forage crops cultivated in the UK by the British Association of Green Crops Driers (BAGCD).

Crops are shown which are currently cultivated, or are of interest to BAGCD for future cultivation. Crops are split for grasses and legumes, with species and common names included.

Grasses		Legumes		
Species	Common name	Species	Common name	
Lolium perenne	Perennial Ryegrass	Medicago sativa	Lucerne/Alfalfa	
Festuca arundinacea	Tall Fescue	Lotus corniculatus	Birdsfoot Trefoil	
Festuca pratensis	Meadow Fescue	Trifolium pratense	Red Clover	
<i>Festuca</i> x <i>Lolium</i> (various)	Festulolium	Pisum sativum	Spring Pea	
Phleum pratense	Timothy grass			
Anthoxanthum odoratum	Sweet Vernal grass			

Current crops cultivated by BAGCD

Crops of interest to BAGCD

Gras	sses	Legumes		
Species	Common name	Species	Common name	
Dactylis glomerata	Cocksfoot	Onobrychis viciifolia	Sainfoin	
Lolium x boucheanum	Hybrid ryegrass	Trifolium repens	White Clover	
Lolium multiflorum	Italian Ryegrass	Galega orientalis	Forage Galega	
Lolium westerwoldicum	Westerwold Ryegrass			

1.2.3 The main difficulties associated with the forage crop industry is nitrogen use efficiency

The industry faces many challenges at present, ranging from high fuel costs and support payment changes, to soil erosion and biodiversity initiatives. Again, these are discussed in Appendix E, where industry, policy and research are detailed in parallel, and discussed regarding the operations of the BAGCD. Undoubtably one of the most prominent challenges felt by BAGCD members is in future N fertiliser use.

In the highest forage producing countries in Europe, N use efficiency (NUE) of fertilisers for all crops is variable, with NUE low in the UK, shown in Figure 1.5. It is important to define NUE and how it can be measured. The definition of NUE has been used sporadically and inaccurately across the years, with limited agreement between studies throughout the decades [66-71; 18]. This is due to the complicated nature in which N acts within the plant system. In reviews by both Xu et al., (2012) [72] and Good et al., (2004) [73] the authors highlight that efficiency can be monitored using different parameters. Some definitions are based on grain yield and others are for biomass, with only the latter relevant to forage crops. Table 1.2 is an amalgamation of the definitions as set out in these papers as well as my own contribution.

Some definitions are interested in the N distribution processes from the environment through the plant, such as nitrogen uptake efficiency (NupE), nitrogen transport efficiency (NTE), and nitrogen remobilization efficiency (NRE). These primarily measure the mechanisms by which the N can be made available from the soil supply to the tissue of interest.

Other terms are associated with how biomass changes in differing conditions, such as nitrogen use efficiency (NUE), nitrogen harvest/usage index (NHI), and nitrogen physiological use efficiency (NpUE). These terms measure of how the mechanisms of each genotype give a phenotype for yield and productivity. These terms can also be adapted for the measurement of nutritional status. For example, NUE can be modified to quantify the amount of a certain compound accumulated within tissue, based on N availability.

16

<u>Table 1.2:</u> The multiple definitions surrounding nitrogen use efficiency (NUE).

Definitions are adapted from those given in Xu et al., (2012) [72] and Good et al., (2004) [73].

Term	Equation	Definition
Nitrogen use efficiency (NUE)	Relevant biomass ÷ N available for plant	Total biomass of tissue of interest (e.g. shoots, grain) for N supplied
Nitrogen harvest/usage index (NHI)	N contents of relevant biomass ÷ N contents of whole plant	Shows N in relevant tissue whilst considering absolute biomass
Uptake efficiency (NupE)	N in plant ÷ N available	Capacity of the roots to uptake N available in soil
Utilization efficiency (NUtE)	Relevant biomass ÷ total N in plant (all tissues)	The percentage of N assimilated into the tissue of interest (e.g. grain)
Nitrogen transport efficiency (NTE)	N in above ground tissue ÷ amount of N taken up by roots	Relates how much N is transported after uptake
Nitrogen physiological use efficiency (NpUE)	Difference in relevant tissue weight with N application and without	Shows net efficiency of plant tissue with N application (e.g. fertilizers)
Agronomic efficiency (AE)	The cost difference of NpUE	The monetary difference between N application and without
Nitrogen remobilization efficiency (NRE)	N remobilized from sink tissue to source tissue	Measures how well plant redistributes N to where it is required (e.g. grain)

NUE, in terms of N fertiliser input compared to crop output, is low in the UK. Figure 1.5 shows a comparison between the NUE of all crops in the UK and the other top forage producing European countries. The UK is close to an overall NUE rate of 50 %, which greatly impacts all crop production but especially forage crops. Grasses and legumes have a slightly larger N content than cereal grain crops and therefore require higher N rates for optimal production [74]. Deficiency seriously limits forage crop production by restricting protein assembly, and primary and secondary metabolism. Forage crops have a high N demand of leaf photosynthetic tissue, and deficiencies impact many growth parameters [75]. This N requirement is exacerbated by the high cutting regimes of forages.

Over the past three decades there has been an increase in policy reforms to curb excessive N fertiliser use [76-79]; again details of these changes are discussed in Appendix E. Most importantly to the dried forage industry is the BAGCD exemption for N fertiliser use in the Fertiliser Manual RB209 8th addition [79]. In this exemption it is stated that BAGCD members can add up to 500 – 700 kg N ha⁻¹ yr⁻¹ to their crops depending on irrigation level; this application rate is significantly higher than the 2018 average N fertiliser use of UK farms for all other crops at 144 - 152 N ha⁻¹ yr⁻¹ [80]. BAGCD sites are found in nitrate vulnerable zones (NVZs), shown in Figure 1.6, which has confounding implications for NUE. NVZs make up approximately 58 % of the land in England, which has increased in total area from 2013 by 1300 km² [81; 82]. These areas are determined by land gradients, ground cover, water sources, soil types, and weather conditions [83]. In Chapter 4 such implications, as well as current assessment methods, will be explored in more detailed.

18

Fertiliser efficiency of all crops in highest forage producing European countries.



<u>Figure 1.5:</u> N fertiliser efficiency of all crop types in the highest forage producing European countries.

Data is for fertiliser efficiency of all crop types in 2015 for those European countries which are the top forage crop producing. Countries are shown as diamonds (using International organization for standardization United Nations coding) for N fertiliser input against total N crop output of tissue of interest (see NUE from Table 1.2). UK is depicted as red diamond. Efficiency is indicated for 50 % (green dotted line), and 90 % (blue dotted line).Source of data is Eurostat [84] with online data codes: aei_pr_gnb and Eurostat calculations.



Figure 1.6: Nitrate vulnerable zones (NVZs) in England.

Map shows total areas in England which are classed are nitrate vulnerable zones, reproduced and adapted from [85] Purple shows areas which are NVZs, with diagonal blocks showing areas where drinking water may be affected due to proximity to NVZs. Red triangles show the sites of British Association of Green Crop Driers (BAGCD) farms. Source: Environment Agency, UK.

1.2.4 Grassland research

As mentioned previously, forage crops can be cultivated under many different intensity systems. Forages can be grown in permanent pastures, which due to their stability are known to be important in preventing soil erosion, and contribute to fixing atmospheric carbon, with forages sequestering 34 % of global carbon stocks [86]. Permanent grasslands are capable of increasing biodiversity-richness and genetic variability [86], and provide crucial ecosystem services including water services, climate regulation, erosion prevention, cultural services, and clearly biomass production [87]. As there is no tillage of the soil, permanent grasslands can have positive effects on diversity, erosion, and emissions [88-92].

Despite the 'natural capital' [93; 94] within permanent pastures, most forage cultivation for agriculture is through high intensity, temporary grasslands, often in rotation with commodity crops. This high input/high output system has been pushed by continued global pressures for animal products [95-100]. However, there has been little research targeted to temporary grasslands, and it is imperative that we better understand temporary grasslands to determine their role in agricultural nutrient cycling when compared to permanent pastures.

Managed, temperate grasslands are of particular interest in biodiversity studies to determine how N fertiliser use of intense production can affect the environment. Significant habitat changes and declines of biodiversity in managed forage crops due to pollution from N fertilisers have been found [101]. These changes have been replicated in both short- and long-term experiments for many species [102-109]. For soil erosion, temporary cultivation may not be as beneficial as permanent pastures as more complex root architecture may not have had a chance to develop [110]. Moreover, many forage crops used in intensive cultivation (see Table 1.1) are associated with increased soil aggregate stability [111], and improve soil organic matter [112], when in permanent grasslands. Furthermore, emissions are known to differ with fertiliser use and the crop cultivated [113; 114], and at present there is limited data for how temporary forage cropland with intensive cutting affects this loss [115; 116].

1.3 STRATEGIES TO IMPROVE NITROGEN USE EFFICIENCY IN FORAGE CROPS

1.3.1 Project aims

It is difficult to draw parallels between permanent grassland research and temporary forage cultivation for dried bulk feed due to a lack of studies aimed specifically at the industry. Forage crops are grown as feed for animal production, and the social, environmental, and economic importance of forage in the UK has been discussed in Capstaff and Miller (2018) [18]. For UK forage growers, having ways to increase the crop's NUE is a primary focus at present. This project concentrated on three strategies for achieving improvements in NUE, in a multifaceted approach. This included assessing the plant itself during growth periods, evaluating the soil changes in forage crop production, and a possible management practice to increase yield with fewer nitrogen inputs. These were conducted in-line with current BAGCD production methods and management practices. Such a project not only tackles improving NUE in forage crops, but also provides valuable insights into temporary, high intensity legume and grasslands.

The overall goal of this project was to learn more about the basic connection between input and biomass in a range of forage crops. The main over-arching theme was to develop ways to test this relationship with the aim of increasing biomass with lowered inputs in the future. Moreover the project was aimed at providing more evidence for farmers regarding their N fertiliser use in the context of the above goal.

1.3.2 Thesis overview, contents, and hypotheses

Chapter 2: Materials and methods

All materials and methods for the thesis are described for Chapters 3 - 6 within this section.

<u>Chapter 3: Marker genes in Festulolium can be used to assess crop</u> and soil N status, to aid with grower decisions to improve future yield

At present farmers cannot reliably determine the N status of their crops [18]. This is predominantly hindered by unreliable and unreproducible tests for soil N levels. Presently farmers take limited samples across their growing area in the hope that this is representative of the N in the whole plot, and moreover through the whole growing season. If instead the farmer could determine their crop N status directly, then they could make a more informed decision as to how they should subsequently treat the plot.

Many genes have been shown to link N availability and expression across many crops [117], but a suite of suitable candidates which could be tailored for this need is lacking. Also, as there is a diverse variety of species and cultivars grown for forage, such a suite of genes would require high conservation. A detailed review of such potential genes available for testing is found in Chapter 3.1 for the following hypothesis:

Hypothesis for Chapter 3: Expression of N status genes is reliably related to crop production and soil N status.

A suite of N status marker genes for forages was investigated to predict crop quality and soil N status. This was developed in glasshouse from model plants to forage crops and then be applied to the field in *Festulolium*.

Chapter 4: Soil sensors can detect N profile changes under forage

crops with different management regimes

Numerous management regimes have effects on grass production [118], but their effect on soil N profiles have not been formally tested for UK forage growers. These management practices include nitrate (NO_3^-) application, defoliation, and cropping method[119]. It is unknown how these practices affect the soil profile around forage crops, and a detailed review is found in Chapter 4.1 for the following hypothesis:

Hypothesis for Chapter 4: Using nitrate-selective sensors in soil columns with grass and lucerne will provide valuable data on plant-soil interactions of management practices.

To assess N profile changes we developed nitrate-selective sensors to track NO_3^- movement through the soil profile for use alongside conventional testing, in soil column systems. The effect of management practices on the forage crop *L. perenne* and *M. sativa* was then measured with the use of these sensors.

Chapter 5: Fulvic acid increases vegetative growth in the forage

legume Medicago sativa, and is associated with influencing microbial

<u>activity</u>

The investigation of potential biostimulants is of great interest in current plant science [120]. A potential application of a humic derivative called fulvic acid (FA) could improve forage crop production [121; 122]. Such an improvement could lower N fertiliser inputs. FA is a reasonably uncharacterised mixture of chemicals, and although it has been shown to increase growth parameters in many species [123; 124], the mode of action is unclear. Additionally, many studies lack correct controls in their investigations, as discussed in Chapter 5.1 in more detail for the following hypothesis:

Hypothesis for Chapter 5: Application of the commercially available biostimulant fulvic acid (FA) improves forage crop production.

To test this, commercially available FA was investigated for its potential to increase yields in the forage crops. Experiments in the glasshouse and field, with the goal to increase vegetative biomass in crops were conducted, with care to include nutritional controls so far unseen in the literature.

<u>Chapter 6: Transcriptome analysis shows preferential enrichment of</u> <u>nodulation regulation and signalling-related genes in *Medicago sativa* following fulvic acid application</u>

Following plant and microbial investigations in Chapter 5, there was evidence that FA was affecting nodulation in *M. sativa* (see Chapter 5.3). To aid in establishing if this link was true, *de novo* transcriptome analysis of both shoot and root tissue using RNA-sequencing was undertaken

following three days of FA application. A review of RNA-sequencing as a technique is found in Chapter 6.1 with the following hypothesis for fulvic acid:

Hypothesis for Chapter 6: RNA-seq will provide evidence of early differentially regulated genes in either shoot or root tissue upon fulvic acid application.

Expression analysis was performed for *M. sativa* following FA application to elucidate the growth increases investigated in the previous chapter.

Chapter 7: General discussion

The results of Chapters 3 – 6 will be summarised in regard to current literature and agriculture.

1.3.3 Contributions to thesis

All experiments in this thesis were conducted by me (N.C.) unless otherwise acknowledged. An undergraduate year-in-industry student, Freddie Morrison (F.M.) from Manchester Metropolitan University contributed to this work under both mine and Prof Miller's supervision. He received specialist supervision from Dr Michael Stephenson, Dr Lionel Hill, and Dr Paul Brett (all Metabolic Biology, JIC). Analysis was also performed by technical staff at both University of East Anglia (Norwich, Norfolk) and Rothamsted Research (Harpenden, Hertfordshire); most notably Dr Juan Carlos Muñoz-Garcia performed NMR with supervision from Prof Yaroslav Khimyak (both School of Pharmacy, University of East Anglia).

My industry collaborators BAGCD have supported this project with seed provision and specialist agricultural knowledge. Some sites have also provided sites for field trials. Specifically, Blankney Estates Ltd. (Blankney, Lincolnshire) have not only assisted in field trials through 2018 - 2019, but moreover samples were analysed to industry specifications at their laboratory British Chlorophyll Company Ltd.: technical assistants were Abigail Ewen, Sam Carruthers, and Hollie Compton. Field trials were performed in 2017 at Dengie Crops Ltd. (Southminster, Essex) with help from their agronomist Andrew Spackman (Farmacy Plc., Dorrington,

25

Lincoln), and in 2018 at both Blankney Estates Ltd. and A Poucher and Sons (Bardney Dairies) Ltd. (Market Rasan, Lincolnshire). I was also grateful to receive seed resources from Mr Ianto Thomas and Dr Danny Thorogood from the Genebank at The Institute of Biological, Environmental and Rural Sciences (Aberystwyth University, Aberystwyth, Wales).

Bioinformatic support was also provided from two Postdoctoral specialist onsite; Marker gene analysis from Dr Alexander Calderwood (Crop Genetics, JIC) and RNAseq analysis from Dr Jitender Cheema (Computational and Systems Biology, JIC).

Contributions of all above collaborators are acknowledged in Chapter 2. All experimental work performed by others and incorporated into this thesis is appropriately and fully acknowledged in the legends pertaining to display items (Figures and Tables).

Chapter 2: Materials and methods

TO NOTE

Throughout this chapter most chemicals used were from Sigma-Aldrich®, a subsidiary of Merck KGaA©, Darmstadt, Germany. For ease in the text if a chemical is available from Sigma-Aldrich then only its catalogue number will be provided. If a chemical was sourced elsewhere then this will be clearly indicated within the text.

Any reference to 'dH₂O' stands for sterile deionised water, and 'EtOH' and 'MeOH' are ethanol and methanol respectively. The abbreviations ' μ L' and 'mL' have been used for microlitre and millilitre respectively to avoid any confusion with the number 1.

As outlined in 1.3.3 any contributions from collaborators is credited in the below methods; for clarity Nicola Capstaff is abbreviated to N.C. and Freddie Morrison to F.M.

2.1 N STATUS MARKER GENES

2.1.1 Arabidopsis growth conditions

Arabidopsis thaliana Columbia-0 (Col-0) was provided by Yi Chen (Metabolic Biology, John Innes Centre). Seeds were surface sterilised with EtOH 70 % (v/v) and grown on 100 mm² plates (R & L Slaughter Ltd., Upminster, UK) with the following modified Murashige and Skoog (MS) medium of pH 5.7 and 1 % agar (Formedium[™], AGA03): ½ MS Basal Salt Micronutrient Solution (M0529), 10 % MES buffer (M2933), 1 x Gamborgs vitamins 1000 (G1019), MgSO₄.7H₂O 0.75 mM (M1880), CaCl₂.2H₂O 1.5 mM (223506), KH₂PO₄ 62.5 mM (P9791), with either KNO₃ (P8291) or Ca(NO₃)₂ (C2786) as N source at 0.3, 0.6, 3, 6, 10, 30 mM. Cation influence was negated with molarity balanced with KCl (P5405) or CaCl₂.2H₂O (C3306) respectively. These media plates were then termed 0.3, 0.6, 3, 6, 10, 30 mM KNO₃/Ca(NO₃)₂ treatments. Seeds plates were vernalised for two days at 4 - 6 °C before being transferred standing vertically to a controlled environment room (CER) with temperature controlled at 23 °C and a photoperiod of 16 h light (90 μ mol m⁻² s⁻¹) and 18 h dark. Plants were grown for 3 weeks before tissue RNA extractions.

2.1.2 RNA extraction

20 x shoot samples were flash frozen in liquid nitrogen in 1.5 mL Eppendorf tubes. Samples were stored at -80 °C until required for RNA extractions. Frozen samples were ground using pellet pestles (Z359947). RNA was extracted using 1 mL TRI Reagent (93289) per 100 mg tissue on ice with 150 μ L 1-Bromo-3-chloropropane (B9673) and shaken vigorously for 15 s. Following incubation on ice for 10 mins, to decrease DNA contamination the extraction was centrifuged at 12 000 g at 4 °C for 10 minutes. RNA from the aqueous phase was precipitated with 400 μ L isopropanol (563935) and 400 μ L of High Salt Precipitation solution: 0.8 M sodium citrate (71497) and 1.2 M NaCl (S3014). Samples were vortexed at high speed for 10 s, incubated at room temperature (RT = ~ 23 - 26 °C) and

centrifuged at 12 000 g at 4 °C for 15 mins. The supernatant was discarded, and the pellet washed with 1.5 mL EtOH 70 % (v/v). Samples were air-dried at RT for 5 mins and contaminated DNA removed using RNase-Free DNase Set (QIAGEN Ltd. 79254) according to the manufacturer's instructions.

RNA quality was measured on a NanoDropTM 2000 spectrophotometer (ThermoFisher Scientific, ND-2000) for a A260/A280 ratio of \geq 1.7, and concentrations were recorded. Samples were stored at -80 °C until required for cDNA synthesis.

2.1.3 cDNA synthesis

cDNA was synthesised in 22 µL final volume reactions with 100 - 200 ng mRNA using SuperScript[™] II Reverse Transcriptase (Invitrogen[™] 18064022, Life Technologies Ltd.) performed according to the manufacturer's instructions with oligo-dT, dNTPs and RNaseOUT[™] Recombinant Ribonuclease Inhibitor (Invitrogen[™] 18418012, 10297018 and 10777019 respectively, Life Technologies Ltd.), and provided 5X first-strand buffer (250 mM Tris-HCl (pH 8.3), 375 mM KCl, 15 mM MgCl₂), and 100 mM DTT. Negative controls for qRT-PCR were also carried out with a no template control (NTC) and a no transcriptase control (NRT). cDNA was stored at -20 °C until use.

2.1.4 Candidate gene identification and primer design

A literature search produced a list of potential N status markers in *A. thaliana*, to be tested as genes of interest in samples. Genes were as follows; *CLCa* (AT5G40890), *NIR* (AT2G15620), *GLN1* (AT5G37600), *GLN2* (AT5G35630), *VSP1* (AT5G24780), *VSP2* (AT5G24770), *TIP1.1* (AT2G36830), *TIP1.2* (AT3G26520), *TIP3.1* (AT1G73190), *TIP3.2* (AT1G17810), *NADH-GOGAT* (AT5G53460) and *RBCS2B* (AT5G38420), and reference genes *ACTIN8* (AT1G49240) and *CYP5* (AT2G29960). Primers were either taken from published work, or designed in Primer3 [125; 126] and checked for properties such as hairpin loops in OligoCalc

30

[127]; primers used can be found in Table 2.1. Expression was confirmed in vegetative tissue including leaves and shoot apex using the Arabidopsis eFP Browser 2.0 [128-130] found in Figure A1, Appendix A.

2.1.5 Quantitative RT-PCR

cDNA was diluted 1:50 for use in quantitative RT-PCR (qRT-PCR) with SYBR® Green JumpStart Taq ReadyMix (S4438) in 20 μ L reactions on 96well plate (4titude Ltd., 4TI-0741_50). Each reaction was performed as below on ice as follows: 10 μ L SYBR ReadyMix, 3 μ L cDNA, 5 μ L dH₂O, and 1 μ L of both Forward and Reverse primer (5 μ M).

Biological reactions were aliquoted in technical duplicates or triplicates, with each gene of interest being testing against one or two reference genes depending on plate capacity with number of treatments. NTC and NRT reactions were also run on the plate.

Reactions were carried out using C1000TM CFX96 Touch[™] Real-Time PCR Detection System (Biorad, Hercules, California, USA). The qRT-PCR program was run using CFX Manager (Biorad, Version 3.1.1517.0823) as follows: lid set to 105 °C, 3 min at 95 °C, followed by 40 cycles of 30 s at 94 °C, 30 s at 58 °C, 30 s at 72 °C, followed by one cycle of 30 s at 50 °C, followed by melt curve analysis (65 - 95 °C, with increments of 0.5 °C) with a plate read throughout. Melt curves were checked before analysis, as well as any detection in NTC or NRT wells.

Expression of the genes of interest were calculated using the arithmetic mean Ct according to analysis for $2^{-.\Delta CT}$ method [131]. Single threshold on CFX Manager was used to determine the arithmetic mean for 2 - 3 technical replicates of both gene of interest and a reference gene for biological samples. The $2^{-.\Delta CT}$ was calculated from the difference between these means and fold changes were expressed relative to the lowest concentration NO₃⁻ treatment as 100 %, for either KNO₃ or Ca(NO₃)₂ experiments. Due to expression ranges potentially being affected by cation molarity, the overall gene pattern of expression was calculated using the geometric mean of KNO3 or Ca(NO₃)₂ fold change percentages for treatment mM using Excel (Microsoft® Office 2016). The geometric

31

standard deviation was calculated by finding the exponential value for the standard deviation of the natural logarithms fold change percentages. Graphs were made in GraphPad Prism 7 (GraphPad Software Inc.).

Gene name	Primer name	Sequence (5′ – 3′)	Source
CLCa	AtCLC F	GGTGCCAAGGTTTCACAC	Nicola Capstaff
	AtCLC R	GGCGAGAAGTTTGTAACCGG	-
NIR	AtNIR F	CCGATTTCACCAACTTGCCA	Nicola Capstaff
	AtNIR R	AAGAACGTCATCAGCAGGGA	
RBCS2B	AtRBCS F	ACCCATTTCTATGTGGTCAATGC	Izumi et al.,
	AtRBCS R	TTCACTTTCAAACAATAGTTCCTCAAC	2012 [132]
NADH-GOGAT	AtNADH F	AGTTGGGAGAAGGATGAAACCGGGAGG	Konishi et al.,
	AtNADH R	GTGATAGTGTTGTGTTTCATCTGGTTAAGG	2014 [133]
GLN1	AtGLN1 F	CATCAACCTTAACCTCTCAGACTCCACT	Ishiyama et
	AtGLN1 R	ACTTCAGCTGCAACATCAGGGTTGCTA	al., 2004 [134]
GLN2	AtGLN2 F	TTCTCCAACATGTCAGATGAGAGTGCCT	Ishiyama et
	AtGLN2 R	CCAGGTGCTTGACCGGTACTCGAACCA	al., 2004 [134]
VSP1	AtVSP1 F	CCCGGAGACCTTGCATCTA	Nicola Capstaff
	AtVSP1 R	ACACCACTTGCGTCAACTTC	
VSP2	AtVSP2 F	ACTCCAAAACCGTGTGCAAA	Nicola Capstaff
	AtVSP2 R	GTAAAGATGCAAGGCCTCCG	-
TIP1.1	AtTIP1.1 F	GTGGAATCGCTGGACTCATC	Ma et al., 2004
	AtTIP1.1 R	TGATTCGAAATTACACAAACGG	[135]
TIP1.2	AtTIP1.2 F	AAGCTGGACGTGGACCAAC	Ma et al., 2004
	AtTIP1.2 R	GCCAGAAACCCATTACGATG	[135]
TIP3.1	AtTIP3.1 F	CCCACCGAACCACCTACC	Mao and Sun, - 2015 [136]
	AtTIP3.1 R	GAACAACGAACAAAAGCA	
TIP3.2	AtTIP3.2 F	ACCACAGTACCCACCAAC	Mao and Sun,
	AtTIP3.2 R	ACATAGGAAATGGCAGGA	2015 [136]
CYP5	AtCYP F	CTTCAGAGCTTTGTGCACAGG	Mao and Sun,
	AtCYP R	AAGCTGGGAATGATTCGATG	2015 [136]
ACT7	AtACT F	GGTCGTACAACCGGTATTGT	Mao and Sun,
-	AtACT R	GAAGAGCATACCCCTCGTA	- 2015 [136]

Table 2.1: Primers used in study for Arabidopsis thaliana.

2.1.6 Candidate gene primer design for forages

After qRT-PCR in *A. thaliana*, potential N status marker gene homologs were identified in a range of temperate legumes and monocots, using blastn, blastp [137-140] and PLAZA 3.0 Dicots and Monocots [141]. If published literature had already used these genes in *M. sativa* or *L. perenne* then these were used, otherwise they were designed in this study. Primers were needed that would cover the largest portion of sequence conserved in all species, and thus hopefully successful in the temperate forage crops used in the UK.

Multiple sequence alignments (MSA) were performed using MAFFT v7 [142] for grasses and legumes separately. The E-INS-I iterative refinement method was used as the MSA included < 50 datasets. Phylogenies were built using Newick format in iTOL v3.4.3 [143]. A consensus cds sequence was added with MSAViewer [144] and from this the most conserved sequence was used for primer design. Again these primers were designed in Primer3 [125; 126] and checked in OligoCalc [127]. Annealing of primers were checked in silico [145]. Primers based on the legume and grass consensus can be found in Table 2.2 and 2.3 respectively.

34

<u>Table 2.2:</u> Primers used in study for *Medicago sativa* testing based on legume consensus.

Gene name	Primer name	Sequence (5' – 3')	Source
CLC	LCCLC F	GAGGTTGCAAGTTGGTGGAG	Nicola Capstaff
	LCCLC R	TCCCAAGACACCACCAATGA	
NIR	LCNIR F	TGGCTTGCCCACAAAATTGG	Nicola Capstaff
	LCNIR R	ACTTGATGCATGGGTCTCCG	
RBCS	LCRBCS F	CCCTCTTGGGCATGCAGTAT	Nicola Capstaff
	LCRBCS R	GCCATGAATGCACTGGTTCG	
NADH-GOGAT	LCNADH F	CACTTCCGACGAAAGACCCA	Nicola Capstaff
	LCNADH R	TGGCTACACGACCCTTGTTC	
GLN	LCGLN F	CGGCCAAATGCCTTATCAGC	Nicola Capstaff
	LCGLN R	AGGATCCATTCAGGAGGGGT	
VSP	LCVSP F	TGCCCTTGTTCACCCATTCC	Nicola Capstaff
	LCVSP R	ACGTGAGTGCCTGGAAAACT	
TIP1	LCTIP1 F	CGGAGATGTTAGCACCAACG	Nicola Capstaff
	LCTIP1 R	CACCCAGACACCTTGAAAGC	
СҮР	LCCYP F	CTTGGGAGTGCCACTTTGTG	Nicola Capstaff
	LCCYP R	ATGGCAATGGAATGGGTGGA	
ACT	LCACT F	ATCCCTCACAATTTCCCGCT	Nicola Capstaff
	LCACT R	AAGCTCAGTCCAAGAGGGGT	

<u>Table 2.3:</u> Primers used in study for *Lolium perenne* testing based on grass consensus or published primers.

Gene name	Primer name	Sequence (5′ – 3′)	Source
CLC	GCCLC F	GTCGCCAGGTCGTTGTATTC	Nicola Capstaff
	GCCLC R	GTGCTCCGCCTCTATAACCT	
NIR	GCNIR F	CACATCATGCGGGTCTTCTG	Nicola Capstaff
	GCNIR R	CCGAGGAAATGGAACGTGTG	
RBCS	GCRBCS F	TTCTTGACCTCCTCCAGCTC	Nicola Capstaff
	GCRBCS R	CGAGGGCATCAAGAAGTTCG	
NADH-GOGAT	GCNADH F	ATTCACACGGCGTTGAACAA	Nicola Capstaff
	GCNADH R	GTACCTCGACCACCACTTCA	
GLN	GCGLN F	TGGACTCGGTGCTGTAGTTT	Nicola Capstaff
	GCGLN R	GGCATCAACATCAGTGGCAT	
VSP	GCVSP F	GACCAGCTTGTTGTACAGCC	Nicola Capstaff
	GCVSP R	GTTGGCCACTACATGCTCG	
TIP1.1	GCTIP1 F	GCGGCAACATCAGCCTCCTCA	Nord-Larsen et
	GCTIP1 R	TCATGACGATCTCGAACACC	— al., 2009 [146]
GADPH	LpGADPH F	CAGGACTGGAGAGGTGG	Petersen et al.,
	LpGADPH R	TTCACTCGTTGTCGTACC	2004 [147]
ACT	LpACT F	GAGAAGATGACCCARATC	Petersen et al.,
	LpACT R	CACTTCATGATGGAGTTG	2004 [147]

2.1.7 Forage crop growth conditions

The forage crops to be tested with designed primers were *Lolium perenne* cv. AberMagic and *Medicago sativa* cv. Daisy, found in Tables 2.4 and 2.5 below. As with *A. thaliana* above in 2.1.1, seeds for *L. perenne* were sterilised, plated and vernalised.

Seeds for *M. sativa* were sterilised as follows. Seeds were scarified prior to sterilisation with concentrated H₂SO₄ followed by six dH₂O washes. Seeds were then sterilised with a sodium hypochlorite 10 % (v/v), Triton X-100 0.05 % (v/v) (X100) solution followed by six dH₂O washes. The final wash included Nystatin 5 μ g/mL (N6261), Amoxicillin 50 μ g/mL (A8523), and was filter sterilised to reduce any fungal or bacterial contamination. The seeds were imbibed in this solution on a slow shake (30 ± 1 rpm) for 2 h at 4 °C, then the wash was replaced for a repeat imbibing period. Finally, seeds were washed in dH₂O and plated on the same media as above N treatments.

For cv. Daisy only, plates were inoculated with *Sinorhizobium meliloti* 1021 and *Sinorhizobium medicae* WSM419, provided by Anne Edwards (Metabolic Biology, John Innes Centre), at 7-10 days post germination to encourage nodulation. Both Sinorhizobium inoculums were prepared from single colony growth in 200 mL TY buffer (5 g Tryptone (T7293), 3 g Yeast Extract (Y1625), 1.325 g CaCl₂.2H₂O (223506), in 1 L of dH2O) at 28 °C for 48 hours in orbital shaker. Cultures were centrifuged for 10 minutes at 5,000 rpm at 4 °C and diluted in FP liquid media to an optical density (OD) of ~ 0.02 at 600 nm using DeNovix® DS-11-FX+. FP liquid media was prepared using 2.5 mL CaCl₂.2H₂O 272.18 mM, 3 mL MgSO₄.7H₂O 162.28 mM (M1880), 3.33 mL KH₂PO₄ 220.45 mM (P9791), 3.33 mL Na₂HPO₄.12H₂O 125.5 mM (71649), 2 mL FeC₆H₅O₇ 10.21 mM (F3388), 1 mL Gibson's Trace (2.86 g H₃BO₃ (B6768), 2.03 g MnSO₄.H₂O (M7634), 220 mg ZnSO₄.7H₂O (Z0251), 80 mg CuSO₄.5H₂O (C8027), 80 mg H₂MoO₄ (232084)), in 1 L dH₂O and pH to 6.3 – 6.7. 2.1.8 Candidate gene testing for forages with RT-PCR

RNA from plants grown on 30 mM KNO₃ plates was extracted and cDNA synthesised as in 2.1.2 and 2.1.3. NTC and NRT controls were again carried out.

RT-PCR was used to check primer suitability. Reactions in 40 – 50 μ L were performed with GoTaq G2 DNA Polymerase (Promega UK Ltd. M7845) as follows; 0.75 - 1 μ L cDNA, 1 μ L dNTPs (10 mM), 8 - 10 μ L GoTaq Green 5 x buffer, 0.25 - 0.4 μ L Forward and Reverse Primer 10 μ M and 0.25 μ L GoTaq Polymerase. Primers used can be found in Tables 2.2 and 2.3. Reactions were performed in a G-Storm GS1 thermocycler (Gene Technologies Ltd., UK) using the following programme: 30 s at 98 °C, followed by 40 cycles of 15 s at 98 °C, 30 s at 58 °C, 20 s at 72 °C, and a final extension of 10 min at 72 °C with holding at 4 °C.

2.1.9 Gel electrophoresis and sequencing

Gel electrophoresis was used to determine expected size DNA fragment bands based on primer design. Reactions were ran using 1.5 % agarose gel (Melford Laboratories Ltd., UK, A20080) with 1 % ethidium bromide with standard laboratory procedure. Correct size bands were excised under U.V. and extracted using QIAquick Gel Extraction Kit (QIAGEN Ltd., UK, 28704) following the manufacturer's instructions with a final volume of 70 μ L. PCR products were sent for sequencing using LightRun GATCbiotech service (Eurofins Genomics, UK); reactions were 5 μ L PCR product, 2.5 μ L Forward primer 10 μ M and 12.5 μ L dH₂O. Primers were redesigned to different ranges of consensus sequences based on sequencing data or failure to PCR bands of expected size. Gene primers which were successful in this procedure were tested for their primer efficiencies.

<u>Table 2.4:</u> Forage grass species used in this study.

Lolium perenneAberMagicABY-S562-2016 (Bred at IBERS)Ianto Thomas, IBERSLolium multiflorumDavinciABY-Bb 2593 (Obtained by IBERS)Danny Thorogood, IBERSFestuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestuca pratensisAberPaddockABY-S603-2008 (Bred at IBERS)Danny Thorogood, IBERSL. mulitflorum x F. arundinaceaHykorDLF Forage Seeds, DK Estates Ltd., UKBlankney Estates Ltd., UKL. mulitflorum x F. arundinaceaLofaDLF Forage Seeds, DK DIF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorum x F. arundinaceaLofaDLF Forage Seeds, DK Driers Ltd., UKDanny Thorogood, IBERSL. perenne x F. pratensisMatrixABY-bAF 25 (Obtained by IBERSDanny Thorogood, IBERSL. perenne x F. pratensisMatrixABY-bAF 25 (Obtained by IBERS from Cates Grain Ltd., NZ)Danny Thorogood, IBERSAnthoxanthum aristatumABY-BS 3150-2001 (Collected Spain/Portugal 1963, IBERS)DannyPhleum pratenseABY-Bd 3342 (Collected DannyDanny	Species name	Cultivar	Code/origin	Source
IBERS)IBERSLolium multiflorumDavinciABY-Bb 2593 (Obtained by IBERS)Danny Thorogood, IBERSFestuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestucaAberPaddockABY-S603-2008 (Bred at IBERS)DannyFestucaAberPaddockABY-S603-2008 (Bred at IBERS)DannyFestucaAberPaddockDLF Forage Seeds, DK Estates Ltd., UKBlankneyL. mulitflorumHykorDLF Forage Seeds, DK DLF Forage Seeds, DKBlankneyL. mulitflorumLofaDLF Forage Seeds, DK DLF Forage Seeds, DKDriers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DK IBERSDriers Ltd., UKL. mulitflorumABY-bBF 67-1986 (Line ex IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS) IBERSThorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERS from Cates Grain Itd., NZ)DannyAnthoxanthumABY-Bs 3150-2001 1963, IBERS)Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	Lolium perenne	AberMagic	ABY-S562-2016 (Bred at	Ianto Thomas,
Lolium multiflorumDavinciABY-Bb 2593 (Obtained by IBERS)Danny Thorogood, IBERSFestuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestucaAberPaddockABY-S603-2008 (Bred at IBERS)DannypratensisIBERSThorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKDannyx F. arundinaceaE.J. Lewis 1986, IBERS)Thorogood, IBERSL. mulitflorumKaY-bBF 67-1986 (Line ex E.J. Lewis 1986, IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by Ltd., NZ)DannyAnthoxanthumABY-BS 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny			IBERS)	IBERS
IBERS)Thorogood, IBERSFestuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestucaAberPaddockABY-S603-2008 (Bred at IBERS)DannypratensisIBERSThorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKDiriers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumKaff DLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumKaff DLF Forage Seeds, DKNorthern Crop Driers Ltd., UKBannyL. mulitflorumABY-bBF 67-1986 (Line ex IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERSDannyx F. pratensisIBERS from Cates Grain IBERSThorogood, IBERSAnthoxanthumABY-Bs 3150-2001 1963, IBERS)Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	Lolium multiflorum	Davinci	ABY-Bb 2593 (Obtained by	Danny
Festuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestucaAberPaddockABY-S603-2008 (Bred at IBERS)DannypratensisIBERS)Thorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DK Estates Ltd., UKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DK E.J. Lewis 1986, IBERS)Northern Crop Driers Ltd., UKL. mulitflorumKABY-bBF 67-1986 (Line ex E.J. Lewis 1986, IBERS)Danny Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERS from Cates Grain Itd., NZ)DannyAnthoxanthumABY-Bs 3150-2001 1963, IBERS)Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny			IBERS)	Thorogood,
Festuca arundinaceaKoraDLF Forage SeedsNorthern Crop Dries Ltd., UKFestucaAberPaddockABY-S603-2008 (Bred atDannypratensisIBERS)Thorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankneyx F. arundinaceaLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKDannyx F. arundinaceaE.J. Lewis 1986, IBERS)Thorogood, IBERSL. mulitflorumKBY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates Grain IBERSThorogood, IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal I963, IBERS)Thorogood, Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny				IBERS
Festuca pratensisAberPaddockABY-S603-2008 (Bred at IBERS)DannypratensisIBERS)Thorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumMatrixABY-bBF 67-1986 (Line ex IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS) IBERSThorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by Ltd., NZ)DannyAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	Festuca arundinacea	Kora	DLF Forage Seeds	Northern Crop
FestucaAberPaddockABY-S603-2008 (Bred at DannyDannypratensisIBERSThorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankney Estates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKDannyx F. arundinacea				Dries Ltd., UK
pratensisIBERS)Thorogood, IBERSL. mulitflorumHykorDLF Forage Seeds, DKBlankneyx F. arundinaceaLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Crop Driers Ltd., UKL. mulitflorumKABY-bBF 67-1986 (Line ex L. perenneDannyK F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by Ltd., NZ)DannyAnthoxanthumABY-BS 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	Festuca	AberPaddock	ABY-S603-2008 (Bred at	Danny
L. mulitflorumHykorDLF Forage Seeds, DKBlankneyx F. arundinaceaEstates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Cropx F. arundinaceaVDriers Ltd., UKL. mulitflorumABY-bBF 67-1986 (Line exDannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates Grain Ltd., NZ)Thorogood, IBERSAnthoxanthumABY-bS 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	pratensis		IBERS)	Thorogood,
L. mulitflorumHykorDLF Forage Seeds, DKBlankneyx F. arundinaceaLofaDLF Forage Seeds, DKNorthern Cropx F. arundinaceaDLF Forage Seeds, DKNorthern Cropx F. arundinaceaABY-bBF 67-1986 (Line exDannyL. mulitflorumABY-bBF 67-1986 (Line exDannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates Grain Ltd., NZ)Thorogood, IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny				IBERS
x F. arundinaceaEstates Ltd., UKL. mulitflorumLofaDLF Forage Seeds, DKNorthern Cropx F. arundinaceaABY-bBF 67-1986 (Line exDannyL. mulitflorumABY-bBF 67-1986 (Line exDannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERS from Cates GrainDannyx F. pratensisIBERS from Cates Grain Ltd., NZ)Thorogood, IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	L. mulitflorum	Hykor	DLF Forage Seeds, DK	Blankney
L. mulitflorumLofaDLF Forage Seeds, DKNorthern Cropx F. arundinaceaDriers Ltd., UKL. mulitflorumABY-bBF 67-1986 (Line exDannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates Grain Ltd., NZ)Thorogood, IBERSAnthoxanthumABY-bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	x F. arundinacea			Estates Ltd., UK
x F. arundinaceaDriers Ltd., UKL. mulitflorumABY-bBF 67-1986 (Line ex E.J. Lewis 1986, IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERS from Cates GrainDannyx F. pratensisIBERS from Cates Grain Ltd., NZ)Thorogood, IBERSAnthoxanthumABY-BS 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	L. mulitflorum	Lofa	DLF Forage Seeds, DK	Northern Crop
L. mulitflorumABY-bBF 67-1986 (Line ex E.J. Lewis 1986, IBERS)Dannyx F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained by IBERS from Cates GrainDannyx F. pratensisIBERS from Cates GrainThorogood, IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	x F. arundinacea			Driers Ltd., UK
x F. pratensisE.J. Lewis 1986, IBERS)Thorogood, IBERSL. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates GrainThorogood, IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	L. mulitflorum		ABY-bBF 67-1986 (Line ex	Danny
L. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates GrainThorogood, Ltd., NZ)IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	x F. pratensis		E.J. Lewis 1986, IBERS)	Thorogood,
L. perenneMatrixABY-bAF 25 (Obtained byDannyx F. pratensisIBERS from Cates GrainThorogood,Ltd., NZ)IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood,Phleum pratenseABY-Bd 3342 (CollectedDanny				IBERS
x F. pratensisIBERS from Cates GrainThorogood,Ltd., NZ)IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/PortugalThorogood,1963, IBERS)IBERSIBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	L. perenne	Matrix	ABY-bAF 25 (Obtained by	Danny
Ltd., NZ)IBERSAnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	x F. pratensis		IBERS from Cates Grain	Thorogood,
AnthoxanthumABY-Bs 3150-2001Dannyaristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny			Ltd., NZ)	IBERS
aristatum(Collected Spain/Portugal 1963, IBERS)Thorogood, IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	Anthoxanthum		ABY-Bs 3150-2001	Danny
1963, IBERS)IBERSPhleum pratenseABY-Bd 3342 (CollectedDanny	aristatum		(Collected Spain/Portugal	Thorogood,
Phleum pratenseABY-Bd 3342 (CollectedDanny			1963, IBERS)	IBERS
	Phleum pratense		ABY-Bd 3342 (Collected	Danny
Taiwan/Japan 2017, IBERS) Thorogood,			Taiwan/Japan 2017, IBERS)	Thorogood,
IBERS				IBERS

Species name	Cultivar	Code/origin	Source
Medicago sativa	Daisy	DLF Forage Seeds, DK	Dengie Crops
			Ltd., UK
Medicago sativa	Luzelle	Oliver Seeds Ltd. (Bred by	Fox Feeds Ltd.,
		INRA/Agri-Obtentions, FR,	UK
		1993)	
Medicago sativa	Gea	DLF Forage Seeds, DK	A Pouchers and
			Sons Ltd., UK
Lotus corniculatus		ABY-AI 592-2013 (Breeder's	Danny
		Line Highgrove x Lotar,	Thorogood,
		IBERS)	IBERS
Onobrychis sativa	Aberystwyth	ABY-Am 361 (Bred at	Danny
	Sanfoin	IBERS)	Thorogood,
			IBERS
Trifolium pratense	AberRuby	ABY-S543-2005 (Bred at	Danny
		IBERS from Sabtoron	Thorogood,
		(S310))	IBERS
Trifolium repens	AberConcord	ABY-S505-1997 (Bred at	Danny
		IBERS)	Thorogood,
			IBERS

<u>Table 2.5:</u> Forage legume species used in this study.

2.1.10 Primer efficiency testing

Marker gene primers in Tables 2.2 and 2.3 were tested in *L. perenne* and *M. sativa* for their primer efficiencies using SYBR® Green JumpStartTM Taq ReadyMixTM (S4438) with either concentrated cDNA or 1:10, 1:100 or 1:1000 dilutions in quadruplicate; 10 µL 2 x SYBR® ReadyMixTM; 5 µL cDNA, 3 µL dH₂O, 1/1 µL F/R primers 5 µM. qPCR run as above. Efficiencies were calculated for each marker gene primer set by taking an average of the individual cDNA quadruplets and plotting against the common logarithm of cDNA concentration. The linear regression of the slope was calculated, and the slope gradient value was used in the following equation to determine the % efficiency:

$$\% = (10^{1/slope gradient value}) \div 2 \times 100$$

Efficiencies that were between 90 – 110 % were designated fit to use in future gene expression assays, but those that failed were redesigned and tested as in 2.1.6 for a slightly shifted portion of consensus sequence. Those with efficiency pass scores were used in a larger forage crop screen. Primers to test across all forages can be found in Tables 2.6 and 2.7 below.

<u>Table 2.6:</u> Primers used in study for legumes based on *Medicago sativa* testing.

Gene name	Primer name	Sequence (5′ – 3′)	Source
CLC	MsCLC F	GGTTATTAGTTTCCGCGGCA	Nicola Capstaff
	MsCLC R	CTCAGCCAACTCAACCCAAC	
NIR	MsNIR F	TAGTTCGCCGGTTCCGTATT	Nicola Capstaff
	MsNIR R	TGCCAGAGAAGAAGCTAGAGAG	
RBCS	MsRBCS F	GGACATGCAAGATGGAACTCC	Nicola Capstaff
	MsRBCS R	AGGACAGCAGAGACTCGAAC	
NADH-GOGAT	MsNADH F	GTCGCCAAATTGCCTCTACA	Nicola Capstaff
	MsNADH R	TGCAGTCCACCAACATCTAGT	
GLN	MsGLN F	TGTCTTTCTGCAACAAGGTGT	Nicola Capstaff
	MsGLN R	CCCACCAACAAGAGACATGC	
VSP	MsVSP F	GAAACACAAAGCCAAAACCACA	Nicola Capstaff
	MsVSP R	ACCTTCTTCCATTACCATTTCCA	
TIP1	MsTIP1 F	CGGAGATGTTAGCACCAACG	Nicola Capstaff
	MsTIP1 R	CACCCAGACACCTTGAAAGC	
CYP20	MsCYP F	GCTACCTTTGTAATGAAGAGGCT	Nicola Capstaff
	MsCYP R	TGCAGGTCGTATTGTAATAGGTC	
ACT2	MsACT F	CGCCGGAATCCAACACAATA	Nicola Capstaff
	MsACT R	GAGGCTCCACTTAACCCAAAG	_

<u>Table 2.7:</u> Primers used in study for grasses based on *Lolium perenne* testing.

Gene name	Primer name	Sequence (5′ – 3′)	Source
CLC	LpCLC F	AGTTCTGGGTGCCGTACTAC	Nicola Capstaff
	LpCLC R	ACAGAAGAGAAGACGAGGGC	
NIR	LpNIR F	GTTTTGCCGTCCTTCTCCG	Nicola Capstaff
	LpNIR R	CCGAGGAAATGGAACGTGTG	
RBCS	LpRBCS F	CAGTACCTGCCGTCGTAGTA	Nicola Capstaff
	LpRBCS R	CGAGGGCATCAAGAAGTTCG	
NADH-GOGAT	LpNADH F	CCTCTGTTTGCGTCCGTTAG	Nicola Capstaff
	LpNADH R	GTACCTCGACCACCACTTCA	
GLN	LpGLN F	GTTGGCCCTTCTGTTGGTAT	Nicola Capstaff
	LpGLN R	TGGACTCGGTGCTGTAGTTT	
VSP	LpVSP F	AGGACCCAGATTTGAAGGAGA	Nicola Capstaff
	LpVSP R	GCCACTACATGCTCGGATTC	
TIP1.1	LpTIP1 F	GCGGCAACATCAGCCTCCTCA	Nord-Larsen et
	LpTIP1 R	TCATGACGATCTCGAACACC	— al., 2009 [146]
GADPH	LpGADPH F	CAGGACTGGAGAGGTGG	Petersen et al.,
	LpGADPH R	TTCACTCGTTGTCGTACC	2004 [147]
ACT	LpACT F	GAGAAGATGACCCARATC	Petersen et al.,
	LpACT R	CACTTCATGATGGAGTTG	2004 [147]

2.1.11 Larger forage screen of developed N status marker primers

To test the viability of consensus primers a larger forage screen was used with the species above in Tables 2.4 and 2.5. Seeds were surface sterilised with EtOH and plated onto water agar (3 g Agar (AGA03, Formedium Ltd. Norfolk, UK) in 200 mL dH₂O); ~ 20 seeds for each cultivar. Seeds plates were vernalised for four days at 4 - 6 °C before being transferred upside-down to a CER with temperature controlled at 23 °C and a photoperiod of 16 h light (90 μ mol m⁻² s⁻¹) and 18 h dark. Plants were germinated for 4 days on plates before being transferred to glasshouse compost pots. Ten seedlings were transplanted to 1 L pots containing a peat mix; 600 L Levington F2 peat, 100 L 4 mm grit, 196 g Exemptor® (GB84080896A, Bayer CropScience Ltd., UK) which is a chloronicotinyl insecticide.

After 3 weeks of growth in pots, the seedlings were flash frozen in liquid $N_{2.}$ RNA extraction, cDNA synthesis, and RT-PCR were carried out with samples as in 2.1.2, 2.1.3 and 2.1.8 respectively with primers from Tables 2.6 and 2.7. PCR products were tested using gel electrophoresis as above to confirm expected band sizes. Percentages of correct bands for each primer was calculated for grasses and legumes. Gel images with percentage calculations can be found in Figure A2, Appendix A.

2.1.12 Developed N status marker primers forage crop qRT-PCR

Gene expression using developed primers were tested in the glasshouse with the grass species *L. perenne* cv. AberMagic and *Festulolium* cv. Hykor and the legume *M. sativa* cv. Daisy and Luzelle. Seeds were sterilised and vernalised as in 2.1.6 and germinated on water agar plates for 3 days. Seedlings were then moved to a glasshouse with 16 h light/8 h dark, 18/15 °C, relative humidity 70 %, and lit with 600 w HPS lamps and planted in p26 seed trays of medium vermiculite and 3 mm sharp sand 50:50 mix. For *M. sativa* cultivars individual plants were inoculated with 0.5 mL of

both *Sinorhizobium meliloti* 1021 and *Sinorhizobium medicae* WSM419 at 7 - 10 days post germination as described in 2.1.7. Plants were watered for 1 week with dH₂O, then subsequently watered with 50 mL treatments of 0.6, 3, 6 mM KNO₃/Ca(NO₃)₂ in the case of *M. sativa* and with treatments of 0.3, 0.6, 3, 6, 10, 30 mM KNO₃/Ca(NO₃)₂ for grasses, as in 2.1.1. Three weeks after treatment the top 6 cm vegetative tissue was cut from 12 plants for RNA extraction and qRT-PCR as in 2.1.2, 2.1.3, 2.1.5, with primers from Tables 2.6 and 2.7. Another 12 plants were oven dried at 50 – 65 °C overnight and biomass measured. Plants were grown for another 3 weeks with dH₂O watering. At the six weeks after treatment 24 plants were cut to the above-ground base and again dried for measured for final biomass.

2.1.13 2018 field sampling of developed N status marker primers

In early May 2018 field sampling was carried out at Blankney Estates Ltd. Fields tested were in the Northwest region of Lincolnshire, England and located within 3 km of the site office (53°06'22.2"N 0°27'48.4"W, 45 m above sea level). Site specific climate data during the last decade was mean annual max/min temperature 13.8/4.9 °C and mean annual rainfall 577 mm. Soil is sandy clay with crop drying fields (BCDs) containing the following in their residual 0 – 45 cm depth topsoil before any fertiliser application for years 2011 – 2018: ammonium 0.56 – 1.6 mg/kg; nitrate 1.72 – 6.04 mg/kg; phosphorus 12.49 – 18.18 mg/kg; potassium 107.66 – 151.38 mg/kg; magnesium 45.0 – 85.42 mg/kg; pH 7.45 – 8.28 (Blankney site office data provided by Andrew Hayden).

The fields sampled are indicated in Figure 2.1 below. Crops grown in fields were either *M. sativa* cv. Asmara or cv. Daisy, or *Festulolium* cv. Hykor (*Lolium perenne x Festuca pratensis*). Sample points are indicated in the figure as estimated 'dots', the first 5 m from corner of field, and the second 5 m from this to approximate centre of field, and third 5 m again from that. Samples taken were as follows:

- Soil samples ~ 20 g of 1 10 cm topsoil, chosen due to practical ease of sampling and borer available, from pooled sample points to EtOH cleaned glass vials, transported at ambient temperature, stored at -20 °C within 6 hours of sampling.
- Vegetative gene expression samples cut ~ 15 g of top 6 cm of vegetative tissue from each individual sample point, flash frozen and transported in waxed bag liquid N₂ on-site, stored at -80 °C within 6 hours of sampling.
- Vegetative tissue nutritional samples collection of ~ 15 g fresh weight was taken at each sample point, placed in a waxed bag for chlorophyll and protein analysis at BCC Ltd., and stored at -20 °C within 2 hours of sampling.

Analysis of these samples was as follows:

- Soil samples KCl extraction to determine nitrate concentration of topsoil, described in 2.1.14 and 2.1.15.
- Vegetative gene expression samples RNA extraction, cDNA synthesis and qRT-PCR as in 2.1.2, 2.1.3 and 2.1.5. For qRT-PCR an average of all sample points expression was calculated, and fold changes were relative to absolute topsoil nitrate values.
- Vegetative nutritional samples protein content analysed using Kjeldahl method as in 2.1.16, chlorophyll content analysed using Soxhlet method as in 2.1.17.

Future yield measurements for the subsequent cut of field in T ha⁻¹, \sim 6 - 21 days post sampling, were provided by Blankney Estates Ltd.



Figure 2.1: Fields sampled at Blankney Estates Ltd. in May 2018.

Crop grown is indicated with coloured box; red is *Medicago sativa* cv. Asmara, brown is *Medicago sativa* cv. Daisy, and purple is *Festulolium* cv. Hykor (*Lolium perenne x Festuca pratensis*). Dots indicate area where soil and vegetative samples were taken for analysis.

2.1.14 Soil extraction with KCl

Soil stored at -20 °C was defrosted to RT and sieved to 5.6 mm. A mixed sample of 10 \pm 0.02 g soil was transferred to a 125 ml Wide Neck bottle (AzlonTM HDPE Wide Neck Round Bottles, AzlonTM BLH0125P) and 50 mL KCl 2.0 M extraction solution added. A blank extraction with no soil was also carried out. This was shaken on an orbital shaker at 5 Hz for 2 hours \pm 10 minutes, and then filtered through a Whatman No.4 filter paper (150 mm diameter, WHA1204320) and filtrate retained. Filtrate was used to determine nitrate concentration as below in 2.1.15.

2.1.15 Spectrophotometric determination of nitrate in soil extractions

A standard nitrate solution of 50 μ g/mL was prepared using KNO₃ in dH₂O. From this, solutions were prepared in 10 mL KCl 2.0 M extraction solution to contain 0, 0.2, 0.4, 0.6, 0.8, 1.0, 1.2, 1.4, 1.6, 1.8, and 2.0 μ g/mL as working standards for calibration regression.

A reduction-diazotisation reagent was prepared within a fume hood by adding the following 'solution a' to 'solution b'; 'solution a' was 400 mg vanadium chloride (208272) to 50 mL HCl 1.0 M shaken gently until dissolved; 'solution b' was 200 mg of sulphanilamide \geq 99 % (S9251) and 10 mg NEDD \geq 98 % (N-(1-naphthyl)ethylenediamine dihydrochloride, 33461, Supelco®) in 400 mL dH₂O.

To standard 3.5 mL cuvettes (Sarstedt Limited, 67.755) 1 mL of either working standard or soil extracted filtrate was added, then 800 μ L of reduction-diazotisation reagent. After 20 hours at RT absorbance was measured at 540 nm. Working standard readings were used to fit a calibration regression, with R² of \geq 0.98. The linear equation was used to determine soil extracted filtrate concentrations, with blank reading subtracted and multiplied by both the dilution factor of KCI solution (5) and the dilution factor of soil sampled (10). This provided the nitrate concentration in mg/kg of the original soil.

2.1.16 Protein analysis using Kjeldahl method

Protein analysis using the Kjeldahl method was carried out at British Chlorophyll Company Limited (Blankney Estates, Navenby). Vegetative tissue from field trials was analysed to determine N content through conversion to NH₃, and therefore calculate protein content. Frozen samples were ground at RT, and ~ 1 g measured on a Whatman No.1 filter paper (125 mm diameter, WHA1201331). This was folded into a tight parcel and reweighed. The sample parcel was added to a 500 mL round bottom Kjeldahl flask for digestion of organically bonded N into NH₄⁺. A catalytic salt mix was added to aid digestion; 35 g of 50:1 sodium sulphate:cupric sulphate in each flask. In a fume cupboard 30 mL H₂SO₄ 95% (GPR grade) was added, swirled gently, and an adapter placed in the flask neck. The flask was heated with a Bunsen burner, gently at first and increasing up to 350 – 380 °C. Digestion took \sim 30 – 60 mins and was deemed complete when the solution turned clear or pale blue with no brown material left. Once colour change was complete, the flask was removed from the fume cupboard to cool to RT, but not allowed to solidify.

When cooled the flask adaptor was removed for distillation and ~ 25 mL dH₂O added to ensure solution didn't set, swirling to dissolve all digested material. Anti-bumping crystals were added to each flask, and the neck of the flask was greased with petroleum jelly to ensure a seal when connecting the flask to the distillation unit. When all joints were sealed the condenser spout was placed above a receiving vessel containing a trap solution; 3 mL hydrochloric acid 1 M (Analar Grade) with a few drops of Methyl red indicator and 150 mL dH₂O in a 500 mL flat bottom round flask. Ensuring the reservoir at the top of the distillation unit was closed, 75 mL caustic soda solution 33 % was added slowly, opening the reservoir to gently add the caustic solution until there was a colour change. Once added the reservoir was closed and water ran through the condenser. The flask was gradually heated again with a Bunsen burner, firstly to the side of the flask then slowly moved to the centre of the flask once the solution began to bubble gently. Heating continued until the mixture started 'banging', then heat was removed. The condenser was disconnected to prevent any of the

heated mixture passing through the distillation unit and contaminating the trap solution, which now contained N converted to NH₃.

The trap solution was titrated with sodium hydroxide 0.1 M until the solution turned from pale pink to yellow. The burette reading prior and after titration was recorded and used to calculate protein as follows:

 $\frac{Titration \, difference \, (mL)}{Weight \, of \, ground \, sample \, (g)} \times 100 = Protein \, in \, sample \, \%$

The protein content (%) could be used to calculate total dry weight % of protein in tissue using the moisture % of original tissue:

 $\frac{Protein in sample \%}{(100 - moisture \%)} \times 100 = Total dry weight protein (\%)$

2.1.17 Chlorophyll analysis using Soxhlet extraction

Chlorophyll analysis using the Soxhlet extraction method was carried out at British Chlorophyll Company Limited (Blankney Estates, Navenby). Vegetative tissue from field trials was analysed to determine crude chlorophyll content. Ground samples from protein analysis in 2.3.10 were used, and ~ 5 g measured on a Whatman No.4 filter paper (150 mm diameter). The paper was folded, and sample parcel fitted into an extraction thimble. This was plugged with cotton wool and placed into a Soxhlet extraction apparatus. This was connected to a 250 mL flat bottom flask containing anti-bumping crystals, with weight recorded. ~ 100 mL of DCM was added to the apparatus, and water ran through the condenser, with heating from a Bunsen burner. Extraction was performed until there was no green colour in the liquid surrounding the thimble. Once extracted, the thimble was removed, and the DCM distilled from flask. Once distilled, the flask was placed in an oven set at 105 °C for 20 minutes. Flask was then removed and cooled to RT. The flask was weighed and recorded, and crude chlorophyll calculated:

 $\frac{Final \ flask \ weight \ (g) - initial \ sample \ weight \ (g)}{Initial \ flask \ weight \ (g)} \times 100 = Crude \ chlorohpyll \ \%$

The crude chlorophyll content (%) could be used to calculate total dry weight % of chlorophyll in tissue using the moisture % of original tissue:
$\frac{Crude \ chlorophyll \ in \ sample \ \%}{(100 - moisture \ \%)} \times 100 = Total \ dry \ weight \ chlorophyll \ (\%)$

2.1.18 Further qPCR of developed N status marker primers in *Festulolium*

Following the initial testing of *Festulolium* cv. Hykor in 2.1.10, and its extensive use on collaborator sites, gene expression was tested with more NO_3^- treatments. Seeds were sterilised, vernalised and planted as above, with plants again watered for 1 week with dH₂O. Treatments were then applied with the following NO_3^- mM concentrations: 1, 5, 8, 12, 15, 17, 19, 21, 23, 26, 29, 32, 35, and 38 mM. To avoid problems with cation balancing for KNO₃ and Ca(NO_3)₂ in 2.1.1 with higher concentration solutions, the treatments were prepared with both nitrate sources in a 1:1 NO_3^- ratio. qRT-PCR was performed with N status marker primers for *NIR*, *NADH-GOGAT*, *RBCS* and *TIP1* as before in 2.1.12.

2.1.19 2019 field sampling of developed N status marker primers

Sampling was again carried out in 2019 at Blankney Estates Ltd. as in 2018 detailed in 2.1.13, this time only for fields cultivating *Festulolium* cv. Hykor. Figure 2.2 below shows the fields sampled. Sampling and analysis were as before.



Figure 2.2: Fields sampled at Blankney Estates Ltd. in April 2019.

Crop grown is indicated with coloured box; purple is *Festulolium* cv. Hykor (*Lolium perenne x Festuca pratensis*). Dots indicate area where soil and vegetative samples were taken for analysis.

2.1.20 Modelling N status marker gene expression

Expression data from glasshouse experiments was investigated to produce a potential model for field analysis. The hypothesis was that expression data could be used to assess either soil nitrate or future yield. The idea to use a Gaussian process approach to the data as well as the initial R script [148] was written by Dr Alexander Calderwood (Crop Genetics, JIC) after discussing the aims of the project with N.C.; initial script included assessing all gene expressions and investigating both soil nitrate and future yield. The script was then edited by N.C. for soil nitrate only as future yield was too variable in the field. Moreover only the genes *NADH-GOGAT*, *NIR* and *TIP1* were included as *RBCS* was also too variable for analysis. This R script for analysing 2018 and 2019 data combined can be found in Table A1, Appendix A. The whole script used common packages installr [149], tidyverse [150], cowplot [151], ggplot2 [152], and data.table [153] to build and plot data.

Simply, the R script used the normalised geomean 2^{ACT} of 2018 glasshouse and field Festulolium NADH-GOGAT, NIR and TIP1 expression data from 2.1.12 and 2.1.13, to generate values for 'mean.delta.ct.norm'; values for 0.3 mM were ignored due to the extreme outlier of TIP1 expression. Glasshouse values are plotted against soil 'NO3' shown in Figure A4, Appendix A, with all 2018 graphs given a red border. The glasshouse values were then used to build a Gaussian model using the package GauPro [154] with the known minimum and maximum values for soil NO_{3⁻} specified. This required values for 'yhat' or the mean of all gene expression values for 'mean.delta.ct.norm', and 'xpred' as predicted soil NO₃⁻, again shown for each gene in Figure A5, Appendix A. The field 'mean.delta.ct.norm' values were then fed into the model to give 'yhat' and the 'xpred' predicted soil NO_3^- calculated based on each gene, as Figure A6, Appendix A. All the 'yhat' values for individual genes were then calculated using the model all together as the series P(D|NO3)' and a plot created against 'predicted equivalent greenhouse NO3', provided in Figure A7, Appendix A. The plot was then compared to actual field extracted NO₃⁻

from sampling to see if there was agreement in the model and the actual values, of which there was for most fields.

All of the above was repeated for 2019 glasshouse and field data from 2.1.18 and 2.1.19 respectively; all graphs can be found in Figures A8, A9, A10, and A11 Appendix A with all 2019 graphs given a blue border.

Finally the R script was run for all samples, with models built using 2018 and 2019 glasshouse data and used to assess 2018 and 2019 field data. Again this was compared to actual extracted soil NO_3^- values, with plots available in Chapter 3 and results discussed.

2.2 NITRATE AVAILABILITY MONITORING USING SOIL SENSORS

2.2.1 Nitrate-selective sensor construction

Nitrate-selective sensors were constructed executing patent for Soil Chemistry Sensors Pub. No. WO/2014/096844 [155] with the following method. A general schematic can be found in Appendix B (Figure B1).

Boxes of 1.25 mL graduated tips (TipOne® S1112-1830, STARLAB) were silanized to prevent membrane leakage by dipping the tip to a depth of ~ 1 cm with Repelcote[™] (Dow Corning® Repelcote VS water repellent, a 10% emulsion of a polydimethylsiloxane fluid, silica filler and non-ionic emulsifier). These were re-racked into original box and dried overnight (o/n) in a fume hood at RT. Two membrane solutions were prepared in sterilised 2 mL glass vials cleaned with 70 % EtOH, as follows;

- Ion-selective membrane solution containing 12 mg tridodecylmethylammonium nitrate (Selectophore[™] ≥ 99.0 %, 91664), 2 mg methyltriphenylphosphonium bromide (98 %, 130079), 46 mg poly(vinyl chloride) (high molecular weight, 81392), 10 mg nitrocellulose (Amersham Hybond ECL, RPN2020D, 0.45 µM, 200 x 200 mm, GE Healthcare), and 130 mg 2-nitrophenyl octyl ether (99 %, 73732) in 2 mL final volume of tetrahydrofuran solvent (EMD Millipore, ≥ 99.9 % 1081100500).
- Reference membrane solution containing 2 mg potassium tetrakis(4-chlorophenyl)borate (Selectophore[™] ≥ 98.0 %, 60591), 45 mg polyethylene glycol 3500 (1546547), and 10 mg nitrocellulose in 2 mL final volume of tetrahydrofuran solvent.

Both solutions were covered with aluminium foil then capped and sealed with parafilm to avoid evaporation of solvent. Solutions were shaken at \sim 150 rpm using an orbital shaker o/n at RT to dissolve reagents thoroughly. The salinized tips were dipped into either ion-selective or reference membrane solution to a depths of \sim 2 cm and left to dry in a fume hood for 48 hours at RT. This left a membrane in the tip of \sim 2 - 3 mm. Two backfill solutions were prepared in 200 mL Duran flasks;

- Ion-selective backfill solution containing 2.202 g potassium nitrate (ReagentPlus[®] ≥ 99.0 %, P8394), and 1.49 g potassium chloride (BioXtra ≥ 99.0 %, P9333) in 200 mL dH₂O.
- Reference backfill solution containing 3.12 M potassium chloride, 20 mg silver chloride (99.999 % trace metals basis, 204382), 1.8 g sodium chloride, and 0.18 g naphthol green B (Technical grade, N7257) in 200 mL dH₂O (note that the naphthol green B dye is only used to colour the solution).

One mL of the corresponding backfill solution was loaded into the top of the corresponding membraned tip. Air bubbles were displaced with gentle flicking of the tips. Backfilled tips were stored in their original rack box half-filled with 100 mM potassium nitrate at RT o/n or until future use.

Sensor wires were prepared by stripping ~ 1 cm of plastic coating from 1.5 m lengths of wire. Black and brown sensor wires were used for ion-selective and reference tips respectively (RS Components Ltd., 192-3998 and 400-0199 respectively). At one end of the sensor wire ~ 7 cm of silver wire 99.9 % 0.7 mm diameter (Palmer Metals, FS-RW070) was clamped to each sensor wire. The silver end of the sensor wires was coated in 50 mM potassium chloride and then threaded through an earplug (RS Components Ltd. 771-4894) to secure into tips on construction. Sensor wires were placed into the top of individual membraned and backfilled tips corresponding to their sensor type using a disposable needle to allow gas to escape. These were secured with black cable ties (RS Components Ltd. 233-455). Finished sensors were sealed/secured depending on future use; sensors to be used in the field would be glued with and those for columns were wrapped in 2 x 5 cm strips of parafilm 'M' 100 mm x 38 m (Slaughter Ltd., 291-0057).

2.2.2 Nitrate ion-selective sensor calibration

Pairs of each sensor type were formed into sensor sets and connected to a GP2 logger (Delta-T Devices Ltd., Cambridge, UK); within sets ionselective sensors were (+) channels and corresponding reference sensor were '-'. All available channel pairs were used in a logger (n=12) unless a Delta-T SM300 soil moisture and temperature sensor was also connected (which reduces channels available to 10).

A DeltaLINK 3.6.2 (Delta-T Devices Ltd.) mM programme was installed on the logger for calibrations; 'Voltage, not powered' and circuit detection and power channel disabled. Sensors were placed into solutions of 300, 30, 3 and 0.3 mM KNO₃ sequentially whilst the programme was running, for at least 5 min each. This measured the electrical potential of sensors in each concentration and recorded them in mV. Data was captured and an average across one-minute period was calculated. This was fitted in a linear regression for each sensor alongside the known $log_{10}(x) NO_3^$ concentration using Excel® 2016 (Microsoft®). This provided a calibration equation for each sensor set:

$$mV = (m \times \log_{10} NO_3^- mM) + c$$

Any sensors with a slope factor 'm' not between 46 - 64 mV were considered not viable for use. Nonviable sensors were reconstructed, usually with only a prepared backfilled tip change required. Viable sensors were stored in 100 mM KNO₃ until use.

2.2.3 Sensor running and data analysis

Sensors were placed in the soil in the field or columns with care taken to limit disturbance to tip membranes. The output voltage for each sensor set was recorded with the same programme as calibrations but at 1 - 30 minutes intervals, depending on resolution desired for the experimental set-up. During experiments, sensors which showed signs of leaking or whose signal behaved erratically were monitored. At the end of an experiment all sensors were recalibrated as in 2.2.2, and those which did not fit into the above criteria were removed from the data analysis.

The data was analysed with a laboratory temperature slope coefficient included to compensate for temperature changes derived from the Nicolsky-Eisenmann relationship [156]; experimental mV is compared to

theoretical calculated mV across a range of temperatures to formulate a linear coefficient of compensation:

Temperature compensation = $(0.4045 \times {}^{\circ}C) + 93.61$

After a few days of settling when the sensors were monitored for stability and viability, a 'resting period' was initiated; about 1 week of programme running with only dH₂O watering of the soil every 2 - 4 days.

The arithmetic mean 'mM' value for 12 hourly periods were calculated using calibration equations and temperature compensation in Excel, at 0000 hours and 1200 hours. These were plotted against time in GraphPad Prism 7 (GraphPad Software Inc.), with soil temperature and moisture measurements plotted at 10-minute intervals. Datasets were analysed using repeated measurement ANOVA in RStudio to determine statistical significance between columns at different depths across time.

Although all columns should have the same starting residual nitrate in soil before the experiment began as they came from the same field site, if there was a significant difference then the delta mM (Δ mM) could be calculated; the mean value for the 0 - 6 day 'resting period' could be subtracted from both 'resting period' values and subsequent 12 hourly 'running period' values. Furthermore, the dH₂O watering 'resting period' helped to negate any differences between columns.

2.2.4 Column experiments of Lolium perenne

monocropping

Silty clay soil was sourced (Church Farm, John Innes Centre, 52°37'59.8836" N 1°10'46.3440"E) and placed into 4 plastic opaque columns (height = 50 cm, inside diameter = 15.4 cm) with 5 drainage holes at base; columns were made by John Humble, John Innes Centre Workshop. Holes for sensors we made at three levels, top (1 cm depth), middle (25 cm depth) and bottom (49 cm depth). Mini suction lysimeters (10 Rhizon SMS, Rhizosphere Research Products B.V., Wageningen, The Netherlands) could also be used to collect soil water samples from the base drainage holes, for conventional soil water analysis. A schematic and

photograph of the column setup can be found in Chapter 4 as Figure 4.2 and Appendix B as Figure B2 respectively. Columns were placed in a glasshouse with 16 h light/8 h dark, 18/15 °C, relative humidity 70 %, and lit with 600 w HPS lamps.

Water holding capacity of each soil column was determined and watered with dH_2O every 2 - 4 days to a similar water capacity to allow for nutrient movement through a soil profile. One to three NO_3^- soil sensors were placed at each of the three levels in all four columns and recorded NO_3^- concentrations as described above in 2.2.3.

At the end of the 'resting period' (6 - 12 days) the experiment began with watering of either H₂O or KNO₃ and planting of *L. perenne* cv. AberMagic seedlings in 3 of 4 columns. An experimental design table is found in Chapter 4 as Table 4.1. KNO₃ treatments were equivalent to 57 kg ha⁻¹ as the standard in UK forage agriculture [157], which is 10.76 g KNO₃ in 1 L for each column. Seeds were surface sterilised with EtOH 70 % (v/v) and then germinated in a 10 cm round petri dish for 6 days before transplanting into columns at a seeding rate of 43.68 kg ha⁻¹, ~ 400 seedlings per column factoring in germination rate.

Columns were watered every 2 - 4 days. After 4 weeks all the plants were cut to their vegetative base to simulate cropping in one column, and the tissue was oven dried overnight and biomass recorded. At 8 weeks all three planted columns were cut to the base and again the tissue was dried overnight, and the biomass recorded. The whole experiment was repeated four times.

For one experiment the soil water was collected from the base drainage holes. Water was collected every 1 - 4 days. This was analysed with the spectrophotometric method described in 2.1.15 for soil extractions, but with 1 µL of sample.

2.2.5 Column experiments of *Lolium perenne* and *Medicago sativa* intercropping

Column experiments were repeated with *L. perenne* and *M. sativa* intercropping. Four opaque columns were again used, with the

experimental design for the set-up found in Chapter 4 as Table 4.2. Nitrate treatment was as before, this time for only 2 of the 4 columns. Seeds were sterilised as above, with a seeding rate of 80:20 grass:legume; 0.66 g of *L. perenne* cv. AberMagic and 0.08 g *M. sativa* cv. Daisy to 3 columns. This planting density simulated that used by the UK forage industry.

After 4 weeks, the plants were cut to the base in two columns and the biomass measured. At 8 weeks, the plants were cut in all 3 columns and the biomass measured. The whole experiment was repeated twice, due to time constraints of project.

For one experiment, the soil water was again collected for conventional soil water analysis from the base drainage holes. Again, as with monocrop columns, this was taken every 1 - 4 days, and stored for analysis at -20 °C. Testing was performed as in 2.1.15, using a 1 µL of sample.

2.3 FULVIC ACID TESTING OF FORAGE CROPS

2.3.1 Fulvic acid materials

Two fulvic acid materials (FAs) were acquired, VitaLink Fulvic (sourced from Holland Hydroponics & Horticulture, UK [158]) and MPXA (F.A.R.M. Co., California, USA [159]). These were called VFA and MFA for all subsequent work. Stock solutions were used for chemical analyses, and diluted solutions based on company guidelines were used in plant applications. VFA stock solution was the solution as purchased, with a 1 % dilution in dH₂O used in applications, with pH 6.0. An MFA stock was made with ~ 1 g MPXA in 1 L dH₂O and a 0.5 % dilution of this was used in applications, with pH 4.8. A year in industry student, Freddie Morrison (F.M.) performed some of the experiments with my supervision and guidance from other experts as indicated.

2.3.2 Initial test of FAs on forage crop vegetative

biomass

An initial screen of 17 species of forage grasses and legumes was carried out with autoclaved treatments of either dH_2O , 1 % VFA or 0.5 % MFA. Information of species tested can be found in the Tables 2.4 and 2.5.

Seeds of all cultivars were sterilised, plated and vernalised as in 2.1.7, with ~ 200 seeds for grasses and ~ 100 seeds for legumes. Seedlings were grown for six days. Similar sized seedlings were transferred to 1 L pots filled with Church Farm soil (see 2.2.4) in triplicate; there were 40 grass seedlings established per pot (apart from *Phleum pratense* where 60 seedlings were used) or 20 legume seedlings per pot (apart from *Onobrychis sativa* and *Medicago sativa* cv. Gea where 10 seedlings were used due to poorer germination rate). These planting densities simulated those used by the UK forage industry. The day after transfer to soil, pots were treated with 10 mL if either dH₂O, VFA or MFA. This liquid application was made evenly over the soil surface. Seedlings were grown in glasshouse

conditions for 21 days, with watering every 3 – 5 days. Any weeds growing in pots were recorded on day 7, 12, and 17 and designated as a 'grassy weed' (e.g. couch grass), or a 'leafy weed' (e.g. buttercup, nettle, chickweed).

At 21 days post-treatment photos were taken for all species apart from *L. mulitflorum* x *F. arundinacea* cv. Lofa and *O. sativa* were seedling establishment was too low. Photos were taken using an iPhone SE with iOS 12 and imported as JPGs into Powerpoint for figures. Plants were cut to the base of plants and the tissue collected and dried in oven at 55 – 60 °C overnight. Biomass was recorded, and the percentage yield difference of FAs compared to NA were calculated in Excel. Soil samples of 10 ± 0.1 g treatment pots were taken for three grasses and three legumes and freeze dried (VirTis, Benchtop SLC model 4KBTZL-150, 2022131) for pH analysis in 2.3.3. The whole initial test of FAs with UK forage crops was carried out twice.

2.3.3 Soil pH measurement

Freeze dried soil samples were sieved to 2 mm into glass Duran flasks containing 50 mL dH₂O and placed on an orbital shaker at RT for 60 \pm 5 mins at 5 Hz. Samples settled for 60 mins and the pH was recorded.

2.3.4 FA Chemical analysis standard solutions

Chemical analysis of FAs required standard solution and were made in conjunction with F.M. To determine the soluble dry weight of each FA they were washed in MeOH 100 % (v/v) and the soluble layer evaporated in low boiling point mode at 60 °C and no heat lamp (Genevac EZ-2 Elite). VFA yielded 0.579 mg/mL and MFA 1.521 mg/mL. VFA is a brown, slightly sticky powder, and MFA a very dark brown sticky gel. Standard concentrations of 0.01, 0.05 and 1 g/mL in dH₂O was made for each for analysis.

2.3.5 FA Inductively Coupled Plasma Spectrometry

Elemental analysis of 0.05 g/mL for total N, C and trace elements using Inductively coupled plasma-optical emission spectrometry (ICP-OES) and Inductively coupled plasma mass spectrometry (ICP-MS) for VFA and MFA standard solutions. These were performed offsite for VFA at Computational and Analytical Sciences, Rothamsted Research, Harpenden and for MFA at Biological Services, UEA, Norwich. Tables 2.8 and 2.9 show the results of this analysis, calculated as mg/L. Some detected contents were deemed negligible when accounting for elemental concentrations in growth media or soil and could be discounted as having an impact on plant growth; for example, a concentration of titanium at 0.0618 mg/L in MFA 0.5 % solution is not likely to affect plant yields. This data, along with any compounds found with Mass Spectral data below was used to formulate nutritional control with similar elemental contents and pH as both FA sources.

<u>Table 2.8:</u> VFA detectable content in mg/L for individual elements by ICP-OES and ICP-MS if 0.5 % solution.

Al 396.153	Ca 315.887	Co 59	Cu 63	Fe 238.204	K 766.490	Mg 279.077
4.33e ⁻¹	19.06	2.55e ⁻³	7.49e ⁻³	4.20e ⁻¹	1.72	8.90
Mn	Ni	ć	7	N		
55	60	5 181.975	2n 66	Na 589.592	Total N	Total C

<u>Table 2.9:</u> MFA detectable content in mg/L for individual elements by ICP-OES and ICP-MS for 0.5 % solution.

Ti	Mo	Cr	Cd	P	Zn
337.280	202.032	267.716	226.502	213.618	213.857
6.18e ⁻²	3.57e ⁻⁴	4.86e ⁻⁴	1.25e ⁻⁴	4.24e ⁻²	4.43e ⁻³
Co	Ni	Cu	Fe	Mn	Total N
228.615	231.604	324.754	259.940	259.372	

2.3.6 FA Thin Layer Chromatography

FA standards of 0.01 g/mL were analysed by F.M. under the guidance of Dr Michael Stephenson (Metabolic Biology, John Innes Centre) using Thin Layer Chromatography (TLC) on TLC Silica gel 60 plates (HS3822 00 00, Merck Millipore Ltd., Watford, UK). Standard solutions were dissolved in the solvents Hexane:Ethylacetate in 2:1, 1:1, 1:2, 1:3, and 100 μ L applied to the plate. Plates were examined under short and long wave UV and subsequently stained with vanillin stain: EtOH 95 % (v/v), sulphuric acid 5% (v/v) and vanillic aldehyde 40 mM (W310700). No movement was found from base line following staining, suggesting FA compounds were not reactive to UV.

Increased standard concentration was tested, 1 g/mL, with solvents mixes of Dichloromethane:MeOH and Ethylacetate:MeOH in 1:1, 1:2 and 1:3 mixes were tried, as well as pure MeOH. Only a very faint 'smudge' was found, suggesting the compounds for both VFA and MFA were extremely polar.

2.3.7 FA Liquid Chromatography Mass Spectrometry

Standard sample of 0.01 g/mL were analysed for their components by Liquid Chromatography Mass Spectrometry (LC-MS IT-TOF 010024700170, Shimadzu) by F.M. under the guidance of Dr Lionel Hill (Metabolic Biology, John Innes Centre). Run information was as follows:

- Column: C18 chain column (Kinetex® 2.6 µm EVO C18 100 Å, LC Column 100 x 2.1 mm).
- Elution: Gradient of acetonitrile (solvent B) versus 0.1 % formic acid in HPLC water (solvent A), run at 0.5 mL min⁻¹ and RT with 3 % B each time point: 0 min; 1 min; 6 min; 6.5 min; 6.6 min; 9 min hold with detection below.
- Detection and data acquisition: Positive polarity ionization with detection with 6.25 Hz; Curved Desolvation Line (CDL) temperature at 250 °C; Heat block temperature at 300 °C; nebulizing gas flow 1.5 L/min. Detections were monitored at 200 – 600 nm for

UV/visible absorbance with positive mode electrospray mass spectrometry of 200 - 2000 m/z full spectra; Tandem mass spectrometry of precursor ions at an isolation width of 3 m/z; Collision-induced dissociation (CID) 50%, and Ion Energy 50%.

In VFA a peak cluster identified as polyethylene glycol (PEG) chains of different lengths was detected, a with a charge envelope from 0.1-1 m/z and repeating ion mass units of 44 Da. LC-MS was repeated with samples, with a standard of 0.01 g/mL PEG 400 (202398). Peak in VFA was confirmed with this standard for its concentration using NIST Atomic Spectra database (v14, National Institute of Standards and Technology, Maryland, USA) [27].

Samples were also run in reverse phase using High Performance Liquid Chromatography (HPLC Dionex Ultimate 3000) with both a Charged Aerosol Detector (CAD Dionex Corona Veo RS). Run information was as follows:

- Column: Hydrophilic Interaction Chromatography (HILIC) column (Luna® 5 µm HILIC 200 Å, LC Column 250 x 4.6 mm).
- Elution: Linear low of acetonitrile (solvent B) versus 0.1 % formic acid in HPLC water (solvent A), run at 10 mL min⁻¹ and RT with 10 % B each time point increasing every 5 mins to 95 % from 10 400 mins; hold at 95 % for detection for 5 mins; decrease to 60 % for detection for 2 mins; increase to 100 % until competition.
- Detection: Partial injection mode with fluorescence detection at wavelengths 220, 250 and 280 nm; fractions pooled and evaporated at 40 °C with CAD peak width limited to 0.1.

In VFA, peak cluster suspected as PEG was again detected, but poor range in chromatograms suggested other compounds were extremely polar as found by TLC.

2.3.8 FA Gas Chromatography Mass Spectrometry

Standard samples of 0.01 g/mL were analysed for their components by Gas Chromatography Mass Spectrometry (Agilent GC-MS Single Quad Mass Spectrometer (7890/5977), Agilent technologies, California, USA) by

F.M. under the guidance of Dr Paul Brett (Metabolic Biology, John Innes Centre). Run information was as follows:

- Column: Zebron ZB5-HT INFERNO column (7HG-G015-02-GGA; length 30 m with 5 m guard, internal diameter 0.25 mm, film 0.1 μm; Phenomenex).
- Carrier gas: Helium at constant flow of 1.1 mL/min.
- Inlet temperature: 250 °C
- Injection volume: 1 μL
- Injection mode: split (10:1)
- Oven temperature: Initial temperature 60 °C

Ramp 10 °C/min to 300 °C

Hold: 300 °C for 5 min

Equilibration time: 0.5 min

- Auxiliary temperature: 290 °C
- Acquisition mode: SCAN between 50 600 m/z
- Sample preparation: Derivatised with MSTFA (N-Methyl-N-(trimethylsilyl) trifluoroacetamide, 394866) instead of dH₂O solution.

Data was acquired with Agilent Masshunter Qualitative Analysis (B.07.00) and peaks were identified by METLIN [160; 161]. In VFA a PEG cluster was again found, as in above LC-MS, (see Figure C3, Appendix C) In MFA citric acid derivatives synonymous to isomers including R-(homo)2-citrate were detected (see Figure C2, Appendix C), likely in a complex with Fe⁽²⁺⁾ which is in high levels comparable to other elements (see Table 2.9). GC-MS was repeated with samples, with standards 0.001 g/mL citric acid (C1909) for MFA, and 0.01 g/mL PEG 400 (202398) for VFA. Peaks in MFA and VFA was confirmed with these standards as above with NIST Atomic Spectra database (v14, National Institute of Standards and Technology, Maryland, USA) [162] (see Figures C2 and C3, Appendix C).

2.3.9 FA control solutions

Controls for VFA and MFA were formulated using the chemical data above in Tables 2.8 and 2.9, and with the above LC-MS and GC-MS analysis (2.3.7 and 2.3.8). Compounds which could impact plant growth when compared to media or soil contents, even in tiny amounts, were accounted for. These were named VC and MC accordingly and can be found in Tables 2.10 and 2.11. All solutions were autoclaved prior to assays.

<u>Table 2.10:</u> VFA control solution (VC) components for 1 % application as in 2.3.1.

Compound is given, with Sigma-Aldrich catalogue number, and mg/L required.

KNO ₃	Sucrose	AICI ₃	CaSO ₄ .2H ₂ O	FeSO ₄ .7H ₂ O	K ₂ SO ₄
P8291	84097	563919	C3771	F8263	P0772
1.95	125.5	7.75	163.7	4.2	13.2
MgSO ₄	MnSO ₄ .xH ₂ O	Zn(NO ₃) ₂ .H ₂ O	NaCl	Na ₂ SO ₄	PEG-400
MgSO ₄ M7506	MnSO ₄ .xH ₂ O 229784	Zn(NO ₃) ₂ .H ₂ O 230006	NaCl S7653	Na ₂ SO ₄ S6547	PEG-400 202938

pH to 6.0

<u>Table 2.11:</u> MFA control solution (MC) components for 0.5 % application as in 2.3.1.

Compound is given, with Sigma-Aldrich catalogue number, and mg/L required.

NH ₄ NO ₃	CuSO ₄ ·5H ₂ O	FeSO ₄ .7H ₂ O	Monosodium citrate
A7455	C8027	F8263	71498
24.5	8	12.45	45.6

pH to 4.8

2.3.10 FA Nuclear Magnetic Resonance

Analysis above was used to produce elemental controls for FAs to use in plant and microbial assays, and Nuclear Magnetic Resonance (NMR) was carried out to elucidate which carbon compound types or ratios are present in the extractions. NMR and analyses of VFA and MFA were carried out by Dr Juan Carlos Muñoz-Garcia with supervision from Prof Yaroslav Khimyak (both School of Pharmacy, University of East Anglia).

¹H-decoupled ¹H-¹³C cross polarization (CP) and CP single pulse (CPSP) solid state NMR experiments were performed at 20 °C using a 7.05 T Bruker Avance III spectrometer equipped with a 4 mm triple resonance probe (Bruker©, Fällanden, Switzerland) operating at frequencies of 300.1 MHz (¹H) and 75.5 MHz (¹³C). Each sample was packed into a zirconia rotor, sealed using a kel-f drive cap, and spun at 12 kHz. A CP contact time of 1 ms and relaxation delay of 5 s were employed, with 90° pulses of 3.5 and 4.5 µs used for ¹H and ¹³C, respectively. All spectra were referenced with respect to Tetramethylsilane (TMS, 87920).

Peak areas were obtained using the automatic integration tool of TopSpin 3.6.1 (Bruker©, Fällanden, Switzerland), and subsequently normalised to relative areas and grouped into different functional groups according to the expected chemical shift regions for soil organic matter [163; 164]. These were as follows; alkyl C (0 - 50 ppm); methoxyl C (50 - 60 ppm); carbohydrate C (60 - 90 ppm); di-O-alkyl C (90 - 110 ppm); aryl C (110 - 142 ppm); phenolic C (142 - 160 ppm); carbonyl C (160 - 200 ppm). The percentage of each functional group was then calculated for the total of each FA.

2.3.11 Glasshouse FA vegetative growth and nodulation assays

Due to the legume yield effect measured in 2.3.2, FAs were tested in *M.* sativa for vegetative yield and nodulation rate. Treatments of VFA and MFA from 2.3.1, controls VC and MC from 2.3.7, and extra controls of no application (NA) and dH_2O were used.

Seeds of *M. sativa* cv. Daisy, Luzelle and Gea were sterilised, vernalised and germinated on water agar as in section 2.1.7. At 9 days post germination, seedlings were transplanted to 60 cell trays (individual cells were 4.5 x 4 x 14 cm dimensions) with Church Farm soil (see 2.2.4) in the glasshouse. At 12 days post germination, cells were randomly assigned to one of 8 treatments. Treatments were then designated a number from 1-8 by a lab colleague, Marco Fioratti (Metabolic Biology, John Innes Centre), and these given to N.C. making treatments blinded when treating plants and subsequent sampling; note NA treatment was not blinded due to practical necessity. Each cell received 1 mL of their specified numbered treatment.

For cultivar Daisy only, four soil samples for each numbered treatment were taken at both seedling transplanting (0 days post treatment) and biomass harvesting (21 days post treatment). Around 10 - 15 g of soil from treatment cells was mixed into 100 mL glass vials and stored at -20 °C.

For all cultivars at 21 days post treatment, individual plants above ~ 4 cm were harvested for each treatment (n = 6 - 12), and vegetative tissue dried in at 55 - 65 °C for biomass. The whole experiment was repeated separately for three individual replicates.

For cultivars Daisy and Luzelle, roots were washed with dH_2O to be free of soil. The numbers of pink and white nodules were scored visually with numbered treatment blinded.

Once vegetative biomass (all cultivars) and nodule count measurements (not including Gea) had been taken the results were unblinded. All treatment data was calculated relative to NA in Excel® 2016. Significance between treatments for p-value < 0.05 was shown with different letters using one-way ANOVA with Tukey testing in GenStat® 18th Edition. Graphs were made in RStudio.

As treatments were unblinded following the above measurements, so the origin of soil samples was known during subsequent assays.

2.3.12 Plate FA vegetative growth and nodulation assays

FAs were tested on agar plates either in sterile conditions or with *Sinorhizobium* inoculation. Seeds of *M. sativa* cv. Daisy, Luzelle and Gea were sterilised (0.5 g each), vernalised, and germinated on water agar as in section 2.1.7. Six treatment plates were prepared with 20 mL addition of treatments in 2.3.18 to a final volume of 200 mL FP media from 2.1.7 with 5 % agar No. 1 (Lab M, MC002). Two days post-germination 5 seedlings of the same size were transferred to each treatment plates. Half of the treatment plates were inoculated with 1 mL of *Sinorhizobium meliloti* 1021 culture prepared as in 2.1.7. Plates were sealed with micropore tape and transferred to CER for 21 days at 23 °C and a photoperiod of 16 h light (90 μ mol m⁻² s⁻¹) and 18 h dark.

At 21 days growth, total nodule number was recorded for both inoculated and sterile treatment plates. Seedling shoots and roots were then separately pooled for each treatment and dried 55 – 60 °C overnight. Biomass was measured the following day. The whole experiment was repeated in triplicate, and mean values calculated. Statistical analysis was performed as in 2.3.11.

2.3.13 *Sinorhizobium* colony forming unit counts

The effect of FAs on *Sinorhizobium meliloti* 1021 growth in cultures was analysed using single colony forming unit (CFU) counts. The strain was preincubated in 100 mL TY media for 2 days at 28 °C shaking at 200 rpm. The preincubated stock was diluted to 0.1 OD at 600 nm to add to treatment flasks.

Treatment flasks of 100 mL TY were set up as follows: NA = no addition; dH₂O = added 10 mL dH₂O; VFA/MFA = added autoclaved 10 mL of either 10 % VFA or 5 % MFA from 2.3.1 stock solutions; VC/MC = added autoclaved 10 mL of either 10 % VC or 5 % MC from Tables 2.10 and 2.11 solutions. Flasks were inoculated with 10 µL of preincubated stock and incubated at 28 °C shaking at 200 rpm. At timepoints of 0, 1, 2, 3, and 4 days dilutions from treatment flasks of 10⁻¹, 10⁻², 10⁻³, 10⁻⁴, 10⁻⁵, 10⁻⁶, 10⁻ ⁷, 10⁻⁸, 10⁻⁹, and 10⁻¹⁰ in 1 mL dH₂O were taken in triplicate. From these dilutions 10 μ L was spotted out on TY agar plates, and plates placed upside-down in 28 °C oven for one day until single colonies had formed in a dilution spot at a rate of ~ 20 – 200. Plates were scanned using a Perfection V550 Photo scanner (Epson, SKU: B11B210301) and CFU recorded for dilution. An example can be found in Figure C6, Appendix C. Rhizobial cell density was calculated for dilution factor and total volume of culture at each timepoint. The whole experiment was repeated in triplicate to compare treatments, with graphs produced in RStudio and ANOVA performed using Genstat as above in 2.3.11.

2.3.14 Soil Microbial Phospholipid Fatty Acids Extraction

Soil stored at -20 °C from glasshouse *M. sativa* cv. Daisy assays in 2.3.11 were defrosted and mixed thoroughly. Each treatment (NA, dH₂O, VFA and VC) had samples from day 0 and day 21, in triplicate. Extraction of phospholipid fatty acids (PLFA) was performed using a modified protocol from Quideau et al., (2016) [165]. Prior to assays, a 5 g of soil sample was freeze dried (VirTis, Benchtop SLC model 4KBTZL-150, 2022131) in EtOH cleaned glass vials (used throughout) to remove all water from the soil. A negative control was also carried out without soil.

An internal standard of 250 μ L methyl decanoate solution 1 % (299030) was added at the beginning of the extraction process. Total lipids were extracted using 4 mL Blyth-Dyer extract; 200 mL H₂HPO₄ 50 mM, 500 mL MeOH. Extractions were vortexed for 1 min, then placed on an orbital shaker for 2 hours in the dark; PTFE tape was used to seal vials before plastic lids. Extractions were centrifuged at 1650 rpm for 15 mins at 4 °C, and supernatant was added to a new vial. The Blyth-Dyer extraction was repeated with the soil sample, and again supernatant added to vial. 1 mL chloroform and 1 mL dH₂O was added to the extraction, and the upper phase was aspirated with N₂ gas. The lower phase, which contained total lipids, was centrifugally evaporated in low boiling point mode at 60 °C and no heat lamp (Genevac EZ-2 Elite). Vials were stored overnight at -20 °C.

Elution of lipids was performed using SPE columns 500 mg 6 mL silica (SUPELCLEAN LC-SI, 57134) Columns were conditioned with 5 mL MeOH, and then 5 mL chloroform. Evaporated extracts were brought to RT, dissolved in 2 mL chloroform, and loaded into a column. A vacuum pump was used for elution at \sim 2 Hg. Neutral lipids were eluted with 5 mL chloroform, and glycolipids eluted with 5 mL acetone. With new vials underneath columns, PLFAs were eluted with 5 mL 5:5:1MeOH:chloroform:dH₂O buffer. The upper phase was again evaporated using the centrifugal evaporator, with a N_2 gas purge.

Transesterification of PLFAs to produce fatty acid methyl esters (FAME) for detected was performed in a 37 °C water bath for 15 mins with 250 µL transesterification reagent; 0.561 g KOH dissolved in 75 mL MeOH, and 25 mL toluene added. At RT 400 µL acetic acid 75 mM and 400 µL chloroform was added. Phases were allowed to separate, and the bottom 300 µL of the bottom phase was added to a new vial. The transesterification process was repeated, with another 300 µL added to the new vial. Samples were again centrifugally evaporated until dry and purged with N₂ gas. Samples were dissolved in 75 µL hexane, and stored at -20 °C until ran on GC-MS. For GC-MS analysis samples were ran using an Agilent GC-MS Single Quad Mass Spectrometer (7890/5977). Run information was as follows:

- Column: Zebron[™] Inferno[™] ZB5-HT column (Phenomenex®; length 30 m, internal diameter 0.32 mm, film 0.25 µm; Phenomenex Inc., California, USA).
- Carrier gas: Hydrogen at constant flow of 1.2 mL/min
- Inlet temperature: 250 °C
- Injection volume: 1 μL
- Injection mode: split-splitless (30:1)
- Oven temperature: Initial temperature 190 °C

Ramp 10 °C/min to 285 °C Hold: 310 °C for 5 min Equilibration time: 0.5 min

- Auxiliary temperature: 280 °C
- Acquisition mode: SCAN between 50 500 m/z

- MS temperature: Source 230 °C
- Quad temperature: 150 °C

Analysis for FAME profiles from PLFA extracts was performed as below using MassHunter Profinder and Mass Profiler Professional.

2.3.15 Analysis of FAME profiles

FAME spectra from 2.3.14 were extracted from all chromatograms in MassHunter Profinder version 2.0 (Agilent B.08.00). The inbuilt wizard for batch recursive feature extraction was used to detect compounds with parameters as follows to avoid erroneous peaks due to noise: no region exclusion; noise cut 100 – 500; RT tolerance 0 – 0.1 min with minimum dot prod = 0.4; score(MFE) filter = 70; expected retention time 0.5 – 1.5 mins; Agile 2 Integration used with set parameters without smoothing; no peak filter; chromatogram format of centroid. Compounds in this wizard are identified with the NIST 17 library (NIST 17 MS Database and MS Search Program v.2.3).

The compounds identified were exported as compound exchange format files and imported into MassHunter Profiler Professional 14.0 (Agilent G3835AA). Again, the inbuilt wizard was used to determine fold difference of peaks with parameters set as follows; no abundance parameter; allowed tolerance of 30 s; normalisation algorithm set to standard as all spectra were similar. A compound list with abundances for each extraction was exported to excel for further analysis.

Most compounds found in all spectra were FAME as expected. Those not were low abundances of 2,6-Bis(tert-butyl)phenol, 1-Octadecanol, and 1-Dodecanol, and likely to be TMS derivative contaminants of the FAME, and thus discounted from analysis. The internal standard was Decanoic acid (C10:0) All identified FAME were assigned into their biomarker class for corresponding microbial community as compiled by Quezada et al., (2007) [166]. However, as some FAME biomarkers overlap in their assignments the final microbial groupings were reduced to 'Common', 'Gram-negative

bacteria', 'Algae/Higher Plants', 'Gram-positive bacteria', 'Fungi', and 'Actinomycetes/Desulfobacter'.

FAME abundance in nmol g⁻¹ was calculated for each methyl ester using its spectra peak area compared to the internal standard decanoic acid (C10:0, retention time 9.199) as:

$$nmol \ g^{-1} = \frac{methyl \ ester \ molarity}{total \ sample \ weight \ (g)} \times C10:0 \ nmol \ L \right)$$

Abundance values were used to calculate percentage of each microbial grouping in samples. Mean percentages of groupings were calculated with standard deviation in Excel, and Principal Component Analysis (PCA) plots created with ClustVis [167] using the NIPALS algorithm [168] within the pcaMethods Bioconductor package [169].

2.3.16 Soil microbial DNA extraction

Soil stored at -20 °C from glasshouse *M. sativa* cv. Daisy assays in 2.3.11 were defrosted and mixed thoroughly. Each treatment (NA, dH₂O, VFA and VC) had samples from day 0 and day 21, in triplicate. Total DNA from soil was extracted using the PowerSoil® DNA Isolation Kit (MO BIO Laboratories, Inc., QIAGEN Ltd., 12888-100) by the manufacturer's instructions with the following changes. Solution C1 was incubated to 60 °C for 10 minutes prior to use. A negative control with no soil was ran alongside the samples. A total of 1.5 g of each soil was used for extractions with 3 x 500 mg in sample preparation, cell lysis, inhibitor removal, and DNA binding steps. The triplicate extractions were mixed into one at the spin column and elution step. Final volume for samples was 200 μ L.

Positive DNA controls for *Sinorhizobium meliloti* were also extracted. Cultures were grown in 200 mL TY media at 28 °C for 2 days at 200 rpm, as in 2.1.7.

DNA quality was measured on a NanoDropTM 2000 spectrophotometer for a A260/A280 ratio of \geq 1.7, and concentrations were recorded.

2.3.17 Soil DNA qPCR

Soil DNA extractions were used in qPCR for inferring any changes in *S. meliloti* dependant on treatment. Published primers used can be found in Table 2.12 below, with efficiencies confirmed between 90 – 110 % as in 2.1.10. DNA extractions were diluted to ~ 2 ng/µL, and qPCR set up as in 2.1.5 with DNA further diluted to 1:50 like cDNA. The qPCR programme was as used previously, but with the annealing temperature set to 56 °C and for a total of 50 cycles.

The transcript expression fold changes were calculated as in 2.1.5, with *S. meliloti* gene of interest *nodC* $2^{A.\Delta CT}$ values calculated for individual reference gene sets, 799/1391 and Eub338/518, and expressed relative to experimental replicate one 'NA' treatment. The geomean was determined and was used to infer any differences in *nodC* expression in soil extractions between 0 day to 21 days depending on treatment. Graphs were made in RStudio.

<u>Table 2.12:</u> Primers used in study for soil DNA extractions for testing *Sinorhizobium meliloti* activity.

Gene name	Primer name	Sequence (5' – 3')	Source	
nodC	SmnodC F	GCCGCTATCTCAATCTACGC	Trabelsi et al., 2009 [170]	
	SmnodC R	TTGAAGCTGGGGACGATAAC		
799/1391	799 F	AACMGGATTAGATACCCKG	Beckers et al., 2016 [171]; Chelius and	
	1391 R	GACGGGCGGTGWGTRCA	Triplett 2001 [172]; Walker and Pace 2007 [173]	
Eub338/518	338 F	ACTCCTACGGGAGGCAGCAG	Fierer et al., 2005 [174]; Lane 1991	
	518 R	ATTACCGCGGCTGCTGG	[175]; Muyzer et al., 1993 [176]	

2.3.18 Field trials with VFA

To assess if yield increases in *M. sativa* from VFA found in both plate and glasshouse experiments was applicable to growers, field trials were carried out over both 2017 and 2018 growing seasons. In 2017 trials were performed at Dengie Crops Ltd. (Southminster, Essex) with the help of Andrew Spackman (Farmacy Plc., Dorrington, Lincoln) in cv. Daisy and Fado. In 2018 the trials were at both Blankney Estates Ltd. (Blankney, Lincolnshire) and A Poucher and Sons (Bardney Dairies) Ltd. (Market Rasan, Lincolnshire) with the cv. Daisy and Gea respectively.

Treatments to be tested were NA, dH_2O , 1 % VFA (see 2.3.1) and 1 % VC (see 2.3.7). Individual experiment design of each plot is in Figure C9, Appendix C. Trials were designed as randomised block treatment plots for 2017 trials, and as randomised full trial treatment plots for 2018 trials. Each trial contained 4 - 6 plots per treatment of areas 4 - 10 m² with buffer zones between plots. As in the glasshouse trials in 2.3.11 treatments were applied and at 21 days post treatment samples were taken for vegetative biomass measurements using a randomly placed 625 cm² sampling quadrat with all tissue taken for biomass. Samples were also taken for protein and chlorophyll for 2018 trial plots as described in 2.1.13, 2.1.16 and 2.1.17. Statistical testing was carried out as in 2.3.11 for measurements, with ANOVA calculated in Genstat for significance between treatments.

2.4 FULVIC ACID RNA-SEQUENCING OF MEDICAGO SATIVA

2.4.1 RNAseq of VFA and VC experimental design

Due to the role *Sinorhizobium meliloti* inoculation plays in the growth effects of VFA, RNAseq was used to investigate transcriptional changes in the shoots and root separately. Figure 2.3 below shows the experimental design for RNAseq investigation with biomass measurements detailed below, with three independent replicates.



<u>Figure 2.3:</u> RNAseq experimental design for comparing transcriptional differences in shoots and roots of *Medicago sativa* cv. Daisy.

Treatments were either 1 % VFA or 1 % VC, with RNA sampling of tissue at day 0 and day 3. Vegetative biomass was measured as in 2.3.11.

2.4.2 RNAseq plant material and RNA extraction

Seeds of *M. sativa* cv. Daisy were sterilised as in 2.1.7. Seeds were sown in full seed trays ($36 \times 22 \times 6 \text{ cm}$) containing Church farm soil as in 2.2.4 at a rate of 20 kg ha⁻¹, or 0.158 g per tray. Trays were watered every 3 -4 days, and at day 12 were treated with autoclaved 1 % VFA or 1 % VC, described in 2.3.1 and 2.2.9 respectively. Plants were sampled for RNA at day 12 and day 15, each referred to as day 0 and day 3 in subsequent analysis regarding their treatment timepoints. For each sample ten plants were pooled, with shoot and root tissue separated. Therefore, each sample consisted of ten biological samples, and the experiment was carried out in triplicate over 3 days to provide three experimental replicates. Tissue was immediately frozen in liquid N₂ and stored at -80 °C until RNA extraction. RNA extraction was carried out using the TRI Reagent (93289) method described in 2.1.2 with a final volume of 50 µL, and stored at -80 °C.

2.4.3 RNA clean-up and quality control

RNA from 2.4.2 was thawed on ice. Samples were purified using the RNeasy MinElute Cleanup Kit (QIAGEN Ltd., 74204) as per the manufacturer's instructions to a final volume of 30 μ L; 25 μ L was stored at -80 °C to be sequenced, and 5 μ L was used to perform quality check using a Bioanalyzer and for any future qRT-PCR confirmation.

Quality check was performed with cleaned RNA samples using the Agilent RNA 6000 Nano Kit (Agilent Technologies, 5067-1511) as per the manufacturer's instructions for a total of 2 chips. This was ran using a 2100 Bioanalyzer Instrument (Agilent Technologies, G2939BA). RNA Integrity Number (RIN) and RNA concentration (ng/µL) were recorded. Samples with а RIN below 7 were checked with NanoDrop™ 2000 spectrophotometer for 260/280 to ensure RNA had not degraded during extraction. Any samples with a concentration above 500 ng/µL were diluted to be between 50 – 500 ng/ μ L with final volume of 23 μ L sent for sequencing. General RNA quality check information and Bioanalyzer gel results can be found in Table D1 and Figure D1, Appendix D.

2.4.4 RNA Sequencing

Library construction and sequencing was performed by Novogene (HK) Company Ltd. (Hong Kong). Library construction was performed using Next® Ultra[™] RNA Library Prep Kit (New England BioLabs Inc., E7530L) and sequenced on 1 lane of a HiSeq[™] 2000 (Illumina, HWI-ST1276) in High Output mode using 150 bp paired end reads and V2 chemistry. Initial quality assessment of sample reads was performed using Phred quality scores with Illumina Q score calculations. Sequencing quality check information can be found in Table D2, Appendix D.

2.4.5 Read alignments

Read alignments were performed by Dr Jitender Cheema (Computational and Systems Biology, John Innes Centre) with aligned data to N.C. for all subsequent analysis.

Reads were aligned to two reference sequences for the close relative *Medicago truncatula*, the fully annotated A17 v.4.0 [177] and the first full assembly of R108 v1.0 [178]; A17 is a more annotated reference but *M. sativa* is more closely related to R108. Assembly used the R package [148] using Bioconductor [49] ballgown [179]. The mapping rate was 60-70 % for A17 alignment and 75 - 85 % for R108 alignment, with aligned transcripts totalling 57585 and 61019 respectively. Although read alignment was adequate in for both references it was decided that de novo transcriptome assembly would be more suitable for this case.

De novo transcriptome assembly was performed with Trinity [180], which used all samples, regardless of tissue, treatment, or timepoint to build. A total of 630599 transcripts were preliminary identified (including isoforms). BUSCO [181] was ran to check benchmarking of assembly using Universal Single-Copy Orthologs. BUSCO checked 430 conserved plant origin benchmarks and found that only 4 were missing from de novo assembly, providing a 93 % quality assessment score. Kallisto [182] was used to align assembly which is less subjective than ballgown mapping; Kallisto is a form of pseudoalignment, where compatibility of reads to

target are tested for likelihood, without a need for alignment. Pseudoalignment percentage scores can be found in Table D3, Appendix D. This provided both Transcripts Per Million (TPM) and Reads Per Kilobase Million (RPKM) for subsequent analysis.

2.4.6 Differential gene expression analysis

Differential gene expression was performed for both shoot and root tissue independently using Degust [183] and performed with all read alignments generated in 2.4.5. Sample multi-dimensional scaling (MDS) plots for 200 genes were generated to compare how experimental replicates clustered for different treatment and timepoints.

Tissue samples were grouped into treatment and timepoint for transcript analysis. Transcripts with both an absolute log fold change of 0.585 (1.5 x fold change) and a false-discovery rate (FDR) adjusted p-value (qvalue) < 0.05 were considered as differentially expressed. The grouping of tissue samples requires all 3 experimental replicates to fit these criteria to be accepted, thus ensuring a very high benchmark was set at the beginning of analysis. Differential expression was checked using voom/Limma method [184] for log Fold Change (logFC) between treatments (VC, VFA) at both timepoints (0, 3); 0VC vs 0VFA and 3VC vs 3VFA. Differential expression was then check for individual treatments between timepoints; 0VC vs 3VC and 0VFA vs 3VFA. An example of this analysis for 0VFA vs 3VFA can be found in Figure D2, Appendix D.

To eliminate any differences caused by random chance or plant development changes over the 3-day timescale, transcripts that were truly differentially expressed based on the VFA treatment were calculated by subtracting 0VC vs 0VFA and 0VC vs 3VC from 0VFA vs 3VFA. It was this differentially expressed (DE) transcript list which was carried forward for the functional annotation.

2.4.7 Functional annotation

DE transcripts from degust analysis in 2.4.6 were imported into the Blast2GO v1.4.4 programme pipeline [185; 186] as FASTA contigs for functional annotation. DE transcripts were checked against NCBI's nonredundant NR database [187] with a BLAST expectation value cut-off of 1.0E⁻³, and hits excepted for no more than 20 sequences. In programme mapping was ran with the EMBL-EBI InterPro library [188] using amino acid mapping [189] with all families, domains, sites, and repeats available tested. Annotation of mapped results was ran using Gene Ontology Annotation Version 2019 [190; 191] with the follow strict parameters; Annotation cut off of 55; GO weight of 5 only; E-value-Hit-Factor restricted to 1.0E⁻⁶; Hit filter set to 500; Evidence Codes weighted from 0.5 to 1 depending on depth of evidence (default software parameters). To quality check once annotation was completed in Blast2GO a manual BLAST algorithm was performed [137; 138; 192] with NCBI database [193; 139; 194; 140] of the most significantly up or downregulated genes. Moreover, any genes lacking any GO annotation through InterPro library were checked for annotation in both QuickGO [195] and UniProt [196] and added to future analysis.

The inbuilt statistical wizard in Blast2GO was used to generate distribution graphs for sequences and hit species. Top GO annotation distributions and gene ontology directed acyclic graphs (DAGs) were also generated for molecular function, biological process and cellular components. Moreover, data was exported to Excel for manual bar charts containing all GO terms.

2.4.8 GO term enrichment testing

To test for enrichment of different categories of de novo *M. sativa* DE transcripts relative to all expressed transcripts found in *M. truncatula* (as the closest relative), the PANTHER Classification System v14.1 was used [197; 198]. GO-Slim graphs were generated for molecular function, biological process, and protein class, and then an Overrepresentation test was performed using the Fisher's exact test [199]. These were run

separately for up and downregulated DE transcripts in both shoot and root. Fold enrichment values and their p-value for each term was recorded.

2.4.9 Validation of RNAseq by qRT-PCR

compared to RPKM logFC of DE transcripts.

Following RNA-seq analysis, qRT-PCR was used to measure expression of a subset of DE transcripts. From the root Overrepresentation test performed in 2.4.8, a subset of seven genes were chosen to confirm with qRT-PCR. Primers were designed for genes using available *M. truncatula* sequences, shown in Table 2.13, and primer efficiencies calculated as in 2.1.10. Primers with 90 – 115 % efficiencies were used in qRT-PCR. Root RNA from 2.4.2 underwent cDNA synthesis as in 2.1.3. qRT-PCR was performed as in 2.1.5 using the reference gene *ACTIN2* in Table 2.6. Mean relative expression was calculated for experimental replicates and
<u>Table 2.13:</u> Primers used in study for *Medicago sativa* qRT-PCR confirmation of RNAseq analysis.

Gene name	Primer name	Sequence (5′ – 3′)	Source
Myb/SANT-like DNA- binding domain protein	Myb/SANT-like F	GGGCCACATCTGCAGTAGTA	Nicola Capstaff
	Myb/SANT-like R	TGCTTGGTGGGATGCTAAGA	
putative CC-NBS-LRR resistance protein	CC-NBS-LRR F	GCAACGCAACTTGAGAGAAT	Nicola Capstaff
	CC-NBS-LRR R	TTCCTCCAAACCCACAGTGT	
LysM domain receptor-like kinase 3	LysM kinase F	CGTCTACCTCCAATGAACAAGT	Nicola Capstaff
	LysM kinase R	GTGTCTTCCCTGCAATGACTAC	_
MtN6	MtN6 F	GAAGAGTTCGGGGAATGCAT	Nicola Capstaff
	MtN6 R	CGTCCTCTCCTATTAATTGGCG	_
nodulin-26	nodulin-26 F	TCAACCAGAACCAGAGCCAT	Nicola Capstaff
	nodulin-26 R	ACCTCATCACAGCCCTAGTT	-
NRT1/ PTR FAMILY 4.5-like	NRT1/PTR 4.5 F	TTCTTTGCTCGTCCTAGGGG	Nicola Capstaff
	NRT1/PTR 4.5 R	CGACGACAATGGAGAGCAAG	-
beta-tubulin subunit	beta-tubulin F	GAATTGTGATTGCTTGCAAGGT	Nicola Capstaff
	beta-tubulin R	TCATTAACCTCTGAATCGTCACG	

This page is intentionally left blank

Chapter 3: Marker genes in Festulolium can be used to assess crop and soil N status, to aid with grower decisions to improve future yield

3.1 INTRODUCTION

3.1.1 Soil N status affects plant NUE and yield

Forage NUE is an important target for increasing forage production, particularly as the protein content of crops is so valuable for animal feed. Protein accumulation is linked to nitrogen (N) status and when the N supply is supra-optimal greater storage occurs. Forage crops often require N in greater amounts due to their increased growth, storage capacity and higher fibre content [200], when compared with grain crops that have been bred for high seed yield and starch content. For forage crops, NUE is the accumulation of vegetative tissue (biomass) harvested compared to the N fertiliser inputs. Principally NUE for forage crops can be based on N utilization efficiency (NUtE) and N uptake efficiency (NUpE) as we are interested in the highest achievable biomass of the shoot and leaves which will form the content to be dried for feed production [72]. Not only does this include biomass, but also the relative N levels in this tissue; it is not enough to only have a high shoot biomass, the crop must also yield optimal amounts of protein N. This total N within the plant is in turn important for the quality of the final animal feed. Moreover, when looking at the effect of fertiliser use, we are interested in how both the biomass and N status change on application and thus NUpE.

Nitrogen availability to animals is predominantly from forage proteins and these are estimated using crude total protein Kjeldahl measurements. Protein is abundant mainly as RuBisCO, with tissue specific proteins such as prolamins within monocot seeds [201; 202], although relative amounts do vary between forage species [23]. This is especially true when comparing the relative contents in grasses such as *Lolium* spp., *Festuca* spp., and *Festulolium* hybrids, with herbaceous legumes particularly lucerne (*Medicago sativa*), which is grown widely in the UK due to its high protein value [22].

Forage crops offer challenges for NUE as there is a requirement for optimal yield of shoot biomass with a high N content (NUtE) while also optimizing N fertilizer acquisition (NUpE) throughout the growing season [203].

Studies have already shown, through ¹⁵N labelling of *Lolium perenne*, how deficiency caused by low N fertiliser application causes an increase in the protein substrate pool whereas the store pool decreased in size and turnover rate [204]. It is important to maintain the N status of the crop throughout the growing season, depending on external abiotic factors, to ensure optimal production. In short, soil N supply must be optimised for individual crop growth [205]. This explicitly highlights the importance of fertiliser studies which assess soil and plant N status, for consequent N composition of forage crop vegetative tissue. Additionally, the intense cutting regimes, or grazing, within the management practice of cultivating forage crops exacerbates this challenge [49; 50; 206].

As NUE in plants is strongly associated with soil N status, many plant genes relating to N acquisition or metabolism may be useful to investigate this link. Genes which are important in carbon metabolism may be of relevance also, due the relationship of C:N for optimal plant growth [207-209]. Moreover genes related to the storage of proteins have been identified in both aboveground [210; 211], and belowground [212-214], tissue, which may be good candidates for exploring. Such genes will be discussed below but many references will be the homologues in *Arabidopsis thaliana* due to their better characterisation.

3.1.2 There is a wide variety of genes associated with NUE

As mentioned above, many genes associated with NUE have been well characterised in the model plant *A. thaliana* [117]. These *A. thaliana* genes, alongside more limited progress in grass research will be discussed below. Resources such as the published draft genome sequences of *L. perenne* [215] and *Brachypodium distachyon* [216] will make future analysis possible in grasses and other forage crops. The *NRTs*, *GS* and *NADH-GOGAT*, *RBCS* and *VSPs*, and *TIPs* gene families will be discussed with their characterisation in model species including *A. thaliana*, as well as any grass or legume related advances.

Nitrate transporters (NRTs)

Plants have evolved two nitrate uptake systems using proton-coupled transporters due to the high variability of available NO_{3} which is affected by soil type, day length and season. Because of their importance, these two nitrate transport systems must be considered when investigating genes associated with NUE. The high affinity transport system (HATS) has transporters with a $K_m \sim 50 \ \mu M$ for NO₃⁻, whereas the low affinity transport system (LATS) transporters have a K_m of ~ 5 mM. Examples of HATS and LATS transporters in A. thaliana are the NRT2 and NRT1 gene families [217]. Transporter kinetics were often identified using *Xenopus* oocytes assays or using mutant plants lacking expression of a gene [218]. An NRT of particular importance is NRT1.1, referred to as CHL1. CHL1 encodes a protein that can operate as a dual-affinity transporter in both HATS and LATS ranges [219]. NRT1.1 has become accepted as the primary nitrate sensing transceptor affecting the expression of numerous genes including *NRT2.1* under different conditions. This regulation extends to posttranslational phosphorylation at T101 by a CIPK23 [220; 221], with the phosphorylation status being the basis for its affinity changes due to structural flexibility [222]. The Medicago truncatula NRT1.3 has become accepted as the legume equivalent of a dual-affinity transporter [223]. Other examples of NRTs have been found across monocot cereal species [224-226] and are related to forage grasses.

There is redundancy in the N uptake system, as with many nutrient uptake pathways. For example, in *A. thaliana* when *NRT2.1* is knocked-out, plants show partial compensation through increased expression of *NRT2.2* with restoration of shoot:root N ratios to wildtype levels [227]. The ability of these primary transporters to affect root architecture is a key to their importance. *NRT2.1* initiates lateral root growth, *NRT1.1* is required for root primordia formation and emergence [228]. Alongside transcription factors such as the MADS-box member *ANR1*, roots are able to stimulate or initiate lateral root growth by sensing not only the levels of N but also in which form the N exists [229]. Control of root architecture is also affected by phytohormones, predominantly cytokinins, abscisic acid and auxin. Cytokinins and abscisic acid repress N uptake and NRT activity, and

auxin stimulates lateral root formation and therefore indirectly uptake [230-232]. Moreover, the interaction with NRT2s and N uptake requires extra genes such as NAR2 to carry out function across many species [233]. Members of the NRT1 family in A. thaliana have been shown to be involved in the translocation of NO_3^- from roots to shoots and vice versa [234-236]. The source-sink dynamic of N distribution is mediated by a complex regulatory exchange between the xylem and the phloem. This intricate network with complicated regulation, makes the study of NRT families a important line of research for plant scientists. Despite the advances still being made in this area in A. thaliana, and the accepted hypothesis that many other plant species will use a similar system, there remains a lack of characterised examples to study in forage crops. Plett et al., (2010), highlighted how the use of phylogenetic analyses in grasses to investigate *NRTs* is challenging, especially with the *NRT2* family [237]. Since then a flurry of research has shown advances in the field [238-244], however the *NRT* network in grasses largely remains elusive.

While the *NRT* family may be an interesting case for candidate N status genes in the future, for now it may be prudent to analyse more downstream genes related to assimilation and storage of N as proteins, rather than genes required for N uptake and distribution.

Nitrogen assimilation (GS and NADH-GOGAT)

Once the root architecture has developed to adequately uptake NO₃⁻ which heavily depends upon the *NRT* network, the N needs to be uploaded into vacuoles or transported to plastids for movement from the root to the resulting vegetative tissue [245]. As N assimilation is energetically costly it is important that this process occurs close to the photosynthetic apparatus where energy is more highly and quickly available [246]. Nitrogen assimilation from nitrate requires reducing power, usually from photosynthetically derived NADPH, in both leaf and root tissue, thus highlighting the complexity of the interactions between the C and N cycles. Carbon fixation provides not only the reducing power for N assimilation, but also the substrates of the GS-GOGAT cycle for amino acid synthesis through ammonium assimilation. In accordance with this, N availability has been shown to greatly affect metabolic profiles in *L. perenne* [247] and *L. multiflorum* [248]. Genes related to the GS-GOGAT cycle and NADH production have been recently examined in a range of forage crops; *Glutamine synthetase* (*GS*) in grasses [249] including *Lolium* [146; 250; 251], and legumes including *Lotus japonicus* [252], *Glycine max* and *M. sativa* [253-255], and *M. truncatula* [256; 257]; *NADH-dependent glutamate synthase* (*NADH-GOGAT*) in the grasses *Lolium* [258-260], *Eleusine coracana* [261], *Phragmites australis* [262], and legumes [263] including *M. sativa* [264] and *M. truncatula* [265]

Nitrogen reduction (NR and NIR)

After N has been acquired by the plant it then requires assimilation into amino acids. The assimilation pathway includes the key enzymes of nitrate reductase (NR) and nitrite reductase (NIR) [266]. These NR and NIR enzymes are able to reduce N sources for amino acid biosynthesis (predominantly the major amino acids are glutamine, glycine, serine). The complex interaction of N metabolism with carbon metabolism means that the reducing capabilities of the cells are affected by diurnal conditions [267]. Illumination of leaves increases the transcript levels and posttranslational activity of both NR and NIR [268], leading to an accumulation of reduced N, which in turn will be converted to the aforementioned amino acids through the day via the GS-GOGAT cycle [269]. This flux in metabolism leads to depletion of stored N and so acquisition of N through the roots is affected. Furthermore enzymes found in the Krebs cycle (e.g. phosphoenolpyruvate carboxylase, citrate synthase, NADP-isocitrate dehydrogenase) are also upregulated to keep up with the increased metabolic output [269]. As the Krebs cycle receives its carbon source from pyruvate produced by glycolysis from the breakdown of hexose generated from photosynthesis, it is easy to see how such an increase of NR and NIR activity can cause an upstream effect on the whole metabolic function of the plant. Because of this, both NR and NIR proteins are regulated by light signals, and the N cycle transcription factor HY5 [270], with NR being regulated further by reversible phosphorylation [268; 271; 272].

The activity of the NR and NIR enzymes, in conjunction with the accumulation of amino acids (predominantly glutamine), can give a representation of overall plant metabolic rate [273]. Moreover, studies in *Lolium* [274-277] and *Medicago* [253; 278-280] has been well-covered in the literature like with *GS* and *NADH-GOGAT* previously, and making the *NR* and *NIR* genes good potential candidates for future study.

Protein storage (RBCS and VSPs)

Upon assimilation N can be stored in the cell. As grains and fruits are of more widespread agricultural significance there has been a comparatively greater research focus in seed storage of proteins and grain filling. However in terms of forage crops it is important to consider the vegetative storage of protein, as the crop is cut usually prior to flowering. There are two main N storage complexes to note here, RuBisCO and vegetative storage proteins (VSPs), requiring the genes *RBC* (large *RBCL* and small *RBCS* subunits) and *VSPs* for their formation.

Around 50 % of the total leaf soluble proteins are found in the chloroplast in the form of RuBisCO [281] encoded by the *RBC* genes [282; 283]. RuBisCO is the key enzyme in carbon fixation and is an active N protein continuously being regenerated in green tissues. It also shares the same pattern as VSPs where higher amounts are accumulated in N surplus and lower amounts in times of starvation [284]. As RuBisCo is found in the chloroplast it does not require actual storage compartments. Due to their role in photosynthesis, *RBC* genes have been studied across the grass family [285-288].

In 1983, Wittenbach first characterised a VSP whose accumulation increased in leaves even after pod removal in soybean [289]. Since then, the prevalence of VSPs as a main mode of protein storage in many crops has been realised [289; 290]. Soybean has two VSPs of sizes 27 kD and 29 kD; *Arabidopsis* has been shown to also have two homologues of these proteins [291]. Work in both these species has shown *VSP* expression to be regulated by jasmonic acid pre-empting its role in stress responses, a likely characteristic of a storage entity. The breakdown of VSPs can be used to cope with the shortening days and colder environment of winter or

be used after herbivory attack to increase substrates to repair the tissue affected.

Found in leaves, stems and seed pods, VSPs that are strictly glycoproteins show increased accumulation when N is in abundance; these VSPs can also be rapidly degraded on N-depletion [292; 293]. Work has shown that their accumulation and depletion could make them a marker for the cessation of N uptake, such as in the senescence of older leaves [294]. *VSP* genes have been particularly well studied for many decades in *M. sativa* [295-299] with limited work in grasses [300-302].

VSPs and nitrate are stored within vacuoles. It is generally accepted in current plant physiology that plants possess two distinct types of vacuoles, although care should be taken since this may not be universal [303]. These two types are termed lytic vacuoles (LVs) and protein storage vacuoles (PSVs), and can occur individually or in combination in a cell [304]. Lytic vacuoles have been likened to the animal lysosome, an organelle containing hydrolytic enzymes to break down molecules for remobilisation and the storage of cell components. Conversely, PSVs are viewed as the storage capacity of cells with roles including turgor pressure of tissue, ion sequestration, secondary metabolism and, as the name suggests reserve protein accumulation. In terms of forage crops it is the PSV type that is of significance for research to improve NUE. Nitrate is stored within such vacuoles with the help of CLC transporters [305], which could be another candidate marker gene for NUE studies.

The trafficking pathway is highly conserved across all eukaryotes [306] and should be viewed as a highly regulated system rather than a passive flow which ends in vacuole formation. The Golgi apparatus is the main site of regulation and its sorting capacity later determines the type of vacuole produced [307]. This organisation is caused by the vacuolar sorting determinant (VSD) present [308]. These sequence motifs act as signatures and direct the Golgi-derived protein to vacuole formation rather than secretion to the cytoplasm, with a lack of receptor leading to mis-sorting [309]. The major difference in sorting to either vacuole is the use of either clathrin-coated vesicles (CCVs) as in LV formation or dense vesicles (DVs) as for PSVs [310; 311].

Although these vacuoles have differing trafficking pathways and characteristics, it is thought that their status in cells can change due to external effects. For example removing the shoot tip of young soybean plants can push LVs into PSVs, as measured by *tonoplast intrinsic protein* (*TIP*) gene expression [312]. The opposite change has been shown to occur in meristematic cells of germinating tobacco seedlings [313]. In other cases vacuoles with both LV and PSV characteristics have been found in young barley and pea seedlings [314]. For older tissue as with mature vegetation in forage crops the nature of the vacuole is likely to be more defined [315]. TIPs are discussed below.

Vacuole protein pathway (TIPs)

Aquaporins (AQPs) are a diverse family of transmembrane pore proteins [316] whose roles are well documented as regulating the transport of a range of substrates including water, gases such as carbon dioxide and ammonia, metalloids and more complex molecules such as organic acids. Maurel *et al.*, (2015), concludes that AQPs are so widespread in their roles of regulating a vast array of key processes such as lateral root development and turgor pressure control of tissues that they are becoming integrated into multiple fields [317]. Different AQPs are localised to plants at specific cellular membranes, and at various times of development. Tonoplast intrinsic proteins (TIPs), are pore proteins with a distinct role in vacuoles (bounded by the tonoplast), and therefore an essential part of the secretory pathway. The localisation of a TIP to the tonoplast and its function as a specific water transporter pore was shown in 1993; TIP1.1 mRNA expression in Xenopus oocytes induced high membrane permeability [318]. Since then, TIPs have long been investigated for their designated association with different vacuoles with some shown to be linked to PSV [319-321]. Research with resurrection grasses [322-325], have led to more insight into how TIPs may function in forage species [146].

Identification of different families of NUE genes provides a platform to choose potential forage N status marker genes. These could be used to

determine N content of tissue more rapidly than conventional testing, with potential insights into soil N status.

3.1.3 Plants marker genes for a variety of parameters

Forage crop breeding programs are fraught with difficulties, due to individual plants having high genotypic and phenotypic heterogeneity in a cultivar. Moreover many species are polyploid, a problem that is exacerbated within hybrid grass populations. Few agronomic traits are linked to a distinct gene or number of genes [326; 327]. Recent studies have focussed on specific legumes [328-332] and grasses [333-335; 332; 336; 337], which can provide a wealth of potential genes of interest for NUE similar to, and including, those described in the last section.

One of the most prominent breeding successes in forage crops was the exploitation of closely related species of *Lolium* and *Festuca* to create hybrid *Festulolium* cultivars [338; 339; 13; 3; 340; 341]; backcrossing using conventional breeding methods of *Festulolium* generated novel hybrids with more stable protein content when compared to parental lines. Advances in phenotyping may make it easier to quantify of characteristics in the breeding programmes, such as high level imaging of crops to determine later traits especially biomass [342].

Additionally, new forage cultivars are being generated through advances in transcriptomic data with more sequences available for breeders to exploit [343-347]. This has included the identification of SNPs in *Lolium* [335; 348] and *Trifolium* [349-351] which may be used in programmes to improve on breeding of new cultivars. Moreover, draft genomes for forage crops are increasingly common [352-355; 346; 356-358; 215; 359-361; 348; 337]. There are many advances using such resources [18] including some in *M. sativa* where QTL and gene analysis have been used to investigate lodging resistance and vigor [362], plant height and regrowth following harvests [363], flowering and stem height [364], and drought tolerance [365]. *Lolium* has similarly been investigated with improvements in biomass increases [366; 367] as well as tracking of such phenotypic traits in programmes [368]. In addition many potential genes have been

identified in model plants that may become the targets for gene editing techniques [369; 370; 336], or the focus of TILLING populations [371; 372; 332], in forage crops.

Despite the above advances within breeding systems, there is a lack of the direct use of genetic information by the grower. Although, new cultivar breeding is crucial for the industry to progress in terms of both yield and efficiency terms, there is a huge wealth of genetic information that might be used by the grower themselves.

A genetics approach assessing crop efficiencies during growth could be investigated for multiple forage species, with comparison to current testing methods. This method could test the expression pattern of NUE related genes to see if these can be used to infer plant nutrient status, particularly as N status marker genes. These may also give information about soil N levels. Soil N is likely to be the biggest factor affecting crop yield, with the next important factors including pathogen responses and drought tolerance. Having one measurement to accurately assess both crop and soil N status may offer the prospect of future yield estimation or a guide for changing management practices to improve yields. The current N status testing methods of both crop and soil are discussed next.

3.1.4 Current testing for soil N lacks spatial and temporal detail

There are very few measurements of NUE in forage crops [373; 374]. At present forage growers cannot easily and reliably determine the N status of their crops. Most commonly used methods are biomass recording at the end of production periods (at cutting times for forage crops), Soil-Plant Analyses Development (SPAD) measurements for chlorophyll, or costly plant tissue nutrient analysis including protein assessments using the Kjeldahl method. The first method of biomass recording is passive, and information can only be used for next cultivation periods, whilst the second two are time-consuming, costly, and may have variable results based on how samples are collected and assayed. Plant samples can only be taken periodically and must be done carefully to increase the accuracy of

measurements across a whole site. Some research has focused on the use of spectral data to evaluate crops [375], but such techniques require further investigation and can give false readings caused by pathogen attack. It may be better if the farmer could determine the crop N status directly and then make an informed decision as to how they should subsequently fertilize the plot. This would enable more efficient targeted fertiliser use, thus increasing forage biomass with lowered costs and is the underpinning aim of precision agriculture. Better strategies of crop testing should be developed to reliably inform the grower of nutrient status to improve NUE efficiency.

For maximum biomass production, it is important to maintain the N status of the crop throughout the growing season and this requires an optimised soil N supply [205], particularly in coordination with cutting regimes. Furthermore, application of too much N fertiliser results in wasteful runoff, and sub-optimal supply results in decreased biomass production; Inefficient fertiliser use is of environmental, agricultural production, and economic concern. Maintaining N supply for maximal yield is limited by two factors: 1) an easy, consistent measure of the crop's status and 2) reliable and reproducible tests for soil N levels [376].

Presently farmers, or fertiliser suppliers, take limited samples of soil across their growing area in the hope that this is representative of the N in the whole plot through the growing season. Not only is the representative N level tested for a field usually spatially limited with only a small section being measured, it is also temporally lacking as costly soil analysis is not performed on each field each year. For example, a grower with a large farm may only have the resources, time and funds to test the N level of each field every other year, at best. In addition, although soil N level is strongly linked to plant production, it may not always be the best indicator of a plant's N status a useful measure of NUE.

3.1.5 Aims of this chapter: Testing N status marker genes in forage crops

This chapter describes work investigating the development of N status marker genes in the forage crops *L. perenne*, *Festulolium* and *M. sativa*. This may be useful for forage growers to assess both their crop's growth, as well as the soil N status. These techniques may be used in breeding programmes. Such method development can be tested in the glasshouse, and then verified in the field to decide on the suitability of its use.

To assist in the choice of N status marker genes to test, there is scope for the use of the molecular markers outlined above, with the future possibility of a PCR test for the optimal time of harvest based on the expression of candidate genes like storage proteins. Such tissue testing of crops can also be used for decisions on the timing of fertilizer applications as the two evaluations can be made around the same time. There is scope to identify a suite of marker genes to help decide when these key decisions are made. The aim was to identify if such a suite could be used in prominent UK forage species in the glasshouse alongside conventional testing methods. This would then be trialled in the field to see if such a method is utilisable by forage farmers.

3.1.6 Materials and Methods

The methods used in his chapter are detailed in Chapter 2, in section '2.1 N STATUS MARKER GENES'. Supplemental data is provided in Appendix A.

3.2 RESULTS

3.2.1 Nitrogen status candidate genes have different expression patterns in *A. thaliana* in relation to NO_3^- supply

A literature search was undertaken to find suitable candidate genes for N status in forage crops. The search began with considering *A. thaliana* genes with good characterisation and included those known to be important in C:N metabolism, or vacuole transport. The *A. thaliana* genes were also checked for expression with parity across vegetative developmental stages using the *Arabidopsis* eFP Browser [130] (Figure A1, Appendix A).

Expression pattern testing of these 12 candidate genes with *A. thaliana* was carried out under different NO_3^- treatments. Nitrate was used as the N source as usually it is the most abundant soil form of N for plants, whilst also being relatively easy to control compared to other N sources. Individual candidate genes were shown to have different patterns of gene expression under NO_3^- response (see Figure 3.1). Most of the genes showed an increase in relative expression with increasing NO_3^- concentration, including *CLC* (Figure 3.1 a), *GLN2* (Figure 3.1 f), and *VSP1* (Figure 3.1 g). Many others also increased in their expression, however reached a peak at 10 mM NO_3^- with a drop in relative expression fold change at 30 mM; see *NIR*, *RBCS2B*, and *NADH-GOGAT* (Figure 3.1 b, c, and d respectively). This suggests that for *A. thaliana* the optimal concentration of NO_3^- for growth is below 30 mM, and above this point there is a negative impact on growth.

Moreover the results showed that these effects are independent of the cation present in NO_3^- treatment, as expression patterns for KNO₃ tests were equivalent to $Ca(NO_3)_2$. All *TIP* genes showed a decrease in their expression fold change in relation to increasing NO_3^- concentration, see Figure 3.1 i)-l). This assays shows how a number of genes all affected by N status of environment can have different patterns of expression. This variation in expression pattern will be investigated in forage crops.





Expression was calculated as the geometric expression of independent Ca(NO₃)₂ and KNO₃ experiments relative to lowest NO₃⁻ concentration (0.3 mM shown as 100 % in pale green). Bars are denoted for increasing concentration using darkening shades of green, and error bars are standard deviation of expression using two reference genes with 12 biological replicates, in two independent experiments. Expression patterns of genes varied dramatically in relation to NO₃⁻ concentration; a)-h) shows those genes which are generally upregulated with increasing NO₃⁻ concentration; *CLCa*, *NIR*, *RBCS2B*, *VSP1*, *VSP2*, *GLN1*, *GLN2*, *NADH-GOGAT*; i)-l) shows genes downregulated; *TIP1.1*, *TIP1.2*, *TIP3.1*, *TIP3.2*.

3.2.2 Model species sequence alignments can be used to investigate forage crops

Through compiling known and predicted leguminous dicot and monocot gene coding sequences it was possible to determine which candidate N status marker genes tested in *A. thaliana* (Figure 3.1) could be closely related to forage crops with a similar sequence for use. A phylogenetic tree was built using the compiled sequences can be seen for *Nitrite Reductase* (*NIR*) sequence alignment in Figure 3.2 below. Comparing known and predicted sequences for related species is important as many forage crops including *M. sativa* and *L. perenne* (shown in bold in figure), and *Festulolium* hybrids do not have fully annotated genomes published at present. Candidate genes with a high degree of sequence similarity to genes fully or partly characterised in model species were more likely to be usable in these crops. The candidate genes with high sequence similarity by nucleotide alignment were *NIR*, *GLN*, *RBCS*, *NADH-GOGAT*, and *TIP1* (*MCP1* in *M. truncatula* homologue for legume sequences). Reference genes were also compared across species with alignments.

Consensus regions from sequence alignments was used to develop primers to test with forage crops in glasshouse conditions. Figure 3.3 shows a pipeline for using the high similarity sequence alignments to find conserved areas to design gRT-PCR primers. The pipeline was used for grasses and legumes separately to generate primers to test the forage crops found in Table 2.4 and 2.5. These primers were tested using PCR to check for predicted band sizes, as shown in Figure A2, Appendix A. The percentage of correct size bands for species indicated how well the conserved sequence area used for primer design was actually conserved across forage grasses and legumes; grass primers were ~ 56 % utilisable in species tested, and legume primers were ~ 71 %. Correct expected size bands were sequenced for both *M. sativa* cv. Daisy and *L. perenne cv.* AberMagic to check sequences matched the expected genes. Primers were checked for annealing efficiencies between 90 - 110 %. These N status marker primers were for the genes NIR, RBCS, NADH-GOGAT, and TIP1 in grasses, and NIR, GLN, RBCS, NADH-GOGAT, and MCP1 in legumes.



<u>Figure 3.2:</u> Phylogeny of temperate forage and related model species for *Nitrite Reductase* (*NIR*).

Species included were many model dicots including *Arabidopsis thaliana* shown in blue, with those that are legumes highlighted with light blue. Monocots were displayed in orange, and *Chlamydomonas reinhardtii* with *NII1* was used as an outlier. The model species *Medicago truncatula* and *Lolium perenne* with interest for forage crop research are shown in bold.



Figure 3.3: Pipeline for candidate gene primer design for forages crops.

Primers to test in forage crops were designed as follows; 1. Multiple sequence alignment was built using MAFFT v7 [142] for a gene using available and predicted legume or monocot sequences (in the above example this is *nitrite reductase* nucleotide coding sequences in monocots closely related to forage grasses); 2. A consensus sequence was added with MSAViewer [144]; 3. Most conserved section of sequences identified; 4. Primers were designed in silico [145; 127; 125; 126]; 5. Primers were tested with RT-PCR for a range of relevant species; 6. Primers for conserved consensus area were tested for efficiencies and used in future analysis of forage crop.

The N status marker gene primers were tested using qRT-PCR in forage crops grown under different NO_3^- treatments, like with *A. thaliana* in Figure 3.1. The forage crops tested were the grasses *L. perenne* and *Festulolium*, and the legume *M. sativa* (cv. Daisy and Luzelle). These species were chosen due to their high yield and common use by UK growers (including BAGCD members), as well as their prominence in temperate forage research globally. If a strong relationship between gene expression and plant N status can be found within glasshouse grown forage crops, then in-field sampled tissue could also be tested.

3.2.3 N status genes also showed different gene expression patterns in grasses in relation to NO₃⁻ supply

Gene expression was tested at three weeks after the NO₃⁻ treatment. At six weeks the total vegetative yield was measured as the aboveground biomass. Grasses were tested under a range of nitrate concentrations like *A. thaliana*, to mimic potential differences in supply occurring during the growing season that result from management practices. For example, if soil acidity is a problem a grower may add less N fertiliser in some fields, causing a likely drop in soil NO₃⁻ concentration. As *M. sativa* is a nodulating legume it was only tested under a small range of NO₃⁻ treatments; these treatments were used to guide for different future yield rates. It should be noted that growers do occasionally add N fertilisers to legumes to aid establishment at the beginning of the growing season, especially in mixed pastures.

Testing of N status maker gene expression of the grasses *L. perenne* and *Festulolium* showed were similar patterns to those seen for the NO_3^- treatments in *A. thaliana*. These expression patterns are shown in Figure 3.4 alongside later biomass vegetative yield. The genes *NIR*, *RBCS*, and *NADH-GOGAT* again showed an increase in their expression with increasing NO_3^- concentration, until a peak at ~ 10 mM whereupon expression decreases. This decrease in expression in the higher concentrations also matches with a slight decrease in later yield shown in t ha⁻¹. As with *A. thaliana*, *TIP1* showed a decrease in expression with increasing NO_3^-

concentration. This downregulation pattern of TIP1 can be useful in conjunction with the other upregulated genes. Conversely, M. sativa cultivars did not have patterns of expression that were easily related to NO_3^- supply and subsequent yields, as shown in Figure A3, Appendix A. As mentioned previously, this may be due to N fixation occurring during low supply, it being a nodulating legume. A lack of clear patterns may be due to the narrow range of NO_3^- concentrations tested in the glasshouse, however, testing anymore may not be relevant to the situation in the field. As the data suggests that N status marker genes would be most useful to test in grasses, it was decided that more glasshouse Festulolium testing should be carried out with an increased number of NO3⁻ treatment concentrations. This was done to obtain more data points within the expression pattern ranges, but also as *Festulolium* is grown across a wide range of UK sites, and frequently in numerous fields on one soil type but with variable nutrient supply. Festulolium was used in experiments due to it being a commercially relevant crop, and although other grasses especially Lolium or Fescue cultivars are interesting to study, the hybrid vigour of *Festulolium* makes it most relevant to the UK forage industry. Fourteen NO_3^- concentrations ranging from 1 – 39 mM were tested, with both 3-week gene expression and 6-week vegetative yield measured as before. The results of this expanded assay are found in Figure 3.5. By increasing the number of concentrations tested it was clear to see that both NIR and NADH-GOGAT in Figure 5 a) and c) have increased expression until around 21 mM treatment, which matches the drop in later yield. For example, for NIR measurements 19, 21, and 23 mM treatments relative expression drops from 4500 % then 1200 % to 250 % respectively, with a change in yield of 7.12 t ha^{-1} , then 7.2 t ha^{-1} , down to 6.3 t ha⁻¹ respectively. *RBCS* also decreased expression at the higher ranges shown in Figure 5 b), however this is at 23 mM and the decrease in transcript is more variable with larger error bars at 26 and 29 mM nitrate. This may be because RBCS is highly expressed in all plant vegetative tissues, and although soil NO₃⁻ concentration does affect its expression, the differences are less apparent and more variable at higher supply. As before, *TIP1* shows decreased expression with increasing NO₃⁻ concentration. The expression patterns found for NIR, NADH-GOGAT, and

TIP1 under different NO_{3}^{-} concentrations in glasshouse conditions were consistent and sufficiently reliable to be used to test field samples.

Lolium perenne cv. AberMagic



Figure 3.4: Gene expression patterns of N status marker genes in grasses *Lolium perenne* and *Festulolium* hybrid vegetative tissue at three weeks. Expression was calculated as the geometric expression percentage of independent $Ca(NO_3)_2$ and KNO_3 experiments relative to lowest NO_3^- concentration (0.3 mM shown as 100 % in pale green). Bars are denoted for increasing concentration using darkening shades of green, and error bars are standard deviation of expression using two reference genes with 12 biological replicates, in two independent experiments. Expression of genes was shown to vary dramatically in relation to NO_3^- concentration with good agreement between species. Genes tested in grasses were; a) and e) is *NIR*; b) and f) is *RBCS*; c) and g) is *NADH-GOGAT*; d) and h) is *TIP1*.





<u>Figure 3.5:</u> Gene expression patterns of N status marker genes in *Festulolium* vegetative tissue at three weeks, with yield photographs.

Expression was calculated as the geometric expression percentage of independent $Ca(NO_3)_2$ and KNO_3 experiments relative to lowest NO_3^- concentration (1 mM shown as 100 % in pale green). Bars are denoted for increasing concentration using darkening shades of green, and error bars are standard deviation of expression using two reference genes with 12 biological replicates, in one independent experiments. Expression pattern of genes varied dramatically in relation to NO_3^- concentration, with later yield also plotted (purple dots). Genes tested were; a) *NIR*; b) *RBCS*; c) *NADH-GOGAT*; d) *TIP1*. Photographs show visually the yield of *Festulolium* effect by different N concentrations at six weeks.

3.2.4 Glasshouse testing of N status marker genes in vegetative tissue could be used to assess field samples using a Gaussian model approach

To determine whether the expression of these candidate N status genes was relevant to samples taken from the field a round of preliminary testing was carried out. The aim was to compare the results from the first glasshouse screen of gene expression patterns, in Figure 3.4, with field samples collected in 2018. At the beginning of the growing season, samples of *Festulolium* cv. Hykor were taken from six individual fields at a BAGCD site. Samples were taken in triplicate for measuring gene expression during growth period shortly after fertiliser application, but also for chlorophyll and protein levels. Soil samples were taken at the same time to analyse extractable NO₃⁻. Raw gene expression values for *NIR*, *NADH-GOGAT* and *TIP1* of 2018 field samples were calculated in relation to two reference genes. Gaussian process models were fitted to the glasshouse data generated in Figure 3.4 and used to analyse this field expression data. The results of this can be found in Appendix A, Figures A4, A5, A6, and A7.

A second round of field sampling was implemented using crops from the 2019 growing season and with more fields available to sample. Again a Gaussian process model approach was used for this dataset, using the glasshouse measurements from Figure 3.5; these can be found in Appendix A, Figures A8, A9, A10, and A11. As this was also successful, the 2018 and 2019 field datasets were combined and analysed together using a model based on all the glasshouse measurements. The results of this analysis are discussed next, with Figures 3.6, 3.7, 3.8, 3.9 and 3.10.

The glasshouse gene transcript measurements from Figures 3.4 (2018 in red) and 3.5 (2019 in blue) were used to calculate individual normalised gene expression patterns relative to the two reference genes (Figure 3.6). This was called the 'mean.delta.ct.norm' for *NADH-GOGAT* ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') respectively. These were plotted against measured soil NO₃⁻ concentrations in mg/kg, referred to as 'NO3'. One can see the gene expression patterns as previously discussed; *NADH-GOGAT* and *NIR*

show increases in relative expression with increasing NO_3^- concentrations, until a drop after ~ 50 mg/kg, which is comparable to 19 – 21 mM NO_3^- treatment, and *TIP1* has a steady decrease in expression relative to increasing concentration.

Figure 3.7 shows how a Gaussian process model were fitted to each gene transcript individually. This generates a mathematical expression for the estimated gene expression level ('yhat') as NO₃⁻ concentration in the soil ('xpred') varies for each gene. y is the mean of all gene expression values for 'mean.delta.ct.norm' as in Figure 3.6. In the case of glasshouse data, 'xpred' is the measured concentration of soil NO₃⁻. However when testing field samples, this will be unknown and the model will interpret the raw geometric expression data for each gene to predict the most likely soil NO₃⁻ concentration, which will be checked against later laboratory analysis.



Soil nitrate (mg/kg)

<u>Figure 3.6:</u> Normalised expression data 'mean.delta.ct.norm' of 2018 and 2019 glasshouse grown *Festulolium*.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data was calculated from glasshouse measurements in relation to two reference genes, to generate values for 'mean.delta.ct.norm' against known soil 'NO3'. Original glasshouse data as relative fold change expression is found in Figure 3.4 for '2018' (red), and Figure 3.5 for '2019' (blue).



<u>Figure 3.7:</u> Gaussian process models of 2018 and 2019 glasshouse grown Festulolium.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data generated from glasshouse data with a Gaussian model applied to individual genes; y-axis ('yhat') shows experimental and predicted values for 'mean.delta.ct.norm' as in above figure; this is required for *NADH-GOGAT* and *NIR* to ensure the model can distinguish the low NO₃⁻ and high NO₃⁻ despite the 'mean.delta.ct.norm' being similar in range. The x-axis ('xpred') is the measured soil NO₃⁻ concentration derived from greenhouse experiments; when field data is analysed with the model it is the 'xpred' which will be deduced from the raw geometric expression data provided. Error region in light grey shows the 95% confidence interval of the model.

Next, all field geometric expression data was assessed against the Gaussian process models generated in Figure 3.7. Raw expression data was uploaded for each field, and compared against the model predicted expression of *NADH-GOGAT* ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') (Figure 3.8) at each candidate NO₃⁻⁻ concentration. The value for the normalised gene expression is shown on each graph as a red dot, at the most likely 'xpred' value. For example, for Field no. 58 the normalised gene expression, for *NADH-GOGAT* was ~ 1, for *NIR* ~ 0.5, and for *TIP1* ~ 2. This gives a predicted soil NO₃⁻⁻, or 'xpred', of ~ 12.5 mg/kg. All fields are placed in order for their 'xpred' value, and so from top left to bottom right the normalised gene expression calculated in the model corresponds to higher predicted soil NO₃⁻⁻; Field no. 58 – 76 are low predicted soil NO₃⁻⁻, Field no. 12b – 67 are in the middle range, and Field no. 49 – 16 are high predicted soil NO₃⁻⁻.

Finally, the analysis of individual gene models shown in Figure 3.8 can be combined to give a single bar chart of each field, provided in Figure 3.9. For each possible NO_3^- concentration (x-axis), this shows the model estimated probability that the gene expression of the three genes arose from a plant grown at this NO_3^- (y-axis), based on the similarity of field gene expression to data generated in the glasshouse. The probability is calculated for each candidate NO_3^- for each field.

Therefore the y-value is an indication of the confidence of the model (the higher and thinner the distribution the greater the confidence in predicted soil NO_3^{-}). For example Field no. 58 shows a relatively tall and thin bar, y ~ 0.4, with an output of x = 7 – 14 mg/kg. This again suggests Field no. 58 has a low soil NO_3^{-} concentration, also suggested for Field no. 12, 31, 93 and 78. In contrast, Field no. 82, 49, 31, and 16 have lower bars skewed across the upper range of predicted soil NO_3^{-} , indicating these field have a high concentration above the range for optimal growth found in glasshouse assays, see Figure 3.5. This analysis shows how raw gene expression of three genes measured with plant tissue, relative to two reference genes, can be used determine a value for predicted soil NO_3^{-} .



Figure 3.8: Analysis of individual gene models for field grown *Festulolium*. Field sample expression data was calculated in relation to two reference genes and the normalised gene expression for each gene calculated. Gene expression at varied NO₃⁻ was predicted using Gaussian process models fit to glasshouse gene expression data, found in Figure 3.7; 'yhat' is the mean values for 'mean.delta.ct.norm', and x-axis 'xpred' is predicted soil NO₃⁻. Red dot shows the gene expression, and maximum probability NO₃⁻ concentration for each field.



Figure 3.9: Analysis of combined gene models for field grown *Festulolium*. Field sample combined normalised gene expression analysed with glasshouse generated Gaussian model. Here y-axis is P(D|NO3)' which is the estimated probability that field gene expression values would be produced at 'xpred' values shown. The x-axis is again 'xpred', here termed 'predicted equivalent greenhouse NO3'. This provides a visual representation of predicted concentrations in the field soil with the higher the 'P(D|NO3)' bar points the more reliable the interpretation of predicted soil NO₃⁻. The x-axis shows if the plots are skewed for x < 25 this means predicted soil NO₃⁻ is low, between x = 25 - 50 then predicted soil NO₃⁻ is within an optimal range for yield, and if x < 50 then concentrations are deemed higher than required for efficient growth, based on glasshouse yield observations.

3.2.5 N status marker genes can be comparable to current conventional testing, providing information of both crop and soil status

As Figure 3.9 shows how gene expression of three N status marker genes can be used to predict soil NO_3^- , it is necessary to compare these to actual soil NO_3^- concentration measurements. Moreover, if predicted soil $NO_3^$ matches actual measurements from soil extractions sampled on the same day as tissue for expression, it will be interesting to see if there is a later correlation to yield of each field measured in the subsequent weeks. Figure 3.10 presents three datasets for field samples. At the bottom of the figure, discrete field 'P(D|NO3)' against 'predicted equivalent greenhouse NO3' is shown from Figure 3.9, in order from low predicted soil NO_3^- to high, indicated with text and purple arrows. Above these are graphs of actual measurements for extractable soil NO_3^- as mg/kg (purple dots), and later yield as t ha⁻¹ (blue squares).

Essentially, if one compares the predicted model generated bar charts for soil NO₃⁻ against the actual extractable soil NO₃⁻ there is a high rate of agreement in the measurements; the model correlates with actual measurements for soil NO₃⁻. Moreover, in Appendix A, Figure A12 shows the actual extractable soil NO₃⁻ correlated to predicted soil NO₃⁻ based on the highest 'P(D|NO3)' bar point, showing an R² = 0.92.



<u>Figure 3.10:</u> Extractable soil NO₃⁻ and *Festulolium* yield for fields from 2018 and 2019, with comparison to combined gene models generated with glasshouse data.

Extractable soil NO_3^- as mg/kg sampled at time of expression sampling is shown in top graph as purple circles ('Soil NO_3'), and crop yield in t ha⁻¹ measured ~ 3 weeks after sampling is shown below in blue squares ('Yield'). At the bottom, 'P(D|NO3)' is the estimated probability distribution across predicted greenhouse equivalent NO_3^- concentration. Plots provided from Figure 3.9 to compare to values for 'Soil NO_3' ' and 'Yield'.

3.3 DISCUSSION

3.3.1 Testing N status candidate genes provides more information of how NO₃⁻ environment can affect NUE associated gene expression

This chapter showed that the candidate N status marker genes in *A. thaliana* have clear patterns of expression between various NO_3^- treatments (Figure 3.1). This has confirmed the expression pattern reported in the literature for some of these genes under different NO_3^- concentrations in an independent study. For example, both *NIR* and *NADH-GOGAT* showed similar expression patterns in *L. perenne* and *Festulolium* to those described in *A. thaliana* [377; 378; 134; 133].

The discovery that TIPs were downregulated pattern was unexpected. As some TIPs are vacuolar markers for protein storage vacuoles (PSVs) and are well documented as mediators of water and ammonia transport [316; 379; 380], it was speculated that they would increase in expression following higher NO₃⁻ supply levels due to increased PSV numbers in the tissue. However as the data does not support the prior assumption that PSV numbers will increase with higher nitrate supply it is possible that although the TIPs are important in vacuolar transport, they may be important for not just PSVs but also LVs [320; 381; 382]. Therefore, turnover of substrates depending on whether the vegetative tissue tested is source or sink. Nevertheless, it is unclear how specific individual TIPs function, with TIP1s and TIP2s both being association with large LVs and PSVs [319; 383], with cellular specificity rather than tissue. More exploration of TIPs in monocots could provide answers to these specificity questions, as *A. thaliana* has been the primary focus of their investigation so far.

In addition to this, the vegetative tissue assessed in this study is only three weeks post germination. This means there may be high differences between expression in young sink leaves and older source leaves. This

would be true for *A. thaliana* studies, but as a single grass leaf has both old and young tissue present on sampling this could be worth further investigation for protein storage in monocots, as well as general water movement. This work further supports the idea that more investigation of TIP function is needed. Areas of TIP function to be addressed before assumptions can be made regarding their specificity across plant tissues include whether TIPs are localised to PSVs or LVs, and how strictly localised they are in certain tissues of different ages.

The development of N status marker genes in forage crops provides tools for future investigations. Firstly, the design and use of primers for *L. perenne*, *Festulolium* and *M. sativa* could be used in gene expression studies testing parameters of fertiliser application, rate, and regimes. These primers are compatible in both glasshouse and field studies, and so could be an excellent tool for breeders in targeted NUE programmes. This was not true for *M. sativa*, where N-fixation by the legume may be complicate the picture. To test this idea, it would be interesting to check the pattern for expression of my suite of marker genes when the legume is grown in aseptic culture when N fixation is not possible and increasing NO₃⁻ is supplied.

The results of the preliminary round of field testing were positive, especially for using plant gene expression data to infer soil NO_3^- status without need for soil extraction. This approach was successful in transferring a technique established in the glasshouse to the field. Success could be measured statistically by how well the model fitted the combined field sample data.

This means that the sampling of vegetative tissue at three points of a field and analysing its expression of only three N status marker genes is enough to accurately determine its actual soil NO_3^- without need for laborious soil extraction. This information could be used to aid management practices, for example apply fertiliser to a low concentration field as the plant is growing to increase yields before harvest.

In addition, it is possible to also compare the predicted model bar charts for soil NO_3^- against the actual future yield of each field in t ha⁻¹. Although this does not correlate as strongly as actual extractable soil NO_3^- , one can
see a general trend where the higher the predicted soil NO₃⁻ then the higher the subsequent later yield of the field (see Figure 3.10). This is not surprising, as a field with a better N status should be able to support higher yield of vegetative crop. However the correlation is not strong enough to support its own model with outliers in yield found including Field no. 12 (2019), 49 (2018), 88 (2019). This is likely due to more convoluting factors affecting yield in the field when compared to glasshouse measurements. Moreover, no correlation for extracted vegetative chlorophyll or protein could be found with yield, shown in Appendix A Figure A13, suggesting more factors are at play in determining yield variations. There also seems to be no significant drop in yield at the higher soil NO₃⁻ concentrations found in glasshouse assays.

3.3.2 Fields are variable for both soil N and crop yield

The high spatial variability of soil N is known to affect its biological communities [384-390]. This heterogeneity is shown in Figure 3.10 where both extractable soil NO_3^- and *Festulolium* yields have large ranges of measurements; soil NO_3^- between fields ranged from 19.44 - 68.5 mg/kg based on three samples per field, and crop yield ranged from 1.28 - 4.03 t ha⁻¹. These large ranges are found on a relatively small farm area, see Chapter 2, Figures 2.1 and 2.2 for fields sampled, despite the fields having similar soil types and texture, and farming management practices, including fertiliser application rates were parallel.

The variation found in soil N levels is likely to be caused by the natural heterogeneity of land even at small scale found especially in temperate regions such as the UK [391-397]. Such variation makes the low frequency of current sample practices for UK dried forage producers (often just 3 - 5 soil cores per field for BAGCD members [398]), inadequate for grower needs where precision farming techniques are becoming more desirable.

The high variation in field yields measured in this study shows how forage cropland and grassland are affected not only by differences on soil nutrient levels especially N heterogeneity [399; 391; 400]; soil NO_3^- and field yield have different coefficients of variation at 40.13 and 29.27 respectively. It

also indicates how other factors are likely to influence forage yield, including soil acidity [401; 104; 402; 403], water availability [404; 405; 205], and disease susceptibility [406-409]. Interestingly correlation of vegetative yield with chlorophyll and protein concentrations found in this study may support the usefulness of fluorescence with drones and SPAD meters for decision support, but the precision of lab-based measurements may not be matched by these field monitoring methods. Moreover, as such factors are interconnected especially when the complication of climate change is included. Many studies are beginning to explore how to model these factors integrating their effects on yield [410; 411].

3.3.3 Refining N status marker gene suite with model provides promising developments for grass research

Despite similarities in cultivation, the measurements for soil N and yield drastically highlight the need for growers to be able to assess individual fields when required, and not be trapped with conventional methods which are costly and time-consuming [376], even if they are necessary to comply with regulatory standards [412; 413; 79; 414; 415]. Moreover as heterogeneity within individual fields is so large methods that allow for rapid sampling at a few points across an area provides a more accurate representation of field soil status.

In this study it has been shown how a relatively non-invasive technique measuring a small amount of plant vegetative tissue from three field points can assess soil N levels rapidly across many fields. Although requiring scientific skill to conduct, the measurements for soil NO₃⁻ have been shown to be robust using gene expression models built using glasshouse measurements. This data could be improved through more testing, particularly in the field. Although some training would be required, the current testing of soil extracts especially using soil suction cups has limited precision and scope. Crop growers including BAGCD members are increasingly wanting to have more exact tools for assessing their crops and soil, so such techniques are gaining more practical need. Such techniques are also incredibly applicable to breeding programmes who already have a

wealth of scientific training. Moreover, there is continued development showing use of handheld PCR kits in the field [416-419] which are easy to use for a small number of parameters; such a technique may be utilisable for the current suite of three N status marker genes alongside two reference genes. Of course the downside of using the suite of marker genes in any crop is that when the PCR method detects a sub-optimal soil nitrate supply this is already too late for the crop and the maximal yield will not be achieved. But this criticism is valid for any methods that detect a response in the plant.

It would also be interesting to see how other cultivars vary in their marker gene expression related to soil N levels. Variation could be seen in different *Festulolium* cultivars due to their breeding history [420; 2] as can be seen in Figure 3.4 with its slight N specificity differences to *L. perenne*. This suite of genes may potentially be important markers for NUE in forage grasses that can be used in breeding programmes.

This page is intentionally left blank

Chapter 4: Soil sensors can detect N profile changes under forage crops with different management regimes

4.1 INTRODUCTION

4.1.1 Nitrogen fertiliser production for crops is

demanding

The Haber-Bosch process for fixing free nitrogen (N_2) from the air and converting it into ammonium (NH_4) is arguably the most dramatic invention in modern human history. Along with the breeding of dwarf varieties of the main cereals and developments in herbicide and pesticide production, this fixation process was part of the Green Revolution which allowed crop production outputs to increase by a dramatic four-fold [421]. Due to its contribution to human history with a huge increase in crop production causing an increase in human population, the technique has been termed the 'detonator of the population explosion' [422].

Fritz Haber and Carl Bosch's process was developed from Haber's 1908 "synthesis of ammonia from its elements" patent [423]. The work won each of them a Nobel Prize. More notable than its scientific importance in the field of chemistry, ammonium fertiliser production changed the impact of the late-industrial age globally by feeding 50 % of the world population through increased grain yields [424]. The process converts atmospheric N₂ to NH₄ under high temperature and pressure and in the presence of a catalyst, typically iron-based. It should be noted that despite this incredible development in fertiliser production, the Haber-Bosch technique was originally developed in Germany with ammunitions as its motivation, and ironically Haber's work can also be linked to the deaths of hundreds of thousands of people [425; 426].

Anthropogenic activities fix 210 Tg N year⁻¹ [427] with total N input for food production globally at 171 Tg N year⁻¹, half of which is accounted for by the animal products industry [428]. The process is resource and financially exhaustive; the practice uses 3 - 5 % of the world's natural gas, and 1 - 2 % of its energy [429]. Recent studies have aimed at increasing efficiency of N fixation by manipulating the Haber-Bosch technique with specialised catalysts and plasma electrode fields [430-435], but still global N demand continues to increase. This is also being stretched by its

potential use in transport fuel and water space heating with further technological advances on the process multiplying [436]. Emissions of N from fertiliser are estimated to be at around 10 % of global outputs [437], with agricultural emissions from animals and manure being responsible for the bulk of this loss [438]. Emissions had previously been shown to be highest in Asia, predominantly China and India, and Europe [439; 9]. However. other areas of the world are now catching up due to increases in synthetic fertiliser use intensity [393; 427]. Western Europe including the UK uses 7.3 % of the world's synthetic N consumption [440], which is disproportionate with its landmass size.

This resource consumption and subsequent emissions have dramatic consequences for the world. The vast energy required for N fertiliser production puts a strain on energy demands, and using a huge supply of natural gas directly contributes to our unsustainable dependence on fossil fuels. This is directly beginning to influence governmental policies and is continuously monitored in countries including the UK [77; 78; 394; 441; 414; 81; 82; 442; 83; 415]. There is a political and financial aspect to using fertiliser more efficiently.

There are serious environmental concerns regarding N fertiliser use. Leaching of N leads to eutrophication of water supplies where algal blooms in rivers and lakes limits sunlight, space and oxygen for aquatic species and therefore leads to high death rates in these ecosystems [443; 444]. Such leaching can also contaminate human drinking water especially in ground water supplies. Forage grass and legume crops have long been thought to have high rates of leaching to the environment [413]. This problem can be exacerbated when they are grown in sandy soils, the choice for many forage crops grown on BAGCD sites. Some estimates are that 60 % of applied N is lost through leaching, run-off, denitrification and consumption by microbial populations. Furthermore areas termed Nitrate Vulnerable Zones (NVZs) make up approximately 58 % of the land in England [77; 414; 81; 82; 445]. These areas include farms used by forage growers such as the BAGCD.

There are also arguments for an impact of N-fixation on human health issues including infantile methemoglobinemia (blue baby syndrome) and

the formation of N-nitroso compounds that are potential carcinogens, formed by commensal bacteria from nitrate, although there is much need for further studies into such effects [446]. The impact of drinking-water being contaminated with leached N on human health may results in decreased life expectancy [446-448]. Assessment studies for the social cost of nitrogen (SCN) have been implemented [445]. Finally there is growing evidence that N emissions, which are 0.5 - 1 % of production inputs [440; 449; 450], may contribute to climate change, as a by-product of fertilizer production [451-457]. Nitrogen oxides (N₂O and NO) may be the single largest contributor to atmospheric greenhouse gases [458; 456; 459; 460]. Ammonia emissions are most relevant to grasslands and have been investigated in the UK [461; 462; 458]. These emissions can be produced as a by-product of N fertiliser use through microbial breakdown. Increases in global demand for protein are estimated as 110 % \pm 7 % [463]. This will undoubtedly exacerbate the current problems with N fertiliser use as an increase in food production is required [464]. The rate of N fertiliser use in the UK compared to global consumption has recently declined slightly by almost 1 %, although consumption still stands at a forecasted 7.13 % for the next decade [449]. The future use of fertiliser is of increasing concern for crop growers worldwide. Environmental concern over N fertiliser use by farmers has been hotly debated due to the potential adverse effects of leaching into ecosystems compared with the economic advantage for production. Moreover, there is growing pressure to impose legislation and regulations on farmers due to these detrimental effects. Negative associations between fertiliser use and crop production are increasingly recognised by the public.

Despite synthetic N fertiliser use being a major factor in the global population boom, these negative consequences have caused its use to be deemed 'too much of a good thing' [465]. A 'greener revolution' is required [466]. This revolution should be directed towards increasing the efficiency with which crops use N fertiliser and decreasing as many adverse effects as possible. One approach is to study N fertiliser supply changes in soils, particularly under grass and forage crops.

4.1.2 Studies have estimated N changes in grassland and forage cropland

Plants are able to take up a variety of N substrates principally in the forms of nitrate and ammonium, but also amino acids and other organic N compounds [266]. Deficiency of N limits crop production by restricting protein assembly, affecting primary and secondary metabolism. This is particularly relevant for the high N demand of the leaf photosynthetic apparatus where deficiency impacts on all growth parameters [75]. Leafy crops including forage grasses and leguminous herbs have a slightly higher N content than cereal grain crops, and require more N for optimal production [74]. Soluble nitrate (NO_3^-) in the soil is the most abundant source of N readily available to the crop [467; 72; 468]. Many forage grasses, such as ryegrass, will preferentially uptake NO₃⁻ as N source particularly at high fertiliser rates [469; 470]. Although usually above 1 mM, the concentration of NO_3^- in plots can vary between areas metres from one another [399; 391; 392; 400; 393]. Nitrate has been shown to have a particularly high degree of variability between samples [471], with a high potential to leach through soil [472-475]. Some estimates calculated that 60 % of applied N is lost through leaching, run-off, denitrification and consumption by microbial populations [476; 477]. There is a lack of research into leaching when grassland is temporary and intensive for high production, and more research is required with a focus on different soil types and management practices [118]. Moreover, N leaching may vary depending on the crop grown, due to differences in the root structure of forage crops [478].

Long-term experiments into N fertiliser leaching between soil plots have been conducted such as Rothamsted's Park Grass Continuous Hay Experiment, which began in 1856 [479; 104; 480]. One prominent study was a 19-year long assessment of leaching using ¹⁵N labelled fertiliser, which found leaching rate of ¹⁵NH₄ and ¹⁵NO₃ to be approximately 13.9 % and 21.9 % of the total N fertiliser application respectively [481]. However, many growers choose to cultivate mixed crops of grasses and/or legumes so the degree of leaching may vary. Other large scale evaluations of how

N application in agriculture affects the surrounding ecosystem have been undertaken. A decrease in grassland plant diversity has been shown following an application of just 2 kg N ha⁻¹ year⁻¹ [482]. Moreover, studies have shown leaching of inorganic N was lower for grassland than that for other arable crops including cereals [483; 484].

One key reason for lower leaching in grasslands is that the N applied is routinely removed as the plant uptakes the required amount, then removed from the field by either grazing or frequent cutting [485; 486]. Although it is known in *Lolium* that defoliation of aboveground vegetative tissue can halt the uptake of NO₃⁻ in hydroponics, which could therefore increase leaching of N fertiliser [119], it is difficult to compare leaching rates in permanent grasslands with cultivated forage crops. More research is needed to compare how specific management practices can influence leaching. For dried forage grass production, research is needed on N uptake in newly seeded species, cutting effects on leaching especially in root response and intercropping with leguminous forages.

By understanding the N cycle and being able to measure NO_3^- levels in the soil one can test how well crops utilise fertiliser. This is of particular interest in forage crops, as these areas of farming are vast with high fertiliser inputs and wide-ranging across many species.

4.1.3 Current testing methods of soil N are complicated to use

At present the measuring of soil N has been argued as unreliable, inaccurate and inconsistent across land and within the industry [376]. Farmers and agronomists aim to make detailed plans for their fertiliser applications and crop rotations, that depend on the local weather to extrapolate a baseline field soil N content. This estimate depends on soil sampling, with soil cores sent to independent laboratories for analysis of ammonium, nitrate and nitrite. This convention has existed for many decades [487-490], and includes calculations for soil nitrogen status [491]. Analysis is performed using KCl soil extracted samples and these are calibrated to known standards with chromatography and colorimetric

tests such as the Griess–Ilosvay diazo-coupling reaction [492; 493]. These measurements are converted to quantify the concentration per hectare. Although the procedures have been used for decades, there are notable problems in their uses. There is no standardised protocol for collecting soil samples and their transport, or how these are stored. Net nitrogen mineralization in soils is affected by abiotic conditions [494], including pH [495; 496], temperature [495; 497-500], moisture [497; 501; 500; 502], and soil texture and chemistry [503; 497; 504; 505; 496]. Therefore, the time taken between collecting a sample and laboratory analysis, including the storage during transport may significantly affect the final extrapolated measurement. Other techniques have tried to improve the precision of readings by removing interference from other ions [506; 507] including the deployment of specific test strips [499], or by optimising extraction and colorimetric buffers to accurately reflect soil concentrations [508-510; 496; 493]. Innovations with plant spectroscopy methods can be used in conjunction with these soil analysis to provide a description of the field environment [511].

Porous ceramic cups or soil lysimeter methods can be used to sample N in soil water on farms, with pH indicators used as a proxy for soil nutrient stoichiometry including N levels [512]. The soil water samples undergo the same chemical analysis as soil samples above and is therefore open to the same criticisms. This requires a large number of samples just for one field, to get a good representative amount for the site. The cups can provide large variation in individual measurements from a single cup sample [513]. Moreover, there is no consensus on when soil samples should be taken, after harvest in the winter or in the spring and how long they should be stored before analysis [376]. The N levels in the soil temporarily vary across a season and spatially across depths [384; 391; 400; 471; 514-516; 388; 389]. Methods such as ceramic cup sampling or soil extractions are labour intensive, with a cost impact for each sample. As reviewed by Shaw et al., (2016), current methods of testing are labour-intensive, expensive, and not conducive to the aims of the modern drive for precision agriculture [517]. In addition, there is limited sharing of data in the UK across different farmlands making extrapolations of soil N availability to crops difficult.

One method is the use of ion-selective sensors such as those used by Shaw *et al.*, (2016) to measure *in-situ* real time NO_3^- levels through networks as an alternative to conventional sampling methods described above. Such ion-selective sensors have been developed in the Miller/Sanders laboratory [155] based on earlier work using ion-selective microelectrodes [518; 156; 519]. These sensors can report in real-time the free NO_3^- in the soil water across different soil types and can be calibrated with standard NO_3^- concentrations. It is also possible to use these sensors at different depths to get a measurement of N across a soil profile. Limitations to their use are the design of economically viable sensors and farmer-friendly interfaces to access and store their outputs.

4.1.4 Ion-selective nitrate sensors may be deployed in trials to assess soil concentrations

Sensors which can be used to detect NO₃⁻ in soil were developed in the laboratory from previous work measuring the concentrations in plant cells. These NO₃⁻-selective sensors can be made cheaply and quickly, see Figure B1 in Appendix B for details of routine production in the laboratory. These sensors have been tested against the conventional soil testing detailed in Section 4.1.3. The good agreement between sensor readings and conventional testing can be found in Figure 4.1, where an R² of 0.99 was observed between the two measurement types.

Figure 4.2 below is an example of NO_3^- -selective sensor deployment in the field (Figures 4.2b and 4.2c), alongside standard soil temperature and moisture measurements (see Figure 4.2a). In this field example, plots were treated with either high or low NO_3^- solutions of 3 or 30 mM KNO₃ at 0 days. One can see the concentration of NO_3^- recorded by sensors increases upon application in the 30 mM treatments, indicated by the blue plots for both 10 minute and 12 hour average data, increasing from ~ 18 mM to ~ 50 mM NO_3^- . The data sampling frequency was every 10 minutes, but the outputs were also plotted to show 12 h mean values. The trends in the data are very similar and so all future data plots used a 12 h sampling average. No significant change was shown in 3 mM plots, as NO_3^-

remains between ~ 10 - 20 mM. Such data shows real-time changes in the soil water NO₃⁻ concentrations in the field and may be correlated with biomass yields, protein, and chlorophyll levels of forage crops.



<u>Figure 4.1:</u> NO₃⁻-selective sensors (Probe) correlated to conventional soil nitrate extraction analysis methods (Assay).

Soil at four known nitrate concentrations was measured for nitrate-N as ppm by both NO₃⁻-selective sensors and through conventional spectrophotometric assay. Nitrate-N ppm are the units commonly used by soil scientists, the sensors can be calibrated to give outputs in mM too. The above graph shows the strong agreement between measurements, with $R^2 = 0.99$. Graph provided by plotting a scatter plot of each type of data values and fitting a linear regression.



<u>Figure 4.2:</u> Field trial example of NO₃⁻-selective sensors in plots treated with different potassium nitrate concentrations.

NO₃⁻-selective sensors were constructed in the laboratory and deployed in the field in two plots at the John Innes Centre, Norwich to a depth of ~ 10 cm. On day 0 plots were treated with either 3 mM (green plots) or 30 mM (blue plots) KNO₃ and sensors prepared as in Section 2.2.3, data-logging along with Delta-T SM300 soil moisture and temperature sensors (Delta-T Devices Ltd., Cambridge, UK). Soil temperature and moisture are shown in a), as dark grey and light grey lines respectively. Ion-selective sensor data is shown for each plot, an average of four sensors with data output logging every 10 minutes in b). A 12 hour average is shown in c). For both b) and c) the errors are shown with lighter bars and are the standard errors of the means. Although field trials provide the most accurate data for N fertiliser changes in grassland soil, the testing of NO_3^- treatment effects on plant growth and soil profile leaching analysis was moved to column experiments in glasshouses. Deploying many sensors at various depths in the field is technically difficult, especially for measurements below 60 cm in the soil profile. Under these deep conditions the weight of the soil in the field can crush the plastic tips used to make the sensors. Soil column experiments are routinely used in plant and soil science. They have been used to monitor leaching of other minerals including sodium, calcium and magnesium [520]. They are particularly useful for root imaging studies which use x-ray technology to monitor changes in root architecture based on water and nutrient conditions [521]. They can also be used to assess how plants root systems interact with one another [522] and test cultivar phenotypes [523]. Although they are still an artificial system when compared to field measurements, they are superior to artificial media based or hydroponic systems when studying root nutrient uptake. Hydroponic experiments require aeration systems, with management of either sterile or artificial microbial populations. Hydroponic and gel-based media systems lack the influence that soil structure has on root growth and nutrient uptake [524]. Therefore, column experiments are a more realistic model system and useful for studying soil profile changes under forage crops. Column experiments can be useful for replicating soil environments, recreating forage grower management practices with more technical ease than field trials. For example, one can study quite efficiently how two different fertiliser regimes behave in the soil profile, with the confounding influence of cultivar root architectural differences monitored directly. Figure 4.3 illustrates some parameters which can be easily varied in column experiments, as well as the types of measurements which can be undertaken. The dimensions of the columns make them technically manageable (e.g. weight) and forage plants can be sown at industry rates with adequate room for root growth.

As stand-alone systems they can have sensors inserted at different depths, see Figure 4.3, without the problems associated with field use. Soil water can also be taken from the bottom drainage holes using lysimeters to check for leached NO_3^- , using the conventional testing discussed in Section 4.1.3.

Experimental replicates can be carried out with practical ease within a feasible timeframe, providing meaningful data more representative of field phenomena than hydroponic or gel-based media can provide.



<u>Figure 4.3:</u> Schematic of column set-up for NO₃⁻-selective sensor experiments.

Columns have dimensions of height = 50 cm and inside diameter = 15.4 cm, see Section 2.2.2 for details. Possible experimental regime changes are marked in green, with the different assays possible shown in blue. Holes are available at three different levels for placement of the NO_3 -selective sensors (shown as tip sensor photo symbol), at the top 1 cm (yellow), the bottom 1 cm (brown), and midway between (orange). Drainage holes were located at the bottom of the column, where soil water can be sampled using a lysimeter indicated with photo symbol of suction 10 Rhizon SMS lysimeter (Rhizosphere Research Products B.V., Wageningen, The Netherlands). 4.1.5 Aims of this chapter: Measuring soil NO₃⁻ changes with sensors in forage crops in response to fertiliser application, defoliation, and intercropping

This chapter describes measuring soil N changes using NO_3^- -selective sensors in response to cultivation of *L. perenne* in soil columns. Soil $NO_3^$ profiles at three depths were assessed under different management practices. These included varying NO_3^- application, defoliation of plant, and intercropping with the forage legume *M. sativa*. The sensors were tested for their agreement with conventional soil water analysis of NO_3^- in column drainage samples. This method was tested in columns to simulate the environment in the field, but with easier replication and standardisation of measurements.

The aim was to assess NO₃⁻-selective sensor suitability in forage crop studies of soil N profile changes using industry management practices. This will provide data for growers of forage crops on how their management practices may influence soil N levels for cultivation and any problems associated with fertiliser leaching.

4.1.6 Materials and Methods

The methods used in this chapter are detailed in Chapter 2, in section '2.2 NITRATE AVAILABILITY MONITORING USING SOIL SENSORS'. Supplemental data is provided in Appendix B.

4.2 RESULTS

4.2.1 Nitrate soil profiles of *Lolium perenne* columns vary with management practices

As a well-studied and important UK grass, L. perenne cv. AberMagic was tested in column experiments. The set-up for four columns was shown in Figure 4.3 with NO₃-selective sensors in each depth, and the management practice for each column is indicated below in Table 4.1. One column had no L. perenne present, 'No crop', and three had L. perenne growing. These had either no NO₃⁻ application, 'Monocrop 1', had a NO₃⁻ application at day 0, 'Monocrop 2', or had NO_3^- application at day 0 plus aboveground tissue defoliation (crop harvest) and biomass measurement at day 28, 'Monocrop 3'. The industry regularly cuts their grass crop every 4 – 8 weeks, so 'Monocrop 3' represented two cuts in a cycle with 'Monocrop 1' and 'Monocrop 2' representing one cut. Seeding density matched industry standards, as did N applications. A settling period was included in experiments in order to check for problems with sensor failure before recording measurements. On day 0 columns were treated accordingly, and the crop grown over the next two months without interference apart from regular watering to near column capacity or saturation level. At the end of the two months in 'Monocrop' columns, the crop's whole vegetative tissue was cut, and the biomass measured, for 'Monocrop 3' day 28 biomass was included in the final total.

During the experiment data was captured periodically, and at the end only those sensors able to recalibrate to the defined standard (see Section 2.2.2) were used in analysis. The analysed data for four replicate columns of this experimental design are shown in Figure 4.4. Four graphs are shown, each corresponding to the column set-up indicated in Table 4.1, with each column profile depth level included. The first 'No crop' graph clearly shows the NO₃⁻ application at day 0 - 6 with an increase in sensor mM NO₃⁻ detected at the top of the column (yellow plot). From days 2 – 4 the NO₃⁻ application began to be detected in the middle of the column with a peak in mM detected at days 20 – 22 (orange plot), indicating leaching

through the soil profile. From day 30 onwards the NO₃⁻ application was detected in the bottom level where mM increases were detected (brown plot). Diurnal changes in NO₃⁻ mM detected can be seen in this graph and all subsequent plots. In the next graph, 'Monocrop 1', no NO₃⁻ application was detected in any level, matching the management practice of no KNO₃ application of the column. The measured end vegetative biomass for this column was low at only 5.7 ± 0.7 g.

The third and fourth graphs show data for 'Monocrop 2' and 'Monocrop 3'. Again, in each of these the NO₃⁻ application was detected in the top level at days 0 – 6, (yellow plot), but quickly depleted with little increase in NO₃⁻ mM detected in the middle level from day 6 onwards. In both graphs the NO₃⁻ application was taken up by the crop, as no detected increase in mM was measured by the middle or bottom sensors. The crop uptake was also shown by the much higher end vegetative biomass measurements compared with 'Monocrop 1', with 10 – 11 g measured for each column. There was one major difference between 'Monocrop 2' and 'Monocrop 3' graphs, when defoliation had taken place there was seemingly a 'burst' of detected NO₃⁻ mM in the middle region only, (orange plot). To assess this burst in more detail further graphs were produced to show the data from each column level separately.

<u>Table 4.1:</u> Column set-up for the *Lolium perenne* monocropping experiment described in Section 2.2.4.

Columns were managed as colour coded and as follows; green for planting with 100 % *L. perenne* seedlings, cv. AberMagic, at seeding rate of 43.7 kg ha⁻¹; blue for day 0 nitrate application with KNO₃ treatments equivalent to 57 kg ha⁻¹ as the standard in UK forage agriculture [157]; purple for day 28 total aboveground vegetative defoliation with biomass measurement. All columns with crops growing were cut on day 56 for biomass totals, and in the case of 'Monocrop 3' this was added to the day 28 measurement.

Column	Forage crop	N application	Cutting of crop
No crop		KNO ₃ application	
Monocrop 1	Lolium perenne		
Monocrop 2	Lolium perenne	KNO ₃ application	
Monocrop 3	Lolium perenne	KNO ₃ application	Defoliation



Figure 4.4: Lolium perenne monocrop column experiment NO₃⁻-selective sensor data.

Data are shown for each column separately, indicated in Table 4.1. NO₃⁻-selective sensor data are shown for top (yellow), middle (orange), and bottom (brown) levels of the columns, as described in Figure 4.3. Data is the 12 hour average of four experimental replicates in GraphPad Prism 7 (GraphPad Software Inc.), with standard error of the mean indicated with thinner lines of similar colour. Coloured vertical bars indicate the treatment for the *L. perenne* crop planted (green), nitrate application at day 0 (blue), or defoliation of total aboveground vegetative biomass at day 28 (purple). End total mean vegetative biomass is shown in g for each column with standard deviation of measurements across four experiments indicated.

4.2.1 Soil sensors detect a nitrate 'burst' in the mid column range following defoliation of *Lolium perenne*

Upon defoliation of *L. perenne* in a monocrop, sensors detected a delayed NO_3^- 'burst' in the middle section of roots, see Figure 4.4 'Monocrop 3'. Therefore, plots were produced to show each column level separately with further analysis. Figure B3 in Appendix B shows graphs for 'No crop' and 'Monocrop 1', indicating differences in detected NO_3^- mM from the graphs in Figure 4.4. Conventional soil water analysis of drainage hole leachate, indicated with black diamonds, showed good agreement with bottom sensor detected NO_3^- mM.

Figure 4.5 shows all levels separately for 'Monocrop 2' and 'Monocrop 3'. Figure 4.6 shows a repeated measurements ANOVA for significant differences between columns at each depth, which showed showed very low significant differences between detected NO_3^- mM in either top (a) or bottom c) levels for 12 h timepoints (red plots). However, the middle level 'burst' of increased NO_3^- noticed in Figure 4.4 was shown to be statistically significant in Figure 4.6 b) following defoliation, (blue plot), with p < 0.05 between days 39 – 48. The 'burst' disappears from day 48 onwards, with no significant change found in the bottom level subsequently (brown plot). Conventional soil water analysis of drainage hole leachate, (indicated with black diamonds), again showed good agreement with bottom sensor detected NO_3^- mM, although in 'Monocrop 3' this was slightly higher than the ion-selective sensor data.





Column set-ups are described in Table 4.1. NO₃⁻-selective sensor data are shown independently for top (yellow), middle (orange), and bottom (brown) levels of the columns. Data are the 12 hour average of four experimental replicates in GraphPad Prism 7 (GraphPad Software Inc.), with standard error of the mean indicated with thinner lines of similar colour. Coloured vertical bars indicate management practice for the *L. perenne* crop planted (green), NO₃⁻ application at day 0 (blue), or defoliation of total aboveground vegetative biomass at day 28 (purple). In bottom level graph the soil water from drainage holes for one experiment was tested as leachate through conventional methods described in 2.1.15, indicated by black diamond symbols.





Repeated measurements ANOVA conducted for differences between 'Monocrop 2' and 'Monocrop 3' in RStudio including time as a longitudinal factor. 12-hour timepoints not statistically significant are in red plots as 'FALSE', and those statistically significant with p < 0.05 are in blue plots as 'TRUE'.

4.2.3 Nitrate soil profiles were assessed when *Lolium* perenne columns were intercropped with *Medicago sativa*

Due to a push in modern agriculture to practice more sustainable intercropping, forage grasses and legumes (*L. perenne* cv. AberMagic and *M. sativa* cv. Daisy) were tested together in column experiments. The setup of four columns is shown in Figure 4.3 with NO₃⁻-selective sensors in each depth, and the management practices for each column are indicated below in Table 4.2. One column had no crop present, labelled 'No crop', and three had *L. perenne* and *M. sativa* growing in an 80:20 mixed cultivation. These columns had either no NO₃⁻ application, 'Intercrop 1', with defoliation at day 28, 'Intercrop 2', or had a NO₃⁻ application at day 0 as well as all aboveground tissue defoliation, cut and measured for biomass on day 28, 'Intercrop 3'. Settling period, data analysis, and total end vegetative tissue measurements were carried out as in the monocropping experiments.

The analysed data for two experimental replicates of this experimental design are found in Figure 4.7. Two replicates were achieved in the available timeframe of the project, which may influence the standard error rate between averages. Again, four graphs are shown, each corresponding to the column set-up indicated in Table 4.2, with each column depth level included. The first 'No crop' graph showed the NO_3^- application at day 0 -6 detected NO_3^- mM at the top of the column (yellow plot), with a later peak in the middle of the column shown at day 18 (orange plot). Slightly earlier from day 26 onwards the NO₃⁻ application was detected in the bottom level. This again indicated the NO₃⁻ application had leached through the soil profile. In the next graph, 'Intercrop 1', no $NO_{3^{-}}$ application was detected in any level, this result is like that seen with 'Monocrop 1'. The measured harvest vegetative biomass for 'Intercrop 1' column was higher than that of 'Monocrop 1' at 8.0 \pm 2.0 g. The graph for 'Intercrop 2' was similar to 'Intercrop 1' with no large change in the NO₃⁻ mM measured at any level, either before or after defoliation. The biomass for 'Intercrop 2' was similar to 'Intercrop 1', with a measurement of 8.7 ± 1.3 g.

The last graph for 'Intercrop 3' was similar to both 'Monocrop 2' and 'Monocrop 3', due to their similar management practices. The NO_3^- application was detected in the top level at days 0 – 6 (yellow plot) and quickly depleted with little increase in NO_3^- mM detected in middle level from day 6 onwards (orange plot). The harvest vegetative biomass measurement for 'Intercrop 3' was also like 'Monocrop 2' and 'Monocrop 3' shown in Figure 4.4. Despite having both an NO_3^- application and defoliation as in 'Monocrop 3', no NO_3^- 'burst' was detected in the middle region. Graphs were again drawn for more detailed analysis of the individual column depths separately.

<u>Table 4.2:</u> Column set-up for the *Lolium perenne* and *Medicago sativa* cv. Daisy intercropping experiment described in Section 2.2.5.

Columns underwent the management practice as colour coded as follows; pink for planting with 80:20 *L. perenne* cv. AberMagic and *M. sativa* cv. Daisy seedlings, at a seeding rate of 43.7 kg ha⁻¹ to match the forage industry standard; blue for day 0 nitrate application with KNO₃ treatments equivalent to 57 kg ha⁻¹ as the standard in UK forage agriculture [157]; purple for day 28 total aboveground vegetative defoliation with biomass measurement. All columns with crops growing were cut on day 56 for harvesting the biomass total, and in the case of 'Monocrop 3' this new growth mass was added to day 28 biomass measurement.

Column	Forage crop	N application	Cutting of crop
No crop		KNO ₃ application	
Intercrop 1	<i>Lolium perenne</i> and <i>Medicago sativa</i>		
Intercrop 2	<i>Lolium perenne</i> and <i>Medicago sativa</i>		Defoliation
Intercrop 3	<i>Lolium perenne</i> and <i>Medicago sativa</i>	KNO ₃ application	Defoliation



<u>Figure 4.7:</u> *Lolium perenne* and *Medicago sativa* intercrop column experiment NO₃⁻-selective sensor data.

Data are shown for each column separately, as indicated in Table 4.2. NO_3^{-} selective sensor data are shown for top (yellow), middle (orange), and bottom (brown) depths in the columns, as described in Figure 4.3. Data are the 12-hourly average of two experimental replicates in GraphPad Prism 7 (GraphPad Software Inc.), with standard error of the mean indicated with thinner lines of a similar colour. Coloured vertical bars indicate the management practices for *L. perenne* and *M. sativa* crops (pink), nitrate application at day 0 (blue), or defoliation of total aboveground vegetative biomass at day 28 (purple). Harvest total vegetative biomass is indicated in g for each column with standard deviation of mean measurements across four experiments indicated.

4.2.4 No nitrate 'burst' was detected following defoliation of *Lolium perenne* when intercropped with *Medicago sativa*

Upon defoliation of *L. perenne* in a monocrop, NO_3^- -selective sensors detected a delayed NO_3^- 'burst' in the middle section of roots, shown in Figure 4.4 (orange plot). This was shown to be statistically significant when compared to non-defoliated crop, shown in Figure 4.6 b) (blue plots). This 'burst' was not detected at any depth under defoliated crops when *L. perenne* was grown alongside *M. sativa*, see Figure 4.7 'Intercrop 3'. Individual depth level graphs were again produced to check the significance of the findings described in Section 4.2.3. Figure B4 in Appendix B shows the graphs for 'No crop' and 'Intercrop 1'. In addition, conventional soil water analysis of drainage hole leachate (indicated with black diamonds) showed good agreement with the measurements reported by the bottom NO_3^- -selective sensors.

Figure 4.8 shows all the depth levels plotted separately for 'Intercrop 2' and 'Intercrop 3'. Figure 4.9 shows a repeated measurements ANOVA for significant differences between columns at each depth, which showed significant differences between the detected NO_3^- mM in top a) level (blue plots) soon after the NO_3^- application from days 0 – 4. Other regions of significant difference are found in the middle b) level at days 10 – 13 where some leached NO_3^- application was detected by the sensors, and then again for a portion of the bottom c) level at days 15 – 31. There was little evidence of a NO_3^- 'burst' following defoliation, with or without a NO_3^- application. Again conventional soil water analysis of drainage hole leachate, (indicated with black diamonds on brown plot), showed good agreement with bottom sensor measurement of NO_3^- mM, although 'Intercrop 3' was slightly higher than NO_3^- -selective sensor data.



<u>Figure 4.8:</u> Intercrop column experiment NO₃-selective sensor data for 'Intercrop 2' and 'Intercrop 3'.

The column set-up was described in Table 4.2. NO₃⁻-selective sensor data are shown independently for top (yellow), middle (orange), and bottom (brown) levels of the columns. Data is the 12-hourly average of two experimental replicates, with standard error of the mean indicated with thinner lines of a similar colour. Coloured vertical bars indicate management practices for *L. perenne,* and *M. sativa* crop planted (pink), nitrate application at day 0 (blue), or defoliation of total aboveground vegetative biomass at day 28 (purple). In the bottom level graph, the soil water from drainage holes for one experiment was tested as leachate using conventional chemical assay methods described in Section 2.1.15, and indicated by black diamond symbol.





Repeated measurements ANOVA conducted for differences between 'Intercrop 2' and 'Intercrop 3' in RStudio including time as a longitudinal factor. 12-hour timepoints not statistically significant are in red plots as 'FALSE', and those statistically significant with p < 0.05 are in blue plots as 'TRUE'.

The differences between the cropping type, with all other management practices kept the same, was compared for 'Monocrop 3' with 'Intercrop 3'. This is shown for all column depth levels in Figure 4.10. These data show for top level (yellow plot) no significance apart from a small portion at the start of the experiment, likely due to the fewer replicates performed for 'Intercrop 3'. For the bottom level (brown plots) NO₃⁻-selective sensor detected mM was not different between plots, although 'Monocrop 3' look slightly higher in NO₃⁻ when observing conventional soil water analysis (black diamond symbols), but this cannot be tested accurately for significance with few data points. It was the middle level where there were differences in the plots, (orange plot). Again, no region of 'burst' in detected NO₃⁻ mM following defoliation can be seen in 'Intercrop 3' between days 34 - 44 when compared to 'Monocrop 3'.



Figure 4.10: Column experiment NO₃⁻-selective sensor data for `Monocrop 3' and `Intercrop 3'.

NO₃⁻-selective sensor data are shown separately for top (yellow), middle (orange), and bottom (brown) levels of the columns. These data are for columns of monocropping and intercropping experiments where KNO₃ application and defoliation occurred. Data are the 12-hourly average of four or two experimental replicates, with standard error of the mean indicated with thinner lines of similar colour. Coloured vertical bars indicate management practices for *L. perenne* only (green), or with *M. sativa* (pink), nitrate application at day 0 (blue), or defoliation of total aboveground vegetative biomass at day 28 (purple). In the bottom level graph, the soil water from drainage holes for one experiment was tested as leachate using the conventional chemical assay method described in Section 2.1.15 and indicated by black diamond symbol.

4.3 DISCUSSION

4.3.1 Soil columns can be used to produce standardised measurements of nitrate profiles

Column experiments allow the users to carry out multiple tests with defined parameters. This is similar to gel-based media systems and hydroponics, but soil columns allow for more environmental fluctuations, including N fertiliser leaching through the profile. Gel-based media systems are sterile and are in an environment with controlled light, temperature and humidity. Hydroponics are not sterile, but it is standard to change hydroponic solutions frequently to avoid unwanted pathogen effects and to provide nutritional repletion for adequate growth. An advantage of the soil column system is that the microbial populations, soil moisture levels, and light and temperature changes are more like those for crops growing in the field. These fluctuations may be less severe than field trials, but columns allow measurements to be standardised in laboratory settings to eliminate some of the climate variability occurring in field trials. This standardisation is important for testing the effect of different management schemes on NO₃⁻ leaching. This is especially true for forage grassland as there is a lack of evidence to suggest how different practices affect the environment [118], including the high intensity, temporary grassland like those schemes used by dried forage growers such as BAGCD.

As management practices can be changed relatively easily in columns, this work has shown that standardised measurements can be achieved in a relatively short time period compared to field trials. All NO_3^- -selective sensor data for column types in this project show very similar measurements across replicate experiments, with minor standard error of the mean differences. It is true that NO_3^- -selective sensors only measure discrete volumes of soil by testing the activity near to the sensors, however as there are multiple sensors at each depth, the data suggests these areas are representative of the whole depth volume. Moreover, the good agreement between experimental replicates, is shown in statistical testing between the measured NO_3^- mM, especially in Figures B3 and B4 of
Appendix B, suggesting the small volume tested by sensors over time is accurate for whole depth measurements. By having a soil based system with testing at numerous layers it is possible to see leaching in real-time after NO_3^- application in 'No crop' columns. The movement of NO_3^- through the soil profile is altered by the uptake of crop roots and by measuring the NO_3^- run-off from the base of the column it is possible to obtain a direct measure of the efficiency of the root system in acquiring NO_3^- . On this basis, the data suggest the forage crops grown in this work are efficient at obtaining NO_3^- .

4.3.2 Soil sensors match conventional measurements, but with higher definition and lower interference

Soil sensors can provide a wealth of data for cost-effective analysis of fields in real-time [517]. With these experiments, data from NO_3^{-} -selective sensors built in the laboratory have been shown to provide similar measurements to conventional testing when analysing drainage hole leached soil water. Sensors have provided dramatically more detail with significantly lower labour when compared with conventional suction lysimeters or drainage water sampling. The NO₃-selective sensor data can provide a more detailed picture of NO₃⁻ uptake by crops and soil microbes. This is particularly true for capturing diurnal changes in NO₃⁻, for example in Figure 4.4 'No crop' plot. Such high definition of data throughout the day over a two and a half month period is unfeasible with current field methods. It is this wealth of data capture alongside low costs and labour which shows their superiority compared to the current testing methods described above. It is hoped that such testing in the future could aid quicker responses by the grower to improve growth conditions and combat leaching and future yield losses; this is very important for the success of precision agriculture [514; 517].

It should be noted that although feasible for short-term field use, see Figure 4.2, there are technical and durability problems with the NO_3^- -selective sensors in their current design form for monitoring throughout an entire season. The NO_3^- -selective membranes are very thin, about 1 –

2 mm thick, and so can easily be damaged especially in sandy soils. This damage can lead to the liquid backfill solution leaking, with evaporation from the top of the sensor providing another potential source of error due to inadequate sealing. Additionally, burying sensors in the field to depths > 50 cm can cause cracking of the plastic tips. The silver wire used for conductivity can corrode if sourced cheaply and the data loggers used can fail due to insufficient power supply. Finally, as the NO₃-selective sensors are measuring soluble NO₃⁻ concentrations they are dependent on the soil water content, and extreme water conditions such as drought or flooding will change the NO_3^- concentration measured. For this reason, it is important to always include soil moisture measurements to monitor rainfall events. These factors mean that well-sourced and constructed sensors with rigorous lab testing work best in clay soils with adequate water drainage systems. Future sensors are likely to be made using solid-state technology like some new types of commercial pH sensors. These can offer the opportunity to avoid damage by soil particles and they may be used at greater depths.

4.3.3 The detected nitrate 'burst' following defoliation of Lolium perenne, but not when cultivated with Medicago sativa

The detected NO₃⁻ mM 'burst' found in the middle section of roots with NO₃⁻-selective sensors following defoliation of *L. perenne*, see Figures 4.4, 4.5 and 4.6, is very intriguing. A lack of NO₃⁻ uptake may have been expected [119], but a potential NO₃⁻ release is worthy of further investigation. There are hydroponics data to suggest N is released by roots at ~ 5.1 - 6.1% of the total plant storage under normal conditions [525], but this was not limited to one portion of the root under stress. It could be possible that this phenomenon is an experimental artefact, however this is unlikely as the experimental replicates show standardised measurements and no burst was observed when the grass and legume were intercropped. Possibly, the grass released NO₃⁻ as a stress response after cutting, or as a by-product of another similar process.

Post-defoliation, it is known that for the remnant aboveground vegetative tissue the uptake of carbon is stronger than for N [526], and therefore a release from the roots of NO_3^- may be due to this change in C:N ratio of tissue [527; 250]. Increasing defoliation of *Lolium* in hydroponics showed decreased plant N uptake and increased plant N remobilization [209], as shoots appear to be the predominant site of whole plant NO₃⁻ reduction in grasses [528]. Moreover, it is known that remobilization of plant Ccontaining compounds in the leaf is coordinated with N availability to the root [529]. However, hydroponic systems are unable to assess specific changes to root portions and importantly the soil microbial contribution is missing. As the detected mM NO_3^- 'burst' disappeared after day 48, shown in Figure 4.6 in 'Monocrop 3', any available N may be taken up by the plant once vegetative regrowth is established following defoliation, which is also shown in its relative end vegetative biomass total. Such changes may not be found in hydroponic analysis due to free movement of soluble NO₃⁻ through solution. Therefore, it may be possible for N-containing compounds to also be released from the plant as a stress response. Changes in both amide and amino acid composition has been found in Lolium xylem sap following defoliation [530], especially asparagine and glutamine, suggesting increased nitrogen assimilation [212; 531]. Moreover N reserves stored as vegetative storage proteins (VSPs) in roots and stem bases are rapidly degraded after defoliation [532]. But there is no published assessment to show if these N-compounds occur in root exudates.

It is documented that grasses in hydroponics can release carbon exudates from *Lolium* roots as a response to cutting [41; 42; 46], and also its close relative *Festuca* [43; 44; 47]. Alcohols and aldehydes can be released with complex profiles which change throughout developmental stage or cutting event for plant [533]. It could be possible that such C-containing compound exudates may cause a process to release NO_3^- in this middle region of the soil column. This process may be mediated by the plant or through microbial activity such as denitrifying bacteria. Carbon compounds and other non-nutrient root exudates are important as host specific recognition signals [534] and this may be the cause for microbial changes and their activity following defoliation. In permanent grassland it was

shown that the composition of microbial populations changed in response to defoliation [535], and although there was no effect on microbial activity, such population changes may affect the soil NO₃⁻ mM pools for this short 'burst' period. Moreover, possibly NO₃⁻ of microbial origin is not able to be taken up by the roots during this time due to changes in plant N uptake described above, so microbial processes are unaffected, and it is a plant only phenomena causing this 'burst'.

As the 'burst' is not found when intercropping with *M. sativa* is performed, then a microbial change caused by legume root presence may be possible; it is known that there is greater diversity in soil microbial populations for legumes when compared with grasses [536; 537]. However, this may be due to the differences in root architecture of legumes when compared to grasses that may be important for this NO₃⁻ 'burst'. Changes in transpiration rate post-defoliation are recorded in many plants [538-542] and these may affect certain portions of the root to different amounts, however more work on root differences is required for such hypotheses. Specific species and plant taxa vary in their root branching shapes [543] with plant root ideotypes having slightly different patterns that may be important for breeders [544]. The differences found in the long tap root grasses, including *Lolium* may be influence the N profile changes when compared to the much more branched legume *Medicago* [545; 546; 478], and thus could be investigated in more intercropping experiments for leaching investigations.

Chapter 5: Fulvic acid increases vegetative growth in the forage legume *Medicago sativa*, and is associated with influencing microbial activity

5.1 INTRODUCTION

5.1.1 Using biostimulants to aid crop yields

During the past 100 years of modern agriculture, crop production has become dominated by intense agricultural practices. These have included concentrated mono-cropping with increasing use of fertilisers, pesticides, and invasive soil treatments including high-tillage [547]. Such practices can have significant negative environmental impacts, including excessive accumulation of nutrients and contamination of drainage water [548; 549; 483; 550]. Intensive agricultural practices can also depreciate soil through erosion, C and N losses, and lowered microbial activity [88; 551; 405; 552; 553]. These practices also affect greenhouse gas emissions across the UK and Europe [554; 555; 452; 453; 556; 456; 557; 558]. Due to the problems with modern, intensive practices, studies with a focus on alternative methods are being investigated. Forage crops cultivated for dried feed have huge N fertiliser inputs, which is discussed in detail in Appendix E: 'UK dried forage production: a review of industry changes and assessment of prospects for both policy and science'. New management practices for forage crops that can reduce the negative impacts of cultivation are required.

Improvements in plant nutrient capture efficiencies are being investigated, with the aim of reducing fertiliser use without impacting on yield. Often this is achieved through the use of breeding [338; 328; 339; 559; 560; 14; 3; 335; 331; 336; 337], or genetic manipulation [561-563; 373; 564-572], to expand productivity. However, there are time limits in both breeding and transgenic techniques in forage crops, which mean application of research to growers is slower than other crops [573]. Studies with faster applicable results include investigating forage crop rotation changes, particularly of forage legumes, on farms [574-580] and changes to tillage management [111; 581-583; 89; 584]. These studies have emphasised a need for yield improvements, but also a push for more sustainable agriculture, focusing on not only the plant, but also the soil and the wider biodiversity reaction to management [103; 105; 585-591].

One strategy which may improve forage plant performance is the use of molecules that act as biostimulants. As defined by Du Jardin in 2015, biostimulants correspond to any substance applied to plants to enhance nutrient efficiency regardless of nutrient content, and thus do not include fertilisers and pesticides [120]. The investigation of potential biostimulants is a focus point in current plant science including in forage crop research, and can incorporate studying the use of biotic protein hydrolysates [592-594], plant growth-promoting rhizobacteria (PGPRs), and beneficial fungi [595-599], algae extracts [600; 601], and inorganic salts (e.g. Al, Co, Na, Se, Si) and amino acids [602; 120; 603; 604]. Such potential biostimulants include humic substances (HS), including specifically the derivative fulvic acid (FA).

5.1.2 Humic substances are potential biostimulants, and include fulvic acid

Humus accounts for ~ 10 % of soil content across terrestrial ecosystems and it is also found in rivers and wastewater, with significantly lower levels in agricultural land [605]. Humus is formed of decaying biological tissue, mainly of plant origin. Insoluble humin and extractable humic substances (HS) are a major component of humus. Extractable HS fractions are humic acid and fulvic acid (FA), and these are considered key soil components as their complex composition may be responsible for facilitating many complex chemical reactions in soil systems [606-609]. All fractions are a poorly characterised mixture of chemicals, and although many have been shown to increase growth parameters in numerous species [123; 610; 611; 122; 612; 613], their mode of action is unclear. There is some debate over which fractions, the high or the low molecular weight, are the most active. Canellas et al., (2010) concluded it is the lowest molecular masses, although indicated that this may be less to do with size, and more their specific bioactive molecule content [122]. The lowest molecular weights are FAs [614; 607], which unlike humic acid are water soluble across a wide range of pH, and so will be the focus of this study.

Despite the wide variety of sources of FA from temperate to tropical soils to Leonardite waste material, there is a uniformity in the gross properties of HS [615-617]. Analyses provide evidence that all FA are complex formulations of C-containing compounds, with potential metal-binding capacities [618; 619; 614; 620; 606; 621; 607; 609; 622]. Biochemical analyses over the decades has demonstrated that these products are supramolecular associations of many compounds whose characteristics depend on their level of aggregation [606]. These compounds are organic, but cannot be classified into other groups such as proteins or starches. The extraction procedure for FAs has varied over the last 200 years, but generally involves acidification to pH 2 followed by precipitation, with the solid precipitate purified by absorption through XAD-8 resin or dialysis [623-626]. The precipitant gained and the method of purification have a large influence on the characteristics of the resulting solution and the extraction methods may differ in their component steps [626-629]. Extraction methods are under ongoing review with patents being awarded that depend on the possible future applications of the humic derivatives. The possible applications for FAs are wide-ranging, from medicinal studies into wound healing [630], to studying complex metal chelation interactions such as for cadmium, lead and copper [614; 631; 632], and crucially for this study in improving crop quality and yield.

5.1.3 Studies have investigated humic and fulvic acids effects on plant growth

Many studies have looked into the benefits of applying HS, as humic acid or FA extractions, in either the growth medium or as a foliar application to a wide range of crops. In *Arabidopsis thaliana* and many cereal crops, HS has been shown to have effects on plant growth including increased root architecture, improved nutrient uptake and yield even under stress, and enhanced access to metals [633-640], with comprehensive examples detailed in numerous reviews [641-643; 610; 644; 613].

A study of particular prominence to forage crops is that of the legumes soybean (*Glycine max*), peanut (*Arachis hypogaea*) and clover (*Trifolium*)

vesiculosum) [645]. This study showed that a sand growth medium supplemented with FA increased nodulation weight, but not size, in a dose-dependent manner. Application of HS in *Pisum sativum* also showed an increase in root nodulation and mycorrhizal colonisation [646]. If such increases were able to improve N fixation in legumes, then this could inturn increase the N storage of their vegetative tissue. The important forage crop *Medicago sativa* (commonly called 'lucerne' and 'alfalfa') has been found to increase in vegetative biomass with FA application, however with variable responses in different conditions [647; 648]. Studies using various HS including FAs have been carried out in other important legumes and forage grasses [124; 649-652], however, these experiments did not include appropriate nutritional controls and more detailed studies are required to assess the effect of FA on forage crops.

5.1.4 Limited evidence for the mode of action of humic substances

Identification of the specific effects of humic substances requires well structured, specific methods that are not always met in published research. Research on FA is often limited by the chemical analysis and frequently using samples which are not easily replicable, because their source is unique [633]. This makes designing appropriate controls for experiments difficult. Due to this lack of analysis many studies often rely on a 'no application' or 'water treatment' for controls, to determine the FAs potential biostimulant effect on a plant [610].

Some studies have linked HS and extractions including FA with hormonelike responses in plants, particularly *A. thaliana* [653; 654; 643; 655; 121], however there is a void of evidence from well-controlled experiments to support this. Other than this tentative hormonal hypothesis as a mode of action, most studies conclude that it is the complex composition of HS that promotes the above changes in physiology, nutrient uptake, or stress response. This is understandable, due to the nature of elucidating a single growth response from substances so complex in their composition. It is

imperative that more studies of FA are carried out with good controls to elucidate the role it may play in improving crop growth.

5.1.5 Aims of this chapter: Assessing the potential biostimulant fulvic acid in forage crops

This chapter describes work investigating the use of FA to promote growth in the cultivation of forage crops. This will be useful for forage growers to assess if inclusion of this treatment can increase the yields of their crops, without increasing N fertiliser inputs. Two readily-available, commercial FAs will be tested across a range of important temperate forage crops. One of the major problems in analysis of FA effects on plant growth is that studies have not attempted to control for nutritional composition of FA treatment. Adequate chemical analysis of the FA used is often lacking and so will be incorporated into this project, including controlling for nutritional composition.

Treatments are first assessed in the glasshouse to establish growth effects on crops, before being trialled in the field. Field trials were at forage grower sites with applications and management using industry standard practices. The aim was to identify if a change in management practice, including FA treatments, could be used in UK forage cultivation under conventional farming methods.

5.1.6 Materials and Methods

The methods used in his chapter are detailed in Chapter 2, in section '2.3 FULVIC ACID TESTING OF FORAGE CROPS'. Supplemental data is provided in Appendix C.

5.2 RESULTS

5.2.1 A screen of two fulvic acid applications found a vegetative yield increase in forage legumes, but not in grasses

Two commercially available fulvic acids (MFA and VFA) sourced from the UK and USA were tested with forage crops to see if they improved their vegetative yield. Fifteen forage crop species and cultivars important in UK cultivation were tested in the glasshouse, with details found in Tables 2.4 and 2.5. Crops were treated either with the FA or with deionised water (dH₂O), and the vegetative yield was measured as the total aboveground biomass after 3 weeks of growth. Figure 5.1 shows yields for 6 cultivars, the grasses *Lolium perenne* cv. AberMagic and *Festuca arundinacea* cv. Kora (a and b), and the legumes *Medicago sativa* cv. Daisy and cv. Gea, *Lotus corniculatus*, and *Trifolium pratense* cv. AberRuby (c, d, e, and f). Grass yields did not visually differ between treatments, whereas MFA and VFA treated legumes all looked notably bushier with more vegetative production.

Biomass measurements for the FAs treatments were calculated relative to the dH₂O (as 100 %) and compared across experiments, as shown in Figure 2.5. There were no significant differences in biomass for grasses with application of FAs, matching the visual observations in Figure 5.1 a) and b). There is a slight statistically significant measurement for *Anthoxanthum aristatum* with MFA as p-value < 0.05, but this difference is small and may not be enough to suggest that FAs are capable of increasing grass yields. A two-way ANOVA showed that neither FA treatment was statistically significant in increasing biomass, indicated with significance letters.

On the other hand, Figure 5.2 shows that legume biomasses were dramatically increased by FA application, particularly *M. sativa* cv. Daisy and Luzelle, and *L. corniculatus*; two-tailed student t-tests indicated in graphs for most measurements of FA applications were p < 0.05, and if a

one-tailed test was used based on visual observations then all measurements are statistically significant for FA application. This was also demonstrated in visual observations of the leguminous crops shown in Figure 5.1 c) – f). A two-way ANOVA showed that both FA treatment were statistically significant in increasing biomass, indicated with significance letters. As *M. sativa* is a widely cultivated forage crop, the vegetative growth increase measured is commercially important and requiring further assessment.

As farm-collected soil was used in these experiments, the effect of FA treatment on weed count rate and final soil pH was compared, and is shown in Figure C1, Appendix C. FA treated pots had higher numbers of leafy, dicot weeds during the first week post treatment when compared to dH₂O control (b) and more grass weeds like couch grass at the start of the second week (a). This supports other documented cases in the literature of HS inducing germination across many plants species [656; 619; 657; 658; 650]. Final pH of soil was also higher in FA treated pots, especially those with MFA treatment. Similarly this matches other published evidence that humic substances can act as soil conditioners, reducing acidity through increasing soil aggregate stability and reducing ionic species [618; 614; 620; 621]. Chemical analysis of both types of FAs was required to determine how similar their compositions and identify any similarities.



<u>Figure 5.1:</u> Vegetative tissue of forage crops following one of two fulvic acid treatments compared to a control.

Treatments were applied to seedlings at 12 days post germination and vegetative yields were assessed at 21 days post treatment. Treatments were deionised water (1. dH₂O in grey), 0.5 % MFA (2. MFA in blue), or 1 % VFA (3. VFA in orange). Fifteen forage crop species/cultivars were tested in total, with the above showing two grasses and four legumes; the grasses were a)-b), *Lolium perenne* cv. AberMagic and *Festuca arundinacea* cv. Kora respectively; the legumes were c)-f), *Medicago sativa* cv. Daisy and cv. Gea, *Lotus corniculatus*, and *Trifolium pratense* cv. AberRuby. Scale bars indicated 1 cm.



<u>Figure 5.2:</u> Vegetative biomass of forage crops following one of two fulvic acid treatments relative to a control.

Treatments were applied to seedlings at 7 days post germination and vegetative yields were assessed at 21 days post treatment. Treatments were deionised water (dH₂O in grey), 0.5 % MFA (MFA in blue), or 1 % VFA (VFA in orange). Fifteen forage crop species/cultivars were tested in total, with the top bar chart showing grass species, and the below bar chart showing legumes. Biomass was measured for two independent experiments with biomass for both FAs calculated relative to dH₂O (shown as 100%). Error bars show standard deviation between experiments. A two-factor ANOVA was also performed for grass and legume cultivars, as displayed in table with significance level of 0.05.

5.2.2 Analysis of fulvic acids found varying chemical composition

Although FAs have shown both in this study and in other literature to affect numerous crops including legumes, it is not clear whether the effect is as a biostimulant, or as a nutritional additive. Analysis of FA composition was necessary to elucidate whether application is working actively to effect growth, or whether it is a fertiliser.

Using ion-coupled plasma (ICP) analysis, a range of metal ions were detected, as well as total N and C contents; these can be found in Tables 2.8 and 2.9. Several ions were found at very low concentrations and varied between FAs; MFA included detectable Fe, with trace amounts of Ti, Mo, Cr, Cd, P, Zn, Co, Ni, Cu, Mn; VFA included detectable Ca, K, Mg, S, and Na, with trace rates of Al, Co, Cu, Fe, Mn, Ni, Zn. FAs included total N of 1.51 and 8.59 mg/L respectively, with both FAs containing C compounds but VFA to a detectable limit of 0.03 mg/L. Although elements including metals were detected, they may not be in high enough concentrations to be a source for the plant; for example in MFA Fe content was 839 ppm, but a normal range in soils is 7000 – 55000 ppm dry weight [659; 660]. It was unclear whether some of these metals were chelated, but they make up a small but significant portion of both types of FAs treatment. These elemental measurements should be included in 'control' treatment solutions. These new nutritional controls were called MC for MFA and VC for VFA, and these were used in future experiments (see below).

NMR was implemented to compare the organic compounds found in MFA and VFA. Spectra of rigid and mobile C components using ¹³C-NMR is shown in Figure 5.3 below, alongside abundances compared to total of each class of compounds. The most striking difference between FA samples was that MFA largely contained alkyl C compounds (75.08 %), whereas VFA contained largely carbohydrates as saccharides (80.32 %). Both shared a substantial portion of their organic matter as carbonyl C (15.86 and 7.32 %). This data shows FAs can have very different compositions of organic matter, which matches their variation in metal ion contents above.



<u>Figure 5.3:</u> Analysis of two fulvic acids (MFA and VFA) using ¹³C-NMR for carbon compounds in evaporated samples.

Analysis was performed on evaporated MFA (blue) and VFA (orange). NMR spectra is shown for both on the left hand side, for rigid and mobile components (red traces), and for rigid components only (blue traces). Below traces there is a guide for which carbon compound type is found at that chemical shift range in ppm as follows; carbonyl C in 160 – 220 ppm (C=O including ketones, aldehydes, acids, and esters); phenolic C in 140 – 160 ppm and aryl C in 110 – 140 ppm (aromatic rings); di-O-alkyl C in 90 – 110 ppm and carbohydrate C in 60 – 90 ppm (primarily alkanes); methoxyl C in 50 – 60 ppm (compounds with methyl group bound to oxygen); alkyl C in 0 – 50 ppm (alkanes). The relative composition of carbon compounds is shown on the right hand side as bar graphs, with values indicated. '*' is an unknown peak in VFA rigid and mobile components trace and is not included in total relative composition. Experiment and data analysis performed by Dr Juan Carlos Muñoz-Garcia, School of Pharmacy, University of East Anglia).

Finally, using mass spectrometry (MS) techniques a small number of compounds were detected in FAs to include in control solutions. In MFA, citric acid derivatives were detected, see GC-MS spectra in Figure C2, Appendix C. Derivatives detected matched to citric acid in both chromatograms with citric monohydrate standard, and with library derivative entries (93.3 % and 19.8 %). These derivatives were isomers of citric acid including free R-(homo)2-citrate, likely to be found in a complex with Fe²⁺, which is in high levels when compared to other elements. It was unsure whether citrate and iron would form a complex in MFA strong enough to remain associated throughout chromatography or whether the complex dissociated during chromatography, with the separated ions citrate, homocitrate, and (homo)2-citrate all detected. It does not functionally matter whether the complex only formed in the spray chamber, or not. What is relevant is whether citrate/homo-citrate from MFA is capable of forming a complex with transition metals in soil conditions, and whether this is beneficial to plants, for example by increasing availability of trace nutrients. It was therefore important that citrate was included in our control solution MC to test such capabilities. It is worth noting citrates are exuded by the roots of some plants for metal chelating effects [661], so it is entirely plausibly that MFA could also be showing this chemistry for metal acquisition.

In VFA, a low weight polyethylene glycol (PEG) cluster was detected, see GC-MS spectra in Figure C3, Appendix C. The cluster detected matched to PEG in both chromatograms with a PEG-400 standard, and with a library entry match (93.43 %). It was unsure whether PEG in VFA was either present in FA starting humic material, added during the extraction process, or if is a by-product of the extraction. PEG is known to induce water stress effects in plants, but only when molecular weights and concentrations present are high enough. Although PEG-400 can penetrate plant tissues when grown in solution, leaves and root systems will not be damaged unless grown in a concentrated PEG solution of at least 100 mL/L and with a low osmotic pressure (-14 bar) applied. Even so, this concentration of PEG is low with only small amounts of local damage occurring [662]. In VFA PEG concentration is only approximately 1.5 mL/L. It is unlikely to be causing a water stress effect, however as this was identified in VFA and

stress effects have been noted in the literature, PEG-400 was included in control solution VC.

Also in VFA, derivatised isothiazolone was detected in very low abundance (PEG:isothiazolone abundance ratio was 20:1), with a database match of 97.2 %. Isothiazolone is a biocide and generally used for water treatment and may have been added to keep the product free from bacterial growth [663]. A benefit of isothiazolone over other biocides is that it stays stable in an acid environment [664] which is important when soil acidity is of increasing concern for growers. It was not deemed necessary to include isothiazolone in our control solution because it was found in such low abundance. Moreover, it was probably added to VFA keep the product free of microbial contamination, however in this study there was no requirement for this extra level of control, as all solutions were sterilised by autoclaving before the plant assays.

This analysis alongside ICP and MS shows how two FAs can have very different compositions, despite having similar starting material and following the same extraction process. Therefore, it is imperative that as much of the contents of FAs as feasible are added to controls in plant assays. This will determine if FAs are acting on plants through a specific pathway with an active ingredient, or just as an extra nutritional addition. The controls for MFA and VFA are termed MC and VC respectively, and will be used in subsequent assays.

5.2.3 Fulvic acid increase growth of *Medicago sativa* and is specific to its composition and not only a nutritional effect

Further biomass yield assays were carried out in the glasshouse with *M.* sativa cultivars using applications of MFA and VFA alongside their control solutions MC and VC (see 5.2.2). Moreover, controls of no application (NA) and deionised water (dH₂O) were also included to ensure there was no water availability factor involved in the growth increases. Figure 5.4 shows visually FA-induced vegetative growth stimulation after three weeks in *M.* sativa cv. Daisy. Both MFA and VFA have increased vegetative biomass compared to controls, with plants taller and bushier with more lateral expansion of shoots and leaves. There was no difference in root biomass between treatments.

Vegetative biomass measurements were recorded in three independent experiments for cv. Daisy, Luzelle, and Gea. Results are shown in Figure 5.5, alongside the grass *L. perenne* cv. AberMagic. Both cv. Daisy and Luzelle showed significantly increased vegetative growth when compared to control solutions; cv. Gea showed more growth but this increase was not statistically significant (see ANOVA letters of Figure 5.5c, with significance level of 0.05). This is likely due to the fact cv. Gea was tested throughout the winter months in the glasshouse, when temperature fluctuated more rapidly, and there was more variation between experiments (as shown in individual sample points). If relative increase in growth in independent experiments is used to assess yield increase, then the biomass of all cultivars was significantly increased with FA application compared to all controls. When using both raw yield values as in Figure 5.5, or if using relative experimental increases, the grass *L. perenne* did not show a significant growth increase with FA treatment.



Figure 5.4: Medicago sativa cv. Daisy seedlings following treatment with fulvic acids or controls.

Treatments were applied to seedlings at 7 days post germination and photographs above were taken at 21 days post treatment. Treatments were; no addition (1. NA in dark grey); deionised water (2. dH₂O in grey); 0.5 % MFA (3. MFA in blue); 0.5 % MC (4. MC in light blue); 1 % VFA (5. VFA in orange); 1 % VC (6. VC in yellow). Scale is provided in cm.



<u>Figure 5.5:</u> Vegetative biomass of *Medicago sativa* cultivars and *Lolium perenne* following treatment with fulvic acids or controls.

Treatments were applied to seedlings at 7 days post germination and vegetative biomass (dry weight in mg) were assessed at 21 days post treatment. Treatments were; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Three cultivars of *M. sativa* were tested, cv. Daisy (a), Luzelle (b), and Gea (c). One cultivar of *L. perenne* was tested, cv. AberMagic (d). Individual seedling biomass was measured for three independent experiments, as shown in black data points (Exp. 1 = circles, Exp. 2 = triangles, Exp. 3 = squares). Box plots show variation across experiments. Multiple comparisons between treatments were conducted using a one-way ANOVA Tukey test shown with letter, with significance level of 0.05.

5.2.4 Fulvic acid application caused an increased number of pink nodules in *Medicago sativa*

As *M. sativa* is a legume, and yield is known to be affected by the amount of nodulation in roots, the number of nodules were investigated in cv. Daisy and Luzelle. This count was done following application of MFA or VFA, or their controls. Figure 5.6 shows a representative visual scoring of nodules for each treatment. All treatments had plants which nodulated to varying degrees. Some roots had only early stage initiating nodules (EIN), whereas others had established white nodules (WN), or mature pink nodules (PN). No late stage nodules were found at this 21 day timepoint, which would be greyish green or brown in colour. EIN were not included in nodule counts as these may have never matured. Both MFA and VFA had visually more PN of large, mature size at 21 days than all other treatments. Total counts across three blinded experiments can be seen in Figure 5.7, alongside percentage of pink nodules. In other words, nodule numbers were counted without knowing the treatments applied. Again, MFA and VFA have more PN compared to all other controls, even if the total nodule number was only slightly increased. This nodulation phenotype is significant in both cultivars for both FAs, (see ANOVA letters in Figure 5.7, with significance level of 0.05).

The pink colour of large PNs is caused by the presence of leghaemoglobin, indicative that bacteria are actively N-fixing *Rhizobium* within the nodule. Therefore, it is possible that this increased rate of N-fixation caused by FA treatment may be the cause of increased vegetative growth. This could mean that FAs do not affect only the plant, but may also influence the N-fixing bacteria such as *Sinorhizobium*. More investigation is needed to identify whether the increased vegetative growth phenotype of FA treatment is found in the absence of *Rhizobium*.



<u>Figure 5.6:</u> *Medicago sativa* cv. Daisy nodules following treatment with fulvic acids (MFA or VFA) or controls.

Treatments were applied to seedlings at 7 days post germination and photographs above were taken at 21 days post treatment. Treatments were; no addition (1. NA in dark grey); deionised water (2. dH₂O in grey); 0.5 % MFA (3. MFA in blue); 0.5 % MC (4. MC in light blue); 1 % VFA (5. VFA in orange); 1 % VC (6. VC in yellow). Nodules are indicated as either early initiating nodules (EIN), white nodules (WN), or pink nodules (PN). Only white and pink nodules were counted as true nodules in for this analysis. Scale included is 1 mm.



<u>Figure 5.7</u>: Nodulation counts of two *Medicago sativa* cultivars following treatment with fulvic acids or controls.

Treatments were applied to seedlings at 7 days post germination and nodules counted at 21 days post treatment. Treatments were; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Two cultivars of *M. sativa* were tested, cv. Daisy (a), cv. Luzelle (b). Individual seedling nodules were counted for three independent experiments, as shown in black data points (Exp. 1 = circles, Exp. 2 = triangles, Exp. 3 = squares). Box plots show variation across experiments. Multiple comparisons between treatments were conducted using a one-way ANOVA Tukey test shown with letters, with significance level of 0.05.

5.2.5 Vegetative growth phenotypes with Fulvic acid application is dependent on *Sinorhizobium* presence

Following on from the observation that FA treatment stimulated vegetative growth of *M. sativa*, it was decided to investigate this effect both in the presence and the absence of *Rhizobia*. To test this, the *Rhizobium Sinorhizobium meliloti* was selected *as M. sativa* is known to be its main host plant. Seedlings of similar sizes were transferred to sterile plates with treatment media and were either inoculated with *S. meliloti* or left sterile. After 21 days, plates were checked for nodulation and changes in both vegetative and root tissue biomass.

The results of this experiment can be found in the dot plots in Figure 5.8 for cv. Daisy only; results for cv. Luzelle and cv. Gea can be found in Figures C4 and C5, Appendix C respectively. Yellow plot areas with open sample dots are sterile plates, and light purple plot areas with closed dots indicate those plates inoculated. Vegetative biomass (a) was significantly increased with FA treatments compared to control when *S. meliloti* was present (inoculated plates), see ANOVA results indicated by letters with significance of 0.05. Average vegetative biomass of 15 seedlings with MFA treatment was 18.2 ± 0.9 mg compared to MC which was 13.7 ± 2.1 mg, and VFA treatment was 16.6 ± 0.2 mg compared to VC which was 12.4 ± 1.9 mg. MFA treated plants had an average vegetative growth increase of 167 %, and VFA an increase of 152 % relative to the no application (NA) control; cv. Luzelle and cv. Gea had average relative increases for MFA treatment of 142 % and 157 % respectively, and for VFA treatment of 146 % and 148 %.

Nodule number (b) was higher in both FA treatments for plants on inoculated plates only, (plants on sterile plates were unable to form nodules without *S. meliloti*). This increase in nodule number was strongly correlated with vegetative biomass ($R^2 = 0.82$). Again this regression was shown in cv. Luzelle and cv. Gea ($R^2 = 0.78$, $R^2 = 0.76$). This data was supported by glasshouse experiments in which more mature PN nodules were present in plants with FA treatment of VFA and MFA, (see Figures 5.6 and 5.7).

There was no significant difference in root biomass (c) with any treatment, regardless of *S. meliloti* presence or absence. Moreover, in sterile conditions only there were no significant differences in any measurement between FAs and their nutritional controls (see Figures 5.8, C4 and C5 in Appendix C). This shows the intrinsic link between nodulation and vegetative yield increase with FA treatment. The results suggest that investigations should be carried out on *S. meliloti* in isolation to see if FAs can affect its growth regardless from the plant.



<u>Figure 5.8:</u> Vegetative and nodule phenotypes of *Medicago sativa* cv. Daisy following treatment with fulvic acids or controls, with or without inoculation of *Sinorhizobium meliloti*.

Two day old seedlings were transferred to media plates containing treatments as follows; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Plates either remained sterile (open dots on yellow background) or inoculated with *Sinorhizobium meliloti* (closed dots on purple). At 21 days their nodule numbers were counted, and biomass determined for both vegetative tissue and full root tissue. Five seedlings were measured for each treatment condition, and total measurements for three independent experiments are shown in charts. Multiple comparisons were conducted using a one-way ANOVA Tukey test shown with letters, with significance level of 0.05.

5.2.5 Sinorhizobium growth is affected by fulvic acids

To determine if FAs can affect growth of *S. meliloti* in the absence of plants, the growth of the bacteria were tested in liquid culture using the standard microbial techniques of colony forming unit (CFU) counts. Cultures were treated with FAs or controls, and inoculated with *S. meliloti*, and the cell density tested over 4 days. An example of CFU counts can be found in Figure C6, Appendix C and the summary results are shown below in Figure 5.9.

There was no effect of FAs on culture cell density at 0 - 1 days, when the microbial population is moving from lag phase to exponential phase of growth. At 2 days both FA cultures had a higher cell density than their controls, with MFA having 6.56×10^9 compared to MC with 4.07×10^9 and VFA having 6.81×10^9 compared to VC with 4.26×10^9 . By 3 days MFA culture cell density did not differ from controls, but VFA had a significantly higher cell density of 1.88×10^{10} . It is possible that MFA was also higher than controls at a timepoint between 2 - 3 days, but this was not measured. At 4 days all cultures were in the microbial death phase as cell density rapidly declines. These results, particularly for VFA, show adding FAs can increase growth of *S. meliloti* in isolated liquid culture.



<u>Figure 5.9:</u> Growth effects of fulvic acid on the growth of *Sinorhizobium meliloti* in TY medium, compared to controls.

TY cultures containing treatments as follows were inoculated with *Sinorhizobium meliloti*; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Average colony forming unit (CFU) counts were obtained from triplicate samples on 0 - 4 days of incubation with shaking 220 rpm at 28 °C. Average counts for three separate experiments were calculated and shown above with standard deviation. Multiple comparisons between treatments were conducted using a one-way ANOVA Tukey test shown with letters.

5.2.6 Soil microbial population structures are affected by fulvic acid

To test if soil microbes are affected by FAs in soil, a screen of changes in population was performed. Soil in which *M. sativa* was cultivated in glasshouse was treated with VFA, a VC control, a dH₂O control, or with no addition (NA) apart from a standard watering regime. Soil samples were then extracted to compare the relative phospholipid fatty acids (PLFAs) within them, both at day 0 and day 21 post treatment (see Chapter 2, section 2.3.14 for details). These PLFAs underwent transesterification to produce fatty acid methyl esters (FAMEs) Each FAME acts as a biomarker and corresponds to a microbial type, as developed in many studies throughout the past decades [665-669; 166; 670-672; 165]; the FAME biomarker assignments from Quezada *et al.*, (2007) [166] were used in this study. The individual FAME relative amounts can be found in Appendix C in Figure C7, and groupings according to biomarker assignments are found in Figure C8.

It is possible to determine if the general microbial community composition has changed following a soil treatment by comparing principal component analysis (PCA) of extracted FAMEs between two timepoints. PCA plots provide statistical analysis of observations to convert variables into values to assess their correlation. In the case of FAME biomarkers, the plots correspond to how correlated the structure individual experiment treatments samples are to each other compared to other treatments. Figure 5.10 shows the results of this technique for VFA compared to controls.

At 0 days (Figure 5.10 a), treatments show overlap in their soil microbial structure, with all treatments having large variation across experiments. The variation in community data is 65 %, and this variation is shared across all treatments. At 21 days (Figure 5.10 b) variation has dramatically reduced for NA, dH₂O and VC as each PCA plot is reduced in size. It can be seen that dH₂O has diverged from other plots, but still overlaps with the other control treatments, suggesting the community composition still matches that of NA and VC. The VFA plot remains significantly larger and

skewed to the left of plot area, suggesting VFA treated soil is able to support a slightly different microbial population than other treatments.

This difference in VFA treatment PCA associated with FAME biomarker assignments is due to two main factors, (see Figure C8). From day 0 to day 21 for the VFA treatment, the relative amounts of gram-negative bacteria increased by ~ 4.09 %, whilst relative amounts of commonly shared FAMEs decreased by ~ 5.19 %. Moreover the relative amount of fungi increased the most when compared to other treatments at 0.72 %. The increase in relative gram-negative bacteria numbers is most interesting as this includes *Sinorhizobium* species. This means that the gram-negative bacteria proliferation matches the FA induced cell density increase found for the *M. sativa* symbiont *S. meliloti*, shown in Figure 5.9. More validation of this idea is needed, correlating FAME soil biomarkers for gram-negative bacteria with the *S. meliloti* increase in liquid culture following VFA treatment. This can be tested by quantifying *S. meliloti* DNA from the soil after VFA treatment.



<u>Figure 5.10:</u> Two-dimensional principal component analysis (PCA) of fatty acid methyl esters (FAMEs) relative contents from soil treated with fulvic acid or controls.

Soil from *Medicago sativa* assays with fulvic acid applications (NA in dark grey; dH_2O in grey; VFA in orange; VC in yellow) was extracted for PLFA content at 0 days (a) and 21 days (b), for three separate experiments (indicated as 1, 2, and 3), and converted into FAMEs. Individual FAME relative abundance (in % of total FAMEs) was calculated from total FAME biomass (nmol g⁻¹ of dry soil). FAMEs were associated with particular organisms [166], and PCA plots calculated, variation of community data for 0 day and 21 days was 65 % and 63.9 % respectively.

5.2.7 Quantification of DNA from soil substrate confirms fulvic acid increases *Sinorhizobium* numbers

To test if the *S. meliloti* population increased in soil treated with FA, total DNA was extracted from soil from the same origin as that used in 5.2.6. DNA was quantified for relative amounts of *S. meliloti* specific gene *nodC* [170] using real-time PCR alongside two reference gene sets 799/1391 [172; 173; 171] and Eub338/518 [175; 176; 174]. Results are found in Figure 5.11. At 0 days all soil has the same level of *S. meliloti* detection regardless of treatment, shown in the fold change of *nodC* being around 1.5 – 2.2 fold change expression for all treatments. At 21 days all treated soils have increased in *nodC* expression, as *S. meliloti* populations have multiplied over time with growth of *M. sativa*. However, VFA shows a significantly increased *nodC* change in expression compared to other treatments at 45.7. This change is indicative of an increase of *S. meliloti* number in this treatment, which provides more data that VFA can increase the growth of *S. meliloti*. This proliferation may be associated with increased nodulation in *M. sativa* due to the plants increased growth following VFA treatment, as shown in Figure 5.5.



<u>Figure 5.11:</u> Quantification of *Sinorhizobium meliloti* DNA using *nodC* gene from soil treated with fulvic acid or controls.

Soil from *Medicago sativa* assays with fulvic acid applications (NA in dark grey; dH₂O in grey; VFA in orange; VC in yellow) was extracted for PLFA content at 0 days and 21 days, for three separate experiments. DNA was extracted from soil samples and geometric expression of *nodC* was calculated using real-time PCR, with fold change relative to two reference sequences, indicated above each bar. Error bars are standard deviation of relative expression of experiments. Multiple comparisons were conducted using a one-way ANOVA Tukey test shown with letters, with significance level of 0.05.

5.2.8 Fulvic acid vegetative growth effect on *Medicago* sativa was recorded in independent field trials, with no change to nutritional content

It is necessary to assess whether FA treatment can give a vegetative growth increase in *M. sativa* in the field as all previous experiments have been conducted in glasshouse conditions. Over 2017 and 2018, trials were conducted at three sites, all of which are designated forage crop cultivation farms. Field plot plans and maps of the trials are found in Figure C9, Appendix C. For each trial, plots were treated with either VFA, the control solution VC, a dH₂O control, or with no addition (NA) at early establishment of *M. sativa* (April – June). Plots were grown in accordance with site usual management practices, with treatment applications the only difference.

Prior to the first harvest of the season (May – July), vegetative biomass was measured for treatment plots, shown in Figure 5.12 a)-d). Although different cultivars were treated on the various sites, VFA increased vegetative biomass in all trials. Biomass increased with the VFA treatment for vegetative tissues, in both shoot and leaf tissues. This increased growth compared to NA was 135 – 165 %, which is similar to the boost found in plate assays of 146 – 152 %, (see Figure 5.8), and slightly lower than measurements from glasshouse experiments of 167 – 185 %, (see Figure 5.5).

Nutritional content of *M. sativa* from each treatment plot was also assessed for the 2018 trials. Figure 5.13 shows samples from Blankney Estates Ltd. trial with total vegetative biomass measurements, along with chlorophyll and total protein levels of each. There was a significant difference in vegetative biomass, but no difference in either chlorophyll or total protein content for any treatment. This was also found for A Poucher and Sons Ltd.; for NA, dH₂O, VFA, and VC plots the average chlorophyll and total protein content was as follows; 2.91 and 17.87 %; 2.61 and 17.96 %; 2.83 and 18.21 %; 2.63 and 17.97 %. This data provides evidence that the effect of VFA is on increasing biomass yield with no change in nutritional content. It is concluded that the VFA stimulatory effects found in the laboratory are applicable in the field.



Figure 5.12: Vegetative biomass of *Medicago sativa* cultivars in independent field trials following treatment with a fulvic acid or controls. Treatments were applied to field plots at beginning of establishment and vegetative yields were assessed before 1st cut of growing season; an area of 625 cm² was sampled and total vegetative tissue dried for biomass (in g). Treatments were; no addition (NA in dark grey); deionised water (dH₂O in grey); 1 % VFA (VFA in orange); and 1 % VC (VC in yellow). Four trials were run over two years. In 2017 trials were performed at a)-b) Dengie Crops Ltd. (Southminster, Essex) with four plots per treatment of cv. Daisy and Fado. In 2018 the trials were at c) Blankney Estates Ltd. (Blankney, Lincolnshire) and d) A Poucher and Sons (Bardney Dairies) Ltd. (Market Rasan, Lincolnshire) with six plots per treatment of cv. Daisy and Gea respectively. Individual samples are shown with black dot points, with boxplots for each treatment in trials. Multiple comparisons were conducted using a one-way ANOVA Tukey test shown with letters, with significance level of 0.05.


<u>Figure 5.13:</u> Vegetative tissue of *Medicago sativa* of first cut following fulvic acid treatment compared to a control, from field trial plots.

Treatments were applied to field plots at beginning of establishment and vegetative yields were assessed before 1st cut of growing season; an area of 625 cm² was sampled and total vegetative tissue dried for biomass (in g), with value indicated. Treatments were; no addition (NA in dark grey); deionised water (dH₂O in grey); 1 % VFA (VFA in orange); and 1 % VC (VC in yellow). Samples are from 2018 trial plots at Blankney Estates Ltd. (Blankney, Lincolnshire) for cv. Daisy. Samples were duplicated for chlorophyll and protein contents, these are provided as total dry weight %. Multiple comparisons were conducted using a one-way ANOVA Tukey test with VFA having a significant increase in biomass samples only.

5.3 DISCUSSION

5.3.1 *Medicago sativa* vegetative yield can be increased by adding fulvic acid into crop management practices

This study demonstrates how applying a low concentration of FA to M. sativa can increase its vegetative yield, as shown by growth on plates (Figure 5.8), in glasshouse conditions (Figures 5.1, 5.2, 5.4, and 5.5), and in the field (Figures 5.12 and 5.13). This data builds on existing indications of a yield effect of HS in *M. sativa* and other forage legumes [645; 647; 648]. Grass species were not increased in vegetative biomass in this project, see Figures 5.1 and 5.2. As a relatively minor addition to current management practice, applying FA to forage legumes may be an exceedingly cost-effective technique in improving yields. As such, an important crop in the UK and globally, any small intervention to increase its yield is of significant economic importance. However, as *M. sativa* is not the only important temperate forage legume, with Trifolium repens and Trifolium pratense also being highly utilised, it is important that more analysis of FAs effect on other species is assessed. This should be performed in the field where possible, with strict adherence to negating any purely nutritional effects of treatments where possible by using appropriate control solutions as in this project.

5.3.2 The vegetative growth increase is associated with microbial changes including *Sinorhizobium* populations

This study has provided evidence that the vegetative yield in *M. sativa* following FA treatment is associated with changes in microbial populations. This is shown by increases in nodulation and in numbers of *Sinorhizobium* species especially (Figures 5.6 - 5.11), but also potentially with changes in other microorganism populations (Figure 5.10). Isothiazolone was found in low levels in VFA but changes in microbial growth were still detected. This means some additions to commercial products may not always have

the desired effect in application due to low concentrations when diluted. This again confirms a need for thorough investigation of commercial treatments including FAs.

The data in this project demonstrates that FAs may increase growth in plants through stimulating the growth of *S. meliloti*, or through affecting both plant and microbe together. It is necessary to see whether FA is able to affect the growth of other soil microbial populations as many organisms can increase plant vegetative yield, such as other *Rhizobium* species but also species of *Streptomyces*, *Bacillus* and arbuscular mycorrhiza fungi [604]. This can be used to test if the growth effect of FA is affecting vegetative biomass by increasing nodulation. Alternatively, FA may increase vegetative biomass in plants which then has an effect on increasing nodulation.

The effect of FAs on increasing microbial cell growth, without plant interaction agrees with other published studies. It has been previously observed that HS are able to increase growth of *Bradyrhizobium liaoningense* in liquid culture [673] and increase growth of general microbial populations in soil microbial cells [674]. However, a study of the yeast *Candida utilis* found no change in growth with HS, so this response may be taxa dependent [675].

An improved symbiotic association of crops with microbes is important in the light of the current emphasis to grow more perennial and annual leguminous crops globally, due to their N-fixing activity [676-679; 578]. The fixing of atmospheric N₂ in legume/grass pastures is estimated to range from 13 to 682 kg N ha⁻¹ yr⁻¹ [680]. *M. sativa* has been estimated to have a fixation rate of up to 350 kg N ha⁻¹ yr⁻¹, this provides an N fixation rate of 0.021 x DM + 16.9 for *M. sativa* (R² = 0.91) [681], despite large differences in the N status of the soil through fertiliser use and geographic location. Such N fixing calculations are important when accounting for tissue N allocation and the soil residual N available for future crops. For *M. sativa* tissue allocations of total plant N is approximately 50 % in shoot, 45 % root, and 5 % nodule [682; 280]. Studies in which legume stubble and root are left in ground post-harvest have shown that a considerable proportion of biomass and fixed N can provide substantial N fertilisation

197

for future crop rotations [683]. This means that the increased vegetative growth in a cultivated legume as shown in this study could have implications for the yields of future crops.

More elucidation of FAs effect on both plant and microbial communities is required. For future plant-focussed research projects, an assessment of potential N-fixing rate of symbiotic bacteria caused by changes from FA treatment could use the acetylene reduction assay or similar microbial bioassays [684; 685; 682; 686]. An analysis of the host plant's transcriptional changes that occur upon application may be interesting to provide information for the mode of action of plant vegetative increase following the treatment; such an analysis will be discussed in the following chapter. In addition to studying changes in plant growth, studies of the soil could determine how FA may affect microbial communities. Studies could use techniques in the field to determine changes in microbial biomass C or soil enzymatic activity assays to determine potential changes in soil function due to the illustrated changes HS may have on soil structure [625; 614; 615; 682; 632; 621; 607; 648]. Using 16S rRNA sequencing could also be used to investigate in more detail the changes in abundance between taxa [687].

5.3.3 Fulvic acids are complex chemicals and require more work to define their mode of action, including field applications

It may be argued that the most notable advancement of this study is the use of chemical analysis of FAs to develop controls for nutritional content of treatments. These controls have been missing from previous studies, and therefore makes accurate interpretation of the data impossible. A lack of controls in previous studies may be why many have surmised there is general hormone-like response by plants upon HS application [653; 654; 643; 655; 121], as such changes in root architecture and nutrient uptake may be a fertiliser effect only. This studies provides evidence that the response is not only a nutritional response, shown in vegetative biomass increases of VFA and MFA application compared to VC and MC respectively,

shown in Figures 5.4, 5.5, 5.8, 5.12, and 5.13. Nevertheless more work is required to begin untangling the mode of action.

This is particularly true when looking at the chemical analysis of both FAs in this study alone; the FAs were sourced from different locations and showed they had largely differing contents, (see Figures 5.3, C2 and C3, Appendix C, and Tables 2.8 and 2.9). There has been a push for standardisation of HS analysis including separation of C-containing groups recently [609; 629]. However, until a standard analysis of HS is used to develop nutritional controls in plant or microbial studies, then no real advances can be made into how these treatments may promote effects.

In addition to this, it is also necessary for more studies to investigate FA in other contexts. Many studies have relied on only glasshouse and pot experiments without field testing, which may affect the abundance and diversity of soil microbial communities [688]. More analysis of soil chemical and physical parameters could also be explored, to conclude if HS has a soil conditioner effect [614]. It should be noted that such an examination may conclude that HS are not biostimulants if they also have such effects [120].

This page is intentionally left blank

Chapter 6: Transcriptome analysis shows preferential enrichment of nodulation regulation and signallingrelated genes in *Medicago sativa* following fulvic acid application

6.1 INTRODUCTION

6.1.1 Using RNA-seq to identify transcriptional changes in plants

In the past decade, RNA-sequencing (RNA-seq) has changed from a complex and challenging tool in the early-stage of conventional use [689; 690] to a critical technique used in many studies that wish to investigate how phenotypic changes occur in specific conditions [691]. Changes in the pattern of total RNA expression can reveal the signals and metabolic changes that underpin a phenotype. As a tool, RNA-seq is incredibly robust and sensitive [692-695] and provides a wealth of data for a researcher, be it a specific line analysis, treatment investigation, or stress assay.

The increased prevalence of using RNA-seq is for multiple reasons. As with all technologies, the cost has significantly reduced over the past ten years, in both financial terms and through decreased time needed to carry it out with recently improved protocols [696]. The wealth of bioinformatic support has improved with early-career researchers developing computational skills for analysis, with many universities and institutes employing dedicated bioinformations to aid this [697], although not without problems in resource and credit allocations [698]. A vast array of programmes has been developed to execute this high-throughput method, from read mapping to transcriptome reconstruction to expression quantification [699; 691]. Furthermore, these are combined with improved reference genomes of many model species, providing substantial in silico evidence which can be coupled with experimental work to build new biological hypotheses. Model legumes have been analysed for transcriptomic changes, including *Medicago truncatula* [700-706; 178; 707; 708] and Lotus japonicus [709; 354; 710; 711; 708].

RNA-seq has been used in many non-model plants using *de novo* analysis or mapping to closely related reference genomes. These include the legume crops chickpea [712-714], pea [715-717], *Medicago lupulina* [717], lentil [718], soybean [719], lupin [720], clover [347], and mung beans [721]. RNA-seq has been used successfully in many studies for

Medicago sativa for comparative line studies [722-724] and stress responses [725-730].

6.1.2 The effect of Fulvic acid on *Medicago sativa* transcriptome in both shoot and root tissue

In Chapter 5, a vegetative biomass yield increase in *M. sativa* was shown following fulvic acid (VFA) treatment. In previous publications the reasons for this are unknown and studies have suggested wide-ranging modes of action for this biostimulant. Therefore, it was necessary to investigate the transcriptional changes that may have occurred in the shoot and root tissue using RNA-seq analysis. As *M. sativa* is a legume, yields are strongly associated with the symbiotic N-fixing symbiotic bacteria termed *Rhizobium* [731].

Deriving from 'rhizo' meaning root [732], the rhizosphere is the root/soil interface where a high microbe population can be found. The rhizosphere is generally defined as the area of soil that is influenced by the activity of a root. The rhizosphere around an expanding root establishes gradients between the plant and surrounding soil that are important for nutrient and water uptake. The secretion of root exudates and subsequent symbiotic relationships with rhizobial microbes that are attracted by such secretions [733; 734] are important for nutrient delivery in the rhizosphere. Organisms which interact with the plant roots include *Rhizobium* [731; 735], but can also include members of other microorganism families, including a high proportion of fungi, protozoa, and nematodes [736; 737]. The legumes are a family of plants, which include *M. sativa*, that can form a symbiotic relationship with Rhizobium, shown in about 90% of the Leguminosae family [738]. In terms of legume and symbiotic Rhizobium, each must recognize each other to establish such a relationship. This is mediated through an infection of the bacteria into the root via the endorhizosphere or ectorhizosphere, with symbiosis being viewed as mutualistic, communalistic, or pathogenic. Symbiosis then proceeds via the development of special structures termed nodules on the root through rhizobial infection [739; 740].

203

Due to the dependence of *M. sativa* vegetative biomass yield on N supply through rhizosphere interactions (see Chapter 3), root tissue will be analysed for transcriptional changes caused by VFA using RNA-seq.

6.1.3 Rhizobium/legume nodulation initiation

For a symbiosis to occur between the legume and a *Rhizobium* they must first communicate with each other to allow infection into the host [731; 735]. This is caused via the pattern of cytoskeletal rearrangement of cells caused by calcium influx [741], root-hair curl around the bacteria to form an infection pocket, and infection thread formation by the *Rhizobium*, all of which lead to infection and colonization [740; 738]. Transcriptomic data in legumes supports the idea that plant defence must be suppressed by *Rhizobium* at thread formation to successfully reach the nodule and be successful [709; 701; 742]. This ultimately provides the transfer of carbon resources from plant to bacteria and nitrogen resources from bacteria to plant within nodules [743].

Plants use chemical signals to attract *Rhizobium*, primarily via flavonoids, which initiate crosstalk and symbiosis between the organisms [744]. Flavonoids have been shown to be specific for certain *Rhizobium*, for example in soybean the secretion of daidzein and genistein induce *Bradyrhizobium japonicum*, whereas *Sinorhizobium meliloti* is inhibited by these and instead induced by the compound luteolin [745]. These chemicals are then perceived by the *Rhizobium* as aglycones, which lead to the induction of *nod* genes by their interaction with the LsyR-type nod gene regulator [746; 744; 747]. The *nod* gene family is large and diverse, although with a high degree of conservation within their promoter region. Usually, as in the case of *NodV*, the flavonoid signal phosphorylates the NodV protein which in turn activates another nod gene, such as *NodW* [748].

These Nod factors can then be recognized by the LysM-type receptors of the host plant [749-754], and subsequently nodule development and bacterial infection can occur [755]. The symbiosis that results is dependent on the degree to which the plant allocates carbon resources to the

204

Rhizobium, and vice versa, the extent that the *Rhizobium* gives N to its host. It is also hypothesized that local immunosuppression of the host root by the *Rhizobium* is essential to set up the symbiosis. This is thought to be achieved by the nod factors suppressing the normal immunity of the legume by inducing the microbe/pathogen-associated molecular pattern (MAMP/PAMP) [756; 757], which in turn affects the plant's pathogen-/pattern-recognition receptors (PRRs). PRRs are able to recognize molecules which are microbe-associated [758-760]. Moreover, other factors that influence the formation of nodules in the legume/*Rhizobium* symbiosis include DNA-binding transcription factors such as the membrane-localized GmbHLHm1 in soybean. Loss-of-function mutants of GmbHLHm1 showed many defective phenotypes such as reduced levels of leghemoglobin, smaller-infection zone of the *Rhizobium*, reduced nodule number and fresh weight [761].

Mutualistic relationships between plants and *Rhizobium* generally involve this sharing of resources, enabling both organisms to gain from their interactions with each other. A common example is that of soybean and *Bradyrhozbium* [762], and a widely studied example of a non-legume symbiosis is the relationship between the cyanobacteria, *Anabaena*, and its host the aquatic fern, *Azolla* [763]. *Medicago sativa* forms preferential symbiosis with *Sinorhizobium meliloti* [764-768].

Some *Rhizobium* can form a symbiotic relationship with a large variety of legumes. For example, the *Rhizobium* strain NGR234 has been shown to nodulate over 200 species and even affect some non-legume species, which enables mutualistic interaction between both organisms as described above [769]. Moreover, there are extra positive interactions found from these mutualisms; for the plant this can include an increase in abiotic and biotic stress tolerance and its ability to grow in soil where other microbes produce toxic compounds that the mutualistic *Rhizobium* are able to degrade [770; 744]. Furthermore, the production of biofilm layers can also protect against pathogens [771].

6.1.4 Aims of this chapter: RNA-seq analysis in *Medicago* sativa to investigate changes in transcription following fulvic acid application

This chapter describes work investigating the transcriptional changes that occur in *M. sativa* upon FA treatment. This will help investigate the causes for the yield increase shown in Chapter 5. The commercial FA referred to as VFA was tested alongside a control solution called VC defined in 2.3.9 and 5.2.2. Following three days of treatment plant tissue was analysed using RNA-seq using shoot and root tissues separately. Transcriptome analysis was performed *de novo* and using related *M. truncatula* reference accessions.

The aim was to identify transcriptomic changes which occurred following VFA treatment in the plant. This can be used alongside the results in Chapter 5 to identify the FA mode of action for yield increases in legumes.

6.1.5 Materials and Methods

The methods used in this chapter are detailed in Chapter 2, in section '2.4 FULVIC ACID RNA-SEQUENCING OF MEDICAGO SATIVA'. Supplemental data is provided in Appendix D.

6.2 RESULTS

6.2.1 VFA induces differential expression of transcripts in *Medicago sativa* after 3 days of treatment

RNA was extracted from shoot and root tissues of *M. sativa* treated with either VFA or the nutritional control VC, on the day of treatment (day 0) or three days after the treatment (day 3). Samples were generated from three experiments and analysed using RNA-seq. Differentially expressed (DE) transcripts were analysed in shoot and root tissues. As *M. sativa* does not have a fully annotated sequenced genome, many studies have used *M. truncatula* genomes as a mapping reference. The references accessions A17 [702; 707] and R108 [178] show good alignment for studying transcripts (60 – 70 % for A17 and 75 – 85 % for R108), but *de novo* transcriptome assembly was performed to negate for bias in subsequent analysis. *De novo* transcriptome assembly was successful for building a scaffold (shown in Table 6.1 below) with similar alignment rates of all transcripts for A17 and R108 references. This was possible due to the high quality of samples used for sequencing, and shown in the sequencing quality checks in Table D2, Appendix D.

Multi-dimensional scaling (MDS) plots were generated to compare all transcripts in experimental replicates. These are provided in Figure 6.1, with both 2D and 3D plots for both shoot and root samples. The MDS plots demonstrate that the transcripts clustered for treatment and timepoints among samples. This is especially clear at day 0 where both VC and VFA were very similar in their clustering as blue and green plots, in both tissues. This might be expected for day 0 when any differences in transcripts will be through random chance and insignificant in later analysis. In both tissue types, both 2D and 3D plots show clearly that not only did day 0 VC and VFA cluster together (blue and green), but moreover day 3 VC and VFA clustered farther away with greater separation (orange and red), indicating developmental stage transcriptional changes. The 3D plots especially indicate in both tissue types that the day 3 VFA clustering

207

is more distant from the rest, suggesting more transcriptional variation from the other samples due to the treatment effect.

Total transcripts were analysed to find the significantly DE transcripts for each treatment between the two timepoints. The determination of significantly DE transcripts required use of both an absolute log fold change of 0.585 (1.5 x fold change), and a false-discovery rate (FDR) adjusted pvalue (q-value) < 0.05, with removal of DE transcripts significantly expressed between treatments at day 0 to negate for false positives due to experimental variance. The number of up-regulated (+) and downregulated (-) DE transcripts in shoots and roots of *M. sativa* following either VFA and/or VC treatment is shown in the Venn diagram below, Figure 6.2. This result shows that most DE transcripts for VFA treatment occurred in the root tissue. There were found to be 1705 upregulated and 241 downregulated DE transcripts in the root, and 140 upregulated and 209 downregulated DE transcripts in the shoot. This difference in DE transcript number is emphasised by differing sizes of the circles in the plot. <u>Table 6.1:</u> Pseudoalignment summary of RNA-seq samples with *de novo* transcriptome assembly.

Treatment	Tissue	Time point	Rep	Reads	Reads pseudoaligned	% pseudoaligned
Control	Shoot	0	1	19731245	13433374	68.08
Control	Shoot	0	2	20017217	13281556	66.35
Control	Shoot	0	3	22060176	13759703	62.37
Control	Shoot	3	1	23125595	14603288	63.15
Control	Shoot	3	2	22410937	14604137	65.17
Control	Shoot	3	3	22566458	15036502	66.63
Fulvic	Shoot	0	1	21948373	15001567	68.35
Fulvic	Shoot	0	2	20015180	13166950	65.78
Fulvic	Shoot	0	3	19550309	12674614	64.83
Fulvic	Shoot	3	1	21916558	13940709	63.61
Fulvic	Shoot	3	2	20834945	13749556	65.99
Fulvic	Shoot	3	3	21961363	14221382	64.76
Control	Root	0	1	19192851	13111977	68.32
Control	Root	0	2	18832338	13193369	70.06
Control	Root	0	3	19868638	14429523	72.62
Control	Root	3	1	21818824	15337857	70.30
Control	Root	3	2	19940906	14437150	72.40
Control	Root	3	3	18121577	12993129	71.70
Fulvic	Root	0	1	21683448	15322780	70.67
Fulvic	Root	0	2	22756103	15965816	70.16
Fulvic	Root	0	3	20587135	14609678	70.97
Fulvic	Root	3	1	19970525	14133478	70.77
Fulvic	Root	3	2	21347709	15176499	71.09
Fulvic	Root	3	3	20853270	14884900	71.38
			Total	501111680	341069494	_
			Mean	20879653	14211228.92	68.15

a) Shoot tissue RNA-seq samples



b) Root tissue RNA-seq samples



<u>Figure 6.1:</u> Multi-dimensional scaling (MDS) plots to compare all transcripts in experimental replicates clustered for treatment and timepoints.

Shoot (a) and root (b) data is shown separately, with the left-hand graphs generated using Degust [183] showing a 2D MDS plot of samples as follows; day 0 VC is blue; day 0 VFA is green; day 3 VC is orange; day 3 VFA is red. Bar charts are % variance of MDS plot. Right-hand graphs show a 3D MDS plot. In all cases MDS is generated with 200 transcripts.



<u>Figure 6.2</u>: Differentially expressed transcripts in *Medicago sativa* shoot and root tissue with treatments of either fulvic acid (VFA, orange) or the control (VC, yellow).

RNA-seq was carried out on whole shoot and root RNA samples taken on day of treatment (day 0) or three days after treatment (day 3). Transcripts from *de novo* transcriptome assembly with both an absolute log fold change of 0.585 (1.5 x fold change) and a false-discovery rate (FDR) adjusted p-value (q-value) < 0.05 were considered as differentially expressed (DE); DE transcripts significantly expressed between treatments at day 0 were removed to negate for false positives due to experimental variance. The above Venn diagram shows upregulated (+) and downregulated (-) DE transcripts for both treatments between day 0 and day 3, including those which are shared (overlapping region).

6.2.2 Significant transcriptional changes occured in roots following VFA treatment

As the above analysis found that shoot and root tissue had DE transcripts due to VFA treatment, both tissue type samples were used in the subsequent analysis. However as shown in Figure 6.2, a far higher amount of expression changes were found in the root than the shoot. Therefore, a focus was placed on root samples as it is likely they would provide more evidence of VFA mode of action than shoot data. Below in Figure 6.3 is a volcano plot that indicates the statistically significant DE transcriptional changes that occured in the root tissue between day 0 and day 3 for VFA treatment. All significant DE transcripts for shoot and root tissues can be found in Appendix D Figure D2. Comparison of 0 VC to day 0 VFA, and 0 VC to day 3 VC, also determined some DE transcripts. These DE transcripts were removed from subsequent analysis of the VFA treatment to ensure only transcripts truly differentially expressed by plants undergoing the VFA treatment are functionally annotated in the analysis.



<u>Figure 6.3:</u> Volcano plot of DE transcripts as log Fold Change (logFC) between 0 day and 3 day for VFA treatment RNA samples.

Root tissue data is shown only for *de novo* RPKM sample comparisons. Transcripts are deemed DE if all experimental replicates have an absolute log fold change of 0.585 and false-discovery rate adjusted q-value < 0.05. Graph modified from Degust [183] and made using voom/Limma method [184].

6.2.3 Differentially expressed transcripts have many homologues related to legume model species, particularly *Medicago truncatula*

Once DE transcripts for VFA treatment effects were collated, their transcript isoforms underwent basic local alignment search tool (BLAST) analysis to find genes either in *M. sativa* or closely related genes. The results are shown below for root tissue samples in Figure 6.4. Most DE isoforms were found to have homologues in closely related leguminous species (Figure 6.4a), most particularly *M. truncatula*, and *Glycine* and *Trifolium* species. Numerous successful BLAST hits were found for DE transcripts (Figure 6.4b), and most had extremely low E-values close to zero (Figure 6.4c), indicating these transcripts were more likely to be significant and not to be hits by chance. This was strong evidence that the transcript isoforms were related to BLAST gene hits.

a) Species Distribution









DE transcripts were processed in Blast2GO [185; 186] and using the pipeline transcripts for BLAST result against NCBI's non-redundant NR database [187] with project results as follows; a) Species distribution of number of BLAST hits; b) Sequence Similarity Distribution for hits against alignment length; c) E-Value Distribution of hits.

6.2.4 Biological processes associated with VFA treatment include oxidation-reduction, gene regulation, metabolism, transport and defence

BLAST results of DE transcripts were analysed for their related Gene Ontology (GO) 'Biological Process' terms. Figure 6.5 displays the GO terms for individual DE transcripts from VFA treatment of *M. sativa* roots. The plot shows the number of DE transcripts with their corresponding biological process GO terms that were either upregulated or downregulated. The GO terms demonstrated that the transcriptional changes within the root are wide-ranging. A 3-day treatment of VFA resulted in changes in genes regulating transcription and translation, including those associated with oxidation-reduction. Metabolism and transport were also affected by VFA treatment, with both carbon metabolism and nitrate assimilation changing in respect to the controls. Changes in C:N metabolism were interesting for the increased vegetative yield phenotype detailed in Chapter 5.

Figure 6.5 also indicated changes in responses to defence, stress, and bacteria. This was likely to be a response to symbiotic bacteria such as *Sinorhizobium meliloti* as at this developmental stage nodulation can begin to be established; it is well documented that important nodulation genes and factors are associated with defence responses through their evolution and function [709; 701; 742; 772; 773].

The same GO trends were found within shoot data for metabolism processes, cellular processes and response to stimuli. However, this was at a reduced level due to a lower number of transcripts being DE, (see Figure 6.2).

216



<u>Figure 6.5:</u> 'Biological Process' GO term hits for individual DE transcripts from VFA treatment of *M. sativa* roots.

GO terms were pulled using Blast2GO programme mapping and ran with the EMBL-EBI InterPro library [188], with manual addition from QuickGO [195] and UniProt [196] databases. Bars show genes upregulated (blue) or downregulated (red), indicating the number of DE transcripts with the associated GO term. Only the top 20 GO terms are shown; total GO terms associated with dataset was 142.

6.2.5 Molecular functions associated with VFA treatment include catalytic activity and binding

BLAST results of significantly DE transcripts were also analysed for their related GO 'Molecular Function' terms. Figure 6.6 is a Direct Acyclic Graph (DAG) of molecular function for roots following VFA treatment after 3 days. This indicated huge changes in proteins with catalytic and binding activities, including those associated with transport. This matched the biological process changes of transcriptional and translational regulating genes shown above in Figure 6.5.

As for roots, analysis of shoot tissue DE transcripts found similar molecular function increases, particularly those of hydrolase catalytic activity and binding associated with oxidation-reduction.



<u>Figure 6.6:</u> Direct Acyclic Graphs (DAGs) of GOSlim 'Molecular Function' associated with DE transcripts from VFA treatment of *Medicago sativa* roots.

DAG was made in Blast2Go with Sequence filter of 50, nodescore filter of 1.0, and nodes are coloured by score value.

6.2.6 Metabolism was significantly enriched by VFA treatment, particularly N metabolism

Having collated all significantly DE transcripts, enrichment tests for both 'Biological Process' and 'Molecular Function' were calculated using Fisher's exact test, shown in Tables 6.2 and 6.3 respectively. All enrichment was based on legume gene IDs for the *M. truncatula* reference so could be slightly different when a *M. sativa* reference is available. However, as analysis negated differences found in day 0 VFA and VC, and those in day 0 VC and day 3 VC, it was likely that these calculated fold enrichments were accurate.

From the data in Table 6.2 metabolic and catabolic processes were upregulated in roots, particularly those associated with N metabolism. Reponses to bacteria were also enriched, matching the idea of an effect on nodulation initiation causing such a change. Moreover, cell wall biogenesis and organisation were enriched, which are key factors required for new root development and nodule growth.

In Table 6.3, molecular function enrichment test shows that in the root nutrient transporter activity was highly upregulated with VFA treatment. Moreover, enrichment of serine hydrolase activity was detected, which has wide-ranging catalytic activity in plants [774; 775]. A change in chitin regulatory genes [776; 777] were particularly interesting with DE as follows for each (logFC, q-value); *agglutinin-2* (5.59, 1.30E⁻⁰⁶); *BAR-domain-containing protein* (3.22, 1.36E⁻⁰⁵); *legume lectin beta domain protein* (5.27, 2.12E⁻⁰⁵); *L-type lectin-domain containing receptor kinase IX.1-like* (2.49, 5.82E⁻⁰⁵); *chitin elicitor receptor kinase* 1-like (2.48, 9.25E⁻⁰⁶); *putative chitinase* (1.49, 2.30E⁻⁰⁵).

Many interesting N metabolism related genes were DE as follows for each (logFC, q-value); *ammonium transporter 3 member 1* (2.97, 5.64E⁻⁰⁵) and 1-like (5.63, 9.68E⁻⁰⁶); *NRT1/PTR 2.6-like* (4.13, 4.76E⁻⁰⁶); *NRT1/PTR 4.5* (3.01, 1.32E⁻⁰⁶) and *NRT1/PTR 4.5-like* (3.73, 1.59E⁻⁰⁶); *Medtr7g098220.1 peptide transporter* (4.09, 3.09E⁻⁰⁶); *Nitrite reductase(NAD(P)H) large unit* (3.64, 6.69E⁻⁰⁷) and *Nitrite reductase(NAD(P)H) small unit* (3.3, 1.79E⁻⁰⁴).

<u>Table 6.2:</u> Fold enrichment of top 20 'Biological Process' for upregulated DE transcripts from VFA treatment of *Medicago sativa* roots.

Enrichment for each category is scored against the *Medicago truncatula* reference and indicated with p-value using Fisher's exact test, using PANTHER analysis [197; 199; 198]. Only DE transcripts with gene ID annotated in mapping were included.

Biological Process	Expected based on reference	Actual fold enrichment	p-value
nitrogen utilization	0.02	87.32	3.59E ⁻⁰⁴
sterol metabolic process	0.46	8.73	1.30E ⁻⁰³
cell wall biogenesis	0.25	11.91	2.30E ⁻⁰³
organic hydroxy compound metabolic process	0.56	7.17	2.62E ⁻⁰³
proteolysis involved in cellular protein catabolic process	1.45	4.14	3.64E ⁻⁰³
cellular protein catabolic process	1.46	4.11	3.78E ⁻⁰³
steroid biosynthetic process	0.09	22.54	4.03E ⁻⁰³
cell wall organization	0.09	21.83	4.28E ⁻⁰³
phytosteroid metabolic process	0.11	18.88	5.60E ⁻⁰³
lipid homeostasis	0.12	16.63	7.08E ⁻⁰³
nitrogen compound metabolic process	0.13	15.19	8.38E ⁻⁰³
terpenoid biosynthetic process	0.13	14.86	8.72E ⁻⁰³
protein catabolic process	1.77	3.39	9.35E ⁻⁰³
macromolecule catabolic process	2.02	2.96	1.69E ⁻⁰³
isoprenoid biosynthetic process	0.20	10.12	1.77E ⁻⁰³
defence response to bacterium	0.21	9.70	1.91E ⁻⁰³
response to bacterium	0.21	9.70	1.91E ⁻⁰³
organic hydroxy compound biosynthetic process	0.21	9.57	1.96E ⁻⁰³
response to external biotic stimulus	0.21	9.44	2.01E ⁻⁰³
response to other organism	0.21	9.44	2.01E ⁻⁰³

<u>Table 6.3:</u> Fold enrichment of top 20 'Molecular Function' for upregulated DE transcripts from VFA treatment of *Medicago sativa* roots.

Enrichment for each category is scored against the *Medicago truncatula* reference and indicated with p-value using Fisher's exact test, using PANTHER analysis [197; 199; 198]. Only DE transcripts with gene ID annotated in mapping were included.

Molecular Function	Expected based on reference	Actual fold enrichment	p-value
serine hydrolase activity	0.19	20.85	5.33E ⁻⁰⁵
ammonium transmembrane transporter activity	0.03	69.86	5.24E ⁻⁰⁴
cation transmembrane transporter activity	1.39	5.04	5.60E ⁻⁰⁴
catalytic activity	21.90	1.69	$1.02E^{-03}$
proton transmembrane transporter activity	0.85	5.86	$1.84E^{-03}$
monovalent inorganic cation transmembrane transporter activity	0.85	5.86	$1.84E^{-03}$
ATPase activity, coupled to transmembrane movement of substances	0.53	7.59	2.14E ⁻⁰³
solute:proton symporter activity	0.54	7.43	2.31E ⁻⁰³
ATPase activity, coupled to movement of substances	0.57	7.02	2.82E ⁻⁰³
ion transmembrane transporter activity	1.86	3.76	2.94E ⁻⁰³
oxidoreductase activity	4.76	2.52	3.18E ⁻⁰³
oligopeptide transmembrane transporter activity	0.31	9.70	4.03E ⁻⁰³
peptidase activity, acting on L-amino acid peptides	1.63	3.68	6.35E ⁻⁰³
peptide transmembrane transporter activity	0.39	7.65	7.67E ⁻⁰³
peptidase activity	1.72	3.49	8.17E ⁻⁰³
amide transmembrane transporter activity	0.41	7.33	8.60E ⁻⁰³
inorganic cation transmembrane transporter activity	1.32	3.79	$1.11E^{-03}$
Unclassified	109.46	0.89	1.62E ⁻⁰²
transporter activity	3.26	2.45	1.75E ⁻⁰²
monooxygenase activity	0.98	4.09	1.77E ⁻⁰²

6.2.7 Enriched DE transcripts in VFA roots are associated with well-known nodulation regulatory and signalling genes

Many DE transcripts from three days after the VFA treatment were found to be significantly enriched for genes associated with responses to other organisms, including bacteria (see Table 6.2). It was noted that a lot of these have also been categorised as specific early symbiotic root nodule genes in *M. truncatula* in a study by Roux *et al.*, (2014) [705]. In this 2014 study they used laser-capture microdissection of roots and nodules coupled with RNAseq to provide a robust list of genes induced early on in nodulation initiation. Table 6.4a, 6.4b, and 6.4c show the significantly DE transcripts from this project which were also defined in their study.

Annotations showed the VFA treatment enriched a vast array of important genes required for the signalling and regulation of nodulation. These included an array of transcription factors (TFs) and domains including *Myb/SANT-like DNA-binding domain protein, RING-H2 finger protein ATL52-like, AP2-like ethylene-responsive transcription factor, WD40 repeat-like protein,* and *zinc finger MYM-type protein 1-like.* Many leucine-rich repeat receptor-like kinases (LRR RLKs) and other receptor kinases were found to be enriched with VFA treatment, for example *LysM domain receptor-like kinase.* Genes required in bacteria and hormone induced plant responses were found, for example *NDR1/HIN1-like protein 10, protein RRP6-like 2,* and *cytokinin hydroxylase-like* transcripts. Finally, many nodulation specific genes were enriched such as *nodulation-signaling pathway (NSP) proteins, NSP-interacting kinases,* and *nodulins.*

223

<u>Table 6.4a:</u> Enriched DE transcripts in VFA roots which are putatively classed as highly preferential nodulation regulatory genes and nodule-associated signalling-related genes as in Roux *et al.*, (2014) [705].

This includes a description of the protein, available gene/protein IDs, the annotation type, and log fold change (logFC) and q-value for each DE transcript.

Gene/Protein ID	Description	Annotation	logFC	q-value	
XP_024635034	Myb/SANT-like DNA-binding domain protein	TF MYB	5.35	1.90E ⁻⁰⁶	
PNX91228	putative CC-NBS-LRR resistance protein	LRR	4.98	1.23E ⁻¹¹	
ABD33274, AES59362, RHN77255	RALF-like protein	Calcium/lipid- binding	4.74	4.63E ⁻⁰⁶	
RIA81513	calnexin	Calcium/lipid- binding	4.67	6.36E ⁻⁰⁹	
RHN49201	wall-associated receptor kinase- like 20	RLK	4.60	3.35E ⁻⁰⁵	
KEH36571, RHN72042	CLAVATA3/ESR (CLE)-related protein	Ser/Thr protein kinase	4.36	2.44E ⁻⁰⁷	
KEH28705, RHN58556	putative LRR-domain, L domain- containing protein	LRR	4.29	6.43E ⁻⁰⁵	
XP_003612592, AES95550, RHN54652	RING-H2 finger protein ATL52- like	TF ZnFg C2H2	4.28	1.63 ⁻⁷⁵	
XP_024641562	AP2-like ethylene-responsive transcription factor	TF AP2/ERF	4.08	1.74 ⁻⁰⁴	
XP_003594815, AES65066, RHN73104	COBRA-like protein 7	COBRA	4.00	1.78E- ⁻ 07	
XP_003598348, AES68599, RHN65475	F-box protein interaction domain protein	F-box protein	3.75	4.90E ⁻⁰⁷	
AES76072, AES76110, RHN52304	NDR1/HIN1-like protein 10	NHL	3.67	2.52E ⁻⁰⁴	
RHN60433	disease resistance protein (TIR- NBS-LRR class)	LRR RLK	3.64	7.66E ⁻⁰⁶	
XP_013443270, KEH17295, RHN51739	cytokinin hydroxylase-like	CK activated	3.60	6.29E ⁻⁰⁶	
XP_013466350, KEH40391, RHN77806	receptor-like protein kinase	RLK	3.55	4.72E ⁻⁰⁶	
XP_003604023, AES74274	COBRA-like protein 1	COBRA	3.53	7.35E ⁻⁰⁷	

<u>Table 6.4b</u>: Enriched DE transcripts in VFA roots which are putatively classed as highly preferential nodulation regulatory genes and nodule-associated signalling-related genes as in Roux *et al.*, (2014) [705], continued from Table 6.4a.

Gene/Protein ID	tein ID Description Annotation		logFC	q-value	
RGB31681	calcium-binding protein	Calcium/lipid- binding	3.49	1.84E ⁻⁰⁶	
RHN72504	probable inactive receptor kinase At2g26730	RLK	3.42	1.32E ⁻⁰⁵	
XP_003613167, AES96125, RHN55010	L-tryptophanpyruvate aminotransferase 1	TAA1-like	3.39	2.54E ⁻⁰⁴	
AES69839	LRR-P-loop containing nucleoside triphosphate hydrolase	LRR	3.32	7.54E ⁻⁰⁶	
AES91737	F-box/kelch-repeat protein	F-box protein	3.32	6.92E ⁻⁰⁶	
XP_024637477	disease resistance protein (TIR- NBS-LRR class)	LRR	3.24	1.65E ⁻⁰⁵	
EXX59026	WD40 repeat-like protein	TF WD	3.22	$1.01E^{-04}$	
XP_024631685, RHN72543	mitogen-activated protein kinase kinase kinase 18-like	STY	3.14	2.55E ⁻⁰⁵	
ABD28520	protein RRP6-like 2	CK activated	3.09	9.33E ⁻⁰⁵	
XP_013451548, KEH25576, RHN50766	ankyrin repeat/protein kinase domain-containing protein 1	TF ERF	3.06	2.97E ⁻⁰⁵	
AES95938	disease resistance protein (TIR- NBS-LRR class), putative	LRR RLK	3.01	1.91E ⁻⁰⁵	
RZB96753	probable LRR receptor-like Ser/Thr-protein kinase	LRR RLK	2.99	5.78E ⁻⁰⁵	
KHN26259	zinc finger MYM-type protein 1- like	TF Zn finger	2.95	1.31E ⁻⁰⁴	
XP_013451184, KEH25223, RHN50327	protein NSP-interacting kinase 1	NSP	2.94	3.50E ⁻⁰⁵	
RHN42361	kinase RLK-Pelle-WAK-LRK10L- 1 family	RLK	2.89	1.29E ⁻⁰⁴	
RIA84146	Ca2+:H+ antiporter	Calcium/lipid- binding	2.87	8.02E ⁻⁰⁵	
AES60803	F-box plant-like protein	F-box protein	2.78	1.71E ⁻⁰⁴	
XP_013457946, KEH31977, RHN63702	putative LRR-containing protein	LRR RLK	2.78	1.60E ⁻⁰⁴	
RIA97789	ARM repeat-containing protein	E3 ligase	2.72	2.11E ⁻⁰⁴	
XP_013445632	G-type lectin S-receptor-like Ser/Thr-protein kinase	Ser/Thr protein kinase	2.71	1.84E ⁻⁰⁴	
AES73438	Plant regulator RWP-RK	NLP	2.70	1.87E ⁻⁰⁴	

<u>Table 6.4c</u>: Enriched DE transcripts in VFA roots which are putatively classed as highly preferential nodulation regulatory genes and nodule-associated signalling-related genes as in Roux *et al.*, (2014) [705], continued from Table 6.4b.

Gene/Protein ID	Description	Annotation	logFC	q-value	
KEH38435	Rpp4C3	CK activated	2.69	1.22E ⁻⁰⁴	
RIA81779	YIF1-domain-containing protein	TF AP2/ERF	2.69	1.26E ⁻⁰⁴	
AES61923, RHN81250	C3HC4-type RING zinc finger protein	TF Zn finger	2.68	1.90E ⁻⁰⁴	
XP_024633471.1	<i>LysM domain receptor-like kinase 3</i>	LysM receptor kinase	2.68	1.41E ⁻⁰⁴	
XP_024625794	<i>putative receptor-like protein kinase</i>	RLK	2.66	4.17E ⁻⁰⁵	
RHN81081	proline-rich protein 1-like	PRP	2.63	5.03E ⁻⁰⁵	
PF04909	nodulin-6	NIP	2.56	2.00E ⁻⁰⁴	
XP_003615114, AES98072, RHN56135	nodulin-26	NIP	2.56	6.80E ⁻⁰⁵	
XP_013450575, RHN49450	L-type lectin-domain containing receptor kinase IX.1-like	RLK	2.49	5.82E ⁻⁰⁵	
XP_013462891, KEH36925, RHN72571	chitin elicitor receptor kinase 1- like	LysM receptor kinase	2.48	9.25E ⁻⁰⁶	
XP_003601076.1, AES71327	nodulation-signaling pathway 2 protein	NSP	2.11	1.43E ⁻⁰⁵	
XP_024641514, AES76606, RHN52721	putative NF-X1-type zinc finger protein NFXL1-like protein	NFX1-type zinc finger	1.75	1.68E ⁻⁰⁵	
XP_013460228, KEH34259, RHN67624	non-specific phospholipase	Phospholipase A2	1.52	4.71E ⁻⁰⁶	
XP_024625319	<i>U-box domain-containing protein 33 isoform X1</i>	MtPUB	1.37	1.14E ⁻⁰⁴	
XP_003631134, AET05610, RHN43936	probable inactive receptor kinase At1g48480	Kinase	1.15	1.75E ⁻⁰⁴	
XP_003616008, AES98966, RHN56723	<i>CBL-interacting</i> serine/threonine-protein kinase 11	Calcium binding, Ser/Thr protein kinase	1.04	1.40E ⁻⁰⁴	
RHN48771	NDR1/HIN1-like protein 1	NHL	0.88	3.71E ⁻⁰⁵	

6.2.8 qRT-PCR confirmed a subset of DE transcripts which are associated with nodulation

A subset of root DE transcripts were confirmed for their expression using qRT-PCR; similar expression patterns between RNA-seq data and qRT-PCR analysis would validate the previous association of VFA with nodulation effects. Three important genes for nodulation-associated signalling and regulation were chosen, *Myb/SANT-like DNA-binding domain protein*, *LysM domain receptor like kinase 3, nodulin-26,* as well as the N metabolism gene *NRT1-PTR family 4.5-like.* The *beta-tubulin* gene was also measured using qRT-PCR as a gene with reduced expression after 3 days of VFA treatment. Figure 6.7 shows a table of transcript ID tested, with RNAseq and qRT-PCR fold change values (Figure 6.7a). Below this, each gene is plotted individually for RNA-seq data (Figure 6.7b), and qRT-PCR analysis (Figure 6.7c).

All genes showed similar expression patterns between different treatments and timepoints. For example, for *Myb/SANT-like DNA binding domain protein* gene in both treatments at day 0 was low in expression, then for day 3 VC increased in transcript number. However, for day 3 VFA transcript level were many folds higher than the other treatment timepoints. Each of these genes were validated for their RNA-seq expression values using qRT-PCR with *ACTIN2* as the reference housekeeping gene.

a) Transcript ID and expression data			RNAseq mean log ₂ FC			qRT-PCR mean FC			
		Day 0		Day 3		Day 0		Day 3	
Transcript ID	Annotation	VC	VFA	VC	VFA	VC	VFA	VC	VFA
TRINITY_DN157351_c1_g1	Myb/SANT-like DNA-binding domain	2.02	0.94	2.18	43.86	0.96	2.39	6.68	33.08
TRINITY_DN415738_c0_g1	LysM domain receptor-like kinase 3	0	0	0	2.67	0.33	0	11.98	30.37
TRINITY_DN164623_c1_g1	nodulin-26	13.37	22.62	64.69	130.75	0.33	0.43	2.12	38.09
TRINITY_DN170549_c0_g1	protein NRT1/ PTR FAMILY 4.5-like	33.76	24.33	107.50	193.20	0.68	2.05	4.41	8.59
TRINITY_DN154692_c3_g1	beta-tubulin subunit	1.86	8.00	1.67	0	0.90	0.63	0.16	0.15

b) RNA-seq expression level







mean log₂FC



c) qRT-PCR expression level



Figure 6.7: Quantitative reverse transcription PCR (gRT-PCR) of a subset of differentially expressed (DE) transcripts to confirm RNA-seq in Medicago sativa

a) Table of ID and Annotation for five DE transcripts obtained from RNA-seq for Medicago sativa root tissue with treatments of either fulvic acid (VFA, orange) or the control (VC, yellow). RNA-seq was carried out on whole root RNA samples taken on day of treatment (day 0) or three days after treatment (day 3). Mean expression level of three independent experimental samples is provided for both RNA-seq and qRT-PCR. b) Bar charts showing mean log₂ fold change in expression (log₂FC) of the five DE transcripts. c) Bar charts showing mean fold change in expression (FC) relative to the reference gene ACTIN2 of the five DE transcripts. In all cases error bars represent standard error of the mean.

6.3 DISCUSSION

6.3.1 *De novo* transcriptome assembly of *Medicago sativa* was successful

The above analysis is based on the successful *de novo* transcriptome assembly of *M. sativa*, shown in Table 6.1. This assembly mirrors other recent studies carried out in the forage crop [722; 723; 725; 724; 726-730] assessing transcriptional changes in relation to other treatments including stress responses. Such studies can provide information to improve the *Medicago sativa* gene index 1.2 [360], which can progress the genetic resources available for future studies in this important global forage crop.

There was a large overlap in the genes detected in *de novo* analysis which were also detected when mapped to A17 [702; 707] or R108 [178] reference accessions directly. As *M. sativa* is not yet fully sequenced and annotated it is likely the tables of BLAST results and GO terms, found in Figures 6.4, 6.5 and 6.6 and Tables 6.2 and 6.3, are not exhaustive. Some DE transcripts without a BLAST *M. sativa* or legume homologue match are therefore missing from this analysis. These DE transcripts may be of particular interest to study in subsequent work to characterise new genes.

6.3.2 Fulvic acid treatment causes increased

transcription for metabolic processes with upregulated transporter activity predominantly in the root tissue

This study shows that FA as VFA can induce substantial transcriptional change in *M. sativa* after only three days. Significant transcriptional changes occurred in both shoot and root tissues, see Figure 6.2. Metabolism and responses to biotic and abiotic stimuli were up-regulated in both tissues after VFA treatment, as discussed in 6.2.4. Molecular functions which were affected following VFA treatment included catalytic and binding activities, particularly for transport and oxidation-reduction

229

processes, as discussed in 6.2.5. The GO analysis provides evidence that VFA very quickly effects crucial pathways in both C and N metabolism, as well as cell wall modification. This rapid transcriptional effect helps to induce the later yield effect in vegetative tissue found in Chapter 5.

The root had higher amounts of DE transcripts than shoots, see Figure 6.2 and Appendix D Figure D2. This could be caused by the method of application, with VFA and VC being applied to the soil. The higher number of DE transcripts may be due to the roots being the site of more transcriptional change as it is the site of VFA uptake, thus it is affected more promptly and excessively. In addition, the enrichment of N metabolism is more likely to be shown in the root as N uptake is in the roots as nitrate and ammonium [72] then transported up to the shoots as the main tissue for assimilation in both higher and lower order plants [778]. Analysis of soil in Chapters 3 have shown soil to be low in nitrate level for no inhibition of nodulation to occur which is ~ 16 mM for M. truncatula [779] and above 10 mM for L. japonicus [780]; soil in this study was ~ 8.8 mM in conventional soil water analysis, and $\sim 5.53 - 9.02$ mM with ion-selective sensor analysis. Changes in N metabolism in roots of legumes is closely associated with increases in nodulation-signalling during initiation of symbiosis [781]. The quick response in transcription in the roots provides evidence for why there is a larger biomass increase after VFA treatment through stimulated N supply to the legume via nodules or uptake by the roots.

6.3.3 Fulvic acid treatment causes enriched upregulation of nodulation regulatory and nodule-associated signalling-related genes

The above transcription enrichment tests shown in Table 6.2 and 6.3, plus the nodulation experiments in Chapter 5, provides strong evidence that VFA can readily induce nodulation when compared to controls. Genes in the root which are significantly induced following VFA treatment overlap with those in studies of early initiation of nodulation in other legumes [700; 709; 703; 719; 720; 716; 706; 714]. Many DE transcripts in Tables 6.4a,
6.4b, and 6.4c have been characterised for their role in nodulation as follows; LysM-type receptor-like kinases are perceive early *Rhizobium* signals [741; 749-752; 747; 782; 783; 754]; many leucine-rich repeat receptor-like kinases including *CLAVATA* protein homologues signal root development and nodulation induction [784-789]; AP2/ERF transcription factors controls nodule number and differentiation [790-793]; nodulin is crucial in early nodule development [794-799; 765; 800; 801; 714; 802; 803] including in *M. sativa* [764; 804-806]. Important chitin regulatory genes are also detected to be changed in their expression by VFA treatment, discussed in 6.2.6. This may affect lipochitooligosaccharide recognition as the key signal in initiating legume-*Rhizobium* symbiosis [807-811]. Other defence-related genes such as *Pathogenesis-related proteins* are increased which can be induced in early symbiotic infection, before adequate *Rhizobium* suppression, rather than being in relation to a pathogen response [709; 701; 742; 753; 782; 735; 773].

The increase in transcription of these genes upon VFA treatment is a strong indication that it is associated with inducing early nodulation in *M. sativa*. This could be by influencing the plant itself in its response to symbiosis, for example a priming effect of VFA for subsequently inducing infection by the symbiont [812-814]. This would match the effect shown in *Pisum* sativum for soil primed with HS increasing nodulation and mycorrhizal colonisation in roots [646]. Fulvic acid may be able to change the C:N metabolic balance of the plant and thus impact on the regulatory mechanisms of promoting symbiotic nodulation processes [815]. Or the effect could be a consequence of the treatment on the symbiont causing a nodule number increase. VFA may contain a specific nutritional aid, not adequately controlled for in VC application, which boasts symbiotic Sinorhizobium growth in soil and thus makes nodulation happen more rapidly [816; 817]. Or similarly, VFA may decrease the inhibitory role of N in soil on nodulation and thus also encourage nodulation to occur with symbiont and plant [818; 779]. This is unlikely due to the low N content of the soil used in testing, but should be considered.

This page is intentionally left blank

Chapter 7: General Discussion

7.1 SUMMARY OF RESULTS WITH IMPLICATIONS

The overall goal of this project was to study connections between N input and final vegetative biomass in a range of forage crops, particularly *Lolium perenne*, a *Festulolium* hybrid, and *Medicago sativa*. Strategies were developed to test this relationship with the aim of increasing biomass with lowered inputs in future forage crop cultivation. In addition, the project succeeded in providing underpinning evidence to help high-intensity, temporary forage growers, such as BAGCD members, plan their crop management regarding N fertiliser use. The results compared to targets for improved forage crops in Figure 1.4 are shown in Figure 7.1. Below is a discussion of the research in the context of the project area hypotheses, along with the final points for discussion on each of the main project areas. Herein, I will cover N status marker genes as assessment tools from Chapter 3; Management practices influencing soil N profiles using NO₃⁻⁻ selective sensors from Chapter 4; Fulvic acid as a biostimulant in forages from Chapters 5 and 6.



<u>Figure 7.1:</u> Results compared to targets for improved forage crops in Figure 1.4.

Adapted from Capstaff and Miller (2018) [18], showing the main results of the thesis compared to targets for improvement in forage crop research.

7.1.1 N status marker genes have use across forage crop industry

Hypothesis for Chapter 3: Expression of N status genes is reliably related to crop production and soil N status.

Through the study of marker genes in Lolium, Festulolium, and Medicago, the results in Chapter 3 provide a greater understanding of how the $NO_{3^{-1}}$ supply in the environment can affect NUE associated gene expression in forage crops. The different patterns of N status marker genes in relation to NO_3^- concentration show that evidence from N metabolism from model plant studies can be applied to less studied crops like forages. This was especially true for the genes NIR and NADH-GOGAT from Arabidopsis studies [377; 378; 134; 133], showing parallel expression in the grasses (see Figures 3.1, 3.4, and 3.5). Being able to relate this information in the field for grasses for optimal plant growth and metabolism to soil N levels is valuable for the high intensity agriculture performed on such crops. In particular, gene expression analysis can help to understand sense and response of NO3⁻ in low or high N supply environments in a range of grasses, in temporary and permanent grassland. Although gene expression testing was not successful in *M. sativa*, the development of confirmed efficiency gene primers is still of use to studies in this legume. For example, the gene primers could be used to assess *M. sativa* response to pathogens known to affect C or N metabolism in legumes [819-822].

High variation of soil N in temperate fields such as those in the UK [391-397], was accounted for by testing of plant tissue. With current soil testing it is conventional to test only 3 – 5 soil cores per field each year randomly across a site. In this study three samples per field were also taken, but within each sample there are around 25 – 50 individual grass leaf blades. Through gene expression analysis of only three marker genes, it is possible to gain a picture of the crop's N status and representative soil NO₃⁻ value across many fields in only a few days. This study also found a general trend where the higher the predicted soil NO₃⁻ from the gene expression analysis then the higher the subsequent later yield of the field in t ha⁻¹ (see Figure 3.10), although this does not correlate as strongly as actual extractable

soil NO₃⁻. This is likely due to yield being affected by more factors, especially that of biotic and abiotic stress differences across years unaccounted in the model. More sampling across more years may provide improved data for using marker genes to accurately predict yield as well as actual extractable soil NO₃⁻.

Grass tissue sampling is quick and easy, as long as samples can be flash frozen and stored adequately before processing. Soil samples can be difficult to standardise, with compaction making such samples difficult to access. Moreover, soil cores are often split into different depth layers as N levels vary from shallow to deep soils. By using leaf tissue, which is known to have networks of systemic signals from root tissue that penetrate these depths, the analysis reflects the integrated true available N to the plant, not just what can be detected in soil cores. The development of handheld PCR kits in the field [416-419], should make such a routine procedure on farms more possible. This development of a suite of NUE status marker genes could also easily be applied to breeding programmes, which are already equipped with the techniques required to analyse such tissue effectively in a short timescale. Furthermore, they could be used alongside soil column experiments discussed in Chapter 4 to further relate plant growth to changes in soil profiles.

It should be noted that there are some limitations with soil columns. It can be difficult to mirror field soil conditions with variables including changes in soil layers and compaction levels. The differences in soil layers from sand to loam to clay can alter water movement, and compaction in UK farms affects N fertiliser applications. To improve this system, any future experiments could try to replicate these extra factors in soil columns, especially by trying to mirror the conditions to match the soil conditions of BAGCD member farms. Although the soil columns in this project do not match such factors, the profile data is superior to other methods such as hydroponics and gel-based media; these methods cannot reproduce such data as water movement is unconstrained. Repeating such experiments with soil layers and compaction included would be more applicable to forage crop cultivation, as could more depths of samples being included for analysis. Moreover, using gene expression analysis on crops grown in

columns to see if these match with the field data could provide more evidence of how well soil columns relate to the situation within the field.

7.1.2 Studying the field soil N availability provides more detail for growers to pursue precision agriculture techniques

Hypothesis for Chapter 4: Using nitrate-selective sensors in soil columns with grass and lucerne will provide valuable data on plant-soil interactions of management practices.

Soil column experiments using NO₃⁻-selective sensors were successful in tracking NO₃⁻ through soil profiles following different management practices. This provided valuable data for real-time soil changes following NO₃⁻ application, defoliation of vegetative tissue, and the role intercropping may play in plant-soil interactions for these practices. The use of NO₃⁻selective sensors could help in the advancement of precision agriculture by providing detailed information of fields [823; 514; 517], where other conventional methods are too labour-intensive, expensive, and provide only one time measurements. Furthermore, this study using soil columns suggest there is little leaching of NO_{3} when forage crops are present with the current management practices. This means that despite the high NO₃⁻ application, little $NO_{3^{-}}$ is actually lost from the base of the column when plants are grown, suggesting that leaching in temporary systems such as BAGCD maintained sites may not be as problematic as previously proposed. This is likely to only be in cases where crops are able to efficiently take up the applied NO_3^- , as is possible in well-maintained glasshouse conditions.

The most striking observation in these measurements was the detection of a NO_3^- 'burst' in the middle soil column region following defoliation of *L. perenne*. Possible reasons for the NO_3^- 'burst' could include decreased root uptake through changes in transpiration rate [538-542], reduced plant N uptake [209], or changes in N composition to other non- NO_3^- forms [212; 531; 530]. However, these seem unlikely as the concentrations of NO_3^- are exceedingly high, reaching around 60 – 75 mM. Such dramatic changes

not found in other levels of the soil profile, and well-conserved across experiments, suggests this is a potential grass defoliation phenomena previously unobserved. It is unlikely that the NO₃⁻ 'burst' would be a problem for growers, as its detection drops overtime suggesting uptake (or reuptake) of the NO₃⁻, with little evidence of leaching to the lower level through either NO₃⁻-selective sensor data or conventional testing. Furthermore, the lack of a detected 'burst' when intercropped with *M. sativa* also provides evidence that this is grass specific, with the legume's own root system or potential to change soil microbial populations being possible reasons for no detection.

It is feasible that C-containing compounds exuded by the root could interfere with the NO₃-selective sensor measurements. However, malate has been tested and shown not to affect readings [518], although, it is possible that there may be other organic anions that interferes with sensor readings. Nevertheless, this would not explain why any potentially interfering anions are not detected in the intercropping system. It seems reasonable to suspect that the lack of detection of the 'burst' is due to root architecture differences, already of note in erosion studies of intercropping grasses and legumes and also found in permanent pastures [88-92]. As the lack of NO_3^- 'burst' detection in intercropping experiments may be due to improved diversity in root architecture of *M. sativa* with *L. perenne*, compared to monocropping of *L. perenne* alone, studies should be carried out with more grasses with cultivars known to have better root systems. For example, measuring for NO_3^- 'burst' across grass species especially Lolium compared to both Festuca and Festulolium hybrids may help to elucidate this as they have different root architectures from breeding programmes. Moreover, if the lack of detection burst in intercropping is due to microbial population changes from legume crop, then testing other forage legumes including *Trifolium* and *Lotus* may provide evidence of this, as discussed in Chapter 4.

7.1.3 New management practices including fulvic acid application could be advantageous in forage crops

Hypothesis for Chapter 5: Application of the commercially available biostimulant fulvic acid (FA) improves forage crop production. &

Hypothesis for Chapter 6: RNA-seq will provide evidence of early differentially regulated genes in either shoot or root tissue upon fulvic acid application.

An increase in vegetative biomass in *Medicago sativa* was observed upon treatment with FA across laboratory and field trials. Through experimental analysis in Chapter 5 and RNA-seq analysis in Chapter 6, it is clear FA also has an impact on soil microorganisms such as Sinorhizobium species, either directly or indirectly. The lack of biomass increases in grasses (see Figures 5.1 and 5.2) suggests FAs role of N-fixing bacteria may be at least partially responsible for the increase of biomass in legumes. It seems very likely that FA may replicate C-containing exudates usually released by plants to aid in symbiosis initiation, which in turn stimulates growth and activity of this bacteria. This leads to increased nodulation signalling which makes nodulation symbiosis more likely, with the end result of higher nodule number which directly increases vegetative growth. There was no increase in protein or chlorophyll concentration suggesting that the enhanced vegetative growth is not due to protein storage but due to increased growth rate. This can be achieved by increase N causes more leaf expansion and, therefore, more C capture via photosynthesis may occur to ensure a C:N balance, which seems to be the case in both laboratory and field conditions. This could be tested through the use of NUE status marker genes linked to C metabolism such as RBCS, GLN, and NADH-GOGAT from Chapter 3. This would mean that although the suite of genes was not conducive for assessing soil N status of Medicago the markers may have use as a proxy for C:N metabolism rate.

Although this study provides strong, indirect evidence for the FA effect on the nodule and bacteria, the above hypotheses for increased vegetative growth through nodulation symbiosis encouragement requires testing. More direct methods could be used particularly in testing legume nodulation mutants, or with inoculation with *Rhizobium* nodulation mutants [824-827]. If when nodulation is not possible, especially at an early initiation stage, FA does not incur a later vegetative phenotype in the plant then a mode of action can be more formally hypothesised. Reporter bacteria and measurements of N fixation such as through acetylene reduction assays will make such ideas testable.

It was thought for many decades that humic substances, like FA, have a crucial active ingredient or 'hormone', such as an auxin-like molecule [653; 654; 607; 643; 655]. However, based on the analysis in this project, no such molecule was detected as both commercial applications were found to be remarkably different from one another. It is possible that many studies have exaggerated results due to a lack of standardised nutritional controls especially for FA elemental contents, and although solutions such as FA do promote growth responses it is likely to be a mixture of many compounds which provide this increased yield. It would be interesting to screen FA in relation to the growth, or inhibition, for a range of bacterial strains, as this may provide more evidence of which classes of compounds are most necessary for the effect. To assess whether FA is a considerable management practices in legume farming, large-scale and new standardised testing of a range of globally sourced FAs is required. By performing an extensive screen of FAs it may be possible to find common nutritional components in solutions, which is not possible in a study of only two compounds. Controls such as those used in this study could then be introduced to further pinpoint particularly important fractions, which may lead to targeted synthesis of the molecules found in such portions for use in later studies.

7.2 OPEN QUESTIONS

Although this research has provided both specific outcomes and general information of relevance to forage growers in the UK, there are still many open questions which could be further investigated.

7.2.1 Improving tools for tackling nitrogen use efficiency need to be accessible to growers

The strategies adopted in this study to assess plants and soil, primarily the suite of N status genes and use of NO₃⁻-selective sensors, provide tools that can be used in research for many applications. More research into grass regrowth post-defoliation in regard to N metabolism using gene expression could be deployed alongside more thorough analysis of soil NO₃⁻ profile changes. Additionally, these tools could be applied to other crops, especially cereals, which are in the same family as forage grasses, or leafy vegetables, both of which are important in the forage industry. Cereals are complicated by the fact that the crop tissue of interest for NUE (defined in Table 1.2), is grain and not whole vegetative biomass, therefore more analysis may be required before deployment in these crops. Leafy vegetables are similar to forage crops as crop tissue of interest is whole vegetative biomass, and so may be more readily applicable with these tools.

The suite of genes tested and confirmed for field use have a range of applications, particularly in prompt assessment of new hybrid cultivars for their metabolic capacity under many parameters, especially for abiotic stresses, such as drought and severe temperature fluctuations.

NO₃⁻-selective sensors look set to provide a whole range of exciting prospects for future studies, especially due to recent hopes that they can be made using solid-state technology, like some new types of commercial pH sensors. This development, along with improvements in data analysis and sharing, will greatly improve data acquisition in soil column experiments, but will also aid in establishing field-based measurements to compare to those shown in this project.

7.2.2 Management practices require thorough testing to demonstrate their effectiveness as grassland research is not standardised

It is clear from this project that new management practice could be implemented in UK forage farming, such as the addition of FA to legume cultivation. FA application could be provided as a seed coat treatment, or mixed in with inoculum already applied to legume seeds. Such practices require large-scale testing in industry, where standardisation of yield measurements can be achieved across multiple studies around the world over many seasons.

In addition, the management practice of intercropping may be a highly effective way to negate leaching in soils, however due to problems with harvest intervals and competition, it is still in the relatively early stages of practice for modern intensive agriculture. Moreover, the soil column data in this study suggests little evidence of NO₃⁻ leaching when forage crops are present, as either a monocrop or as an intercrop, with increased biomass of *L. perenne* when *M. sativa* is present. As animal feeds can be a mix of grasses and legumes, forage crop cultivation is therefore an ideal system to test more how leaching is affected by rooting systems.

A key problem with this is the range of forage crop cultivation methods found in the UK and Europe, discussed in Chapter 1. Soil columns are one way to relatively quickly assess management practices whilst changing parameters. Yet, a crucial problem will still remain in how to apply results to forage croplands and grasslands. It may be necessary for studies on leaching of N fertiliser to be carried out in both temporary and permanent field systems, for data to be truly meaningful in each circumstance. This is important when using research to assign policy to agriculture – it is clear a one-fits-all solution to forage cultivation in regard to fertiliser application is unlikely. Nevertheless, the improvements described in this project should contribute to more rapid analysis of forage crops in multiple cultivation systems. Below are the main findings for this thesis work, followed by the recommended future testing based on them.

Main findings:

- Festulolium testing during growth using suite of N status marker genes was able to accurately assess both soil NO₃⁻ levels and future yield, instead of conventional testing.
- No detected soil changes of nitrate level were found for intercropping experiments following defoliation, with intercropping of forage grass with legumes, such as *M. sativa*, with decreased N fertiliser application gaining similar biomass.
- An FA application showed improved yield in legumes, particularly in *M. sativa*, and was linked to nodulation affect.

Recommended future testing:

- Testing of more grass species using N status marker genes can prove their use in fields, especially with more field trials included for applicability to growers and breeders.
- Further analysis for NO₃⁻ leaching in field can be carried out with NO₃⁻-selective sensors, as unobserved as a problem in soil columns with current management practices.
- Applying FA to legume forage crops, such as with testing in seed inoculum as applied by seed suppliers currently, can be further tested for investigating the shown increased vegetative growth.

7.2.3 Fertiliser use must be curtailed ahead of resource depletion

As forage crop cultivation methods in the dried forage industry of the UK are one example of the huge levels of N fertiliser application in agriculture, it is clear the sector requires vast changes promptly. N fertiliser use is energy and resource exhaustive, so research in agricultural practices with high N use with the aim to decrease N use are vital to study, so N usage is reduced substantially and quickly. N fertiliser resource depletion due to tot-al energy resource strains are on the horizon, so such research should be prioritised to curtail use in the UK. Through studying a resource demanding systems such as forage crops in high intensity, temporary grasslands, it may be possible to apply use efficiency advances in other crop systems. The results of this project have shown how yields can be measured for forage crops in relation to their N application, with each providing a potential way to decrease fertiliser applications. This is particularly true for the rise in precision agriculture, with the tools of N status marker genes and NO_3 -selective sensor soil data improving when and where N fertiliser is applied in the future.

7.3 CONCLUDING REMARKS

Temperate forage crops are exceptionally important crops, and when grown for dried forage production as high-intensity, temporary cropland they can have substantial fertiliser applications. As an increase in efficiency is required for this agricultural practice, the aim of this thesis was to investigate three strategies to increase yields and assessments in UK forage crops.

Through validating a suite of N status marker genes for use in forage crops, field testing has provided more evidence of how crop production is related to soil N levels. This has provided data directly applicable to grower and breeder needs, and in addition, provided knowledge for N metabolism studies in grasses.

Nitrate-selective sensors have provided extensive data of soil N profile changes following different management practices. This data has been shown to be superior to conventional testing. Additionally, the detection of a NO_3^- 'burst' following defoliation of grasses localised to one area of soil columns requires further investigation.

Lastly, the implementation of a new management practice of fulvic acid application has shown to be very useful in increasing vegetative biomass in lucerne. This yield increase was shown in many growth conditions, including the field. As nodulation was also affected, in physiological experiments and transcriptome analysis, there is scope to investigate fulvic acid's role in both plant and microbial growth.

This project has provided practical information to aid forage crop growers in farming more efficiently, especially with providing tools to assess both plant and soil conditions. This work provides new avenues of work into management practices of forages which are underrepresented in current research.

This page is intentionally left blank

Appendix A: Chapter 3 supplemental figures



<u>Figure A1:</u> Arabidopsis eFP Browser [128-130] images for expression candidate N status genes.

Genes were as follows; *CLCa* (AT5G40890); *NIR* (AT2G15620); *GLN1* (AT5G37600); *GLN2* (AT5G35630); *VSP1* (AT5G24780); *VSP2* (AT5G24770); *TIP1.1* (AT2G36830); *TIP1.2* (AT3G26520); *TIP3.1* (AT1G73190); *TIP3.2* (AT1G17810); *NADH-GOGAT* (AT5G53460); *RBCS2B* (AT5G38420); reference genes *ACTIN8* (AT1G49240) and *CYP5* (AT2G29960).



<u>Figure A2:</u> Forage crop RT-PCR screen of developed N status marker gene primers electrophoresis gel photos.

PCR conducted as in Section 2.1.11 with numbering as indicated in key.

Medicago sativa cv. Daisy



<u>Figure A3</u>: Gene expression patterns of N status marker genes in *Medicago sativa*, cv. Daisy and Luzelle, vegetative tissue at three weeks. Expression was calculated as the geometric expression percentage of independent CaNO₃ and KNO₃ experiments relative to lowest NO₃⁻ concentration (0.6 mM shown as 100 % in pale green). Bars are denoted for increasing concentration using darkening shades of green, and error bars are standard deviation of expression using two reference genes with 10 biological replicates, in two independent experiments. Expression of genes showed to vary dramatically in relation to NO₃⁻ concentration with agreement between cultivars. Genes tested in Daisy and Luzelle were; a) and f) is *NIR*; b) and g) is *RBCS*; c) and h) is *NADH-GOGAT*; d) and i) is *GLN*; e) and j) is *MCP1* (*Medicago truncatula TIP1* homologue).

<u>Table A1:</u> R Script outlined for Gaussian Process (GP) fitting model of N status marker gene expression.

Script uses both 2018 and 2019 glasshouse data detailed in Section 2.1.20.

```
#Packages installed and opened, theme set to 'theme bw'
install.packages("installr")
library(installr)
install.packages("tidyverse")
install.packages("cowplot")
install.packages("ggplot2")
install.packages("data.table")
install.packages("GauPro")
library(tidyverse)
library(cowplot)
library(ggplot2)
library(data.table)
library(GauPro)
theme set(theme bw())
#Functions assigned for normalising gene expression
geoMean <- function(x) {</pre>
  return(exp(mean(log(x))))
}
load data <- function(file.name) {# read data and formatting</pre>
  df <- fread(file.name)</pre>
  names(df)[names(df) == '2.delta.ct'] <- 'two.delta.ct'</pre>
  df$gene <- factor(df$gene, levels=c('NADH', 'NIR', 'TIP'))</pre>
  df$hk.gene <- factor(df$hk.gene, levels=c('ACT', 'GADPH'))</pre>
  df$cv <- factor(df$cv, levels=c('hykor'))</pre>
  df[, mean.delta.ct:=geoMean(two.delta.ct), by=.(environment, cv,
NO3, treatment, gene, rep)]
 df$hk.gene <- NULL
  df$two.delta.ct <- NULL
 df <- unique(df)</pre>
  df$sample id <- paste(df$environment, df$cv, df$treatment,
'NO3=', df$NO3, df$rep, sep='-')
  return(df)
}
my.norm <- function(x) {</pre>
 y < -x / mean(x)
  return(y)
}
standardize <- function(v) {</pre>
 v <- (v - min(v)) / (max(v)-min(v))
}
unstandardize <- function(w, vmax, vmin) {</pre>
 v <- w*(vmax-vmin)+vmin</pre>
}
opt.rescale <- function(curr.opt.df, r) {</pre>
 curr.opt.df$expt.val <- curr.opt.df$mean.delta.ct.norm * r</pre>
```

```
curr.opt.df$dnorm <- dnorm(curr.opt.df$expt.val,</pre>
mean=curr.opt.df$yhat, sd=curr.opt.df$se, log=T)
  ll <- sum(curr.opt.df$dnorm)</pre>
#Importing data and assign above factors
getwd
setwd('/RMG2018 2019')
df <- load data('RAWdataboth.csv')</pre>
df$year <- as.factor(df$year)</pre>
df.h <- df[df$cv=='hykor', ]</pre>
df.h[, mean.delta.ct.norm:=my.norm(mean.delta.ct),
by=.(environment, cv, treatment, year, gene)]
#Plot of 'mean.delta.ct.norm' versus 'NO3'
ggplot(df.h, aes(x=NO3, y=mean.delta.ct.norm, color=year))+
  geom point(alpha=0.5, size=2)+
  facet wrap(~environment+gene, scales='free y', ncol=3)
rm(df)
#Functions assigned for building model with glasshouse data using
'yhat' and 'xpred', including min and max
p.list <- list()</pre>
qp.list <- list()</pre>
pred.list <- list()</pre>
df.g <- df.h[df.h$environment=='greenhouse']</pre>
df.g$standard.NO3 <- standardize(df.g$NO3)</pre>
xmax <- max(df.g$NO3)</pre>
xmin <- min(df.g$NO3)</pre>
for (curr.gene in unique(df.g$gene)) {
 print(curr.gene)
 curr.df <- df.g[df.g$environment=='greenhouse' &</pre>
df.g$gene==curr.gene]
  x <- matrix(curr.df$standard.NO3)</pre>
  y <- matrix(curr.df$mean.delta.ct.norm)</pre>
  ymax <- max(y)</pre>
  ymin <- min(y)</pre>
  y=standardize(y)
  gp <- GauPro(X=x,Z=y, verbose=2)</pre>
 xpred <- seq(min(df.g$standard.NO3), max(df.g$standard.NO3),</pre>
length.out=50)
  yhat <- gp$pred(xpred)</pre>
  UL <- gp$pred(xpred) + 2* gp$pred(xpred,T)$se</pre>
  LL <- gp$pred(xpred) - 2* gp$pred(xpred,T)$se
  out.df <- data.frame('gene'=curr.gene, 'NO3'=x,</pre>
'mean.delta.ct.norm'=y)
  out.df.pred <- data.frame('gene'=curr.gene, 'xpred'=xpred,</pre>
'yhat'=yhat, 'UL'=UL, 'LL'=LL)
  out.df$NO3 <- unstandardize(out.df$NO3, xmax, xmin)</pre>
  out.df$mean.delta.ct.norm <-</pre>
unstandardize(out.df$mean.delta.ct.norm, ymax, ymin)
  out.df.pred$xpred <- unstandardize(out.df.pred$xpred, xmax,</pre>
xmin)
```

```
out.df.pred$yhat <- unstandardize(out.df.pred$yhat, ymax, ymin)</pre>
  out.df.pred$UL <- unstandardize(out.df.pred$UL, ymax, ymin)</pre>
  out.df.pred$LL <- unstandardize(out.df.pred$LL, ymax, ymin)</pre>
#Plot of GP fitting models of each gene with 'yhat' and 'xpred'
 p <- ggplot(out.df.pred, aes(x=xpred, y=yhat))+</pre>
    geom line()+
    geom ribbon(aes(ymin=LL, ymax=UL), alpha=0.3)+
    geom point(data=out.df, aes(x=NO3, y=mean.delta.ct.norm))+
    ggtitle(curr.gene)
  p.list[[curr.gene]] <- p</pre>
  qp.list[[curr.gene]] <- gp</pre>
 pred.list[[curr.gene]] <- out.df.pred</pre>
plot_grid(plotlist=p.list)
#Save GP fitting models for use in field data
saveRDS(gp.list, file='./ok greenhouse modelsboth.rds')
saveRDS(pred.list, file='./ok greenhouse models preddfboth.rds')
#Load the GP fitting and run with field data
pred.list <- readRDS('./ok greenhouse models preddfboth.rds')</pre>
pred.df <- do.call('rbind', pred.list)</pre>
pred.df$se <- (pred.df$UL - pred.df$yhat) / 2</pre>
df.f <- df.h[df.h$environment=='field']</pre>
field.points <- split(df.f, by=c('NO3', 'rep'))</pre>
out.df.list <- list()</pre>
PlotList <- list()</pre>
curr.no3 <- unique(pred.df$xpred)[1]</pre>
for (i in 1:length(field.points)) {
 curr.df <- field.points[[i]]</pre>
 measured.NO3 <- curr.df$NO3[1]</pre>
 sample.label <- paste0(curr.df$NO3[1], ' ', curr.df$year[1],</pre>
' ', curr.df$rep[1])
 print(sample.label)
 curr.no3 <- unique(pred.df$xpred)[1]</pre>
 curr.df.list <- list()</pre>
  i=1
  for (curr.no3 in unique(pred.df$xpred)) {
    curr.pred.df <- pred.df[pred.df$xpred==curr.no3, ]</pre>
    curr.opt.df <- merge(curr.pred.df[, c('gene', 'yhat', 'se')],</pre>
curr.df[, c('gene', 'mean.delta.ct.norm')], by='gene')
    best.LL <- list()</pre>
    best.LL[['objective']] <- opt.rescale(curr.opt.df=curr.opt.df,</pre>
r=1)
    x = seq(min(pred.df$xpred), max(pred.df$xpred), length.out=50)
    prior = dnorm(curr.no3, mean=measured.NO3, sd=10, log=T)
    best.LL[['objective']] <- best.LL[['objective']] + prior</pre>
    out.df <- data.frame('label'=sample.label,</pre>
'expt.NO3'=measured.NO3, 'assessed.NO3'=curr.no3,
'LML'=best.LL[['objective']])
    curr.df.list[[j]] <- out.df</pre>
    j <- j+1
```

```
out.df <- do.call('rbind', curr.df.list)</pre>
  out.df.list[[i]] <- out.df</pre>
  curr.df$best.assessed <-</pre>
out.df$assessed.NO3[out.df$LML==max(out.df$LML)]
}
out.df <- data.table(do.call('rbind', out.df.list))</pre>
out.df <- unique(out.df)</pre>
out.df$ML <- exp(out.df$LML)</pre>
out.df[, evidence:=sum(ML), by=.(label)]
out.df$ML <- out.df$ML / out.df$evidence</pre>
#Plotting of field values for individual genes using GP fitting
model using 'yhat' and 'xpred'
p.grid <- plot grid(plotlist = PlotList)</pre>
p.grid
#Plotting of field values for combined genes using GP fitting
model using P(D|NO3)' for relative values of 'yhat' for each
gene, and 'predicted equivalent greenhouse NO3'
test.df <- out.df[out.df$label=='22.24 2018 1',]</pre>
ggplot(out.df, aes(x=assessed.NO3, y=ML))+
  geom bar(stat='identity') +
  facet wrap(~label)+
  xlab('predicted equivalent greenhouse NO3')+
  ylab('P(D | NO3)')
```



<u>Figure A4:</u> Normalised expression data 'mean.delta.ct.norm' of 2018 glasshouse grown *Festulolium*.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data was calculated from glasshouse measurements in relation to two reference genes, to generate values for 'mean.delta.ct.norm' against known soil 'NO3'. Details found in Sections 2.1.12 and 2.1.13.



Figure A5: Gaussian model of 2018 glasshouse grown Festulolium.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data generated from glasshouse data with a Gaussian model applied to individual genes; y-axis is 'yhat' is the mean of all gene expression values for 'mean.delta.ct.norm' as in above figure, with minimum and maximum value provided; this is required for *NADH-GOGAT* and *NIR* to ensure the model can distinguish the low NO₃⁻ and high NO₃⁻ despite the 'mean.delta.ct.norm' being similar in range. The x-axis is termed 'xpred' and is the predicted soil NO₃⁻ from actual values provided; when field data is analysed with the model it is the 'xpred' which will be deduced from the raw geometric expression data provided. Error region in light grey ribbon shows the upper and lower limit of the model, with the small the region indicating the smaller the range of error between predicted soil NO₃⁻ results. Details found in Section 2.1.20.



<u>Figure A6:</u> Analysis of individual gene models for 2018 field grown *Festulolium*.

Field sample expression data from 2018 was calculated in relation to two reference genes and the normalised gene expression for each gene calculated using a glasshouse tissue generated Gaussian model expression, found in Figure A5; 'yhat' is the mean values for 'mean.delta.ct.norm', and x-axis 'xpred' is predicted soil NO₃⁻. Details found in Section 2.1.20.



<u>Figure A7:</u> Analysis of combined gene models for 2018 field grown *Festulolium*.

Field sample combined normalised gene expression analysed with glasshouse 2018 generated Gaussian model. Here y-axis is 'P(D|NO3)' which is a symbolic derivative for simple expressions of individual gene 'xpred' values shown in Figure 3.8, thus indicating a single area of expression to relate to x. The x-axis is again 'xpred', here termed 'predicted equivalent greenhouse NO3'. This provides a visual representation of predicted concentrations in the field soil with the higher the 'P(D|NO3)' bar points the more reliable the interpretation of predicted soil NO₃⁻.



<u>Figure A8:</u> Normalised expression data 'mean.delta.ct.norm' of 2019 glasshouse grown *Festulolium*.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data was calculated from glasshouse measurements in relation to two reference genes, to generate values for 'mean.delta.ct.norm' against known soil 'NO3'. Details found in Sections 2.1.12 and 2.1.13.



Figure A9: Gaussian model of 2019 glasshouse grown Festulolium.

NADH-GOGAT ('NADH'), *NIR* ('NIR') and *TIP1* ('TIP') normalised expression data generated from glasshouse data with a Gaussian model applied to individual genes; y-axis is 'yhat' is the mean of all gene expression values for 'mean.delta.ct.norm' as in above figure, with minimum and maximum value provided; this is required for *NADH-GOGAT* and *NIR* to ensure the model can distinguish the low NO₃⁻ and high NO₃⁻ despite the 'mean.delta.ct.norm' being similar in range. The x-axis is termed 'xpred' and is the predicted soil NO₃⁻ from actual values provided; when field data is analysed with the model it is the 'xpred' which will be deduced from the raw geometric expression data provided. Error region in light grey ribbon shows the upper and lower limit of the model, with the small the region indicating the smaller the range of error between predicted soil NO₃⁻ results. Details found in Section 2.1.20.



<u>Figure A10:</u> Analysis of individual gene models for 2019 field grown *Festulolium*.

Field sample expression data from 2019 was calculated in relation to two reference genes and the normalised gene expression for each gene calculated using a glasshouse tissue generated Gaussian model expression, found in Figure A5; 'yhat' is the mean values for 'mean.delta.ct.norm', and x-axis 'xpred' is predicted soil NO₃⁻. Details found in Section 2.1.20.



Figure A11: Analysis of combined gene models for 2019 field grown *Festulolium*.

Field sample combined normalised gene expression analysed with glasshouse 2019 generated Gaussian model. Here y-axis is 'P(D|NO3)' which is a symbolic derivative for simple expressions of individual gene 'xpred' values shown in Figure 3.8, thus indicating a single area of expression to relate to x. The x-axis is again 'xpred', here termed 'predicted equivalent greenhouse NO3'. This provides a visual representation of predicted concentrations in the field soil with the higher the 'P(D|NO3)' bar points the more reliable the interpretation of predicted soil NO₃⁻.





Extractable soil NO₃⁻ as mg/kg sampled at time of expression sampling is shown correlated to predicted soil NO₃⁻ based on the highest 'P(D|NO3)' bar point for each field from Figure 3.9. 'P(D|NO3)' is the estimated probability distribution across predicted greenhouse equivalent NO₃⁻ concentration. Linear regression is calculated as R² = 0.92.



<u>Figure A13:</u> *Festulolium* yield for fields from 2018 and 2019, with comparison to vegetative chlorophyll and total protein.

Crop yield in t ha⁻¹ was measured ~ 3 weeks after gene expression sampling is shown correlated to vegetative tissue a) chlorophyll (green triangles) and b) total protein (blue diamonds) sampled at time of expression sampling. Chlorophyll and total protein extraction measured as in Sections 2.1.16 and 2.1.17. Linear regression is shown with $R^2 = 0.32$ for chlorophyll and $R^2 = 0.14$ for total protein.

Appendix B: Chapter 4 supplemental figures






Figure B2: Photograph of soil column set-up for nitrate-selective sensor experiments. Details described in Section 2.2.4.



<u>Figure B3:</u> Lolium perenne monocrop column experiment NO₃⁻-selective sensor data for 'No crop' and 'Monocrop 1'.

Column set-ups are described in Table 4.1. NO₃⁻-selective sensor data are shown independently for top (yellow), middle (orange), and bottom (brown) levels of columns, as described in Figure 4.3. Data is the 12-hourly average of four experimental replicates plotted in GraphPad Prism 7 (GraphPad Software Inc.), with standard errors of the means indicated with thinner lines of a similar colour. Coloured vertical bars indicate management practice of *L. perenne* crop planted (green) and nitrate application at day 0 (blue). In the bottom level graph the soil water from drainage holes for one experiment was tested as leachate using the conventional chemical assay methods described in Section 2.1.15, and indicated by black diamond symbols.



<u>Figure B4:</u> *Lolium perenne* and *Medicago sativa* intercrop column experiment NO₃⁻-selective sensor data for `No crop' and `Intercrop 1'.

Column set-up are described in Table 4.2. NO_3^- -selective sensor data are shown independently for top (yellow), middle (orange), and bottom (brown) depth levels of the columns, as described in Figure 4.3. Data is the 12-hourly average of two experimental replicates, with standard errors of the means indicated with thinner lines of a similar colour. Coloured vertical bars indicate the management practices for the *L. perenne* and *M. sativa* crops (pink), nitrate application at day 0 (blue). In the bottom level graph the soil water from drainage holes from one experiment was tested as leachate using the conventional chemical assay methods described in Section 2.1.15, and indicated by black diamond symbols.

This page is intentionally left blank

Appendix C: Chapter 5 supplemental figures



<u>Figure C1:</u> Weed counts and soil pH of pots with forage crops following one of two fulvic acid treatments relative to a control.

Treatments were deionised water (dH₂O in grey), 0.5 % MFA (MFA in blue), or 1 % VFA (VFA in orange). Treatments were applied to seedlings at 7 days post germination and weed counts were assessed at 7, 12, and 17 days post treatment in every pot. Grass weeds (a) were those identified as couch grass or similar, and leafy weeds (b) were those such as buttercups, nettles, chickweeds, or similar dicot seedling. Total weeds were calculated for all treatment pots. Soil pH was tested at the end of the experiment, with soil tested from treatment pots of six different forage species, three grass species and three legume species. Average measurements for two independent experiments are shown in bar charts above. Error bars show standard error between experiments.





BLUE (lower spectra) = Database result for citric

GC-MS was performed as in Section 2.3.8, with both citric acid monohydrate standard and MFA at a concentration of 0.01 g/mL. The most abundant peak (14.089) was clearly citric acid, when compared with standard and tested by NIST Atomic Spectra database [162]; c) and d) show high similarity to library entries for two derivatives of citric acid, 93.3 % and 19.8 % respectively. Different derivatives of other citrates may have been due to chelation with metals with mass loss of metals during the experiment. The abundance peaks gave an accounted mass of MFA:citric = 7.2:10. Experiment performed by Freddie Morrison under supervision of Paul Brett, with graph by N.C.



<u>Figure C3:</u> Gas chromatogram spectra of (a) PEG-400 standard and b) VFA, with c) NIST Atomic Spectra database 1A v14 matches for poly(ethylene glycol) (heptaethylene glycol).

GC-MS was performed as in Section 2.3.8, with both poly(ethylene glycol)-400 standard and VFA at a concentration of 0.01 g/mL. The staggered peaks across chromatogram was clearly a poly(ethylene glycol), when compared with standard and tested by NIST Atomic Spectra database [162]; c) show high similarity to library entries for poly(ethylene glycol) (heptaethylene glycol) of 93.43 %. The abundance peaks gave an accounted mass of VFA:PEG = 6.9:10. Experiment performed by Freddie Morrison under supervision of Paul Brett, with graph by N.C.



<u>Figure C4:</u> Vegetative and nodule phenotypes of *Medicago sativa* cv. Luzelle following treatment with fulvic acids or elemental controls, with or without inoculation of *Sinorhizobium meliloti*.

Two day old seedlings were transferred to media plates containing treatments as follows; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Plates either remained sterile (open dots on yellow background) or inoculated with *Sinorhizobium meliloti* (closed dots on purple). At 21 days their nodule numbers were counted, and biomass determined for both vegetative tissue and full root tissue. Five seedlings were measured for each treatment condition, and total measurements for three independent experiments are shown in charts. Multiple comparisons were conducted using a one-way ANOVA Tukey test shown with letters, with significance level of 0.05.



<u>Figure C5:</u> Vegetative and nodule phenotypes of *Medicago sativa* cv. Gea following treatment with fulvic acids or controls, with or without inoculation of *Sinorhizobium meliloti.*

Two day old seedlings were transferred to media plates containing treatments as follows; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Plates either remained sterile (open dots on yellow background) or inoculated with *Sinorhizobium meliloti* (closed dots on purple). At 21 days their nodule numbers were counted, and biomass determined for both vegetative tissue and full root tissue. Five seedlings were measured for each treatment condition, and total measurements for three independent experiments are shown in charts. Multiple comparisons were conducted using a oneway ANOVA Tukey test shown with letters, with significance level of 0.05.



<u>Figure C6:</u> CFU counting of *Sinorhizobium meliloti* for growth effects of fulvic acid in TY medium, compared to elemental controls.

TY cultures containing treatments as follows were inoculated with *Sinorhizobium meliloti*; no addition (NA in dark grey); deionised water (dH₂O in grey); 0.5 % MFA (MFA in blue); 0.5 % MC (MC in light blue); 1 % VFA (VFA in orange); 1 % VC (VC in yellow). Average colony forming unit (CFU) counts were obtained from triplicate samples on 0 - 4 days of incubation with shaking 220 rpm at 28 °C. The above example is from 3 days.



<u>Figure C7:</u> Fatty acid methyl esters (FAMEs) relative contents from soil treated with fulvic acid or controls.

Soil from *Medicago sativa* assays with fulvic acid applications (NA; dH₂O; VFA; VC) was extracted for PLFA content at 0 days and 21 days, for three separate experiments, and converted into FAMEs. Individual FAME relative abundance (in % of total FAMEs) was calculated from total FAME biomass (nmol g⁻¹ of dry soil), as shown above.



<u>Figure C8:</u> Fatty acid methyl esters (FAMEs) biomarker assignments to microbial popoulation relative contents from soil treated with fulvic acid or controls.

Soil from *Medicago sativa* assays with fulvic acid applications (NA; dH₂O; VFA; VC) was extracted for PLFA content at 0 days and 21 days, for three separate experiments, and converted into FAMEs. Individual FAME relative abundance (in % of total FAMEs) was calculated from total FAME biomass (nmol g⁻¹ of dry soil). FAMEs were associated with organism as in literature [166], as shown above with a) relative total FAME biomarker assignment for each treatment on both days, and b) the change in FAME biomarkers between 0 and 21 days for each treatment.



Figure C9: Field plots for fulvic acid trials performed in 2017 and 2018.

Treatments were applied to field plots at beginning of establishment and vegetative yields were assessed before 1st cut of growing season. Plot areas are provided, and treatments were; no addition (NA in dark grey); deionised water (dH₂O in grey); 1 % VFA (VFA in orange); and 1 % VC (VC in yellow). In 2017 trials were performed at Dengie Crops Ltd. (Southminster, Essex) with four plots per treatment of both cv. Daisy and Fado, shown in a). In 2018 the trials were at both Blankney Estates Ltd. (Blankney, Lincolnshire) and A Poucher and Sons (Bardney Dairies) Ltd. (Market Rasan, Lincolnshire) with six plots per treatment of cv. Daisy and Gea respectively, shown in b) and c). Maps were generated using Ordnance Survey OpenData OS Open Greenspace – GB (data type: vector, supply format: GML 3): Contains OS data © Crown copyright and database rights (2019).

Appendix D: Chapter 6 supplemental figures

Sample no.	Treatment	Tissue	Timepoint (day)	Experiment replicate	Total amount (ng)	Concentration (ng/µL)	RIN	Nanodrop 260/280
1	Fulvic	Shoot	0	1	8295.9873	360.6951	6.2	1.88
2	Fulvic	Root	0	1	3466.6543	150.7241	7.9	N/A
3	Control	Shoot	0	1	7392.2	321.4	7.1	N/A
4	Control	Root	0	1	8167.8543	355.1241	6.7	1.9
5	Fulvic	Shoot	3	1	6739	293	8.2	N/A
6	Fulvic	Root	3	1	7221.4894	313.9778	4.2	1.98
7	Control	Shoot	3	1	6920.7	300.9	4.5	1.93
8	Control	Root	3	1	8474.0648	368.4376	6.8	1.87
9	Fulvic	Shoot	0	2	12387.8	538.6	7.7	N/A
10	Fulvic	Root	0	2	4898.2065	212.9655	9.2	N/A
11	Control	Shoot	0	2	7399.1	321.7	7.8	N/A
12	Control	Root	0	2	2078.5008	90.3696	9.5	N/A
13	Fulvic	Shoot	3	2	8795.3403	382.4061	8.8	N/A
14	Fulvic	Root	3	2	1340.52763	58.28381	9.1	N/A
15	Control	Shoot	3	2	7813.9441	339.7367	7.7	N/A
16	Control	Root	3	2	2769.3334	120.4058	10	N/A
17	Fulvic	Shoot	0	3	14812	644	7.7	N/A
18	Fulvic	Root	0	3	7420.5452	322.6324	8.7	N/A
19	Control	Shoot	0	3	7024.2	305.4	7.9	N/A
20	Control	Root	0	3	4492.9787	195.3469	8.4	N/A
21	Fulvic	Shoot	3	3	16010.3	696.1	8.9	N/A
22	Fulvic	Root	3	3	4206.9392	182.9104	9.2	N/A
23	Control	Shoot	3	3	14356.6	624.2	8.4	N/A
24	Control	Root	3	3	7120.8	309.6	9.9	N/A

<u>Table D1:</u> RNA quality check for RNAseq samples in Section 2.4.3.



Figure D1: Gel from Bioanalyzer 2100 run of RNAseq samples.

Details are in Section 2.4.3. Lane numbers match sample numbers in Table D1 above.

count of Phred value > 20 or > 30 respectively divided by total base count and GC(%) was bases G and C divided by total count.									
Sample no.	Raw reads	Clean Reads	Raw data	Clean data	Effective (%)	Error (%)	Q20 (%)	Q30 (%)	GC (%)
1	22973960	22535108	6892188000	6760532400	98.09	0.03	97.02	91.88	41.25
2	22902670	22522045	6870801000	6756613500	98.34	0.03	96.9	91.65	41.71
3	20829575	20201808	6248872500	6060542400	96.99	0.03	97.12	92.09	42.11
4	20231682	19865807	6069504600	5959742100	98.19	0.03	96.47	90.83	41.98
5	22912434	22562546	6873730200	6768763800	98.47	0.03	96.91	91.68	42.72
6	21217369	20736820	6365210700	6221046000	97.74	0.03	96.55	90.96	41.36
7	24176064	23832798	7252819200	7149839400	98.58	0.03	96.74	91.29	42.62
8	22920551	22616180	6876165300	6784854000	98.67	0.03	96.94	91.72	41.53
9	20995814	20723523	6298744200	6217056900	98.7	0.03	96.99	91.84	42.1
10	23979006	23681565	7193701800	7104469500	98.76	0.03	96.5	90.85	41.38
11	20951309	20639160	6285392700	6191748000	98.51	0.03	97.06	91.95	41.88
12	19831106	19614353	5949331800	5884305900	98.91	0.03	96.98	91.87	41.95
13	21828245	21521049	6548473500	6456314700	98.59	0.03	97.33	92.61	42.45
14	22555556	22140153	6766666800	6642045900	98.16	0.03	97.03	92	41.66
15	23481991	22987941	7044597300	6896382300	97.9	0.03	96.94	91.7	42.38
16	21101817	20618591	6330545100	6185577300	97.71	0.03	97.07	92.14	41.91
17	20483130	20160497	6144939000	6048149100	98.42	0.03	97.22	92.34	42.12
18	21714169	21311860	6514250700	6393558000	98.15	0.03	97.32	92.62	41.36
19	23035824	22710308	6910747200	6813092400	98.59	0.03	97.31	92.56	42.75
20	21097923	20501325	6329376900	6150397500	97.17	0.03	97.19	92.28	41.25
21	22941886	22582560	6882565800	6774768000	98.43	0.03	97.09	92.09	42.97
22	22043314	21640054	6612994200	6492016200	98.17	0.03	96.88	91.66	41.9
23	23621835	23184386	7086550500	6955315800	98.15	0.03	97.19	92.22	42.3
24	19159745	18843797	5747923500	5653139100	98.35	0.03	97.06	92.11	41.98

Effective Rate (%) is percentage of clean reads divided by raw reads, Error (%) is the base error rate, Q20 and Q30 relate to base

<u>Table D2:</u> Sequencing quality check for RNAseq samples in Section 2.4.5.

285

Sample no.	Treatment	Tissue	Timepoint (day)	Experimental rep	Reads	Reads pseudoaligned	% pseudoaligned
1	Fulvic	Shoot	0	1	21948373	15001567	68.35
2	Fulvic	Root	0	1	21683448	15322780	70.67
3	Control	Shoot	0	1	19731245	13433374	68.08
4	Control	Root	0	1	19192851	13111977	68.32
5	Fulvic	Shoot	3	1	21916558	13940709	63.61
6	Fulvic	Root	3	1	19970525	14133478	70.77
7	Control	Shoot	3	1	23125595	14603288	63.15
8	Control	Root	3	1	21818824	15337857	70.30
9	Fulvic	Shoot	0	2	20015180	13166950	65.78
10	Fulvic	Root	0	2	22756103	15965816	70.16
11	Control	Shoot	0	2	20017217	13281556	66.35
12	Control	Root	0	2	18832338	13193369	70.06
13	Fulvic	Shoot	3	2	20834945	13749556	65.99
14	Fulvic	Root	3	2	21347709	15176499	71.09
15	Control	Shoot	3	2	22410937	14604137	65.17
16	Control	Root	3	2	19940906	14437150	72.40
17	Fulvic	Shoot	0	3	19550309	12674614	64.83
18	Fulvic	Root	0	3	20587135	14609678	70.97
19	Control	Shoot	0	3	22060176	13759703	62.37
20	Control	Root	0	3	19868638	14429523	72.62
21	Fulvic	Shoot	3	3	21961363	14221382	64.76
22	Fulvic	Root	3	3	20853270	14884900	71.38
23	Control	Shoot	3	3	22566458	15036502	66.63
24	Control	Root	3	3	18121577	12993129	71.70
				Total	501111680	341069494	-
				Mean	20879653.3	14211229	68.15

Table D3: Pseudoalignment summary of RNA-seq samples from de novo transcriptome analysis in Section 2.4.5.



<u>Figure D2:</u> Differential transcript expression using voom/Limma method [184] between 0 day (green) and 3 day (red) for VFA treatment RNA samples (see 2.4.1) using Degust [183] as in Section 2.4.6.

Shoot and root tissue are analysed seperately and transcripts are only deemed differentially expressed if all experimental replicates have an absolute log fold change of 0.585 and false-dscovery q-value < 0.05.

Appendix E: 'UK dried forage production: a review of industry changes and assessment of prospects for both policy and science'

E.1 GENERAL INTRODUCTION

In the UK, dried forage has been produced since the 1930s, with most producers being members of the British Association of Green Crop Driers (BAGCD) [8]. The BAGCD is an association of numerous farms based in Cambridgeshire, Devon, Essex, Hertfordshire, Lincolnshire, Perthshire and Yorkshire. In total they farm approximately 7500 hectares of land across all these sites, with only one farm as permanent pasture, with total production worth ~ \pounds 8 – 9 million per annum [486]. The fundamentals of the grass-drying process arose at Cambridge between 1925 – 1927 [61-63] and as a general description these producers grow various grass and herbaceous legume forage species, both stand-alone or in mixed cropping; Table E1 below shows those crops cultivated by BAGCD at present, and those forage crops not currently grown, but which either were historically grown or may be in the future.

Forage crops are grown across the standard UK growing season and are cut to be high temperature dried in large rotary drum dryers at ~ 800 °C; unlike in hay or silage production swaths, which are left to wilt post-cutting for only a few hours before being high temperature dried to a moisture content of 12 %. The dried product, also termed `Dehy' due to its dehydration, is then milled and pelleted, and used in cattle, equine, poultry, and pet feeds.

The nutritional value of dried forage has similar nutritional content to silage forage, but as it is in a dry, pelleted form, these nutritional levels are retained for longer. Figure E1 shows the ranges for different nutritional parameters of BAGCD produced dried grass, along with two independent ranges for silage grass. One can see for each parameter the nutritional composition is similar for all, however dried forage can be kept for many years with low nutritional changes [64] compared to some silage forage, which over only 6 months can have quality changes in Crude Protein (CP), Neutral Detergent Fibre (NDF), and Acid Detergent Fibre (ADF) [828]. Table E1: Forage crops currently cultivated by BAGCD members across the UK, or those not cultivated but popular for other forage crop growers. These are grouped into either grass or legumes, with the species and common name included. cultivars being drilled shown for the most widely sown species.

Current crops cultivated by BAGCD							
Gras	sses	Legumes					
Species	Common name	Species	Common name				
Lolium perenne	Perennial	Medicago sativa	Lucerne/Alfalfa				
	Ryegrass						
Festuca	Tall Fescue	Lotus	Birdsfoot Trefoil				
arundinacea		corniculatus					
Festuca	Meadow Fescue	Trifolium	Red Clover				
pratensis		pratense					
<i>Festuca</i> x	Festuca x Festulolium		Spring Pea				
<i>Lolium</i> (various)							
Phleum	Timothy grass						
pratense							
Anthoxanthum	Sweet Vernal						
odoratum	grass						
Crops of interest to BAGCD							
Gras	sses	Legumes					
Species	Common name	Species	Common name				
Dactylis glomerata	Cocksfoot	Onobrychis viciifolia	Sainfoin				
Lolium x boucheanum	Hybrid ryegrass	Trifolium repens	White Clover				
Lolium multiflorum	Italian Ryegrass	Galega orientalis	Forage Galega				
Lolium	Westerwold						
westerwoldicum	Ryegrass						



<u>Figure E1:</u> Typical nutritional value of different processed forage grass, dried grass (blue) [64] and two examples of silage grass (grey) [829; 830].

Bars are the typical ranges for each of the following nutritional parameters in either MJ/kg Dry Matter (DM) or % DM; (A) Metabolisable Energy (ME); (B) Fermentable Metabolisable Energy (FME); (C) Crude Protein (CP); (D) Ash; (E) Neutral Detergent Fibre (NDF); (F) Acid Detergent Fibre (ADF); (G) Sugar.

E.2 HISTORICAL CULTIVATION IN THE UK

The use of wild forage pastures around 9000 – 7000 BC coincided with the domestication of animals across Europe, Asia, and North Africa [6]. Developments in haylage production, along with both forage crop domestication from ~ 1000 BC onwards and forage cultivation globally, enabled the rise of the agricultural intensification period of the 13th – 19th Centuries. This increase in agricultural intensification was particularly prominent throughout the UK [7; 6; 18].

In the 1920s the technology for drying forage crops was developed [61; 62], and in the 1930s dried forage became available for animal production [63]. As forage crop cultivation is intrinsic to intensive agriculture, it is in turn essential for supporting a growing population. The process of high temperature drying enables the feed from dried forage to last longer than normal fresh fodder, haylage or silage but retain its nutritional content (as in Figure E1 above). Therefore, high temperature drying contributes to larger scale farms with high feed demands from increased livestock numbers, increasing final output efficiencies for the population. This was particularly evident in a post-World War II push to produce efficient, low waste animal feed that could be stockpiled if necessary in future decades with proclamations "the drying of grass, lucerne and other crops will rapidly become a corner-stone of British Agriculture" [65]. From the 1950s to 1970s, across the UK forage growers were taking advantage of the new drying technology and buying rotary machines such as in Figure E2, making the industry big business. During this time a global association was set up for forage driers called the Commission Intersyndicale des Déshydrateurs Européens (CIDE) of which BAGCD is a member, and international conferences under the title of International Green Crop Drying Congress (IGCDC).

292



<u>Figure E2:</u> Assorted crop drying advertisements from 1970s [831; 832], showing the huge gain of interest in the agricultural techniques first developed from 1925 [62].

By 1975, the UK forage drying industry was widespread, as shown in Figure E3A, with the location of BAGCD members with active driers. Such a trend was occurring internationally with the 1973 First IGCDC having over 300 delegates in attendance from 23 countries, including many across Western Europe, but also North America and New Zealand [831]. This period was marked by increased animal product prices as social demand was high in the UK population, but agriculture would increase efficiencies required to keep up with demand until 1980s.

More efficient agricultural practices were developed throughout the century ranging from crop rotation and weed control, to confined animal feeding; these practices would eventually combine into very low-priced products. This type of farming is now termed intensive agriculture, due to the development of organic- and extensive- farming culture in the latter half of the century [833]. Consequently, as animal feed demand was high this meant "the feeding of dried green crops could play an important role [and] Nowhere is this more marked than within the 9 countries of the European Economic Community" and so demand for drying remained high [831; 832]. The only glimpse of negativity at the 1973 Congress was fuel supply was becoming a costly problem for agriculture, but as this was global and effecting all industries it was not hugely discussed.



<u>Figure E3:</u> UK and ROI forage growers and driers belonging to the British Association of Green Crop Driers (BAGCD) in (A) 1975 [832], (B) 1980s -2000s [4; 834; 398], and (C) 2018 [8; 398].

Maps were generated using Ordnance Survey OpenData OS Open Greenspace – GB (data type: vector, supply format: GML 3): Contains OS data © Crown copyright and database rights (2018). BAGCD member sites were estimated and marked onto maps with Microsoft PowerPoint.

Towards the end of 1970s, the principle of rising energy costs was starting to worry forage crop driers. Moreover, factors such as land use disagreements, newly refined production and storage of haylage, silage, and also the use of cereal supplements for livestock, were also cause for concern [835]. However, the forage drying industry continued with the assistance of the Common Agricultural Policy (or CAP) from 1962 which gave subsidies to the six founding European Economic Community (EEC) Member States, and in 1973 the UK joined, which was later to become the European Union (EU) in 1993. The CAP is financed through national payments by member states through both the European Agricultural Guarantee Fund (EAGF) and European Agricultural Fund for Rural Development (EAFRD), supporting market changes and rural development respectively [836]. The CAP meant during this time crop driers would get approximately €66 per hectare, and these subsidies would be based on the crops per hectare/animals per head that a farm produced, but without a need for the crop in question to be in demand, and not a commodity crop like wheat. This meant that regardless of worries of the forage crop driers,

European CIDE members (including BAGCD) were guaranteed a minimum price for their work every year irrespective of the market and costs. By 1989, CIDE involvement was down to 14 countries, but there was strong feeling that the industry would continue to grow, especially the lucerne market, as "there was huge investment in some 300 drying plants, with the aim of turning 'our' wonder protein plant, lucerne, into a marketable product" [4]; Figure E3B shows a decline in driers in the UK with fewer BAGCD members, despite it still being a relatively lively industry internationally.

From the 1990s onwards, reorganisations of the CAP would severely affect UK Crop Driers. The 1992 "MacSharry Reforms" and the millennium report "Agenda 2000" were the beginning of lowered subsidies to farmers, with increased or new demands for food safety, animal welfare, and the protection of environment including biodiversity and limiting leaching of fertilisers [837]. However, it was not until 2003 when the European Commission (EC) began to reform the entire system by removing coupled subsidies altogether and instead having a single farm payment, whereby industries with high input fuel demands, like crop drying, would be most affected [838]. The EC began to look at the average three-year activity of a farm, which would be the basis for its farm single payments. Farms had to keep up with regulations to receive their payment, which meant growing competitive crops. As forage crops are not commodity crops and therefore not as competitive, their farm single payments dropped significantly when compared to their original CAP subsidies. It was necessary to ensure that UK growers were setting a strong case for the new regulations. From 1992 onwards BAGCD amended their memorandum principles to "encourage the improvement of the fertility of the soil, the betterment of grassland, and the home production of grass and dried green fodder...for the benefit of the community at large" [834]. These amendments also aligned with the sustainable farming practices that the EC was also trying to implement.

By 2005 more changes had occurred due to the Secretary of State's Guidance for Vegetable Matter Drying Processes 2005 Crown Report [839], which further affected UK growers. The initial plan from the Crown Report was to abolish all the specific subsidies that UK growers received for both

295

growing and producing dried fodder. However, as BAGCD had already stated clear principles for sustainable farming practices, a compromise between the industry and government was sought. It was agreed the government would divide the aid between the farms and the crop driers, with halve going to the farm payment and the other half to the crop driers. This meant BAGCD members had to have EU spot checks on quality of fodder, which if not of a high enough standard, would disallow or reduce their subsidies. Occasionally, there were problems where if the regulations were implemented literally, then they would negatively affect the feasibilities of UK forage farming. For example, under EU law growing grass for five years becomes permanent pasture, and the BAGCD was worried about this definition so negotiations were required.

The key issues surrounding these laws are the UK-specific changes to Nitrogen (N) fertiliser usage, and apart from these changes, there have been limited deviations from common agricultural law in the UK dried forage industry. Similar crops have been grown, as in Table E1, and the largest recent change has been in the 2013 CAP reform [840]. With this reform there was more emphasis on environmental issues such as provisions to stop mono-cropping. Rules that enforce requirements for producers with a big enough farmed area to grow at least 3 crops, together with the provision a farmer's largest crop cannot occupy more than 75 % of their total land were set in place. As before, this kind of legislation is sometimes not harmonious with a streamlined production process, however as shown in Table E1, BAGCD members grow a variety of species. There was a trend towards more growth of lucerne as the equine feeds industry developed. The remaining farms use crop rotations of forage crops along with break crops of wheat, maize or oil seed. Cuts are taken throughout the spring/summer growing season every 6 - 8 weeks (usually 4 – 7 cuts depending on weather conditions and growth).

Nonetheless, as can be seen in Figure E3C, the number of UK forage drying sites has been decreasing. The main reason for a drop off in production is due to European competition, however increases in fuel prices, restrictions to N fertiliser applications, and implementation of biodiversity, greening, and health and safety policies are contributing factors [398]. UK fodder is

296

expensive to produce, with the French and Dutch controlling much of the market. Moreover, a drop off in CIDE numbers, due to European growers not seeing benefits from membership, has left a lot of the once thriving international industry disconnected. Forage cropping, of which the drying industry is an important asset, still accounts for a substantial proportion of EU agricultural output, as shown in Figure E4. Again, this is due to its importance in animal production, which is set to continue growing globally [841]. The future of UK forage drying is unsure, and its prospects especially in terms of EU politics will be discussed later.



<u>Figure E4:</u> Total European Union output for all crops in 2016 from Eurostat data [841].

This includes a substantial portion for forage crops (bright blue), comparable to the production of fruit and wine crops, and bioenergy crops if sub-split from 'INDUSTRIAL' crops (not shown). Here 'Forage Plants' are defined as bulk fodder crops such as vegetative legumes and grasses, fodder maize, and fodder roots including forage beet (*Beta vulgaris* spp.).

In addition to both UK and EU policy, there has also been huge land use change globally involving the destruction of forest land to produce agricultural land [842-844], with a 3 % expansion of world pastures and croplands from 1985 to 2005 [845]. This land use change increases soil erosion and loss of nutrients decreases biodiversity indexes and is particularly evident in the tropics where deforestation has ravaged rainforests [846; 847]. However as can be seen in Figure E5, FAO data of the last few decades show an increase in forest areas in Western European countries [843]. This includes the UK, and its dried forage competitors of France and the Netherlands. A new analysis of satellite data also suggests that tree canopy cover has increased across the globe between 1982 - 2016, although land degradation to bare ground is also apparent [848].

However, the suggested increase in forest cover, at least for Western European countries, is because the conversion to agricultural land happened pre-1982, by preindustrial communities [849]. Thus, small increases in recent years look impressive but not when compared to the original landscapes of each country. Therefore, degradation of land has been long underway in the UK, with serious negative consequences and some subsequent effects for farming and ecology are discussed below. This means it is imperative the UK government take seriously Afforestation policy and does not only apply regulations to tropical countries where deforestation is a recent practice, but also temperate climates. This includes reacting to guidelines set by the EC for the green direct payment scheme in relation to Ecological Focus Areas (EFA) on arable land [850], and also continuing to improve the current governments Greening Policy [851].



Extent of forest in the Western Europe, 1990 - 2015

<u>Figure E5:</u> Extent of forest of the UK and surrounding Western European Countries from 1990 - 2015, data from from the "The Global Forest Resources Assessment 2015: Desk reference" [843].

Note beforehand for ease of design both France and Germany are read using the right axis (500 ha) and the rest are read using the left axis (1000 ha) In the original report this data is in "Table E2. Extent of forest 1990-2015", found on pp. 9-14, and is under copyrighted © FAO, 2015

E.3 POLICY IMPLICATIONS OF NITROGEN FERTILISER USE IN DRIED FORAGE PRODUCTION

As touched upon in the above section, a large amount of legislation changes for the forage crop industry have been in restrictions on N fertiliser usage. Nitrogen has long been known to be fundamentally important for plant growth. Deficiency seriously limits crop production through restricting protein assembly, affecting both primary and secondary metabolic pathways. This is particularly true for the high N demand of the leaf photosynthetic apparatus where deficiency impacts on all growth parameters [75]. Leafy crops like vegetables, grasses and leguminous herbs have a slightly larger N content than cereal grain crops and therefore require more N for optimal production [74]. These crops include those grasses and legumes grown for high temperature drying such as those produced by the BAGCD for high N protein concentration forage in Table E1.

The development of Haber-Bosch process for fixing free nitrogen (N_2) from the air and converting it into ammonium (NH_4^+) is arguably the most dramatic event in modern human history. This fixation process was part of the Green Revolution that allowed crop production outputs to increase four-fold [421]. Fritz Haber and Carl Bosch's process was developed from Haber's 1908 "synthesis of ammonia from its elements" patent [423]. The work won each of them a Nobel Prize. More notable than its scientific advance of chemistry, ammonium fertilizer production changed the impact of the late-industrial age globally by feeding 50 % of the world population through both increased grain yields, but by also improved forage crop yields which in turn supported a growing animal production industry [424]. This population boom is shown in Figure E6, adapted from UN statistics from 1700 to 2015 [852; 853]. Nowadays, anthropogenic activities fix 210 Tg of N annually [427], a figure set to continue for the foreseeable future [440], with total N input for food production globally at 171 Tg N year⁻¹, half of which is accounted for by the animal products industry [428].

Global population in billions



<u>Figure E6:</u> Global population since 1700 to the present day; adapted from "World population to 2300" and "World Populations Prospects: The 2017 Revision" [852; 853].

'HB' is the development of the Haber-Bosch process in 1908.
There are some momentous problems with N fertiliser use. The process is resource and financially exhaustive; the practice requires huge quantities of energy that for the UK is mostly generated through reserves of natural gas [854]. Global N demand is also being stretched by its potential use in transport fuel and space heating with further technological advances on the process multiplying [436]. Emissions of N from fertiliser is estimated to be at around 10 % [437] with the bulk of these caused by agricultural emissions from animals and manure [452; 438]. Previously, emissions have been shown to be highest in Asia, predominantly China and India, and Europe [439]. However, other areas of the world are now catching up due to further increases in synthetic fertilizer use intensity [393]. As of 2015 Western Europe used 7.3 % of the world's N consumption [440], disproportionate with its landmass size; the growth in N fertiliser use has decreased by almost 1 % for the area, however consumption still stands at a forecasted 7.13 % [449].

Due to the importance of ensuring N fertiliser is applied responsibly, a lot of policy has been introduced both through the EC and UK government. Arguably, the most significant of these for the dried forage industry has been implementation of the Fertiliser Manual RB209 8th addition [79]. The publication set-out stringent acceptable levels of N fertiliser use across growing seasons built on several policies; the 1991 Nitrate Directive to protect N fertiliser from agriculture polluting ground and surface waters [76], the 1998 Action Programme for Nitrate Vulnerable Zones covering England and Wales [77], and the subsequent 2008 Nitrate Pollution Prevention Regulations for England [78]. Amended annually, this guide provides guidelines for fertiliser management of each specific crop type, including grasses and forage crops. As a document it was originally very strict about the amount of knowledge farmers were expected to have about N use, and desired in-depth testing of N soil content through season, and proof of application needs. The RB209 did promote many sustainability practices by offering growers the opportunity to gain Environmental Stewardship (ES) points that could be converted into funding if they conform to the rules [79].

In both the RB209, and the 2008 Implementation of Nitrates Directive in England report that initiated many of its guidelines, the recommended maximum N application rate for dried forage growers was defined as 370 kg ha⁻¹ over a growing season [412; 79]. This figure uses both the initial application rates for grassland, particularly for dairy-grazed grasslands, and the extra allowance of 40 kg ha⁻¹ permitted after cuts [79]. Based on DEFRA statistics this would be enough to achieve a yield of 11.3 tonnes DM ha⁻¹, however this estimate did not consider the level of cutting the dried forage undergoes. The intensive cutting anywhere from 4 – 7 times between spring and autumn means that the forage crop, whether grass or legume, is subjected to high N demands. Especially affected is the chlorophyll content, which is an important characteristic of dried forage appearance as well as a by-product extracted for other industries. This means if BAGCD members stuck to the guidelines they would not reach their usual yields and production of the dried fodder would be lower than the CAP presumed, thus not increasing their single farm payment until years later.

Therefore, the BAGCD commissioned its own report to appeal to DEFRA to reconsider its guidelines stating that the N max "limit was developed for a different grass production system in mind (i.e. grass production on dairy farms)" and therefore is not in-line with dried forage needs [486]. The request was accepted and included in subsequent legislation from 2013 onwards [414], as detailed below in Figure E7:

"Grass grown for dehydration or chlorophyll production: You can use nitrogen up to the level recommended in writing by a FACTS-qualified adviser if you're growing grass to achieve a protein content of at least 16% in the dried product. If the land is irrigated, you must not use more than 700 kg of nitrogen per hectare. If the land isn't irrigated, you must not use more than 500 kg of nitrogen per hectare."

Crown© (2013). "Guidance on complying with the rules for Nitrate Vulnerable Zones in England for 2013 to 2016 ", (ed.) Defra. (London, UK).

<u>Figure E7:</u> The vital exemption for BAGCD members in the RB209, outlined by the Crown© [414].

This exemption means yields by dried forage growers can be maintained despite huge nitrogen requirements for the crop during the growing and cutting season. Such high guideline usage does mean UK dried forage growers must show evidence routinely to DEFRA and, as mentioned, be signed off by a FACTqualified adviser (a body offering training for agricultural guidelines run by BASIS (registered charity No. 1077006)). The only other strict requirement of the RB209 currently is that all farmers must stick to the Code of Good Agricultural Practice (the CoGAP), which sets out how to limit nitrate leaching and must be followed to gain the basic payment scheme of the CAP [413]. In addition to abiding these guidelines, BAGCD also set up their own trials to optimise and hopefully limit N fertiliser use, as this is also of excessive cost to the grower. However due to their high N max limit, the dried forage producers tend to be left out of annual reporting as an exempt group [442]. The average application of fertilisers for English farms is currently 113 kg N ha⁻¹ (along with 19 kg phosphate ha⁻¹ and 26 kg potash ha⁻¹) [415], whereas BAGCD members can apply up to 700 kg N ha⁻¹[83].

E.4 IMPORTANT RESEARCH DEVELOPMENTS FOR THE INDUSTRY

The above policy changes have been largely affected by scientific developments in many fields related to agriculture, from ecology and conservation biology to plant cell biology and genetics. Three of the principal areas where research has affected the UK dried forage production industry are discussed below; Nitrogen Fertiliser, Leaching and Emissions; 'Soil Health' and Forage Crop Cultivation; and Biodiversity and Greening Policies.

E.4.1 Nitrogen fertiliser, leaching and emissions

As extensively discussed in the previous section the use of N fertilisers is of imperative concern to the UK government. Fertiliser resource consumption and subsequent emissions from their uses have dramatic consequences for the world. The vast energy required puts a strain on our current energy demands and using a huge supply of natural gas directly contributes to our unsustainable dependence on fossil fuels. This directly influences governmental policies and is constantly monitored in developed countries [440; 449; 450] and in the UK specifically [442]. There is increasing emphasis to ensure the N component of manure is adequately exploited, especially for degraded soils [450]. Increased global demand for protein estimates of 110 % \pm 7 % [463] will undoubtedly exacerbate the current problems with N fertilisers as an increase in food production is required. Due to this, there are community, political and financial aspects to using less synthetic fertilizer overall and manage those used more efficiently. Interest in N fertiliser use include the topics of both the leaching and gaseous emissions from N fertiliser use. Below N leaching and emissions are discussed separately.

Leaching of N leads to eutrophication of water supplies where algal blooms in rivers and lakes limits sunlight, space and oxygen for aquatic species and therefore leads to high death rates in these ecosystems. Such leaching can also contaminate human drinking water especially in ground water supplies. Forage grass and legume crops have long been known to have high rates of leaching to the environment [413]. This problem can be exacerbated when they are grown in sandy soils, the choice for many forage crops. Some estimates calculated that 60 % of applied N is lost through leaching, run-off, denitrification and consumption by microbial populations [476; 477]. Furthermore, areas termed Nitrate Vulnerable Zones (NVZs) make up approximately 58 % of the land in England, with an increase in total area of 1300 km² since 2013 [81; 82]. NVZs are determined by land gradients, ground cover (especially fallow or uncultivated land), water sources, soil types and erosion intensity, and weather conditions [83]. Figure E8 shows the most recent mapping of NVZ zones across the UK [82], with many of these in BAGCD farming areas. Not only are there arguments for limiting N leaching due to effects on soil degradation, soil pH and biodiversity (terrestrial and aquatic shown in Figure E8), but also for the impact of drinking-water containing leached N on human health, resulting in decreased life expectancy [446-448]. This has driven assessment studies into the social cost of nitrogen (SCN) [445].

Proposed Nitrate Vulnerable Zones in 2017 Designation methods





<u>Figure E8:</u> Nitrate vulnerable zones (NVZs) map of the UK for 2017, directly reproduced from the "Review of Nitrate Vulnerable Zone designations for implementation in 2017: Environment Agency report and recommendations to DEFRA: Report Number HOEV151604/R" [82].

In the original report this map is "Figure 4.2 - Proposed 2017 NVZ and the methods that designate each area", found on pp. 34, and is under copyright: ©Crown copyright and database rights 2016 Ordnance Survey 100024198 and ©Environment Agency copyright and/or database rights 2016. Research into fertiliser effects on grass has long been carried out including long-term experiments such as Rothamsted's Park Grass Continuous Hay Experiment, which began in 1856 [479; 104; 480]. This included a 19year assessment of leaching using ¹⁵N labelled fertiliser, finding leaching of $^{15}\mathrm{NH_4}$ and $^{15}\mathrm{NO_3}$ to be approximately 13.9 % and 21.9 % of total N application, respectively [481]. Many forage grasses, such as ryegrass, will preferentially uptake nitrate (NO_3) as N source particularly at high fertiliser rates [469; 470]. However, as a grower may choose to cultivate mixed crops of grasses and/or legumes this preference may vary. For a long time, studies have also looked at the direct influence of N sources on final product of dried forage yield, from its early industry [63] to those carried out now on many of the BAGCD current sites. Studies have shown that the leaching of inorganic N is lower for grassland than that of arable land [483; 484]. This is largely due to the N applied being routinely removed as the plant uptakes the required amount, and then furthered removed by the plants after cutting, as discussed previously [485; 486]. However, as dried forage cropping is often seeded each year in the UK, and is not a permanent pasture, it may have a leaching potential more like arable land in discrete periods.

There are strong suggestions that N emissions may contribute to climate change, with nitrogen oxides (N₂O and NO) being the single largest contributor to atmospheric greenhouse gases [459]. Such emissions can be produced as a by-product of N fertiliser use through microbial breakdown, both on farmland (direct) or off farmland due to leaching (indirect). Figure E9 shows the trend of UK nitrogen oxides emissions along with ammonia (NH₃) and non-methane volatile organic compounds [855]. Although such emissions have been decreasing, the majority of emissions are of agricultural origin through N fertilisers, at around 75 % [456]. In many other countries emission quantities are not decreasing.

308





<u>Figure E9:</u> Trend in UK air pollutant emissions in million tonnes including Nitrogen Oxides (NO and NO₂), Ammonia (NH₃) and Non-Methane Volatile Organic Compounds (NMVOCs).

Data is from DEFRA, and is for the years between 1980 and 2016 [855].

Emissions are not just related to the quantity of fertiliser applied, but also the type of fertiliser and the crop [113; 114]. At present there is limited data for grassland emissions grown specifically for cutting and drying [115; 116], but it can be presumed to be similar to the typical agricultural emission amounts in order to adequately reduce UK greenhouse gas contributions. Care should be taken to ensure that trying to reduce emissions does not reduce the efficiency of production. For example, through simulations it was found that reducing N₂O emissions by ~ 4 % would require a N fertiliser application reduction of ~ 10 % which would decrease yield $\sim 2 \%$ [856]. This means that although the policy to decrease emissions may be successful, such as those outlined in various reports [857-859; 557], it may be detrimental to the intensive agriculture system, thus less fertiliser may be applied to the land, but more land is utilised to grow a similar amount of crop. Practices such as low or zero tillage may be able to aid with this emissions problem [582; 88; 90; 92], but more crop and cultivation specific evidence is required. Many countries are signatories of the Kyoto agreement [860], which details the need to decrease emissions. However, care must be taken on a country to county basis, so that adhering to N fertiliser limits does not decrease the efficiency of the system. It is therefore necessary to have management-based studies into N application rates with parallel, accurate quantification of emissions for each arable land type, including forage crops grown for drying [861; 113].

E.4.2 'Soil health' and forage crop cultivation

Across Europe between 1700 and 1950 around 70 % of the original forests and wild grasslands were converted to cropland, including forage cropland [101]. With these changes came new cultivation problems as soils become more vulnerable to erosion and nutrient losses. The status of UK agricultural soils is quickly moving up the list as successive governments are prioritising 'ecosystem services' [413]. Ecosystem services have a long history in both ecological economics and agricultural sciences [862] with the term first coined in 1981 after many decades of development [863; 864]. Figure E9 is taken directly from the Millennium Ecosystem Assessment, and explains interactions between all types of ecosystem services, as well as how they impact on the well-being of a community [101]. By looking at Figure E10 one can see that the soil is a crucial service provider which can be separated into four major categories; supporting, provisioning, regulating, and culturally. Supporting through its structure and formation, provisioning through plant growth, regulating especially in flood and erosion defence, and culturally as aesthetic landscape topology is important for recreational and spiritual practices. Another term used increasingly in UK policy and economics is 'natural capital' [93; 94] and this summates the ecosystem services such as soil, with addition of the available stocks or biodiversity of an area. Consequently, the UK government has appointed the independent advisory Natural Capital Committee (NCC) [865], which works in collaboration with the international Natural Capital Coalition organisation, which also includes the United Nations and EC, as well as discrete representatives of America, Switzerland, the Netherlands, China [866].



<u>Figure E10:</u> 'Ecosystem services' and their link to 'Constituents of Wellbeing', directly reproduced from the "Ecosystems and Human Well-being: Synthesis" [101].

In the original report this diagram is "Figure A. Linkages between Ecosystem Services and Human Well-being", found on pp. vi, and is Copyright © 2005 World Resources Institute. All rights reserved under International and Pan-American Copyright Conventions. No part of this book may be reproduced in any form or by any means without permission in writing from the copyright holder: World Resources Institute, 10 G Street NE, Suite 800, Washington, DC 20002.

'Soil health' is a limited term to denote a soil system providing optimal ecosystem services through its biological, chemical, and physical characteristics. It has been extensively reviewed in terms of universal soil quality factors [867], those specifically assigned to agricultural systems [868] and for parameters of sustainability testing [869]. The main areas of soil health that are threatened in the UK are the decline of soil carbon in arable and peat soils, soil degradation (especially for peatland with subsequent carbon emissions), contamination of land with heavy metals, and a lack of soil monitoring schemes [397]. Specific reviews for both Wales and Scotland [870; 395] are available, as well as EU directives [871]. It is felt by the current UK government that the CAP encouraged farming that led to these problems through intensive agriculture, which has depleted nutrient and humus levels, and eroded or compacted soils [93]. As with all UK agriculture, dried forage growers are facing the same soil health threats. Two of the main UK soil health research questions at present are how to decrease soil erosion levels in arable and grassland, and how to increase the soil organic matter (SOM) of these depleted areas [405].

Through meta-analysis it has been shown that the erosion rate of agricultural land and orchards are similar to that of bare land, approximately between $1000 - 10000 \text{ mg/km}^2$ for their highest probability density whereas areas such as forests and shrubland are lower [551]. This is partly due to a lack of established plants; arable agricultural systems lack the development of complex root architecture under most intensive management practices. Permanent grassland is better able to prevent erosion, as well as sequestering 34 % of global carbon stocks due to their stability [86]. An in-depth review has demonstrated the importance of vegetative cover for controlling, limiting or in some cases, reducing soil erosion [872]. The cost in £ ha⁻¹ of productive land loss through rotation changes or becoming a non-productive area due to soil erosion has been calculated for many UK crops, showing how broad the problem is across farming [552].

As with N fertiliser use, forage crops used for drying are highly cultivated and should therefore be managed like other intensively farmed land until more evidence is found to suggest otherwise. However, promising work on the use of grass and/or legume mixes to reduce soil erosion is being established, which is particularly relevant for forage growers who already trial such mixes. As soil erosion is linked to a more complex root architecture for many plant species, work has shown that crops such as ryegrass (*Lolium perenne*), are particularly good at reducing erosion due to their fibrous rooting pattern [110]. BAGCD crops (Table E1 and E2), of Lucerne (*Medicago sativa*), and red clover (*Trifolium pratense*), are also associated with increased soil aggregate stability when continuously cultivated and could be promising for combating erosion [111].

Depletion of SOM is not only exacerbated by soil erosion, but has a direct impact on future crop yields [873]. This could mean that the breeding of crops with improved root architecture could not only improve their erosion limiting potential, but also increase SOM [874]. Long-term testing of soil health for SOM is also complicated by the impact of climate change [875]. Studies have shown that use of lucerne-grass mix can increase SOM, especially when coupled with lower tillage [876]. As seen with grass and clover leys, which have long been known to improve erosion problems and increase SOM [112]. For forage crops to become useful for limiting these soil health problems, it is important they are cultivated with a long establishment period, using minimal tillage [553]. For these reasons the use of appropriate minimum tillage and the reintroduction of grass buffer zones are part of many governments' policy, including the UK [93]. Buffer zones are also being introduced to aid biodiversity levels, which will be discussed in the next section.

E.4.3 Biodiversity and greening policies

As with the sections above, N fertiliser leaching and emissions in conjunction with soil health problems have a negative impact on biodiversity, which is also declining due to conversion of the land for different uses. These factors have been observed to effect biodiversity globally [844]. Although the conversion of natural UK woodland to agricultural cropland was prehistoric and preindustrial as discussed above, the negative consequences of land use change are current. This is especially true when coupled with more recent changes to UK agriculture including the abandonment of mixed farming systems in favour of intensive cropping, grazing regimes, and pesticide/fertiliser usage. The causes of UK biodiversity decline in cropland include many drivers explained in Burkmar and Bell's 'Drivers of Biodiversity Loss' report for the Field Studies Council. These can be listed as habitat loss, destruction and fragmentation; pollution from farming practices including N fertiliser usage; biotic exchange of invasive species through agricultural trade; and climate change influence especially regarding GHG emissions [877]. In the collaborative State of Nature 2016 publication it was reported biodiversity is declining in the UK from 1970 to the present day in all land use areas, apart from debatably in urban and total marine (excluding fish stocks) populations [878]. It is likely the decline in biodiversity in non-agricultural areas is still broadly affected by the same drivers as cropland, including agriculture's role in climate change.

Semi-natural, managed, temperate grasslands have been designated as very significant examples of rapid decline of biodiversity due to pollution from N fertilisers (and phosphorus; not discussed in this report), and significant habitat changes [101]. It has been shown that the application of N fertiliser on UK managed grassland can decrease diversity in both short- and long-term experiments for animal, plant and microbial species [102-109]. There are some disagreements whether the form of N fertiliser being used varies this decrease [879; 880]. Declines or changes in microbial species diversity have also been linked to declines in human health [881]. Biodiversity is not just important for mitigating declines and changes in individual populations, it also provides crucial information to quantify other ecosystem services in an environment [882; 590]. Thus, one can determine problems in other areas, such as soil pH or long-term decreased flood defence, are dependent on species population variations. Such insight has led the NCC to increase its push to quantify natural capital as a complex entity, with biodiversity interlinked to ecosystem services [883; 884].

315

However, it is important to recognise that pushes to increase biodiversity numbers may be inherently flawed for agricultural systems. By increasing biodiversity through many initiatives, you are likely to decrease system efficiencies, which could mean that more land would be required for agriculture to produce the same crop yields giving a cyclical feedback that reduces biodiversity. Research is needed to find a compromise between system efficiency and biodiversity directives; the balance between these two systems is a concept termed 'ecoagriculture' [587]. Permanent grasslands are capable of increasing biodiversity-richness and genetic variability, including traditionally managed European mixed grasslands [86], which are similar in composition to the forage cropping of BAGCD although less intensive in terms of tillage and rotations. For example, a huge driver of biodiversity decline is the wide use of herbicides for unwanted weeds, so management systems that lower weed populations but do not affect non-targeted, wider biodiversity is required. Approaches that naturally suppress weed population such as low tillage [885], or studies of productive grass and/or legume seed mixes with competitive root architecture [886; 887], are promising for forage growers. Moreover, grassland has high rates of biodiversity stabilisation after ceasing longterm N fertiliser use [888], and so current forage cropland may be fast recovering.

UK agriculture must comply to many EU rules including the Birds and Habitats Directives and Natura 2000 Network, detailed as part of the EU LIFE programme [589]. The EU has also been imperative in allocating funds to 800 LIFE-Nature Projects, with a total of \in 1.3 billion between 1992 – 2005. This funding aim to maintain and restore natural habitats whilst working with both conservation and farming groups. The UK government also has a long-term commitment of funding biodiversity projects, as shown in Figure E11. However, Figure E11 also reveals since 2008 the % of GDP spent on biodiversity projects has decreased, reflecting a change in government priorities due to the transition from a Labour to Conservative cabinet.

Expenditure on biodiversity in the UK 2000 - 2017



<u>Figure E11:</u> 'Expenditure of UK biodiversity', data from the "UK Biodiversity Indicators 2018" [889].

In the original report this diagram is "Figure E2i. Expenditure on biodiversity in the UK, 2000/01 to 2016/17.", however here NGO data have not been included, and is Copyright © Crown 2018.

E.5 THE FUTURE, INCLUDING POTENTIAL POST-EU EXIT CHANGES

This report has discussed the past production of UK forage cropping for dried feed, and the present problems that the agricultural industry face at large, especially in terms of policy restrictions. The future of UK agriculture is currently uncertain, especially with its exit from the EU (Brexit). Such a change will affect the policies made by future governments, which are unknown.

The EU single market means that not only is agriculture heavily subsidised by the CAP, but it is also protected from outside markets [586] through the Common Market Organisation (CMO). The CMO does this through intervention at problematic agricultural times, marketing of products especially in terms of safety, and managing both internal competition rules but also international licenses, tariffs and processing of products [890; 891]. This means there are already strong focuses on protecting farming livelihoods especially through the EC's Agricultural Markets Task Force. This organisation has been imperative to the recent proposed legislation on protecting the business rights of farmers, but also with specific objectives to "maintain market stability, enhance agricultural producers' income, [and] improve agricultural competitiveness" [892]. The chance of improving UK agricultural profits may be lost through the Brexit movement. Critical issues such as the protection of farmer business and trading rights maybe overlooked compared to all the other political objectives the UK government deems important.

The ecosystem benefits of restoring grassland globally are well documented [101]. Even when a mixed forage cropland is cultivated rather than permanent grassland, there are benefits and it is still preferential to the cultivated cereal monocrops. Mixed forage crops can bring improvements in soil health and biodiversity as discussed above whilst still retaining their commercial value. The main problem with moving away from monocropping is the reliance of the livestock industry on the use of concentrate feeds, such as cereal grain and biproducts, rather than forage crop vegetative biomass. Concentrates deliver high nutritional value

318

especially in crucial developmental animal stages (immature livestock, pregnancy, laying, milking, etc.), but come with a higher land use affect because they compete for land area that is also used for to produce cereal grain for human consumption. Forage crops however, can be grown on lower grade land not suitable for cereal production. The current global feedstock rations provided to livestock are shown in Figure E12 below, with fodder crops having a small portion of inputs [893]. Therefore, if more intensive cereal cropland was transformed to mixed forage production, to provide a similar level of animal products more land would be required. A solution may come from a switch by many western populations to more plant-based diets. Such a lifestyle change could conceive a lower reliance on cereal concentrates, enabling livestock to still be integral to diet and culture, but produce increased positive impacts for ecosystem services. However, the problem with such an approach is the lack of knowledge of how much cultivated grassland equates to ecosystem service provision of permanent grassland.



Figure E12: Global livestock feed rations, adapted from "GLEAM 2.0 - Assessment of greenhouse gas emissions and mitigation potential" [893].

In the original report this diagram is "Global livestock feed intake" and is Copyright © FAO 2018.

Of prominent importance for the future is not only the N emissions discussed above, but also total GHG emissions from agriculture and forestry of which net carbon losses globally where they contribute to 24 % of total output [894]. These GHG emissions include CO_2 . This is important as although CO_2 has a lower 'warming' potential than other GHGs, such as nitrogen oxides (NO, N₂O), it is more publicly recognised due to a longer history of journalistic coverage. This means general industry policies on total GHGs, and not just N emissions, are huge political drivers, and thus may continue being the focus for successive UK governments rather than more detailed policies for specific agricultural land-uses including forage cropland. Governments need to understand that a 'one size fits all' policy cannot be applied to agricultural policy.

Although the challenges of curbing N fertilisers use, increasing soil health, and decreasing the negative effects of soil erosion and biodiversity should be high on the UK government's agenda, it is important that numerous policies are not at the detriment to farmer's livelihoods. Although policies such as the Greening initiative should be continuously improved, it is imperative that government also continue to fund agricultural science to explore these complex interactions. Agriculture is heavily restricted and regulated, which should continue to improve the wider community, but care should be taken to ensure farming is still appreciated and not low on priority lists.

REFERENCES

- FAO, and IFIF (2010) Good practices for the feed industry Implementing the codex alimentarius code of practice on good animal feeding [Online]. Rome, Italy. Available: <u>http://www.fao.org/docrep/012/i1379e/i1379e.pdf</u> [Accessed 17/09/2019].
- Kopecký, D., Loureiro, J., Zwierzykowski, Z., Ghesquière, M., and Doležel, J. (2006) Genome constitution and evolution in Lolium × Festuca hybrid cultivars (Festulolium). Theoretical and Applied Genetics 113, 731-42. doi: 10.1007/s00122-006-0341-z
- 3. **Ghesquière, M., Humphreys, M.W., and Zwierzykowski, Z.** (2010) "Festulolium," in Fodder Crops and Amenity Grasses. Handbook of Plant Breeding, eds. **B. Boller, U.K. Posselt & F. Veronesi**. (New York, NY: Springer New York), 288-311.
- 4. **IGCDC, and BAGCD** (1989) "Proceedings of the fourth international green crop drying congress", in: *Proceedings of Dri-Crops 89: Fourth International Green Crop Drying Congress.* (Churchill College, University of Cambridge, UK: Agra Europe Ltd., London.).
- 5. **Eurostat** (2019) "Main crops by NUTS 2 regions (ef_lus_main)", (ed.) E. Commision.).
- Hirata, M. (2011) "Forage crop production," in *The role of food,* agriculture, forestry and fisheries in human nutrition, ed. V.R. Squires. (Oxford: Encyclopedia of Life Support Systems), 181–202.
- 7. **Williams, T.E.** (1980) *Herbage production: grasses and leguminous forage crops.* Oxford, UK: Blackwell Scientific Publications.
- BAGCD (2018) Welcome to BAGCD [Online]. UK. Available: <u>http://www.bagcd.org/</u> [Accessed 20/06/2018].
- 9. **Krausmann, F., Erb, K.-H., Gingrich, S., Lauk, C., and Haberl, H.** (2008) Global patterns of socioeconomic biomass flows in the year 2000: A comprehensive assessment of supply, consumption and constraints. *Ecological Economics* **65,** 471-87. doi: 10.1016/j.ecolecon.2007.07.012
- Makkar, H.P.S. (2018) Review: Feed demand landscape and implications of food-not feed strategy for food security and climate change. *animal* 12, 1744-54. doi: 10.1017/S175173111700324X
- 11. **Katepa-Mupondwa, F.M., Christie, B.R., and Michaels, T.E.** (2002) An improved breeding strategy for autotetraploid alfalfa (*Medicago sativa* L.). *Euphytica* **123,** 139-46. doi: 10.1023/A:1014488307000
- 12. Wilkins, P.W., and Humphreys, M.O. (2003) Progress in breeding perennial forage grasses for temperate agriculture. *The Journal of Agricultural Science* **140**, 129-50. doi: 10.1017/S0021859603003058
- 13. **Humphreys, M.O.** (2005) Genetic improvement of forage crops past, present and future. *The Journal of Agricultural Science* **143**, 441-8. doi: 10.1017/S0021859605005599
- Boller, B., and Green, S.L. (2010) "Genetic resources," in *Fodder Crops* and Amenity Grasses. Handbook of Plant Breeding, ed. B. Boller, Posselt, U., Veronesi, F. (New York, NY: Springer), 13-38.
- 15. **N.I.M.S.S.** (2017) *NE1010:* Breeding and genetics of forage crops to improve productivity, quality, and industrial uses [Online]. USA: NERA. Available: <u>https://www.nimss.org/projects/1314</u> [Accessed 30/11/2017].
- 16. DEFRA (2019) Farming statistics: Final crop areas, yields, livestock populations and agricultural workforce at June 2018 - United Kingdom [Online]. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/805476/structure-jun2018final-uk-31may19.pdf</u> [Accessed 08/09/2019].

- Kilpatrick, J., Heywood, C., Smith, C., Wilson, L., Procter, C., and Spink, J. (2008) Addressing the land use issues for non-food crops, in response to increasing fuel and energy generation opportunities [Online]. Hereford, UK. Available: <u>https://www.forestresearch.gov.uk/documents/5075/NNFCC_ADAS_Addr</u> <u>essing the land use issues for non-</u> <u>food crops in response to in Sm15ni2.pdf</u> [Accessed 17/09/2019].
- Capstaff, N.M., and Miller, A.J. (2018) Improving the yield and nutritional quality of forage crops. *Frontiers in Plant Science* 9. doi: 10.3389/fpls.2018.00535
- 19. **Osbourn, D.F.** (1980) "The feeding value of grass and grass products," in *Grass: its production and utilization,* ed. **W. Holmes**. (Oxford, UK: Blackwell Scientific Publications), 70-124.
- Weimer, P.J. (1996) Why don't ruminal bacteria digest cellulose faster? Journal of Dairy Science 79, 1496-502. doi: 10.3168/jds.S0022-0302(96)76509-8
- 21. Jung, H.G., and Vogel, K.P. (1986) Influence of lignin on digestibility of forage cell wall material. *Journal of animal science* **62**, 1703-12. doi: 10.2527/jas1986.6261703x
- 22. Ruckle, M., Meier, M., Frey, L., Eicke, S., Kölliker, R., Zeeman, S., and Studer, B. (2017) Diurnal leaf starch content: An orphan trait in forage legumes. *Agronomy* **7**, 16. doi: 10.3390/agronomy7010016
- 23. Wallace, R.J., Onodera, R., and Cotta, M.A. (1997) "Metabolism of nitrogen-containing compounds," in *The Rumen Microbial Ecosystem*, eds. P.N. Hobson & C.S. Stewart. (Dordrecht: Springer Netherlands), 283-328.
- 24. **Lees, G.L.** (1992) Condensed tannins in some forage legumes: their role in the prevention of ruminant pasture bloat. *Basic Life Science* **59**, 915-34. doi: 10.1007/978-1-4615-3476-1_55
- 25. **Piluzza, G., Sulas, L., and Bullitta, S.** (2014) Tannins in forage plants and their role in animal husbandry and environmental sustainability: a review. *Grass and Forage Science* **69**, 32-48. doi: 10.1111/gfs.12053
- 26. **Mueller-Harvey, I.** (2006) Unravelling the conundrum of tannins in animal nutrition and health. *Journal of the Science of Food and Agriculture* **86,** 2010-37. doi: 10.1002/jsfa.2577
- 27. **Barry, T.N., and Manley, T.R.** (1984) The role of condensed tannins in the nutritional value of *Lotus pedunculatus* for sheep. 2. Quantitative digestion of carbohydrates and proteins. *Br J Nutr* **51,** 493-504. doi: 10.1079/BJN19840055
- Hatfield, R.D., Jung, H.J.G., Broderick, G., and Jenkins, T.C. (2007) "Nutritional chemistry of forages," in *Forages: The Science of Grassland Agriculture*, eds. R.F. Barnes, C.J. Nelson, K.J. Moore & M. Collins. 6th ed (Ames, Iowa: Blackwell Publishing), 467-85.
- 29. Clapham, W.M., Foster, J.G., Neel, J.P.S., and Fedders, J.M. (2005) Fatty acid composition of traditional and novel forages. *Journal of Agricultural and Food Chemistry* **53**, 10068-73. doi: 10.1021/jf0517039
- 30. **Harfoot, C.G., Hazlewood, G.P.** (1988) "Lipid metabolism in the rumen," in *The Rumen Microbial Ecosystem*, ed. **P.M. Hobson**. (London: Elsevier), 285-322.
- Wood, J.D., Richardson, R.I., Nute, G.R., Fisher, A.V., Campo, M.M., Kasapidou, E., Sheard, P.R., and Enser, M. (2004) Effects of fatty acids on meat quality: a review. *Meat Science* 66, 21-32. doi: 10.1016/S0309-1740(03)00022-6
- Van Elswyk, M.E., and McNeill, S.H. (2014) Impact of grass/forage feeding versus grain finishing on beef nutrients and sensory quality: The U.S. experience. *Meat Science* 96, 535-40. doi: 10.1016/j.meatsci.2013.08.010

- 33. Elgersma, A., Tamminga, S., and Ellen, G. (2006) Modifying milk composition through forage. *Animal Feed Science and Technology* **131**, 207-25. doi: 10.1016/j.anifeedsci.2006.06.012
- Chilliard, Y., Glasser, F., Ferlay, A., Bernard, L., Rouel, J., and Doreau, M. (2007) Diet, rumen biohydrogenation and nutritional quality of cow and goat milk fat. *European Journal of Lipid Science and Technology* 109, 828-55. doi: 10.1002/ejlt.200700080
- 35. Boufaïed, H., Chouinard, P.Y., Tremblay, G.F., Petit, H.V., Michaud, R., and Bélanger, G. (2003) Fatty acids in forages. I. Factors affecting concentrations. *Canadian Journal of Animal Science* **83**, 501-11. doi: 10.4141/A02-098
- 36. **Elgersma, A., Ellen, G., van der Horst, H., Muuse, B.G., Boer, H., and Tamminga, S.** (2003) Influence of cultivar and cutting date on the fatty acid composition of perennial ryegrass (*Lolium perenne* L.). *Grass and Forage Science* **58**, 323-31. doi: 10.1046/j.1365-2494.2003.00384.x
- 37. Dewhurst, R.J., Scollan, N.D., Youell, S.J., Tweed, J.K.S., and Humphreys, M.O. (2001) Influence of species, cutting date and cutting interval on the fatty acid composition of grasses. *Grass and Forage Science* **56**, 68-74. doi: 10.1046/j.1365-2494.2001.00247.x
- 38. **Brugger, D., and Windisch, W.M.** (2015) Environmental responsibilities of livestock feeding using trace mineral supplements. *Animal Nutrition* **1**, 113-8. doi: 10.1016/j.aninu.2015.08.005
- 39. Zhu, Y.-G., Pilon-Smits, E.A.H., Zhao, F.-J., Williams, P.N., and Meharg, A.A. (2009) Selenium in higher plants: understanding mechanisms for biofortification and phytoremediation. *Trends Plant Sci* **14**, 436-42. doi: 10.1016/j.tplants.2009.06.006
- 40. **Freeman, K.W., Girma, K., Arnall, D.B., Mullen, R.W., Kent, K.L., Teal, R.K., and Raun, W.R.** (2007) By-plant prediction of corn forage biomass and nitrogen uptake at various growth stages using remote sensing and plant height. *Agronomy* **99,** 530-6. doi: 10.2134/agronj2006.0135
- 41. Morvan-Bertrand, A., Pavis, N., Boucaud, J., and Prud'Homme, M.P. (1999) Partitioning of reserve and newly assimilated carbon in roots and leaf tissues of *Lolium perenne* during regrowth after defoliation: assessment by 13C steady-state labelling and carbohydrate analysis. *Plant, Cell & Environment* 22, 1097-108. doi: 10.1046/j.1365-3040.1999.00485.x
- 42. **Paterson, E., and Sim, A.** (1999) Rhizodeposition and C-partitioning of *Lolium perenne* in axenic culture affected by nitrogen supply and defoliation. *Plant and Soil* **216**, 155-64. doi: 10.1023/A:1004789407065
- 43. **Paterson, E., and Sim, A.** (2000) Effect of nitrogen supply and defoliation on loss of organic compounds from roots of *Festuca rubra*. *Journal of Experimental Botany* **51,** 1449-57. doi: 10.1093/jxb/51.349.1449
- 44. **Paterson, E., Thornton, B., Sim, A., and Pratt, S.** (2003) Effects of defoliation and atmospheric CO2 depletion on nitrate acquisition, and exudation of organic compounds by roots of *Festuca rubra*. *Plant and Soil* **250**, 293-305. doi: 10.1023/a:1022819219947
- 45. **Dawson, L.A., Thornton, B., Pratt, S.M., and Paterson, E.** (2004) Morphological and topological responses of roots to defoliation and nitrogen supply in *Lolium perenne* and *Festuca ovina*. *New Phytologist* **161,** 811-8. doi: 10.1111/j.1469-8137.2004.00979.x
- 46. **Thornton, B., Paterson, E., Midwood, A.J., Sim, A., and Pratt, S.M.** (2004) Contribution of current carbon assimilation in supplying root exudates of *Lolium perenne* measured using steady-state 13C labelling. *Physiologia Plantarum* **120,** 434-41. doi: 10.1111/j.0031-9317.2004.00250.x

- Paterson, E., Thornton, B., Midwood, A.J., and Sim, A. (2005) Defoliation alters the relative contributions of recent and non-recent assimilate to root exudation from *Festuca rubra*. *Plant, Cell & Environment* 28, 1525-33. doi: 10.1111/j.1365-3040.2005.01389.x
- 48. **Deru, J., Schilder, H., Van der Schoot, J.R., and Van Eekeren, N.** (Year) "No trade-off between root biomass and aboveground production in *Lolium perenne*", in: *Breeding in a World of Scarcity*, eds. I. Roldán-Ruiz, J. Baert & D. Reheul: Springer International Publishing), 289-92.
- Meuriot, F., Avice, J.C., Simon, J.C., Laine, P., Decau, M.L., and Ourry, A. (2004) Influence of initial organic N reserves and residual leaf area on growth, N uptake, N partitioning and N storage in alfalfa (*Medicago sativa*) during post-cutting regrowth. *Annals of Botany* 94, 311-21. doi: 10.1093/aob/mch144
- 50. **Meuriot, F., Decau, M.-L., Morvan-Bertrand, A., Prud'Homme, M.-P., Gastal, F., Simon, J.-C., Volenec, J.J., and Avice, J.-C.** (2005) Contribution of initial C and N reserves in *Medicago sativa* recovering from defoliation: impact of cutting height and residual leaf area. *Functional Plant Biology* **32**, 321-34. doi: 10.1071/FP04151
- 51. **Zhang, J., Yin, B., Xie, Y., Li, J., Yang, Z., and Zhang, G.** (2015) Legume-cereal intercropping improves forage yield, quality and degradability. *PLoS One* **10**, e0144813-e. doi: 10.1371/journal.pone.0144813
- 52. **Wang, Z.-Y., and Ge, Y.** (2006) Recent advances in genetic transformation of forage and turf grasses. *In Vitro Cellular & Developmental Biology Plant* **42**, 1-18. doi: 10.1079/IVP2005726
- 53. Abberton, M.T., Marshall, A.H., Humphreys, M.W., Macduff, J.H., Collins, R.P., and Marley, C.L. (2008) "Genetic improvement of forage species to reduce the environmental impact of temperate livestock grazing systems," in *Advances in Agronomy*. Academic Press), 311-55.
- 54. **van der Heijden, S.A.G., and Roulund, N.** (Year) "Genetic gain in agronomic value of forage crops and turf: A review", in: *Sustainable Use of Genetic Diversity in Forage and Turf Breeding*, ed. C. Huyghe: Springer Netherlands), 247-60.
- 55. **Rasmussen, S., Parsons, A.J., and Jones, C.S.** (2012) Metabolomics of forage plants: a review. *Annals of Botany* **110,** 1281-90. doi: 10.1093/aob/mcs023
- 56. Araújo, S.S., Beebe, S., Crespi, M., Delbreil, B., González, E.M., Gruber, V., Lejeune-Henaut, I., Link, W., Monteros, M.J., Prats, E., Rao, I., Vadez, V., and Patto, M.C.V. (2015) Abiotic Stress Responses in Legumes: Strategies Used to Cope with Environmental Challenges. *Critical Reviews in Plant Sciences* 34, 237-80. doi: 10.1080/07352689.2014.898450
- 57. **Jacob, C., Carrasco, B., and Schwember, A.R.** (2016) Advances in breeding and biotechnology of legume crops. *Plant Cell, Tissue and Organ Culture* **127**, 561-84. doi: 10.1007/s11240-016-1106-2
- 58. **Amundsen, K., Sarath, G., and Donze-Reiner, T.** (2017) Editorial: Genomic approaches for improvement of understudied grasses. *Frontiers in Plant Science* **8**. doi: 10.3389/fpls.2017.00976
- 59. Kumar, T., Bao, A.K., Bao, Z., Wang, F., Gao, L., and Wang, S.M. (2018) The progress of genetic improvement in alfalfa (*Medicago sativa* L.). *Czech Journal of Genetics and Plant Breeding* 54. doi: 10.17221/46/2017-CJGPB
- 60. **Yang, J., Wang, Z., Jiang, Y., and Fei, S.** (2019) Editorial: Genetics and genomics of polyploid plants. *Frontiers in Plant Science* **10**. doi: 10.3389/fpls.2019.00934
- 61. **Woodman, H.E., Blunt, D.L., and Stewart, J.** (1926) Nutritive value of pasture: I. Seasonal variations in the productivity, botanical and chemical

composition, and nutritive value of medium pasturage on a light sandy soil. *The Journal of Agricultural Science* **16**, 205-74. doi: 10.1017/S0021859600018244

- 62. **Woodman, H.E., Blunt, D.L., and Stewart, J.** (1927) Nutritive value of pasture: II. Seasonal variations in the productivity, botanical and chemical composition, and nutritive value of pasturage on a heavy clay soil. *The Journal of Agricultural Science* **17**, 209-63. doi: 10.1017/S0021859600018487
- 63. **Holmes, W.** (1948) The intensive production of herbage for crop-drying: Part I. A study of the productivity of two annual crops and two leys and of their responses in yield and chemical composition to applications of nitrogenous manure. *The Journal of Agricultural Science* **38**, 425-36. doi: 10.1017/S0021859600006225
- 64. **BAGCD** (2018) *Analysis* [Online]. UK. Available: <u>http://www.bagcd.org/analysis/</u> [Accessed 20/06/2018].
- 65. **Slade, R.E.** (1949) *Green crop dryers' research association year book 1949.* The Farmers Weekly,.
- 66. **Siddiqi, M.Y., and Glass, A.D.M.** (1981) Utilization index: A modified approach to the estimation and comparison of nutrient utilization efficiency in plants. *Journal of Plant Nutrition* **4,** 289-302. doi: 10.1080/01904168109362919
- Moll, R.H., Kamprath, E.J., and Jackson, W.A. (1982) Analysis and interpretation of factors which contribute to efficiency of nitrogen utilization. *Agronomy* 74, 562-4. doi: 10.2134/agronj1982.00021962007400030037x
- 68. **Berendse, F., and Aerts, R.** (1987) Nitrogen-use-efficiency: A biologically meaningful definition? *Functional Ecology* **1,** 293-6.
- 69. **Masclaux-Daubresse, C., Daniel-Vedele, F., Dechorgnat, J., Chardon, F., Gaufichon, L., and Suzuki, A.** (2010) Nitrogen uptake, assimilation and remobilization in plants: challenges for sustainable and productive agriculture. *Annals of Botany* **105,** 1141-57. doi: 10.1093/aob/mcq028
- 70. **Hirose, T.** (2011) Nitrogen use efficiency revisited. *Oecologia* **166,** 863-7. doi: 10.1007/s00442-011-1942-z
- 71. **Han, M., Okamoto, M., Beatty, P.H., Rothstein, S.J., and Good, A.G.** (2015) The genetics of nitrogen use efficiency in crop plants. *Annu Rev Genet* **49,** 269-89. doi: 10.1146/annurev-genet-112414-055037
- 72. Xu, G., Fan, X., and Miller, A.J. (2012) Plant nitrogen assimilation and use efficiency. *Annual Review of Plant Biology* **63**, 153-82. doi: 10.1146/annurev-arplant-042811-105532
- 73. **Good, A.G., Shrawat, A.K., and Muench, D.G.** (2004) Can less yield more? Is reducing nutrient input into the environment compatible with maintaining crop production? *Trends Plant Sci* **9**, 597-605. doi: 10.1016/j.tplants.2004.10.008
- 74. Lawlor, D.W., Lemaire, G., and Gastal, F. (2001) "Nitrogen, plant growth and crop yield," in *Plant Nitrogen*, eds. P.J. Lea & J.-F. Morot-Gaudry. (Berlin, Heidelberg: Springer Berlin Heidelberg), 343-67.
- 75. Maire, V., Martre, P., Kattge, J., Gastal, F., Esser, G., Fontaine, S., Soussana, J.F. (2012) The coordination of leaf photosynthesis links C and N fluxes in C3 plant species. *PLoS One* **7**, e38345. doi: 10.1371/journal.pone.0038345.
- 76. **European Commission** (1991) *The nitrates directive* [Online]. Brussels, Belgium. Available: <u>https://eur-lex.europa.eu/legal-</u> content/EN/TXT/?uri=CELEX:31991L0676 [Accessed 07/08/2018].
- 77. **DEFRA** (1998) *Programme for nitrate vulnerable zones (England and Wales) regulations 1998* [Online]. London, UK. Available:

http://www.legislation.gov.uk/uksi/1998/1202/contents/made [Accessed 20/06/2018].

- 78. DEFRA (2008) Nitrate pollution prevention regulations 2008 [Online]. London, UK. Available: <u>http://www.legislation.gov.uk/uksi/2008/2349/contents/made</u> [Accessed 20/06/2018].
- 79. **DEFRA** (2010) *Fertiliser manual Rb209* [Online]. London, UK. Available: <u>https://books.google.co.uk/books?id=cHs4bwAACAAJ</u> [Accessed 20/06/2018].
- 80. **DEFRA** (2019) *British survey of fertiliser practice 2018 annual report* [Online]. London, UK. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/upl</u> <u>oads/attachment_data/file/806643/fertiliseruse-report2018-06jun19.pdf</u> [Accessed 10/09/2019].
- 81. **DEFRA** (2015) *Environmental management guidance: Nutrient management: Nitrate Vulnerable Zones* [Online]. London, UK: DEFRA National Statistics Release. Available: <u>https://www.gov.uk/guidance/nutrient-management-nitrate-vulnerable-zones</u> [Accessed 15/09/2017].
- 82. **Environment Agency** (2016) "Implementation of the nitrate pollution prevention regulations 2015 in England: Review of nitrate vulnerable zone designations for implementation in 2017: Environment agency report and recommendations to DEFRA", (ed.) Environment Agency. (Bristol, UK: Environment Agency,).
- 83. **Crown**© (2018) "The guide to cross compliance in England 2018", (ed.) DEFRA. (London, UK).
- 84. **Eurostat** (2019) "Agri-environmental indicator gross nitrogen balance (aei_pr_gnb)", (ed.) E. Commision.).
- 85. **Environment Agency** (2016) "Check for drinking water safeguard zones and NVZs", (ed.) Environment Agency. (Bristol, UK: Environment Agency, DEFRA, Crown©,).
- 86. **European Environment Agency** (2015) *EU 2010 biodiversity baseline adapted to the MAES typology (2015)* [Online]. Copenhagen, Denmark. Available: <u>https://www.eea.europa.eu//publications/eu-2010-</u> <u>biodiversity-baseline-revision</u> [Accessed 08/08/2018].
- 87. Bengtsson, J., Bullock, J.M., Egoh, B., Everson, C., Everson, T., O'Connor, T., O'Farrell, P.J., Smith, H.G., and Lindborg, R. (2019) Grasslands—more important for ecosystem services than you might think. *Ecosphere* **10**, e02582. doi: 10.1002/ecs2.2582
- 88. Mangalassery, S., Sjögersten, S., Sparkes, D.L., Sturrock, C.J., Craigon, J., and Mooney, S.J. (2014) To what extent can zero tillage lead to a reduction in greenhouse gas emissions from temperate soils? *Scientific Reports* 4, 4586. doi: 10.1038/srep04586
- Mangalassery, S., Mooney, S.J., Sparkes, D.L., Fraser, W.T., and Sjögersten, S. (2015) Impacts of zero tillage on soil enzyme activities, microbial characteristics and organic matter functional chemistry in temperate soils. *European Journal of Soil Biology* 68, 9-17. doi: 10.1016/j.ejsobi.2015.03.001
- Mangalassery, S., SjÖGersten, S., Sparkes, D.L., and Mooney, S.J. (2015) Examining the potential for climate change mitigation from zero tillage. *The Journal of Agricultural Science* 153, 1151-73. doi: 10.1017/S0021859614001002
- 91. Cooper, R.J., Hama-Aziz, Z., Hiscock, K.M., Lovett, A.A., Dugdale, S.J., Sünnenberg, G., Noble, L., Beamish, J., and Hovesen, P. (2017) Assessing the farm-scale impacts of cover crops and noninversion tillage regimes on nutrient losses from an arable catchment.

Agriculture, Ecosystems & Environment **237,** 181-93. doi: 10.1016/j.agee.2016.12.034

- 92. Huang, Y., Ren, W., Wang, L., Hui, D., Grove, J.H., Yang, X., Tao, B., and Goff, B. (2018) Greenhouse gas emissions and crop yield in notillage systems: A meta-analysis. *Agriculture, Ecosystems & Environment* 268, 144-53. doi: 10.1016/j.agee.2018.09.002
- 93. **Crown**© (2018) *A green future: Our 25 year plan to improve the environment* [Online]. London, UK. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/upl</u> <u>oads/attachment_data/file/693158/25-year-environment-plan.pdf</u> [Accessed 01/08/2018].
- 94. **Office for National Statistics, and DEFRA** (2018) *UK natural capital: ecosystem accounts for urban areas* [Online]. Available: <u>https://www.ons.gov.uk/releases/urbanecosystemaccounts</u> [Accessed 17/09/2019].
- 95. Alexandratos, N., Bruinsma, J., Bödeker, G., Schmidhuber, J., Broca, S., Shetty, P., and Ottaviani, M.G. (2006) World agriculture: Towards 2030/2050. Interim report. Prospects for food, nutrition, agriculture and major commodity groups [Online]. Rome, Italy: F.a.A.O.o.t.U. Nations. Available: URL: http://www.fao.org/fileadmin/user_upload/esag/docs/Interim_report_AT 2050web.pdf [Accessed 16/04/2018].
- 96. **Thornton, P.K.** (2010) Livestock production: recent trends, future prospects. *Philosophical Transactions of the Royal Society B: Biological Sciences* **365**, 2853-67. doi: 10.1098/rstb.2010.0134
- 97. Alexandratos, N., and Bruinsma, J. (2012) *World agriculture towards* 2030/2050: the 2012 revision [Online]. Rome, Italy: Food and Agriculture Organization of the United Nations. Available: URL: <u>http://www.fao.org/docrep/016/ap106e/ap106e.pdf</u> [Accessed 16/04/2018].
- 98. Herrero, M., Grace, D., Njuki, J., Johnson, N., Enahoro, D., Silvestri, S., and Rufino, M.C. (2013) The roles of livestock in developing countries. *animal* 7, 3-18. doi: 10.1017/S1751731112001954
- 99. Cramer, H., Kessler, C.S., Sundberg, T., Leach, M.J., Schumann, D., Adams, J., and Lauche, R. (2017) Characteristics of Americans choosing vegetarian and vegan diets for health reasons. *Journal of Nutrition Education and Behavior* 49, 561-7. doi: 10.1016/j.jneb.2017.04.011
- 100. Ramankutty, N., Mehrabi, Z., Waha, K., Jarvis, L., Kremen, C., Herrero, M., and Rieseberg, L.H. (2018) Trends in global agricultural land use: Implications for environmental health and food security. *Annual Review of Plant Biology* 69, 789-815. doi: 10.1146/annurev-arplant-042817-040256
- 101. **Millennium Ecosystem Assessment** (2005) *Ecosystems and human well-being: Synthesis* [Online]. Washington, DC: I. Press. Available: <u>https://www.millenniumassessment.org/documents/document.356.aspx.</u> <u>pdf</u> [Accessed 19/08/2018].
- 102. **Morris, M.G.** (1992) Responses of *Auchenorhyncha* (homoptera) to fertiliser and liming treatments at Park Grass, Rothamsted. *Agriculture, Ecosystems & Environment* **41**, 263-83. doi: 10.1016/0167-8809(92)90115-R
- 103. **Tilman, D., Reich, P.B., Knops, J., Wedin, D., Mielke, T., and Lehman, C.** (2001) Diversity and productivity in a long-term grassland experiment. *Science* **294**, 843-5. doi: 10.1126/science.1060391
- 104. Silvertown, J., Poulton, P., Johnston, E., Edwards, G., Heard, M., and Biss, P.M. (2006) The Park Grass Experiment 1856–2006: its

contribution to ecology. *Journal of Ecology* **94,** 801-14. doi: 10.1111/j.1365-2745.2006.01145.x

- 105. **Tilman, D., Reich, P.B., and Knops, J.M.H.** (2006) Biodiversity and ecosystem stability in a decade-long grassland experiment. *Nature* **441**, 629-32. doi: 10.1038/nature04742
- Weigelt, A., Weisser, W.W., Buchmann, N., and Scherer-Lorenzen, M. (2009) Biodiversity for multifunctional grasslands: Equal production in high-diversity low-input and low-diversity high-input systems. *Biogeosciences* 6, 1675-706.
- 107. Liang, Y., Wu, L., Clark, I.M., Xue, K., Yang, Y., Van Nostrand, J.D., Deng, Y., He, Z., McGrath, S., Storkey, J., Hirsch, P.R., Sun, B., and Zhou, J. (2015) Over 150 years of long-term fertilization alters spatial scaling of microbial biodiversity. *mBIO* 6, e00240-15. . doi: 10.1128/mBio.00240-15.
- Kidd, J., Manning, P., Simkin, J., Peacock, S., and Stockdale, E. (2017) Impacts of 120 years of fertilizer addition on a temperate grassland ecosystem. *PLoS One* **12**, e0174632. doi: 10.1371/journal.pone.0174632
- 109. **Evans, E.C., and Sanderson, R.A.** (2018) Long-term fertilizer regimes have both direct and indirect effects on arthropod community composition and feeding guilds. *Journal of Applied Entomology* **142,** 230-40. doi: 10.1111/jen.12410
- 110. **De Baets, S., Poesen, J., Meersmans, J., and Serlet, L.** (2011) Cover crops and their erosion-reducing effects during concentrated flow erosion. *CATENA* **85**, 237-44. doi: 10.1016/j.catena.2011.01.009
- 111. **Raimbault, B.A., and Vyn, T.J.** (1991) Crop rotation and tillage effects on corn growth and soil structural stability. *Agronomy Journal* **83,** 979-85. doi: 10.2134/agronj1991.00021962008300060011x
- 112. **Fullen, M.A.** (1998) Effects of grass ley set-aside on runoff, erosion and organic matter levels in sandy soils in East Shropshire, UK. *Soil and Tillage Research* **46**, 41-9. doi: 10.1016/S0167-1987(98)80106-2
- 113. **Philibert, A., Loyce, C., and Makowski, D.** (2012) Quantifying uncertainties in N₂O emission due to N fertilizer application in cultivated areas. *PLoS One* **7**, e50950. doi: 10.1371/journal.pone.0050950
- 114. Buckingham, S., Anthony, S., Bellamy, P.H., Cardenas, L.M., Higgins, S., McGeough, K., and Topp, C.F.E. (2014) Review and analysis of global agricultural N₂O emissions relevant to the UK. *Science* of *The Total Environment* **487**, 164-72. doi: 10.1016/j.scitotenv.2014.02.122
- 115. **Dobbie, K.E., and Smith, K.A.** (2003) Impact of different forms of N fertilizer on N₂O emissions from intensive grassland. *Nutrient Cycling in Agroecosystems* **67,** 37-46. doi: 10.1023/a:1025119512447
- 116. Cardenas, L.M., Thorman, R., Ashlee, N., Butler, M., Chadwick, D., Chambers, B., Cuttle, S., Donovan, N., Kingston, H., Lane, S., Dhanoa, M.S., and Scholefield, D. (2010) Quantifying annual N₂O emission fluxes from grazed grassland under a range of inorganic fertiliser nitrogen inputs. *Agriculture, Ecosystems & Environment* **136**, 218-26. doi: 10.1016/j.agee.2009.12.006
- 117. **Miller, A.J., Fan, X., Orsel, M., Smith, S.J., and Wells, D.M.** (2007) Nitrate transport and signalling. *Journal of Experimental Botany* **58**, 2297-306. doi: 10.1093/jxb/erm066
- 118. **Qi, A., Holland, R.A., Taylor, G., and Richter, G.M.** (2018) Grassland futures in Great Britain Productivity assessment and scenarios for land use change opportunities. *Science of The Total Environment* **634**, 1108-18. doi: 10.1016/j.scitotenv.2018.03.395
- 119. **Clement, C.R., Hopper, M.J., Jones, L.H.P., and Leafe, E.L.** (1978) The Uptake of Nitrate by Lolium perenne from Flowing Nutrient Solution:

II. Effect of light, defoliation, and relationship to CO₂ flux. *Journal of Experimental Botany* **29**, 1173-83.

- 120. **du Jardin, P.** (2015) Plant biostimulants: Definition, concept, main categories and regulation. *Scientia Horticulturae* **196,** 3-14. doi: 10.1016/j.scienta.2015.09.021
- 121. **Canellas, L.P., and Olivares, F.L.** (2014) Physiological responses to humic substances as plant growth promoter. *Chemical and Biological Technologies in Agriculture* **1**, 3. doi: 10.1186/2196-5641-1-3
- 122. Canellas, L.P., Olivares, F.L., Aguiar, N.O., Jones, D.L., Nebbioso, A., Mazzei, P., and Piccolo, A. (2015) Humic and fulvic acids as biostimulants in horticulture. *Scientia Horticulturae*. doi: 10.1016/j.scienta.2015.09.013
- 123. Verlinden, G., Pycke, B., Mertens, J., Debersaques, F., Verheyen, K., Baert, G., Bries, J., and Haesaert, G. (2009) Application of humic substances results in consistent increases in crop yield and nutrient uptake. *Journal of Plant Nutrition* **32**, 1407-26. doi: 10.1080/01904160903092630
- 124. Verlinden, G., Coussens, T., De Vliegher, A., Baert, G., and Haesaert, G. (2010) Effect of humic substances on nutrient uptake by herbage and on production and nutritive value of herbage from sown grass pastures. *Grass and Forage Science* **65**, 133-44. doi: 10.1111/j.1365-2494.2009.00726.x
- 125. **Koressaar, T., and Remm, M.** (2007) Enhancements and modifications of primer design program Primer3. *Bioinformatics* **23**, 1289-91. doi: 10.1093/bioinformatics/btm091
- 126. Untergasser, A., Cutcutache, I., Koressaar, T., Ye, J., Faircloth, B.C., Remm, M., and Rozen, S.G. (2012) Primer3—new capabilities and interfaces. *Nucleic Acids Research* **40**, e115-e. doi: 10.1093/nar/gks596
- Kibbe, W.A. (2007) OligoCalc: an online oligonucleotide properties calculator. *Nucleic Acids Research* 35, W43-W6. doi: 10.1093/nar/gkm234
- 128. Nakabayashi, K., Okamoto, M., Koshiba, T., Kamiya, Y., and Nambara, E. (2005) Genome-wide profiling of stored mRNA in *Arabidopsis thaliana* seed germination: epigenetic and genetic regulation of transcription in seed. *The Plant Journal* **41**, 697-709. doi: 10.1111/j.1365-313X.2005.02337.x
- 129. Schmid, M., Davison, T.S., Henz, S.R., Pape, U.J., Demar, M., Vingron, M., Scholkopf, B., Weigel, D., and Lohmann, J.U. (2005) A gene expression map of Arabidopsis thaliana development. *Nat Genet* **37**, 501-6. doi: 10.1038/ng1543
- 130. Winter, D., Vinegar, B., Nahal, H., Ammar, R., Wilson, G.V., and Provart, N.J. (2007) An "Electronic Fluorescent Pictograph" browser for exploring and analyzing large-scale biological data sets. *PLoS One* **2**, e718. doi: 10.1371/journal.pone.0000718
- 131. **Livak, K.J., and Schmittgen, T.D.** (2001) Analysis of relative gene expression data using real-time quantitative PCR and the $2-\Delta\Delta$ CT method. *Methods* **25**, 402-8. doi: 10.1006/meth.2001.1262
- 132. **Izumi, M., Tsunoda, H., Suzuki, Y., Makino, A., and Ishida, H.** (2012) RBCS1A and RBCS3B, two major members within the *Arabidopsis RBCS* multigene family, function to yield sufficient Rubisco content for leaf photosynthetic capacity. *Journal of Experimental Botany* **63**, 2159-70. doi: 10.1093/jxb/err434
- 133. Konishi, N., Ishiyama, K., Matsuoka, K., Maru, I., Hayakawa, T., Yamaya, T., and Kojima, S. (2014) NADH-dependent glutamate synthase plays a crucial role in assimilating ammonium in the *Arabidopsis* root. *Physiologia Plantarum* **152**, 138-51. doi: 10.1111/ppl.12177

- 134. Ishiyama, K., Inoue, E., Watanabe-Takahashi, A., Obara, M., Yamaya, T., and Takahashi, H. (2004) Kinetic properties and ammonium-dependent regulation of cytosolic isoenzymes of glutamine synthetase in *Arabidopsis*. *J Biol Chem* **279**, 16598-605. doi: 10.1074/jbc.M313710200
- 135. **Ma, S., Quist, T.M., Ulanov, A., Joly, R., and Bohnert, H.J.** (2004) Loss of TIP1;1 aquaporin in *Arabidopsis* leads to cell and plant death. *The Plant Journal* **40,** 845-59. doi: 10.1111/j.1365-313X.2004.02265.x
- 136. **Mao, Z., and Sun, W.** (2015) *Arabidopsis* seed-specific vacuolar aquaporins are involved in maintaining seed longevity under the control of *ABSCISIC ACID INSENSITIVE 3*. *Journal of Experimental Botany* **66**, 4781-94. doi: 10.1093/jxb/erv244
- 137. Altschul, S.F., Gish, W., Miller, W., Myers, E.W., and Lipman, D.J. (1990) Basic local alignment search tool. *J Mol Biol* **215**, 403-10. doi: 10.1016/s0022-2836(05)80360-2
- 138. Altschul, S.F., Madden, T.L., Schäffer, A.A., Zhang, J., Zhang, Z., Miller, W., and Lipman, D.J. (1997) Gapped BLAST and PSI-BLAST: a new generation of protein database search programs. *Nucleic Acids Research* 25. doi: 10.1093/nar/25.17.3389
- Zaretskaya, I., Johnson, M., McGinnis, S., Raytselis, Y., Merezhuk, Y., and Madden, T.L. (2008) NCBI BLAST: a better web interface. *Nucleic Acids Research* 36, W5-W9. doi: 10.1093/nar/gkn201
- NCBI Resource Coordinators (2016) Database resources of the National Center for Biotechnology Information. *Nucleic Acids Research* 44, D7-D19. doi: 10.1093/nar/gkv1290
- 141. Proost, S., Van Bel, M., Vaneechoutte, D., Van de Peer, Y., Inzé, D., Mueller-Roeber, B., and Vandepoele, K. (2015) PLAZA 3.0: an access point for plant comparative genomics. *Nucleic Acids Research* 43, D974-D81. doi: 10.1093/nar/gku986
- 142. Katoh, K., and Standley, D.M. (2013) MAFFT multiple sequence alignment software version 7: improvements in performance and usability. *Molecular Biology and Evolution* **30**, 772-80. doi: 10.1093/molbev/mst010
- Letunic, I., and Bork, P. (2007) Interactive Tree Of Life (iTOL): an online tool for phylogenetic tree display and annotation. *Bioinformatics* 23, 127-8. doi: 10.1093/bioinformatics/btl529
- 144. Yachdav, G., Wilzbach, S., Rauscher, B., Sheridan, R., Sillitoe, I., Procter, J., Lewis, S.E., Rost, B., and Goldberg, T. (2016) MSAViewer: interactive JavaScript visualization of multiple sequence alignments. *Bioinformatics* 32, 3501-3. doi: 10.1093/bioinformatics/btw474
- 145. **Stothard, P.** (2000) The sequence manipulation suite: JavaScript programs for analyzing and formatting protein and DNA sequences. *Biotechniques* **28**, 1102, 4. doi: 10.2144/00286ir01
- 146. Nord-Larsen, P.H., Kichey, T., Jahn, T.P., Jensen, C.S., Nielsen, K.K., Hegelund, J.N., and Schjoerring, J.K. (2009) Cloning, characterization and expression analysis of tonoplast intrinsic proteins and glutamine synthetase in ryegrass (*Lolium perenne* L.). *Plant Cell Reports* **28**, 1549-62. doi: 10.1007/s00299-009-0754-7
- 147. **Petersen, K., Didion, T., Andersen, C.H., and Nielsen, K.K.** (2004) MADS-box genes from perennial ryegrass differentially expressed during transition from vegetative to reproductive growth. *Journal of Plant Physiology* **161,** 439-47. doi: 10.1078/0176-1617-01212
- 148. **R Development core team** (2004) *R: A language and enviroment for statistical computing.* [Online]. Vienna, Austria.: R foundation for statistical computing. Available: <u>http://www.r-project.org</u> [Accessed 10/07/2019].

- 149. Galili, T., Rowlingson, B., Hejblum, B., Dason;, Schonbrodt, F., Grothendieck, G., Daroczi, G., Andrew;, H., James;, Leeper, T., VitoshKa;, Xie, Y., Friendly, M., Rohmeyer, K., Menne, D., Hunt, T., Hiramura, T., Boessenkool, B., Godfrey, J., Allard, T., Chen, C., Hill, J., Park, C.-Y., and Nachtmann, G. (2019) Package 'installr': Using R to Install Stuff (Such As: R, 'Rtools', 'RStudio', 'Git', and More!) [Online]. CRAN. Available: <u>https://github.com/talgalili/installr/</u> [Accessed 10/07/2019].
- 150. **Wickman, H.** (2017) *tidyverse: Easily Install and Load the 'Tidyverse'* [Online]. CRAN. Available: <u>http://tidyverse.tidyverse.org</u>, <u>https://github.com/tidyverse/tidyverse</u> [Accessed 11/07/2019].
- 151. **Wilke, C.O.** (2019) *cowplot: Streamlined Plot Theme and Plot Annotations for 'ggplot2'* [Online]. CRAN. Available: <u>https://wilkelab.org/cowplot</u> [Accessed 11/07/2019].
- 152. **Wickham, H.** (2016) *ggplot2: Elegant Graphics for Data Analysis.* New York, USA: Springer-Verlag.
- 153. Dowle, M., Srinivasan, A., Gorecki, J., Chirico, M., Stetsenko, P., Short, T., Lianoglou, S., Antonyan, E., Bonsch, M., Parsonage, H., Ritchie, S., Ren, K., Tan, X., Saporta, R., Seiskari, O., Dong, X., Lang, M., Iwasaki, W., Wenchel, S., Broman, K., Schmidt, T., Arenburg, D., Smith, E., Cocquemas, F., Gomez, M., Chataignon, P., Groves, D., Possenriede, D., Parages, F., Toth, D., Yaramaz-David, M., Perumal, A., Sams, J., Morgan, M., and Quinn, M. (2019) data.table: extension of 'data.frame' [Online]. CRAN. Available: https://CRAN.R-project.org/package=data.table [Accessed 10/07/2019].
- 154. **Erickson, C.** (2017) "GauPro: Gaussian Process Fitting". v0.2.2 ed.: C. Erickson).
- 155. **Miller, A.J., Le Besnerais, P-H., Hugo, M.** (2014) *Soil chemistry sensor - PCT/GB2013/053377 - WO/2014/096844 - G01N 27/403* (2006.01). PCT/GB2013/053377.
- 156. **Miller, A.J.** (1995) Ion-selective microelectrodes for measurement of intracellular ion concentrations. *Methods Cell Biol* **49**, 275-91.
- 157. **Robins, J.G., and Lovatt, A.J.** (2016) Cultivar by environment effects of perennial ryegrass cultivars selected for high water soluble carbohydrates managed under differing precipitation levels. *Euphytica* **208,** 571-81. doi: 10.1007/s10681-015-1607-9
- 158. **HydroGarden Wholesale Supplies Ltd.** (2016) *VitaLink Fulvic* [Online]. Available: <u>http://www.hydrogarden.com/</u> [Accessed 09/07/2019].
- 159. **F.A.R.M. Co.** (2017) *MPXA*® *HumicBased Agricultural Products* [Online]. Available: <u>https://www.humicsbyfarmco.com/agricultural.html</u> [Accessed 09/07/2019].
- 160. Smith, C.A., O'Maille, G., Want, E.J., Qin, C., Trauger, S.A., Brandon, T.R., Custodio, D.E., Abagyan, R., and Siuzdak, G. (2005) METLIN: a metabolite mass spectral database. *Therapeutic Drug Monitoring* 27, 747-51. doi: 10.1097/01.ftd.0000179845.53213.39
- 161. Guijas, C., Montenegro-Burke, J.R., Domingo-Almenara, X., Palermo, A., Warth, B., Hermann, G., Koellensperger, G., Huan, T., Uritboonthai, W., Aisporna, A.E., Wolan, D.W., Spilker, M.E., Benton, H.P., and Siuzdak, G. (2018) METLIN: a technology platform for identifying knowns and unknowns. *Analytical chemistry* **90**, 3156-64. doi: 10.1021/acs.analchem.7b04424
- P.J. Linstrom, and Mallard, W.G. (eds.). (2018) NIST Chemistry WebBook, NIST Standard Reference Database Number 69. Gaithersburg MD: National Institute of Standards and Technology.
- 163. **Mathers, N.J., and Xu, Z.** (2003) Solid-state 13C NMR spectroscopy: characterization of soil organic matter under two contrasting residue

management regimes in a 2-year-old pine plantation of subtropical Australia. *Geoderma* **114**, 19-31. doi: 10.1016/S0016-7061(02)00339-7

- 164. Mathers, N.J., Jalota, R.K., Dalal, R.C., and Boyd, S.E. (2007) 13C-NMR analysis of decomposing litter and fine roots in the semi-arid Mulga Lands of southern Queensland. *Soil Biology and Biochemistry* **39**, 993-1006. doi: 10.1016/j.soilbio.2006.11.009
- Quideau, S.A., McIntosh, A.C.S., Norris, C.E., Lloret, E., Swallow, M.J.B., and Hannam, K. (2016) Extraction and analysis of microbial phospholipid fatty acids in soils. *JoVE*, e54360. doi: 10.3791/54360
- 166. Quezada, M., Buitrón, G., Moreno-Andrade, I., Moreno, G., and López-Marín, L.M. (2007) The use of fatty acid methyl esters as biomarkers to determine aerobic, facultatively aerobic and anaerobic communities in wastewater treatment systems. *FEMS Microbiology Letters* 266, 75-82. doi: 10.1111/j.1574-6968.2006.00509.x
- Metsalu, T., and Vilo, J. (2015) ClustVis: a web tool for visualizing clustering of multivariate data using Principal Component Analysis and heatmap. *Nucleic Acids Research* 43, W566-W70. doi: 10.1093/nar/gkv468
- 168. **Wold, H.** (1966) "Estimation of principal components and related models by iterative least squares," in *Multivariate analysis,* ed. **P.R. Krishnajah**. (New York: Academic Press), 391-420.
- Stacklies, W., Redestig, H., Scholz, M., Walther, D., and Selbig, J. (2007) pcaMethods--a bioconductor package providing PCA methods for incomplete data. *Bioinformatics* 23, 1164-7. doi: 10.1093/bioinformatics/btm069
- 170. Trabelsi, D., Pini, F., Aouani, M.E., Bazzicalupo, M., and Mengoni, A. (2009) Development of real-time PCR assay for detection and quantification of *Sinorhizobium meliloti* in soil and plant tissue. *Lett Appl Microbiol* 48, 355-61. doi: 10.1111/j.1472-765X.2008.02532.x
- 171. Beckers, B., Op De Beeck, M., Thijs, S., Truyens, S., Weyens, N., Boerjan, W., and Vangronsveld, J. (2016) Performance of 16s rDNA primer pairs in the study of rhizosphere and endosphere bacterial microbiomes in metabarcoding studies. *Frontiers in microbiology* 7, 650-. doi: 10.3389/fmicb.2016.00650
- 172. **Chelius, M.K., and Triplett, E.W.** (2001) The diversity of archaea and bacteria in association with the roots of Zea mays L. *Microb Ecol* **41**, 252-63. doi: 10.1007/s002480000087
- Walker, J.J., and Pace, N.R. (2007) Phylogenetic composition of Rocky Mountain endolithic microbial ecosystems. *Applied and environmental microbiology* 73, 3497-504. doi: 10.1128/AEM.02656-06
- 174. Fierer, N., Jackson, J.A., Vilgalys, R., and Jackson, R.B. (2005) Assessment of soil microbial community structure by use of taxon-specific quantitative PCR assays. *Applied and environmental microbiology* **71**, 4117-20. doi: 10.1128/aem.71.7.4117-4120.2005
- Lane, D.J. (1991) "16S/23S rRNA sequencing," in *Nucleic acid techniques in bacterial systematics*, ed. M.G. E. Stackebrandt. (Chichester, New York: John Wiley & Sons), 115-75.
- 176. **Muyzer, G., de Waal, E.C., and Uitterlinden, A.G.** (1993) Profiling of complex microbial populations by denaturing gradient gel electrophoresis analysis of polymerase chain reaction-amplified genes coding for 16S rRNA. *Applied and environmental microbiology* **59**, 695-700.
- 177. Tang, H., Krishnakumar, V., Bidwell, S., Rosen, B., Chan, A., Zhou, S., Gentzbittel, L., Childs, K.L., Yandell, M., Gundlach, H., Mayer, K.F.X., Schwartz, D.C., and Town, C.D. (2014) An improved genome release (version Mt4.0) for the model legume Medicago truncatula. *BMC Genomics* 15, 312. doi: 10.1186/1471-2164-15-312

- 178. Moll, K.M., Zhou, P., Ramaraj, T., Fajardo, D., Devitt, N.P., Sadowsky, M.J., Stupar, R.M., Tiffin, P., Miller, J.R., Young, N.D., Silverstein, K.A.T., and Mudge, J. (2017) Strategies for optimizing BioNano and Dovetail explored through a second reference quality assembly for the legume model, Medicago truncatula. *BMC Genomics* 18, 578. doi: 10.1186/s12864-017-3971-4
- 179. Frazee, A.C., Pertea, G., Jaffe, A.E., Langmead, B., Salzberg, S.L., and Leek, J.T. (2015) Ballgown bridges the gap between transcriptome assembly and expression analysis. *Nature Biotechnology* **33**, 243. doi: 10.1038/nbt.3172
- 180. Grabherr, M.G., Haas, B.J., Yassour, M., Levin, J.Z., Thompson, D.A., Amit, I., Adiconis, X., Fan, L., Raychowdhury, R., Zeng, Q., Chen, Z., Mauceli, E., Hacohen, N., Gnirke, A., Rhind, N., di Palma, F., Birren, B.W., Nusbaum, C., Lindblad-Toh, K., Friedman, N., and Regev, A. (2011) Full-length transcriptome assembly from RNA-Seq data without a reference genome. *Nature Biotechnology* 29. doi: 10.1038/nbt.1883
- 181. Kriventseva, E.V., Zdobnov, E.M., Simão, F.A., Ioannidis, P., and Waterhouse, R.M. (2015) BUSCO: assessing genome assembly and annotation completeness with single-copy orthologs. *Bioinformatics* **31**, 3210-2. doi: 10.1093/bioinformatics/btv351
- 182. Bray, N.L., Pimentel, H., Melsted, P., and Pachter, L. (2016) Nearoptimal probabilistic RNA-seq quantification. *Nature Biotechnology* 34, 525. doi: 10.1038/nbt.3519
- 183. **Powell, D.R.** (2014) *Degust (formerly DGE-Vis) : Visualize, explore and appreciate RNA-seq differential gene-expression data.* [Online]. Monash University, Australia: Victorian bioinformatics consortium. Available: <u>http://degust.erc.monash.edu/</u> [Accessed 09/07/2019].
- 184. Law, C.W., Chen, Y., Shi, W., and Smyth, G.K. (2014) voom: Precision weights unlock linear model analysis tools for RNA-seq read counts. *Genome biology* 15, R29. doi: 10.1186/gb-2014-15-2-r29
- Conesa, A., Götz, S., García-Gómez, J.M., Terol, J., Talón, M., and Robles, M. (2005) Blast2GO: A universal tool for annotation, visualization and analysis in functional genomics research. *Bioinformatics* 21. doi: 10.1093/bioinformatics/bti610
- 186. Gotz, S., Garcia-Gomez, J.M., Terol, J., Williams, T.D., Nagaraj, S.H., Nueda, M.J., Robles, M., Talon, M., Dopazo, J., and Conesa, A. (2008) High-throughput functional annotation and data mining with the Blast2GO suite. *Nucleic Acids Research* **36**, 3420-35. doi: 10.1093/nar/gkn176
- 187. **Pruitt, K.D., Tatusova, T., and Maglott, D.R.** (2005) NCBI Reference Sequence (RefSeq): a curated non-redundant sequence database of genomes, transcripts and proteins. *Nucleic Acids Research* **33**, D501-D4. doi: 10.1093/nar/gki025
- 188. Mitchell, A.L., Attwood, T.K., Babbitt, P.C., Blum, M., Bork, P., Bridge, A., Brown, S.D., Chang, H.Y., El-Gebali, S., Fraser, M.I., Gough, J., Haft, D.R., Huang, H., Letunic, I., Lopez, R., Luciani, A., Madeira, F., Marchler-Bauer, A., Mi, H., Natale, D.A., Necci, M., Nuka, G., Orengo, C., Pandurangan, A.P., Paysan-Lafosse, T., Pesseat, S., Potter, S.C., Qureshi, M.A., Rawlings, N.D., Redaschi, N., Richardson, L.J., Rivoire, C., Salazar, G.A., Sangrador-Vegas, A., Sigrist, C.J.A., Sillitoe, I., Sutton, G.G., Thanki, N., Thomas, P.D., Tosatto, S.C.E., Yong, S.Y., and Finn, R.D. (2019) InterPro in 2019: improving coverage, classification and access to protein sequence annotations. *Nucleic Acids Research* 47, D351-d60. doi: 10.1093/nar/gky1100

- 189. Carbon, S., Ireland, A., Mungall, C.J., Shu, S., Marshall, B., Lewis, S., AmiGO Hub, and Web Presence Working Group (2009) AmiGO: online access to ontology and annotation data. *Bioinformatics* 25, 288-9. doi: 10.1093/bioinformatics/btn615
- 190. The Gene Ontology Consortium, Ashburner, M., Ball, C.A., Blake, J.A., Botstein, D., Butler, H., Cherry, J.M., Davis, A.P., Dolinski, K., Dwight, S.S., Eppig, J.T., Harris, M.A., Hill, D.P., Issel-Tarver, L., Kasarskis, A., Lewis, S., Matese, J.C., Richardson, J.E., Ringwald, M., Rubin, G.M., and Sherlock, G. (2000) Gene ontology: tool for the unification of biology. Nat Genet 25, 25-9. doi: 10.1038/75556
- The Gene Ontology Consortium (2019) The Gene Ontology Resource: 20 years and still GOing strong. *Nucleic Acids Research* 47, D330-d8. doi: 10.1093/nar/gky1055
- 192. Camacho, C., Coulouris, G., Avagyan, V., Ma, N., Papadopoulos, J., Bealer, K., and Madden, T.L. (2009) BLAST+: architecture and applications. *BMC Bioinformatics* **10**, 421. doi: 10.1186/1471-2105-10-421
- 193. Benson, D.A., Karsch-Mizrachi, I., Lipman, D.J., Ostell, J., and Wheeler, D.L. (2005) GenBank. *Nucleic Acids Research* **33**, D34-D8. doi: 10.1093/nar/gki063
- 194. Benson, D.A., Cavanaugh, M., Clark, K., Karsch-Mizrachi, I., Lipman, D.J., Ostell, J., and Sayers, E.W. (2013) GenBank. *Nucleic Acids Research* **41**, D36-D42. doi: 10.1093/nar/gks1195
- 195. Binns, D., Dimmer, E., Huntley, R., Barrell, D., O'Donovan, C., and Apweiler, R. (2009) QuickGO: a web-based tool for Gene Ontology searching. *Bioinformatics* 25, 3045-6. doi: 10.1093/bioinformatics/btp536
- 196. **The UniProt Consortium** (2018) UniProt: a worldwide hub of protein knowledge. *Nucleic Acids Research* **47**, D506-D15. doi: 10.1093/nar/gky1049
- 197. **Thomas, P.D., Campbell, M.J., Kejariwal, A., Mi, H., Karlak, B., Daverman, R., Diemer, K., Muruganujan, A., and Narechania, A.** (2003) PANTHER: a library of protein families and subfamilies indexed by function. *Genome Research* **13**, 2129-41. doi: 10.1101/gr.772403
- 198. **Mi, H., Dong, Q., Muruganujan, A., Gaudet, P., Lewis, S., and Thomas, P.D.** (2010) PANTHER version 7: improved phylogenetic trees, orthologs and collaboration with the Gene Ontology Consortium. *Nucleic Acids Research* **38**, D204-10. doi: 10.1093/nar/gkp1019
- 199. Thomas, P.D., Kejariwal, A., Guo, N., Mi, H., Campbell, M.J., Muruganujan, A., and Lazareva-Ulitsky, B. (2006) Applications for protein sequence-function evolution data: mRNA/protein expression analysis and coding SNP scoring tools. *Nucleic Acids Research* **34**, W645-50. doi: 10.1093/nar/gkl229
- 200. **Parsons, A.J., Harvey, A., and Woledge, J.** (1991) Plant-Animal Interactions in a Continuously Grazed Mixture. I. Differences in the Physiology of Leaf Expansion and the Fate of Leaves of Grass and Clover. *Journal of Applied Ecology* **28**, 619-34. doi: 10.2307/2404572
- Shewry, P.R., and Halford, N.G. (2002) Cereal seed storage proteins: structures, properties and role in grain utilization. *Journal of Experimental Botany* 53, 947-58. doi: 10.1093/jexbot/53.370.947
- 202. Pedrazzini, E., Mainieri, D., Marrano, C.A., and Vitale, A. (2016) Where do Protein Bodies of Cereal Seeds Come From? *Frontiers in Plant Science* 7. doi: 10.3389/fpls.2016.01139
- 203. Piccini, C., Di Bene, C., Farina, R., Pennelli, B., and Napoli, R. (2016) Assessing Nitrogen Use Efficiency and Nitrogen Loss in a Forage-Based System Using a Modeling Approach. *Agronomy* 6, 23. doi: 10.3390/agronomy6020023

- 204. **Lehmeier, C.A., Wild, M., and Schnyder, H.** (2013) Nitrogen Stress Affects the Turnover and Size of Nitrogen Pools Supplying Leaf Growth in a Grass. *Plant Physiology* **162** 2095-105. doi: 10.1104/pp.113.219311
- 205. Hofer, D., Suter, M., Buchmann, N., and Luscher, A. (2017) Nitrogen status of functionally different forage species explains resistance to severe drought and post-drought overcompensation. *Agriculture, Ecosystems & Environment* 236, 312-22. doi: 10.1016/j.agee.2016.11.022
- 206. Liu, Y., Yang, X., Tian, D., Cong, R., Zhang, X., Pan, Q., and Shi, Z. (2018) Resource Reallocation of Two Grass Species During Regrowth After Defoliation. *Frontiers in Plant Science* 9, 1767-. doi: 10.3389/fpls.2018.01767
- Jaradat, A.A., Johnson, J.M.F., Weyers, S.L., and Barbour, N.W. (2009) Determinants and prediction of carbon/nitrogen ratio in five diverse crop plants. *Communications in Soil Science and Plant Analysis* 40, 2688-711. doi: 10.1080/00103620903173756
- 208. Molero, G., Tcherkez, G., Araus, J.L., Nogués, S., and Aranjuelo, I. (2014) On the relationship between C and N fixation and amino acid synthesis in nodulated alfalfa *Medicago sativa*. *Functional Plant Biology* 41, 331-41. doi: 10.1071/FP13189
- 209. Meuriot, F., Morvan-Bertrand, A., Noiraud-Romy, N., Decau, M.-L., Escobar-Gutiérrez, A.J., Gastal, F., and Prud'homme, M.-P. (2018) Short-term effects of defoliation intensity on sugar remobilization and N fluxes in ryegrass. *Journal of Experimental Botany* 69, 3975-86. doi: 10.1093/jxb/ery211
- 210. **Muntz, K.** (1998) Deposition of storage proteins. *Plant Mol Biol* **38**, 77-99. doi: 10.1023/A:1006020208380
- 211. Avice, J.C., Dily, F.L., Goulas, E., Noquet, C., Meuriot, F., Volenec, J.J., Cunningham, S.M., Sors, T.G., Dhont, C., Castonguay, Y., Nadeau, P., Bélanger, G., Chalifour, F.P., and Ourry, A. (2003) Vegetative storage proteins in overwintering storage organs of forage legumes: roles and regulation. *Canadian Journal of Botany* **81**, 1198-212. doi: 10.1139/b03-122
- 212. **Ourry, A., Bigot, J., and Boucaud, J.** (1989) Protein Mobilization from Stubble and Roots, and Proteolytic Activities during Post-clipping Regrowth of Perennial Ryegrass. *Journal of Plant Physiology* **134**, 298-303. doi: 10.1016/S0176-1617(89)80246-9
- Ourry, A., Kim, T.H., and Boucaud, J. (1994) Nitrogen Reserve Mobilization during Regrowth of *Medicago sativa* L. (Relationships between Availability and Regrowth Yield). *Plant Physiology* **105**, 831-7. doi: 10.1104/pp.105.3.831
- 214. **Bewley, J.D.** (2002) Root storage proteins, with particular reference to taproots. *Canadian Journal of Botany* **80**, 321-9. doi: 10.1139/b02-025
- 215. Byrne, S.L., Nagy, I., Pfeifer, M., Armstead, I., Swain, S., Studer, B., Mayer, K., Campbell, J.D., Czaban, A., Hentrup, S., Panitz, F., Bendixen, C., Hedegaard, J., Caccamo, M., and Asp, T. (2015) A synteny-based draft genome sequence of the forage grass *Lolium perenne*. *The Plant Journal* 84, 816-26. doi: 10.1111/tpj.13037
- 216. The International Brachypodium Initiative, Vogel, J.P., Garvin, D.F., Mockler, T.C., Schmutz, J., Rokhsar, D., Bevan, M.W., Barry, K., Lucas, S., Harmon-Smith, M., Lail, K., Tice, H., Schmutz, J., Grimwood, J., McKenzie, N., Bevan, M.W., Huo, N., Gu, Y.Q., Lazo, G.R., Anderson, O.D., Vogel, J.P., You, F.M., Luo, M.-C., Dvorak, J., Wright, J., Febrer, M., Bevan, M.W., Idziak, D., Hasterok, R., Garvin, D.F., Lindquist, E., Wang, M., Fox, S.E., Priest, H.D., Filichkin, S.A., Givan, S.A., Bryant, D.W., Chang, J.H., Mockler, T.C., Wu, H., Wu, W., Hsia, A.-P., Schnable, P.S., Kalyanaraman,

A., Barbazuk, B., Michael, T.P., Hazen, S.P., Bragg, J.N., Laudencia-Chingcuanco, D., Vogel, J.P., Garvin, D.F., Weng, Y., McKenzie, N., Bevan, M.W., Haberer, G., Spannagl, M., Mayer, K., Rattei, T., Mitros, T., Rokhsar, D., Lee, S.-J., Rose, J.K.C., Mueller, L.A., York, T.L., Wicker, T., Buchmann, J.P., Tanskanen, J., Schulman, A.H., Gundlach, H., Wright, J., Bevan, M., Costa de Oliveira, A., da C. Maia, L., Belknap, W., Gu, Y.Q., Jiang, N., Lai, J., Zhu, L., Ma, J., Sun, C., Pritham, E., Salse, J., Murat, F., Abrouk, M., Haberer, G., Spannagl, M., Mayer, K., Bruggmann, R., Messing, J., You, F.M., Luo, M.-C., Dvorak, J., Fahlgren, N., Fox, S.E., Sullivan, C.M., Mockler, T.C., Carrington, J.C., Chapman, E.J., May, G.D., Zhai, J., et al. (2010) Genome sequencing and analysis of the model grass *Brachypodium distachyon*. *Nature* 463, 763. doi: 10.1038/nature08747

- Okamoto, M., Vidmar, J.J., and Glass, A.D.M. (2003) Regulation of NRT1 and NRT2 Gene Families of Arabidopsis thaliana: Responses to Nitrate Provision. *Plant and Cell Physiology* 44, 304-17. doi: 10.1093/pcp/pcg036
- 218. **Huang, N.C., Liu, K.H., Lo, H.J., and Tsay, Y.F.** (1999) Cloning and functional characterization of an Arabidopsis nitrate transporter gene that encodes a constitutive component of low-affinity uptake. *Plant Cell* **11**, 1381-92. doi: 10.1105/tpc.11.8.1381
- Léran, S., Edel, K.H., Pervent, M., Hashimoto, K., Corratgé-Faillie, C., Offenborn, J.N., Tillard, P., Gojon, A., Kudla, J., and Lacombe, B. (2015) Nitrate sensing and uptake in *Arabidopsis* are enhanced by ABI2, a phosphatase inactivated by the stress hormone abscisic acid. *Science Signaling* 8, ra43-ra. doi: 10.1126/scisignal.aaa4829
- 220. Ho, C.H., Lin, S.H., Hu, H.C., and Tsay, Y.F. (2009) CHL1 functions as a nitrate sensor in plants. *Cell* **138**, 1184-94. doi: 10.1016/j.cell.2009.07.004
- Bouguyon, E., Brun, F., Meynard, D., Kubeš, M., Pervent, M., Leran, S., Lacombe, B., Krouk, G., Guiderdoni, E., and Zažímalová, E. (2015) Multiple mechanisms of nitrate sensing by *Arabidopsis* nitrate transceptor NRT1. 1. *Nat Plants* 1. doi: 10.1038/nplants.2015.15
- 222. **Parker, J.L., and Newstead, S.** (2014) Molecular basis of nitrate uptake by the plant nitrate transporter NRT1.1. *Nature* **507,** 68-72. doi: 10.1038/nature13116
- 223. Morere-Le Paven, M.C., Viau, L., Hamon, A., Vandecasteele, C., Pellizzaro, A., Bourdin, C., Laffont, C., Lapied, B., Lepetit, M., Frugier, F., Legros, C., and Limami, A.M. (2011) Characterization of a dual-affinity nitrate transporter MtNRT1.3 in the model legume *Medicago truncatula*. *Journal of Experimental Botany* **62**, 5595-605. doi: 10.1093/jxb/err243
- 224. Tang, Z., Fan, X., Li, Q., Feng, H., Miller, A.J., Shen, Q., and Xu, G. (2012) Knockdown of a rice stelar nitrate transporter alters long-distance translocation but not root influx. *Plant Physiology* 160, 2052-63. doi: 10.1104/pp.112.204461
- 225. Garnett, T., Conn, V., Plett, D., Conn, S., Zanghellini, J., Mackenzie, N., Enju, A., Francis, K., Holtham, L., Roessner, U., Boughton, B., Bacic, A., Shirley, N., Rafalski, A., Dhugga, K., Tester, M., and Kaiser, B.N. (2013) The response of the maize nitrate transport system to nitrogen demand and supply across the lifecycle. *New Phytologist* **198**, 82-94. doi: 10.1111/nph.12166
- 226. Guo, T., Xuan, H., Yang, Y., Wang, L., Wei, L., Wang, Y., and Kang, G. (2014) Transcription Analysis of Genes Encoding the Wheat Root Transporter NRT1 and NRT2 Families During Nitrogen Starvation. *Journal* of Plant Growth Regulation 33, 837-48. doi: 10.1007/s00344-014-9435-z

- 227. Li, W., Wang, Y., Okamoto, M., Crawford, N.M., Siddiqi, M.Y., and Glass, A.D.M. (2007) Dissection of the *AtNRT2.1 AtNRT2.2* Inducible High-Affinity Nitrate Transporter Gene Cluster. *Plant Physiology* **143**, 425. doi: 10.1104/pp.106.091223
- 228. **Guo, F.Q., Wang, R., Chen, M., and Crawford, N.M.** (2001) The Arabidopsis dual-affinity nitrate transporter gene AtNRT1.1 (CHL1) is activated and functions in nascent organ development during vegetative and reproductive growth. *Plant Cell* **13**, 1761-77. doi: 10.11054/tpc.010126
- 229. Filleur, S., Walch-Liu, P., Gan, Y., and Forde, B.G. (2005) Nitrate and glutamate sensing by plant roots. *Biochem Soc Trans* **33**, 283-6. doi: 10.1042/bst0330283
- 230. **Sakakibara, H., Takei, K., and Hirose, N.** (2006) Interactions between nitrogen and cytokinin in the regulation of metabolism and development. *Trends Plant Sci* **11,** 440-8. doi: 10.1016/j.tplants.2006.07.004
- 231. **Kiba, T., Kudo, T., Kojima, M., and Sakakibara, H.** (2011) Hormonal control of nitrogen acquisition: roles of auxin, abscisic acid, and cytokinin. *Journal of Experimental Botany* **62**, 1399-409. doi: 10.1093/jxb/erq410
- 232. **Merewitz, E.B., Gianfagna, T., and Huang, B.** (2011) Photosynthesis, water use, and root viability under water stress as affected by expression of SAG12-ipt controlling cytokinin synthesis in *Agrostis stolonifera*. *Journal of Experimental Botany* **62**, 383-95. doi: 10.1093/jxb/erg285
- 233. Nacry, P., Bouguyon, E., and Gojon, A. (2013) Nitrogen acquisition by roots: physiological and developmental mechanisms ensuring plant adaptation to a fluctuating resource. *Plant and Soil* **370**, 1-29. doi: 10.1007/s11104-013-1645-9
- 234. Lin, S.H., Kuo, H.F., Canivenc, G., Lin, C.S., Lepetit, M., Hsu, P.K., Tillard, P., Lin, H.L., Wang, Y.Y., Tsai, C.B., Gojon, A., and Tsay, Y.F. (2008) Mutation of the Arabidopsis NRT1.5 nitrate transporter causes defective root-to-shoot nitrate transport. *Plant Cell* **20**, 2514-28. doi: 10.1105/tpc.108.060244
- 235. Fan, S.-C., Lin, C.-S., Hsu, P.-K., Lin, S.-H., and Tsay, Y.-F. (2009) The *Arabidopsis* Nitrate Transporter NRT1.7, Expressed in Phloem, Is Responsible for Source-to-Sink Remobilization of Nitrate. *Plant Cell* **21**, 2750. doi: 10.1105/tpc.109.067603
- 236. Li, J.Y., Fu, Y.L., Pike, S.M., Bao, J., Tian, W., Zhang, Y., Chen, C.Z., Zhang, Y., Li, H.M., Huang, J., Li, L.G., Schroeder, J.I., Gassmann, W., and Gong, J.M. (2010) The *Arabidopsis* nitrate transporter NRT1.8 functions in nitrate removal from the xylem sap and mediates cadmium tolerance. *Plant Cell* 22, 1633-46. doi: 10.1105/tpc.110.075242
- 237. Plett, D., Toubia, J., Garnett, T., Tester, M., Kaiser, B.N., and Baumann, U. (2010) Dichotomy in the NRT Gene Families of Dicots and Grass Species. *PLoS One* **5**, e15289. doi: 10.1371/journal.pone.0015289
- 238. Bagchi, R., Salehin, M., Adeyemo, O.S., Salazar, C., Shulaev, V., Sherrier, D.J., and Dickstein, R. (2012) Functional Assessment of the *Medicago truncatula* NIP/LATD Protein Demonstrates That It Is a High-Affinity Nitrate Transporter. *Plant Physiology* **160**, 906. doi: 10.1104/pp.112.196444
- 239. Hu, B., Wang, W., Ou, S., Tang, J., Li, H., Che, R., Zhang, Z., Chai, X., Wang, H., Wang, Y., Liang, C., Liu, L., Piao, Z., Deng, Q., Deng, K., Xu, C., Liang, Y., Zhang, L., Li, L., and Chu, C. (2015) Variation in NRT1.1B contributes to nitrate-use divergence between rice subspecies. Nat Genet 47, 834-8. doi: 10.1038/ng.3337
- 240. **Taulemesse, F., Le Gouis, J., Gouache, D., Gibon, Y., and Allard, V.** (2015) Post-flowering nitrate uptake in wheat is controlled by N status at
flowering, with a putative major role of root nitrate transporter NRT2.1. *PLoS One* **10**, e0120291. doi: 10.1371/journal.pone.0120291

- 241. Plett, D., Holtham, L., Baumann, U., Kalashyan, E., Francis, K., Enju, A., Toubia, J., Roessner, U., Bacic, A., Rafalski, A., Dhugga, K.S., Tester, M., Garnett, T., and Kaiser, B.N. (2016) Nitrogen assimilation system in maize is regulated by developmental and tissuespecific mechanisms. *Plant Mol Biol* **92**, 293-312. doi: 10.1007/s11103-016-0512-5
- 242. Wang, W., Hu, B., Yuan, D., Liu, Y., Che, R., Hu, Y., Ou, S., Liu, Y., Zhang, Z., Wang, H., Li, H., Jiang, Z., Zhang, Z., Gao, X., Qiu, Y., Meng, X., Liu, Y., Bai, Y., Liang, Y., Wang, Y., Zhang, L., Li, L., Sodmergen, Jing, H., Li, J., and Chu, C. (2018) Expression of the Nitrate Transporter Gene OsNRT1.1A/OsNPF6.3 Confers High Yield and Early Maturation in Rice. *Plant Cell* **30**, 638-51. doi: 10.1105/tpc.17.00809
- 243. David, L.C., Girin, T., Fleurisson, E., Phommabouth, E., Mahfoudhi, A., Citerne, S., Berquin, P., Daniel-Vedele, F., Krapp, A., and Ferrario-Méry, S. (2019) Developmental and physiological responses of *Brachypodium distachyon* to fluctuating nitrogen availability. *Scientific Reports* **9**, 3824. doi: 10.1038/s41598-019-40569-8
- 244. **Wang, J., Huner, N., and Tian, L.** (2019) Identification and molecular characterization of the *Brachypodium distachyon* NRT2 family, with a major role of BdNRT2.1. *Physiologia Plantarum* **165,** 498-510. doi: 10.1111/ppl.12716
- 245. **Orsel, M., Filleur, S., Fraisier, V., and Daniel-Vedele, F.** (2002) Nitrate transport in plants: which gene and which control? *Journal of Experimental Botany* **53**, 825-33. doi: 10.1093/jexbot/53.370.825
- 246. **Smirnoff, N., and Stewart, G.R.** (1985) Nitrate assimilation and translocation by higher plants: Comparative physiology and ecological consequences. *Physiologia Plantarum* **64,** 133-40. doi: 10.1111/j.1399-3054.1985.tb02326.x
- 247. **Rasmussen, S., Parsons, A.J., Fraser, K., Xue, H., and Newman, J.A.** (2008) Metabolic profiles of *Lolium perenne* are differentially affected by nitrogen supply, carbohydrate content, and fungal endophyte infection. *Plant Physiology* **146**, 1440. doi: 10.1104/pp.107.111898
- 248. Pan, L., Meng, C., Wang, J., Ma, X., Fan, X., Yang, Z., Zhou, M., and Zhang, X. (2018) Integrated omics data of two annual ryegrass (*Lolium multiflorum* L.) genotypes reveals core metabolic processes under drought stress. *BMC Plant Biology* 18, 26. doi: 10.1186/s12870-018-1239-z
- 249. Swarbreck, S.M., Defoin-Platel, M., Hindle, M., Saqi, M., and Habash, D.Z. (2011) New perspectives on glutamine synthetase in grasses. *Journal of Experimental Botany*
- 62, 1511-22. doi: 10.1093/jxb/erq356
- 250. **Jiang, Y., Li, Y., Nie, G., and Liu, H.** (2016) Leaf and Root Growth, Carbon and Nitrogen Contents, and Gene Expression of Perennial Ryegrass to Different Nitrogen Supplies. *Journal of the American Society for Horticultural Science* **141**, 555-62. doi: 10.21273/jashs03883-16
- 251. **Brunharo, C.A.C.G., Takano, H.K., Mallory-Smith, C.A., Dayan, F.E., and Hanson, B.D.** (2019) Role of glutamine synthetase isogenes and herbicide metabolism in the mechanism of resistance to glufosinate in *Lolium perenne* L. spp. *multiflorum* biotypes from Oregon. *Journal of Agricultural and Food Chemistry*. doi: 10.1021/acs.jafc.9b01392
- 252. Orea, A., Pajuelo, P., Pajuelo, E., Quidiello, C., Romero, J.M., and Marquez, A.J. (2002) Isolation of photorespiratory mutants from *Lotus japonicus* deficient in glutamine synthetase. *Physiologia Plantarum* **115**, 352-61. doi: 10.1034/j.1399-3054.2002.1150304.x

- 253. Lacuesta, M., Gonzalez-Moro, B., Gonzalez-Murua, C., and Muñoz-Rueda, A. (1990) Temporal study of the effect of phosphinothricin on the activity of glutamine synthetase, glutamate dehydrogenase and nitrate reductase in *Medicago sativa* L. *Journal of Plant Physiology* **136**, 410-4. doi: 10.1016/S0176-1617(11)80028-3
- 254. Ortega, J.L., Moguel-Esponda, S., Potenza, C., Conklin, C.F., Quintana, A., and Sengupta-Gopalan, C. (2006) The 3' untranslated region of a soybean cytosolic *glutamine synthetase* (*GS1*) affects transcript stability and protein accumulation in transgenic alfalfa. *The Plant Journal* **45**, 832-46. doi: 10.1111/j.1365-313X.2005.02644.x
- 255. **Yang, S., Zu, Y., Li, B., Bi, Y., Jia, L., He, Y., and Li, Y.** (2019) Response and intraspecific differences in nitrogen metabolism of alfalfa (*Medicago sativa* L.) under cadmium stress. *Chemosphere* **220,** 69-76. doi: 10.1016/j.chemosphere.2018.12.101
- 256. Glevarec, G., Bouton, S., Jaspard, E., Riou, M.T., Cliquet, J.B., Suzuki, A., and Limami, A.M. (2004) Respective roles of the glutamine synthetase/glutamate synthase cycle and glutamate dehydrogenase in ammonium and amino acid metabolism during germination and postgerminative growth in the model legume *Medicago truncatula*. *Planta* **219**, 286-97. doi: 10.1007/s00425-004-1214-9
- 257. **Seabra, A.R., and Carvalho, H.G.** (2015) Glutamine synthetase in *Medicago truncatula*, unveiling new secrets of a very old enzyme. *Frontiers in Plant Science* **6**. doi: 10.3389/fpls.2015.00578
- 258. Swarbreck, S.M., Sudderth, E.A., St.Clair, S.B., Salve, R., Castanha, C., Torn, M.S., Ackerly, D.D., and Andersen, G.L. (2011) Linking leaf transcript levels to whole plant analyses provides mechanistic insights to the impact of warming and altered water availability in an annual grass. *Global Change Biology* **17**, 1577-94. doi: 10.1111/j.1365-2486.2010.02359.x
- 259. **Guo, Q., Turnbull, M.H., Song, J., Roche, J., Novak, O., Späth, J., Jameson, P.E., and Love, J.** (2017) Depletion of carbohydrate reserves limits nitrate uptake during early regrowth in *Lolium perenne* L. *Journal of Experimental Botany* **68**, 1569-83. doi: 10.1093/jxb/erx056
- 260. Roche, J., Turnbull, M., Guo, Q., Novak, O., Späth, J., Gieseg, S., Jameson, P., and Love, J. (2017) Coordinated nitrogen and carbon remobilization for nitrate assimilation in leaf, sheath and root and associated cytokinin signals during early regrowth of Lolium perenne.
- 261. Gupta, N., Gupta, A.K., Gaur, V.S., and Kumar, A. (2012) Relationship of nitrogen use efficiency with the activities of enzymes involved in nitrogen uptake and assimilation of finger millet genotypes grown under different nitrogen inputs. *The Scientific World Journal* 2012, 10. doi: 10.1100/2012/625731
- 262. **Chu, X., Yang, L., Wang, S., Liu, J., and Yang, H.** (2017) Physiological and metabolic profiles of common reed provide insights into plant adaptation to low nitrogen conditions. *Biochemical Systematics and Ecology* **73**, 3-10. doi: 10.1016/j.bse.2017.04.003
- 263. Betti, M., García-Calderón, M., Pérez-Delgado, C.M., Credali, A., Estivill, G., Galván, F., Vega, J.M., and Márquez, A.J. (2012) Glutamine synthetase in legumes: recent advances in enzyme structure and functional genomics. *Int J Mol Sci* 13, 7994-8024. doi: 10.3390/ijms13077994
- 264. **Younesi, O., and Moradi, A.** (2015) Effect of salinity on nodulation, glutamine synthetase and glutamate synthase activity in nodules of alfalfa (*Medicago sativa* L.). *Cercetari Agronomice in Moldova* **48,** 61-70. doi: 10.1515/cerce-2015-0053
- 265. Limami, A.M., Glevarec, G., Ricoult, C., Cliquet, J.B., and Planchet, E. (2008) Concerted modulation of alanine and glutamate metabolism in

young *Medicago truncatula* seedlings under hypoxic stress. *Journal of Experimental Botany* **59**, 2325-35. doi: 10.1093/jxb/ern102

- 266. **Miller, A.J., and Cramer, M.D.** (2005) Root nitrogen acquisition and assimilation. *Plant and Soil* **274,** 1-36. doi: 10.1007/s11104-004-0965-1
- 267. Stitt, M., Müller, C., Matt, P., Gibon, Y., Carillo, P., Morcuende, R., Scheible, W.R., and Krapp, A. (2002) Steps towards an integrated view of nitrogen metabolism. *Journal of Experimental Botany* **53**, 959-70. doi: 10.1093/jexbot/53.370.959
- 268. Kaiser, W.M., Weiner, H., and Huber, S.C. (1999) Nitrate reductase in higher plants: A case study for transduction of environmental stimuli into control of catalytic activity. *Physiologia Plantarum* **105**, 384-9. doi: 10.1034/j.1399-3054.1999.105225.x
- 269. Scheible, W.-R., Krapp, A., and Stitt, M. (2000) Reciprocal diurnal changes of phosphoenolpyruvate carboxylase expression and cytosolic pyruvate kinase, citrate synthase and NADP-isocitrate dehydrogenase expression regulate organic acid metabolism during nitrate assimilation in tobacco leaves. *Plant, Cell & Environment* **23**, 1155-67. doi: 10.1046/j.1365-3040.2000.00634.x
- 270. Huang, L., Zhang, H., Zhang, H., Deng, X.W., and Wei, N. (2015) HY5 regulates nitrite reductase 1 (NIR1) and ammonium transporter1;2 (AMT1;2) in Arabidopsis seedlings. Plant Science 238, 330-9. doi: 10.1016/j.plantsci.2015.05.004
- Kaiser, W.M., and Huber, S.C. (2001) Post-translational regulation of nitrate reductase: mechanism, physiological relevance and environmental triggers. *Journal of Experimental Botany* 52, 1981-9. doi: 10.1093/jexbot/52.363.1981
- 272. **MacKintosh, C., and Meek, S.E.** (2001) Regulation of plant NR activity by reversible phosphorylation, 14-3-3 proteins and proteolysis. *Cellular and Molecular Life Sciences* **58**, 205-14. doi: 10.1007/pl00000848
- 273. Foyer, C.H., Parry, M., and Noctor, G. (2003) Markers and signals associated with nitrogen assimilation in higher plants. *Journal of Experimental Botany* 54, 585-93. doi: 10.1093/jxb/erg053
- 274. Goodman, P.J., Fothergill, M., and Hughes, D.M. (1974) Variation in Nitrate Reductase, Nitrite, and Nitrite Reductase in Some Grasses and Cereals. Annals of Botany 38, 31-7. doi: 10.1093/oxfordjournals.aob.a084798
- 275. **Goodman, P.J., Hughes, D.M., Fothergill, M., and Ellis, P.B.** (1974) Genetic variation in the activity of nitrate reductase and nitrite reductase in *Lolium. Plant and Soil* **41**, 435-43. doi: 10.1007/BF02185807
- 276. **Middleton, K.R., and Smith, G.S.** (1979) A comparison of ammoniacal and nitrate nutrition of perennial ryegrass through a thermodynamic model. *Plant and Soil* **53**, 487-504. doi: 10.1007/BF02140720
- 277. Foito, A., Byrne, S.L., Hackett, C.A., Hancock, R.D., Stewart, D., and Barth, S. (2013) Short-term response in leaf metabolism of perennial ryegrass (Lolium perenne) to alterations in nitrogen supply. *Metabolomics* **9**, 145-56. doi: 10.1007/s11306-012-0435-3
- 278. Vázquez, M., Barea, J., and Azcón, R. (2001) Impact of soil nitrogen concentration on *Glomus* spp.-*Sinorhizobium* interactions as affecting growth, nitrate reductase activity and protein content of *Medicago sativa*. *Biology and Fertility of Soils* **34**, 57-63. doi: 10.1007/s003740100373
- 279. Horchani, F., Prévot, M., Boscari, A., Evangelisti, E., Meilhoc, E., Bruand, C., Raymond, P., Boncompagni, E., Aschi-Smiti, S., Puppo, A., and Brouquisse, R. (2011) Both plant and bacterial nitrate reductases contribute to nitric oxide production in *Medicago truncatula* nitrogen-fixing nodules. *Plant Physiology* **155**, 1023. doi: 10.1104/pp.110.166140

- 280. Yang, H., Unkovich, M., McNeill, A., and Wang, X. (2011) Symbiotic N₂ fixation and nitrate utilisation in irrigated lucerne (*Medicago sativa*) systems. *Biology and Fertility of Soils* 47, 377-85. doi: 10.1007/s00374-011-0543-7
- 281. **Feller, U., Anders, I., and Mae, T.** (2008) Rubiscolytics: fate of Rubisco after its enzymatic function in a cell is terminated. *Journal of Experimental Botany* **59**, 1615-24. doi: 10.1093/jxb/erm242
- 282. Lawlor, D.W., Boyle, F.A., Kendall, A.C., and Keys, A.J. (1987) Nitrate nutrition and temperature effects on wheat: enzyme composition, nitrate and total amino acid content of leaves. *Journal of Experimental Botany* **38**, 378-92. doi: 10.1093/jxb/38.3.378
- 283. Ishida, H., Yoshimoto, K., Izumi, M., Reisen, D., Yano, Y., Makino, A., Ohsumi, Y., Hanson, M.R., and Mae, T. (2008) Mobilization of rubisco and stroma-localized fluorescent proteins of chloroplasts to the vacuole by an ATG gene-dependent autophagic process. *Plant Physiology* 148, 142-55. doi: 10.1104/pp.108.122770
- 284. Schultz, J.C., Appel, H.M., Ferrieri, A.P., and Arnold, T.M. (2013) Flexible resource allocation during plant defense responses. *Frontiers in Plant Science* **4**, 324. doi: 10.3389/fpls.2013.00324
- 285. **Sinha, N.R., and Kellogg, E.A.** (1996) Parallelism and diversity in multiple origins of c4 photosynthesis in the grass family. *American Journal of Botany* **83,** 1458-70. doi: 10.1002/j.1537-2197.1996.tb13940.x
- 286. **Ewing, R.M., Jenkins, G.I., and Langdale, J.A.** (1998) Transcripts of maize *RbcS* genes accumulate differentially in C3 and C4 tissues. *Plant Mol Biol* **36,** 593-9. doi; 10.1023/a:1005947306667
- 287. **Kubien, D.S., and Sage, R.F.** (2004) Low-temperature photosynthetic performance of a C4 grass and a co-occurring C3 grass native to high latitudes. *Plant, Cell & Environment* **27,** 907-16. doi: 10.1111/j.1365-3040.2004.01196.x
- 288. Sathish, P., Withana, N., Biswas, M., Bryant, C., Templeton, K., Al-Wahb, M., Smith-Espinoza, C., Roche, J.R., Elborough, K.M., and Phillips, J.R. (2007) Transcriptome analysis reveals season-specific *rbcS* gene expression profiles in diploid perennial ryegrass (*Lolium perenne* L.). *Plant Biotechnol J* 5, 146-61. doi: 10.1111/j.1467-7652.2006.00228.x
- Wittenbach, V.A. (1983) Purification and characterization of a soybean leaf storage glycoprotein. *Plant Physiology* **73**, 125. doi: 10.1104/pp.73.1.125
- 290. **Staswick, P.E.** (1990) Novel regulation of vegetative storage protein genes. *Plant Cell* **2**, 1-6. doi: 10.1105/tpc.2.1.1
- 291. Berger, S., Bell, E., Sadka, A., and Mullet, J.E. (1995) *Arabidopsis thaliana* Atvsp is homologous to soybean VspA and VspB, genes encoding vegetative storage protein acid phosphatases, and is regulated similarly by methyl jasmonate, wounding, sugars, light and phosphate. *Plant Mol Biol* **27**, 933-42. doi: 10.1007/BF00037021
- 292. **Staswick, P.E.** (1989) Developmental Regulation and the Influence of Plant Sinks on Vegetative Storage Protein Gene Expression in Soybean Leaves. *Plant Physiology* **89,** 309. doi: 10.1104/pp.89.1.309
- 293. **Staswick, P.E.** (1994) Storage proteins of vegetative plant tissues. Annual Review of Plant Physiology and Plant Molecular Biology **45**, 303-22. doi: 10.1146/annurev.pp.45.060194.001511
- 294. **Rossato, L., Dantec, C.L., Laine, P., and Ourry, A.** (2002) Nitrogen storage and remobilization in *Brassica napus* L. during the growth cycle: identification, characterization and immunolocalization of a putative taproot storage glycoprotein. *Journal of Experimental Botany* **53**, 265-75. doi: 10.1093/jexbot/53.367.265

- 295. Cunningham, S.M., and Volenec, J.J. (1996) Purification and Characterization of Vegetative Storage Proteins from Alfalfa (*Medicago* sativa L.) Taproots. Journal of Plant Physiology 147, 625-32. doi: 10.1016/S0176-1617(11)81469-0
- 296. Cunningham, S.M., and Volenec, J.J. (1998) Seasonal carbohydrate and nitrogen metabolism in roots of contrasting alfalfa (*Medicago sativa* L.) cultivars. *Journal of Plant Physiology* 153, 220-5. doi: 10.1016/S0176-1617(98)80069-2
- 297. Justes, E., Thiébeau, P., Avice, J.C., Lemaire, G., Volenec, J.J., and Ourry, A. (2002) Influence of summer sowing dates, N fertilization and irrigation on autumn VSP accumulation and dynamics of spring regrowth in alfalfa (*Medicago sativa* L.). *Journal of Experimental Botany* **53**, 111-21. doi: 10.1093/jexbot/53.366.111
- 298. Meuriot, F., Avice, J.-C., Decau, M.-L., Simon, J.-C., Lainé, P., Volenec, J.J., and Ourry, A. (2003) Accumulation of N reserves and vegetative storage protein (VSP) in taproots of non-nodulated alfalfa (*Medicago sativa* L.) are affected by mineral N availability. *Plant Science* 165, 709-18. doi: 10.1016/S0168-9452(03)00225-5
- 299. Meuriot, F., Noquet, C., Avice, J.-C., Volenec, J.J., Cunningham, S.M., Sors, T.G., Caillot, S., and Ourry, A. (2004) Methyl jasmonate alters N partitioning, N reserves accumulation and induces gene expression of a 32-kDa vegetative storage protein that possesses chitinase activity in *Medicago sativa* taproots. *Physiologia Plantarum* **120**, 113-23. doi: 10.1111/j.0031-9317.2004.0210.x
- 300. **Bakken, A.K., Macduff, J.H., and Collison, M.** (1998) Dynamics of nitrogen remobilization in defoliated *Phleum pratense* and *Festuca pratensis* under short and long photoperiods. *Physiologia Plantarum* **103**, 426-36. doi: 10.1034/j.1399-3054.1998.1030316.x
- 301. Louahlia, S., Macduff, J.H., Ourry, A., Humphreys, M., and Boucaud, J. (1999) Nitrogen reserve status affects the dynamics of nitrogen remobilization and mineral nitrogen uptake during recovery of contrasting cultivars of *Lolium perenne* from defoliation. *New Phytologist* 142, 451-62. doi: 10.1046/j.1469-8137.1999.00419.x
- 302. Louahlia, S., Laine, P., MacDuff, J.H., Ourry, A., Humphreys, M., and Boucaud, J. (2008) Interactions between reserve mobilization and regulation of nitrate uptake during regrowth of *Lolium perenne* L.: putative roles of amino acids and carbohydrates. *Botany* **86**, 1101-10. doi: 10.1139/B08-066
- 303. **Zouhar, J., and Rojo, E.** (2009) Plant vacuoles: where did they come from and where are they heading? *Current Opinion in Plant Biology* **12**, 677-84. doi: 10.1016/j.pbi.2009.08.004
- 304. **Frigerio, L., Hinz, G., and Robinson, D.G.** (2008) Multiple Vacuoles in Plant Cells: Rule or Exception? *Traffic* **9**, 1564-70. doi: 10.1111/j.1600-0854.2008.00776.x
- 305. van der Leij, M., Smith, S.J., and Miller, A.J. (1998) Remobilisation of vacuolar stored nitrate in barley root cells. *Planta* 205, 64-72. doi: 10.1007/s004250050297
- 306. Jürgens, G. (2004) Membrane trafficking in plants. Annual Review of Cell and Developmental Biology 20, 481-504. doi: 10.1146/annurev.cellbio.20.082503.103057
- 307. Matheson, L.A., Hanton, S.L., and Brandizzi, F. (2006) Traffic between the plant endoplasmic reticulum and Golgi apparatus: to the Golgi and beyond. *Current Opinion in Plant Biology* 9, 601-9. doi: 10.1016/j.pbi.2006.09.016
- 308. Pereira, C., Pereira, S., and Pissarra, J. (2014) Delivering of proteins to the plant vacuole--an update. *Int J Mol Sci* 15, 7611-23. doi: 10.3390/ijms15057611

- 309. Craddock, C.P., Hunter, P.R., Szakacs, E., Hinz, G., Robinson, D.G., and Frigerio, L. (2008) Lack of a vacuolar sorting receptor leads to nonspecific missorting of soluble vacuolar proteins in *Arabidopsis* seeds. *Traffic* **9**, 408-16. doi: 10.1111/j.1600-0854.2007.00693.x
- 310. **Pearse, B.M.** (1976) Clathrin: a unique protein associated with intracellular transfer of membrane by coated vesicles. *PNAS* **73**, 1255-9. doi: 10.1073/pnas.73.4.1255
- 311. **Hohl, I., Robinson, D.G., Chrispeels, M.J., and Hinz, G.** (1996) Transport of storage proteins to the vacuole is mediated by vesicles without a clathrin coat. *Journal of Cell Science* **109**, 2539.
- 312. **Murphy, K.A., Kuhle, R.A., Fischer, A.M., Anterola, A.M., and Grimes, H.D.** (2005) The functional status of paraveinal mesophyll vacuoles changes in response to altered metabolic conditions in soybean leaves. *Functional Plant Biology* **32,** 335-44-2005. doi: 10.1071/FP05006
- Zheng, H., and Staehelin, L.A. (2011) Protein storage vacuoles are transformed into lytic vacuoles in root meristematic cells of germinating seedlings by multiple, cell type-specific mechanisms. *Plant Physiology* 155, 2023. doi: 10.1104/pp.110.170159
- 314. **Olbrich, A., Hillmer, S., Hinz, G., Oliviusson, P., and Robinson, D.G.** (2007) Newly formed vacuoles in root meristems of barley and pea seedlings have characteristics of both protein storage and lytic vacuoles. *Plant Physiology* **145,** 1383. doi: 10.1104/pp.107.108985
- 315. Park, M., Kim, S.J., Vitale, A., and Hwang, I. (2004) Identification of the Protein Storage Vacuole and Protein Targeting to the Vacuole in Leaf Cells of Three Plant Species. *Plant Physiology* **134**, 625-39. doi: 10.1104/pp.103.030635
- 316. **Johanson, U., and Gustavsson, S.** (2002) A New Subfamily of Major Intrinsic Proteins in Plants. *Molecular Biology and Evolution* **19**, 456-61. doi: 10.1093/oxfordjournals.molbev.a004101
- 317. Maurel, C., Boursiac, Y., Luu, D.-T., Santoni, V., Shahzad, Z., and Verdoucq, L. (2015) Aquaporins in Plants. *Physiol Rev* **95**, 1321-58. doi: 10.1152/physrev.00008.2015
- 318. **Maurel, C., Reizer, J., Schroeder, J.I., and Chrispeels, M.J.** (1993) The vacuolar membrane protein gamma-TIP creates water specific channels in *Xenopus oocytes. The EMBO journal* **12,** 2241-7.
- 319. Jauh, G.Y., Fischer, A.M., Grimes, H.D., Ryan, C.A., Jr., and Rogers, J.C. (1998) delta-Tonoplast intrinsic protein defines unique plant vacuole functions. *PNAS* **95**, 12995-9. doi: 10.1073/pnas.95.22.12995
- Jauh, G.-Y., Phillips, T.E., and Rogers, J.C. (1999) Tonoplast Intrinsic Protein Isoforms as Markers for Vacuolar Functions. *Plant Cell* **11** 1867-82. doi: 10.1105/tpc.11.10.1867
- 321. **Rojas-Pierce, M.** (2013) Targeting of tonoplast proteins to the vacuole. *Plant Science* **211,** 132-6. doi: 10.1016/j.plantsci.2013.07.005
- 322. Vander Willigen, C., Pammenter, N.W., Mundree, S.G., and Farrant, J.M. (2004) Mechanical stabilization of desiccated vegetative tissues of the resurrection grass *Eragrostis nindensis*: does a TIP 3;1 and/or compartmentalization of subcellular components and metabolites play a role? *Journal of Experimental Botany* 55, 651-61. doi: 10.1093/jxb/erh089
- 323. VanBuren, R., Bryant, D., Edger, P.P., Tang, H., Burgess, D., Challabathula, D., Spittle, K., Hall, R., Gu, J., Lyons, E., Freeling, M., Bartels, D., Ten Hallers, B., Hastie, A., Michael, T.P., and Mockler, T.C. (2015) Single-molecule sequencing of the desiccationtolerant grass *Oropetium thomaeum*. *Nature* 527, 508-11. doi: 10.1038/nature15714
- 324. VanBuren, R., Wai, C.M., Zhang, Q., Song, X., Edger, P.P., Bryant, D., Michael, T.P., Mockler, T.C., and Bartels, D. (2017) Seed

desiccation mechanisms co-opted for vegetative desiccation in the resurrection grass *Oropetium thomaeum*. *Plant, Cell & Environment* **40**, 2292-306. doi: 10.1111/pce.13027

- 325. Yobi, A., Schlauch, K.A., Tillett, R.L., Yim, W.C., Espinoza, C., Wone, B.W.M., Cushman, J.C., and Oliver, M.J. (2017) *Sporobolus stapfianus*: Insights into desiccation tolerance in the resurrection grasses from linking transcriptomics to metabolomics. *BMC Plant Biology* **17**, 67. doi: 10.1186/s12870-017-1013-7
- 326. **Poehlman, J.M.** (1987) "Breeding Forage Crops," in *Breeding Field Crops*. (Dordrecht: Springer Netherlands), 625-79.
- 327. Vogel, K.P., and Pedersen, J.F. (1993) "Breeding Systems for Cross-Pollinated Perennial Grasses," in *Plant Breeding Reviews*, ed. J. Janick. (Oxford, UK: John Wiley & Sons, Inc.), 251-74.
- 328. Jahufer, M.Z.Z., Cooper, M., Ayres, J.F., and Bray, R.A. (2002) Identification of research to improve the efficiency of breeding strategies for white clover in Australia - a review. *Australian journal of agricultural research* **53**, 239-57. doi: 10.1071/AR01110
- 329. **Riday, H., and Brummer, E.C.** (2007) Narrow Sense Heritability and Additive Genetic Correlations in Alfalfa subsp. *falcata*. *The Journal of the Iowa Academy of Science* **114**, 28-34.
- 330. Collins, R.P., Helgadóttir, Á., Frankow-Lindberg, B.E., Skøt, L., Jones, C., and Skøt, K.P. (2012) Temporal changes in population genetic diversity and structure in red and white clover grown in three contrasting environments in northern Europe. *Annals of Botany* **110**, 1341-50. doi: 10.1093/aob/mcs058
- 331. Luo, K., Jahufer, M.Z.Z., Wu, F., Di, H., Zhang, D., Meng, X., Zhang, J., and Wang, Y. (2016) Genotypic variation in a breeding population of yellow sweet clover (*Melilotus officinalis*). *Frontiers in Plant Science* 7. doi: 10.3389/fpls.2016.00972
- 332. Manzanares, C., Yates, S., Ruckle, M., Nay, M., and Studer, B. (2016) TILLING in forage grasses for gene discovery and breeding improvement. *New Biotechnology* 33, 594-603. doi: 10.1016/j.nbt.2016.02.009
- 333. **de Araüjo, M.R.A., Coulman, B.E., and Rakow, G.** (2002) Genetic variation, heritability and progeny testing in meadow bromegrass. *Plant Breeding* **121,** 417-24. doi: 10.1046/j.1439-0523.2002.739124.x
- 334. **Piano, E., Annicchiarico, P., Romani, M., and Pecetti, L.** (2007) Effect of the number of parents and their combining ability on the performance of synthetic varieties in tall fescue. *Australian journal of agricultural research* **58**, 1100-5. doi: 10.1071/AR06424
- 335. Blackmore, T., Thorogood, D., Skøt, L., McMahon, R., Powell, W., and Hegarty, M. (2016) Germplasm dynamics: the role of ecotypic diversity in shaping the patterns of genetic variation in *Lolium perenne*. *Scientific Reports* **6**, 22603. doi: 10.1038/srep22603
- 336. Thorogood, D., Yates, S., Manzanares, C., Skot, L., Hegarty, M., Blackmore, T., Barth, S., and Studer, B. (2017) A novel multivariate approach to phenotyping and association mapping of multi-locus gametophytic self-incompatibility reveals S, Z, and other loci in a perennial ryegrass (*Poaceae*) population. *Frontiers in Plant Science* 8, 1331. doi: 10.3389/fpls.2017.01331
- 337. Harper, J., De Vega, J., Swain, S., Heavens, D., Gasior, D., Thomas, A., Evans, C., Lovatt, A., Lister, S., Thorogood, D., Skøt, L., Hegarty, M., Blackmore, T., Kudrna, D., Byrne, S., Asp, T., Powell, W., Fernandez-Fuentes, N., and Armstead, I. (2019) Integrating a newly developed BAC-based physical mapping resource for *Lolium perenne* with a genome-wide association study across a *L. perenne* European ecotype collection identifies genomic contexts associated with

agriculturally important traits. *Annals of Botany* **123**, 977-92. doi: 10.1093/aob/mcy230

- 338. **Thomas, H., and Humphreys, M.O.** (1991) Progress and potential of interspecific hybrids of *Lolium* and *Festuca*. *The Journal of Agricultural Science* **117**, 1-8. doi: 10.1017/S0021859600078916
- 339. **Humphreys, M.W., Canter, P.J., and Thomas, H.M.** (2003) Advances in introgression technologies for precision breeding within the *Lolium* -*Festuca* complex. *Annals of Applied Biology* **143,** 1-10. doi: 10.1111/j.1744-7348.2003.tb00263.x
- 340. Macleod, C.J.A., Humphreys, M.W., Whalley, W.R., Turner, L., Binley, A., Watts, C.W., Skøt, L., Joynes, A., Hawkins, S., King, I.P., O'Donovan, S., and Haygarth, P.M. (2013) A novel grass hybrid to reduce flood generation in temperate regions. *Scientific Reports* 3, 1683. doi: 10.1038/srep01683
- 341. Humphreys, M.W., O'Donovan, S.A., Farrell, M.S., Gay, A.P., and Kingston-Smith, A.H. (2014) The potential of novel *Festulolium* (2n = 4x = 28) hybrids as productive, nutrient-use-efficient fodder for ruminants. *Food and Energy Security* **3**, 98-110. doi: 10.1002/fes3.50
- 342. Walter, A., Studer, B., and Kölliker, R. (2012) Advanced phenotyping offers opportunities for improved breeding of forage and turf species. *Annals of Botany* **110**, 1271-9. doi: 10.1093/aob/mcs026
- 343. Barrett, B., Griffiths, A., Schreiber, M., Ellison, N., Mercer, C., Bouton, J., Ong, B., Forster, J., Sawbridge, T., Spangenberg, G., Bryan, G., and Woodfield, D. (2004) A microsatellite map of white clover. *Theoretical and Applied Genetics* **109**. doi: 10.1007/s00122-004-1658-0
- 344. **Barrett, B., Baird, I., and Woodfield, D.** (2009) "White Clover Seed Yield: A Case Study in Marker-Assisted Selection," in *Molecular Breeding of Forage and Turf*. (New York, NY: Springer New York), 241-50.
- 345. Dias, P.M.B., Brunel-Muguet, S., Dürr, C., Huguet, T., Demilly, D., Wagner, M.-H., and Teulat-Merah, B. (2011) QTL analysis of seed germination and pre-emergence growth at extreme temperatures in *Medicago truncatula*. *TAG. Theoretical and Applied Genetics. Theoretische Und Angewandte Genetik* **122**, 429-44. doi: 10.1007/s00122-010-1458-7
- 346. **Pfeifer, M., Martis, M., Asp, T., Mayer, K.F.X., Lübberstedt, T., Byrne, S., Frei, U., and Studer, B.** (2013) The Perennial Ryegrass GenomeZipper: Targeted Use of Genome Resources for Comparative Grass Genomics. *Plant Physiology* **161,** 571-82. doi: 10.1104/pp.112.207282
- 347. Yates, S.A., Swain, M.T., Hegarty, M.J., Chernukin, I., Lowe, M., Allison, G.G., Ruttink, T., Abberton, M.T., Jenkins, G., and Skøt, L. (2014) De novo assembly of red clover transcriptome based on RNA-Seq data provides insight into drought response, gene discovery and marker identification. *BMC Genomics* **15**, 453. doi: 10.1186/1471-2164-15-453
- 348. Velmurugan, J., Mollison, E., Barth, S., Marshall, D., Milne, L., Creevey, C.J., Lynch, B., Meally, H., McCabe, M., and Milbourne, D. (2016) An ultra-high density genetic linkage map of perennial ryegrass (*Lolium perenne*) using genotyping by sequencing (GBS) based on a reference shotgun genome assembly. *Annals of Botany* **118**, 71-87. doi: 10.1093/aob/mcw081
- 349. Cogan, N.O., Drayton, M.C., Ponting, R.C., Vecchies, A.C., Bannan, N.R., Sawbridge, T.I., Smith, K.F., Spangenberg, G.C., and Forster, J.W. (2007) Validation of in silico-predicted genic SNPs in white clover (*Trifolium repens* L.), an outbreeding allopolyploid species. *Molecular Genetics and Genomics* 277. doi: 10.1007/s00438-006-0198-5
- 350. Lawless, K.A., Drayton, M.C., Hand, M.C., Ponting, R.C., Cogan, N.O., Sawbridge, T.I., Smith, K.F., Spangenberg, G.C., and Forster,

J.C. (2009) "Interpretation of SNP haplotype complexity in white clover (*Trifolium repens* L.), an outbreeding species," in *Molecular Breeding of Forage and Turf*, eds. **T. Yamada & G. Spangenberg**.).

- 351. Nagy, I., Barth, S., Mehenni-Ciz, J., Abberton, M.T., and Milbourne, D. (2013) A hybrid next generation transcript sequencing-based approach to identify allelic and homeolog-specific single nucleotide polymorphisms in allotetraploid white clover. *BMC Genomics* 14, 100. doi: 10.1186/1471-2164-14-100
- 352. Draper, J., Mur, L.A.J., Jenkins, G., Ghosh-Biswas, G.C., Bablak, P., Hasterok, R., and Routledge, A.P.M. (2001) *Brachypodium distachyon*: A new model system for functional genomics in grasses. *Plant Physiology* **127**, 1539-55. doi: 10.1104/pp.010196
- 353. Sato, S., Isobe, S., Asamizu, E., Ohmido, N., Kataoka, R., Nakamura, Y., Kaneko, T., Sakurai, N., Okumura, K., Klimenko, I., Sasamoto, S., Wada, T., Watanabe, A., Kohara, M., Fujishiro, T., and Tabata, S. (2005) Comprehensive Structural Analysis of the Genome of Red Clover (*Trifolium pratense* L.). *DNA Research* 12. doi: 10.1093/dnares/dsi018
- 354. Cannon, S.B., Sterck, L., Rombauts, S., Sato, S., Cheung, F., Gouzy, J., Wang, X., Mudge, J., Vasdewani, J., Schiex, T., Spannagl, M., Monaghan, E., Nicholson, C., Humphray, S.J., Schoof, H., Mayer, K.F., Rogers, J., Quetier, F., Oldroyd, G.E., Debelle, F., Cook, D.R., Retzel, E.F., Roe, B.A., Town, C.D., Tabata, S., de Peer, Y., and Young, N.D. (2006) Legume genome evolution viewed through the *Medicago truncatula* and *Lotus japonicus* genomes. *PNAS* 103. doi: 10.1073/pnas.0603228103
- 355. Varshney, R.K., Chen, W., Li, Y., Bharti, A.K., Saxena, R.K., Schlueter, J.A., Donoghue, M.T., Azam, S., Fan, G., Whaley, A.M., Farmer, A.D., Sheridan, J., Iwata, A., Tuteja, R., Penmetsa, R.V., Wu, W., Upadhyaya, H.D., Yang, S.P., Shah, T., Saxena, K.B., Michael, T., McCombie, W.R., Yang, B., Zhang, G., Yang, H., Wang, J., Spillane, C., Cook, D.R., May, G.D., and Xu, X. (2012) Draft genome sequence of pigeonpea (*Cajanus cajan*), an orphan legume crop of resource-poor farmers. *Nature Biotechnology* **30**. doi: 10.1038/nbt.2022
- 356. Varshney, R.K., Song, C., Saxena, R.K., Azam, S., Yu, S., Sharpe, A.G., Cannon, S., Baek, J., Rosen, B.D., Tar'an, B., Millan, T., Zhang, X., Ramsay, L.D., Iwata, A., Wang, Y., Nelson, W., Farmer, A.D., Gaur, P.M., Soderlund, C., Penmetsa, R.V., Xu, C., Bharti, A.K., He, W., Winter, P., Zhao, S., Hane, J.K., Carrasquilla-Garcia, N., Condie, J.A., Upadhyaya, H.D., and Luo, M.C. (2013) Draft genome sequence of chickpea (*Cicer arietinum*) provides a resource for trait improvement. *Nature Biotechnology* **31**. doi: 10.1038/nbt.2491
- 357. Li, X., Wei, Y., Acharya, A., Jiang, Q., Kang, J., and Brummer, E.C. (2014) A saturated genetic linkage map of autotetraploid alfalfa (*Medicago sativa* L.) developed using genotyping-by-sequencing is highly syntenous with the *Medicago truncatula* genome. *G3: Genes Genomes Genetics* **4**, 1971-9. doi: 10.1534/g3.114.012245
- 358. Tang, H., Krishnakumar, V., Bidwell, S., Rosen, B., Chan, A., Zhou, S., Gentzbittel, L., Childs, K.L., Yandell, M., Gundlach, H., Mayer, K.F., Schwartz, D.C., and Town, C.D. (2014) An improved genome release (version Mt4.0) for the model legume *Medicago truncatula*. *BMC Genomics* 15, 312. doi: 10.1186/1471-2164-15-312
- 359. De Vega, J.J., Ayling, S., Hegarty, M., Kudrna, D., Goicoechea, J.L., Ergon, Å., Rognli, O.A., Jones, C., Swain, M., Geurts, R., Lang, C., Mayer, K.F.X., Rössner, S., Yates, S., Webb, K.J., Donnison, I.S., Oldroyd, G.E.D., Wing, R.A., Caccamo, M., Powell, W., Abberton,

M.T., and Skøt, L. (2015) Red clover (*Trifolium pratense* L.) draft genome provides a platform for trait improvement. *Scientific Reports* **5**, 17394. doi: 10.1038/srep17394

- 360. O'Rourke, J.A., Fu, F., Bucciarelli, B., Yang, S.S., Samac, D.A., Lamb, J.F.S., Monteros, M.J., Graham, M.A., Gronwald, J.W., Krom, N., Li, J., Dai, X., Zhao, P.X., and Vance, C.P. (2015) The *Medicago* sativa gene index 1.2: a web-accessible gene expression atlas for investigating expression differences between *Medicago sativa* subspecies. *BMC Genomics* 16, 502. doi: 10.1186/s12864-015-1718-7
- 361. Hirakawa, H., Kaur, P., Shirasawa, K., Nichols, P., Nagano, S., Appels, R., Erskine, W., and Isobe, S.N. (2016) Draft genome sequence of subterranean clover, a reference for genus *Trifolium*. *Scientific Reports* 6, 30358. doi: 10.1038/srep30358
- 362. McCord, P., Gordon, V., Saha, G., Hellinga, J., Vandemark, G., Larsen, R., Smith, M., and Miller, D. (2014) Detection of QTL for forage yield, lodging resistance and spring vigor traits in alfalfa (*Medicago sativa* L.). *Euphytica* **200**, 269-79. doi: 10.1007/s10681-014-1160-y
- 363. **Robins, J.G., Bauchan, G.R., and Brummer, E.C.** (2007) Genetic Mapping Forage Yield, Plant Height, and Regrowth at Multiple Harvests in Tetraploid Alfalfa (*Medicago sativa* L.). *Crop Science* **47,** 11-8. doi: 10.2135/cropsci2006.07.0447
- 364. Herrmann, D., Barre, P., Santoni, S., and Julier, B. (2010) Association of a *CONSTANS-LIKE* gene to flowering and height in autotetraploid alfalfa. *Theoretical and Applied Genetics* **121**, 865-76. doi: 10.1007/s00122-010-1356-z
- 365. **Maghsoodi, M., Razmjoo, J., and Gheysari, M.** (2017) Application of biochemical markers for the assessment of drought tolerance in alfalfa (*Medicago sativa* L.) cultivars. *Grassland Science* **63**, 3-14. doi: 10.1111/grs.12144
- 366. **Warnke, E., Barker, E., Brilman, A., Young, C., and Cook, L.** (2002) Inheritance of *superoxide dismutase* (*Sod-1*) in a perennial x annual ryegrass cross and its allelic distribution among cultivars. *Theoretical and Applied Genetics* **105**, 1146-50. doi: 10.1007/s00122-002-1007-0
- 367. Li, W., Katin-Grazzini, L., Gu, X., Wang, X., El-Tanbouly, R., Yer, H., Thammina, C., Inguagiato, J., Guillard, K., McAvoy, R.J., Wegrzyn, J., Gu, T., and Li, Y. (2017) Transcriptome Analysis Reveals Differential Gene Expression and a Possible Role of Gibberellins in a Shade-Tolerant Mutant of Perennial Ryegrass. *Frontiers in Plant Science* 8, 868. doi: 10.3389/fpls.2017.00868
- 368. **Diekmann, K., Hodkinson, T.R., and Barth, S.** (2012) New chloroplast microsatellite markers suitable for assessing genetic diversity of *Lolium perenne* and other related grass species. *Annals of Botany* **110**, 1327-39. doi: 10.1093/aob/mcs044
- 369. Bonhomme, M., Andre, O., Badis, Y., Ronfort, J., Burgarella, C., Chantret, N., Prosperi, J.M., Briskine, R., Mudge, J., Debelle, F., Navier, H., Miteul, H., Hajri, A., Baranger, A., Tiffin, P., Dumas, B., Pilet-Nayel, M.L., Young, N.D., and Jacquet, C. (2014) High-density genome-wide association mapping implicates an F-box encoding gene in *Medicago truncatula* resistance to *Aphanomyces euteiches*. *New Phytologist* 201, 1328-42. doi: 10.1111/nph.12611
- 370. Slavov, G.T., Nipper, R., Robson, P., Farrar, K., Allison, G.G., Bosch, M., Clifton-Brown, J.C., Donnison, I.S., and Jensen, E. (2014) Genome-wide association studies and prediction of 17 traits related to phenology, biomass and cell wall composition in the energy grass *Miscanthus sinensis*. *New Phytologist* **201**, 1227-39. doi: 10.1111/nph.12621

- 371. Carelli, M., Calderini, O., Panara, F., Porceddu, A., Losini, I., Piffanelli, P., Arcioni, S., and Scotti, C. (2013) Reverse genetics in Medicago truncatula using a TILLING mutant collection. Methods in Molecular Biology 1069, 101-18. doi: 10.1007/978-1-62703-613-9_9
- 372. Dalmais, M., Antelme, S., Ho-Yue-Kuang, S., Wang, Y., Darracq, O., d'Yvoire, M.B., Cezard, L., Legee, F., Blondet, E., Oria, N., Troadec, C., Brunaud, V., Jouanin, L., Hofte, H., Bendahmane, A., Lapierre, C., and Sibout, R. (2013) A TILLING Platform for Functional Genomics in *Brachypodium distachyon*. *PLoS One* 8, e65503. doi: 10.1371/journal.pone.0065503
- 373. **Zhang, Y., Mian, M.A.R., and Bouton, J.H.** (2006) Recent molecular and genomic studies on stress tolerance of forage and turf grasses. *Crop Science* **46**, 497-511. doi: 10.2135/cropsci2004.0572
- 374. **Stewart, D., Shepherd, L.V.T., Hall, R.D., and Fraser, P.D.** (2011) Crops and Tasty, Nutritious Food – How Can Metabolomics Help? *Annual Plant Reviews* **43**, 181-217. doi: 10.1002/9781444339956.ch7
- 375. **Foster, A.J., Kakani, V.G., and Mosali, J.** (2017) Estimation of bioenergy crop yield and N status by hyperspectral canopy reflectance and partial least square regression. *Precision Agriculture* **18**, 192-209. doi: 10.1007/s11119-016-9455-8
- 376. **Knight, M.** (2006) *Soil mineral nitrogen testing : Practice and interpretation* [Online]. Norfolk, UK: A.C.O. Seeds. Available: <u>https://cereals.ahdb.org.uk/media/365643/rr58 complete research review.pdf</u> [Accessed 16/04/2018].
- 377. **Takahashi, M., Sasaki, Y., Ida, S., and Morikawa, H.** (2001) Nitrite reductase gene enrichment improves assimilation of NO2 in *Arabidopsis*. *Plant Physiology* **126**, 731. doi: 10.1104/pp.126.2.731
- 378. Lancien, M., Martin, M., Hsieh, M.-H., Leustek, T., Goodman, H., and Coruzzi, G.M. (2002) *Arabidopsis* glt1-T mutant defines a role for NADH-GOGAT in the non-photorespiratory ammonium assimilatory pathway. *The Plant Journal* 29, 347-58. doi: 10.1046/j.1365-313X.2002.01218.x
- 379. Loque, D., Ludewig, U., Yuan, L., and von Wiren, N. (2005) Tonoplast intrinsic proteins AtTIP2;1 and AtTIP2;3 facilitate NH3 transport into the vacuole. *Plant Physiology* **137**, 671-80. doi: 10.1104/pp.104.051268
- 380. Wudick, M.M., Luu, D.-T., and Maurel, C. (2009) A look inside: localization patterns and functions of intracellular plant aquaporins. *New Phytologist* **184**, 289-302. doi: 10.1111/j.1469-8137.2009.02985.x
- 381. Gattolin, S., Sorieul, M., and Frigerio, L. (2010) Tonoplast intrinsic proteins and vacuolar identity. *Biochem Soc Trans* 38, 769-73. doi: 10.1042/bst0380769
- 382. **Gattolin, S., Sorieul, M., and Frigerio, L.** (2011) Mapping of tonoplast intrinsic proteins in maturing and germinating *Arabidopsis* seeds reveals dual localization of embryonic TIPs to the tonoplast and plasma membrane. *Molecular Plant* **4**, 180-9. doi: 10.1093/mp/ssq051
- 383. Reinhardt, H., Hachez, C., Bienert, M.D., Beebo, A., Swarup, K., Voß, U., Bouhidel, K., Frigerio, L., Schjoerring, J.K., Bennett, M.J., and Chaumont, F. (2016) Tonoplast aquaporins facilitate lateral root emergence. *Plant Physiology* **170**, 1640. doi: 10.1104/pp.15.01635
- Jackson, R.B., and Caldwell, M.M. (1993) Geostatistical patterns of soil heterogeneity around individual perennial plants. *Journal of Ecology* 81, 683-92. doi: 10.2307/2261666
- 385. **Jackson, R.B., and Caldwell, M.M.** (1996) Integrating resourse heterogeneity and plant plasticity: modelling nitrate and phosphate uptake in a patchy soil environment. *Journal of Ecology* **84,** 891-903. doi: 10.2307/2960560

- 386. Reynolds, H.L., Hungate, B.A., Chapin Iii, F.S., and D'Antonio, C.M. (1997) Soil heterogeneity and plant competition in an annual grassland. *Ecology* 78, 2076-90. doi: 10.1890/0012-9658(1997)078[2076:SHAPCI]2.0.CO;2
- 387. Enwall, K., Throbäck, I.N., Stenberg, M., Söderström, M., and Hallin, S. (2010) Soil Resources Influence Spatial Patterns of Denitrifying Communities at Scales Compatible with Land Management. *Applied and environmental microbiology* **76**, 2243. doi: 10.1128/AEM.02197-09
- 388. Keil, D., Meyer, A., Berner, D., Poll, C., Schutzenmeister, A., Piepho, H.P., Vlasenko, A., Philippot, L., Schloter, M., Kandeler, E., and Marhan, S. (2011) Influence of land-use intensity on the spatial distribution of N-cycling microorganisms in grassland soils. *FEMS Microbiol Ecol* 77, 95-106. doi: 10.1111/j.1574-6941.2011.01091.x
- 389. **Giles, M., Morley, N., Baggs, E., and Daniell, T.** (2012) Soil nitrate reducing processes drivers, mechanisms for spatial variation, and significance for nitrous oxide production. *Frontiers in microbiology* **3**, 407. doi: 10.3389/fmicb.2012.00407
- 390. **Duran, B.E.L., Duncan, D.S., Oates, L.G., Kucharik, C.J., and Jackson, R.D.** (2016) Nitrogen Fertilization Effects on Productivity and Nitrogen Loss in Three Grass-Based Perennial Bioenergy Cropping Systems. *PLoS One* **11**, 1-13. doi: 10.1371/journal.pone.0151919
- 391. Bogaert, N., Salomez, J., Vermoesen, A., Hofman, G., Van Cleemput, O., and Van Meirvenne, M. (2000) Within-field variability of mineral nitrogen in grassland. *Biology and Fertility of Soils* 32, 186-93. doi: 10.1007/s003740000233
- 392. Elkateb, T., Chalaturnyk, R., and Robertson, P.K. (2003) An overview of soil heterogeneity: quantification and implications on geotechnical field problems. *Canadian Geotechnical Journal* **40**, 1-15. doi: 10.1139/t02-090
- 393. Dentener, F., Drevet, J., Lamarque, J.F., Bey, I., Eickhout, B., Fiore, A.M., Hauglustaine, D., Horowitz, L.W., Krol, M., Kulshrestha, U.C., Lawrence, M., Galy-Lacaux, C., Rast, S., Shindell, D., Stevenson, D., Van Noije, T., Atherton, C., Bell, N., Bergman, D., Butler, T., Cofala, J., Collins, B., Doherty, R., Ellingsen, K., Galloway, J., Gauss, M., Montanaro, V., Müller, J.F., Pitari, G., Rodriguez, J., Sanderson, M., Solmon, F., Strahan, S., Schultz, M., Sudo, K., Szopa, S., and Wild, O. (2006) Nitrogen and sulfur deposition on regional and global scales: A multimodel evaluation. *Global Biogeochemical Cycles* 20. doi: 10.1029/2005GB002672
- 394. Emmett, B.A.R., B.; Chamberlain, P.M.; Rowe, E.; Spurgeon, D.; Brittain, S.A.; Frogbrook, Z.; Hughes, S.; Lawlor, A.J.; Poskitt, J.; Potter, E.; Robinson, D.A.; Scott, A.; Wood, C.; Woods, C. (2010) Countryside Survey: Soils Report from 2007 [Online]. Available: http://nora.nerc.ac.uk/id/eprint/9354 [Accessed 14/06/2018].
- 395. **Dobbie, K.E., Bruneau, P.M.C., and Towers, W.** (2011) *The State of Scotland's Soil* [Online]. Available: www.sepa.org.uk/land/land_publications.aspx [Accessed 08/08/2018].
- 396. **Pritchard, O.G., Hallett, S.H., and Farewell, T.S.** (2014) Soil impacts on UK infrastructure: current and future climate. *Proceedings of the Institution of Civil Engineers Engineering Sustainability* **167,** 170-84. doi: 10.1680/ensu.13.00035
- 397. Environmental Audit Committee (2016) Soil Health: First Report of Session 2016–17 [Online]. London, UK. Available: <u>https://www.parliament.uk/business/committees/committees-a-</u> z/commons-select/environmental-audit-committee/inquiries/parliament-2015/soil-health/ [Accessed 09/08/2018].

- 398. **Harding, E.** 19/04/2018 (2018) *RE: Personal Communication.* Type to N.M. Capstaff.
- 399. **Caldwell, M.M., and Pearse, R.P.** (1994) *Exploitation of Environmental Heterogeneity by Plants: Ecophysiological Processes Above- and Belowground.* Massachusetts, US: Academic Press.
- 400. **Hutchings, M.J., John, E.A., and Wijesinghe, D.K.** (2003) Towards understanding the consequances of soil heterogeneity for plant populations and communities. *Ecology* **84,** 2322-34. doi: 10.1890/02-0290
- 401. **Hill, M.O., and Carey, P.D.** (1997) Prediction of yield in the Rothamsted Park Grass Experiment by Ellenberg indicator values. *Journal of Vegetation Science* **8**, 579-86. doi: 10.2307/3237210
- 402. Valkama, E., Rankinen, K., Virkajärvi, P., Salo, T., Kapuinen, P., and Turtola, E. (2016) Nitrogen fertilization of grass leys: Yield production and risk of N leaching. *Agriculture, Ecosystems & Environment* 230, 341-52. doi: 10.1016/j.agee.2016.05.022
- 403. Holland, J.E., Bennett, A.E., Newton, A.C., White, P.J., McKenzie, B.M., George, T.S., Pakeman, R.J., Bailey, J.S., Fornara, D.A., and Hayes, R.C. (2018) Liming impacts on soils, crops and biodiversity in the UK: A review. *Science of The Total Environment* 610-611, 316-32. doi: 10.1016/j.scitotenv.2017.08.020
- 404. Erice, G., Irigoyen, J.J., Sánchez-Díaz, M., Avice, J.-C., and Ourry, A. (2007) Effect of drought, elevated CO2 and temperature on accumulation of N and vegetative storage proteins (VSP) in taproot of nodulated alfalfa before and after cutting. *Plant Science* **172**, 903-12. doi: 10.1016/j.plantsci.2006.12.013
- 405. Gregory, A.S., Ritz, K., McGrath, S.P., Quinton, J.N., Goulding, K.W.T., Jones, R.J.A., Harris, J.A., Bol, R., Wallace, P., Pilgrim, E.S., and Whitmore, A.P. (2015) A review of the impacts of degradation threats on soil properties in the UK. *Soil Use and Management* **31**, 1-15. doi: 10.1111/sum.12212
- 406. **Arojju, S.K., Milbourne, D., Conaghan, P., Hodkinson, T.R., and Barth, S.** (Year) "Extent of Crown Rust Infection in a Perennial Ryegrass (*Lolium perenne* L.) Association Mapping Population", in: *Breeding in a World of Scarcity*, eds. I. Roldán-Ruiz, J. Baert & D. Reheul: Springer International Publishing), 47-52.
- 407. **Rogers, J.K., Walker, N.R., and Young, C.A.** (2016) The effect of endophytic fungi on nematode populations in summer-dormant and summer-active tall fescue. *Journal of nematology* **48**, 87-94.
- 408. **McKenna, P., Cannon, N., and Conway, J.** (2018) Soil mineral nitrogen availability predicted by herbage yield and disease resistance in red clover (*Trifolium pratense*) cropping. *Nutrient Cycling in Agroecosystems* **112,** 303-15. doi: 10.1007/s10705-018-9947-1
- 409. Yoshihara, Y., and Isogai, T. (2019) Does genetic diversity of grass improve yield, digestibility, and resistance to weeds, pests and disease infection? Archives of Agronomy and Soil Science 65, 1623-9. doi: 10.1080/03650340.2019.1566719
- 410. **Höglind, M., Van Oijen, M., Cameron, D., and Persson, T.** (2016) Process-based simulation of growth and overwintering of grassland using the BASGRA model. *Ecological Modelling* **335**, 1-15. doi: 10.1016/j.ecolmodel.2016.04.024
- 411. **Van Oijen, M., Bellocchi, G., and Höglind, M.** (2018) Effects of climate change on grassland biodiversity and productivity: The need for a diversity of models. *Agronomy* **8**. doi: 10.3390/agronomy8020014
- 412. **Crown**© (2008) "Implementation of the Nitrates Directive in England; Seventh Report of Session 2007-08 ", (ed.) House of Commons

Environment Food and Rural Affairs Committee. (London, UK: The Stationery Office by Order of the House).

- 413. **Crown**© (2009) "Protecting our Water, Soil and Air: A Code of Good Agricultural Practice for farmers, growers and land managers", (ed.) DEFRA. (Norwich, Norfolk: TSO).
- 414. **Crown**© (2013) "Guidance on complying with the rules for Nitrate Vulnerable Zones in England for 2013 to 2016 ", (ed.) DEFRA. (London, UK).
- 415. **DEFRA** (2018) *Fertiliser usage on farms: Results from the Farm Business Survey (FBS), England 2016/17* [Online]. London, UK. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/712349/fbs-fertiliseruse-statsnotice-31may18.pdf</u> [Accessed 08/08/2018].
- 416. **Schaad, N.W., and Frederick, R.D.** (2002) Real-time PCR and its application for rapid plant disease diagnostics. *Canadian Journal of Plant Pathology* **24,** 250-8. doi: 10.1080/07060660209507006
- 417. Mölsä, M., Hemmilä, H., Katz, A., Niemimaa, J., Forbes, K.M., Huitu, O., Stuart, P., Henttonen, H., and Nikkari, S. (2015) Monitoring biothreat agents (*Francisella tularensis*, *Bacillus anthracis* and *Yersinia pestis*) with a portable real-time PCR instrument. *Journal of Microbiological Methods* **115**, 89-93. doi: 10.1016/j.mimet.2015.05.026
- 418. Li, T.-J., Chang, C.-M., Chang, P.-Y., Chuang, Y.-C., Huang, C.-C., Su, W.-C., and Shieh, D.-B. (2016) Handheld energy-efficient magnetooptical real-time quantitative PCR device for target DNA enrichment and quantification. *Npg Asia Materials* **8**, e277. doi: 10.1038/am.2016.70
- 419. Nguyen, P.L., Sudheesh, P.S., Thomas, A.C., Sinnesael, M., Haman, K., and Cain, K.D. (2018) Rapid detection and monitoring of *Flavobacterium psychrophilum* in water by using a handheld, fieldportable quantitative PCR system. *Journal of Aquatic Animal Health* **30**, 302-11. doi: 10.1002/aah.10046
- 420. **Momotaz, A., Forster, J.W., and Yamada, T.** (2004) Identification of cultivars and accessions of *Lolium*, *Festuca* and *Festulolium* hybrids through the detection of simple sequence repeat polymorphism. *Plant Breeding* **123,** 370-6. doi: 10.1111/j.1439-0523.2004.00962.x
- 421. **Smil, V.** (2011) Nitrogen cycle and world food production. *World Agriculture* **2**, 9-11.
- 422. **Smil, V.** (1999) Detonator of the population explosion. *Nature* **400**, 415. doi: 10.1038/22672
- 423. **Haber, F.** (2002) *The synthesis of ammonia from its elements Nobel Lecture, June 2, 1920.*
- 424. Erisman, J.W., Sutton, M.A., Galloway, J., Klimont, Z., and Winiwarter, W. (2008) How a century of ammonia synthesis changed the world. *Nature Geoscience* **1**, 636. doi: 10.1038/ngeo325
- 425. **Galloway, J.N., Leach, A.M., Bleeker, A., and Erisman, J.W.** (2013) A chronology of human understanding of the nitrogen cycle. *Philosophical Transactions of the Royal Society B: Biological Sciences* **368**. doi: 10.1098/rstb.2013.0120
- 426. **Barona, A., Etxebarria, B., Aleksanyan, A., Gallastegui, G., Rojo, N., and Diaz-Tena, E.** (2018) A unique historical case to understand the present sustainable development. *Science and Engineering Ethics* **24**, 261-74. doi: 10.1007/s11948-017-9891-5
- 427. Fowler, D., Coyle, M., Skiba, U., Sutton, M.A., Cape, J.N., Reis, S., Sheppard, L.J., Jenkins, A., Grizzetti, B., Galloway, J.N., Vitousek, P., Leach, A., Bouwman, A.F., Butterbach-Bahl, K., Dentener, F., Stevenson, D., Amann, M., and Voss, M. (2013) The global nitrogen cycle in the twenty-first century. *Philosophical Transactions of the Royal*

Society B: Biological Sciences **368**, 20130165. doi: 10.1098/rstb.2013.0165

- 428. Liu, J., Ma, K., Ciais, P., and Polasky, S. (2016) Reducing human nitrogen use for food production. *Scientific Reports* **6**, 30104. doi: 10.1038/srep30104
- 429. **Nishibayashi, Y.** (2017) *Nitrogen fixation.* Basel, Switzerland: Springer International Publishing.
- 430. Kuriyama, S., Arashiba, K., Nakajima, K., Matsuo, Y., Tanaka, H., Ishii, K., Yoshizawa, K., and Nishibayashi, Y. (2016) Catalytic transformation of dinitrogen into ammonia and hydrazine by irondinitrogen complexes bearing pincer ligand. *Nature Communications* **7**, 12181. doi: 10.1038/ncomms12181
- 431. Manabe, R., Nakatsubo, H., Gondo, A., Murakami, K., Ogo, S., Tsuneki, H., Ikeda, M., Ishikawa, A., Nakai, H., and Sekine, Y. (2017) Electrocatalytic synthesis of ammonia by surface proton hopping. *Chemical Science* 8, 5434-9. doi: 10.1039/C7SC00840F
- 432. Milton, R.D., Cai, R., Abdellaoui, S., Leech, D., De Lacey, A.L., Pita, M., and Minteer, S.D. (2017) Bioelectrochemical Haber–Bosch process: An ammonia-producing H₂/N₂ fuel cell. *Angewandte Chemie International Edition* 56, 2680-3. doi: 10.1002/anie.201612500
- 433. Mehta, P., Barboun, P., Herrera, F.A., Kim, J., Rumbach, P., Go, D.B., Hicks, J.C., and Schneider, W.F. (2018) Overcoming ammonia synthesis scaling relations with plasma-enabled catalysis. *Nature Catalysis* **1**, 269-75. doi: 10.1038/s41929-018-0045-1
- 434. **Ashida, Y., Arashiba, K., Nakajima, K., and Nishibayashi, Y.** (2019) Molybdenum-catalysed ammonia production with samarium diiodide and alcohols or water. *Nature* **568**, 536-40. doi: 10.1038/s41586-019-1134-2
- 435. **Hawtof, R., Ghosh, S., Guarr, E., Xu, C., Mohan Sankaran, R., and Renner, J.N.** (2019) Catalyst-free, highly selective synthesis of ammonia from nitrogen and water by a plasma electrolytic system. *Science Advances* **5**, eaat5778. doi: 10.1126/sciadv.aat5778
- 436. Giddey, S., Badwal, S.P.S., and Kulkarni, A. (2013) Review of electrochemical ammonia production technologies and materials. *International Journal of Hydrogen Energy* 38, 14576-94. doi: 10.1016/j.ijhydene.2013.09.054
- 437. Anderson, N., Strader, R., and Davidson, C. (2003) Airborne reduced nitrogen: ammonia emissions from agriculture and other sources. *Environment International* **29**, 277-86. doi: 10.1016/S0160-4120(02)00186-1
- 438. Battye, W., Aneja, V.P., and Roelle, P.A. (2003) Evaluation and improvement of ammonia emission inventories. *Atmospheric Environment* 37, 3873-83. doi: 10.1016/S1352-2310(03)00343-1
- 439. Bouwman, A.F., Lee, D.S., Asman, W.A.H., Dentener, F.J., Van Der Hoek, K.W., and Olivier, J.G.J. (1997) A global high-resolution emission inventory for ammonia. *Global Biogeochemical Cycles* **11**, 561-87. doi: 10.1029/97GB02266
- 440. **FAO** (2015) *World fertilizer trends and outlook to 2018* [Online]. Rome, Italy. Available: <u>http://www.fao.org/3/a-i4324e.pdf</u> [Accessed 01/08/2018].
- 441. **DEFRA** (2011) "Defra's Evidence Investment Strategy: 2010–2013 and Beyond: 2011 Update", (ed.) F.a.R.A. Department for Environment. (London, UK).
- 442. **Crown**© (2018) "The British Survey of Fertiliser Practice: Fertiliser Use on Farm Crops for Crop Year 2017", (ed.) DEFRA. (York, UK).
- 443. **Harper, D.** (1992) "What is eutrophication?," in *Eutrophication of freshwaters: Principles, problems and restoration,* ed. **D. Harper**. (Dordrecht: Springer Netherlands), 1-28.

- 444. **Dodds, W.K., and Smith, V.H.** (2016) Nitrogen, phosphorus, and eutrophication in streams. *Inland Waters* **6**, 155-64. doi: 10.5268/IW-6.2.909
- 445. Keeler, B.L., Gourevitch, J.D., Polasky, S., Isbell, F., Tessum, C.W., Hill, J.D., and Marshall, J.D. (2016) The social costs of nitrogen. *Science Advances* **2**, e1600219. doi: 10.1126/sciadv.1600219
- 446. Ward, M.H., de Kok, T.M., Levallois, P., Brender, J., Gulis, G., Nolan, B.T., and VanDerslice, J. (2005) Workgroup report: Drinkingwater nitrate and health—Recent findings and research needs. *Environmental Health Perspectives* **113**, 1607-14. doi: 10.1289/ehp.8043
- Brink, C., van Grinsven, H., with contributing authors, Jacobsen, B.H., Rabl, A., Gren, I.M., Holland, M., Klimont, Z., Hicks, K., Brouwer, R., Dickens, R., Willems, J., Termansen, M., Velthof, G., Alkemade, R., van Oorschot, M., and Webb, J. (2011) "Costs and benefits of nitrogen in the environment," in *The European Nitrogen Assessment*, eds. M.A. Sutton, C.M. Howard, J.W. Erisman, G. Billen, A. Bleeker, P. Grennfelt, H. van Grinsven & B. Grizzetti. (Cambridge, UK: Cambridge University Press).
- Sutton, M.A., Bleeker, A., Howard, C.M., Bekunda, M., Grizzetti, B., 448. van de Vries, W., Grinsven, H.J.M., Abrol, Y.P., Adhya, T.K., Billen, G., Davidson, E.A., Datta, A., Diaz, R., Erisman, J.W., Liu, X.J., Oenema, O., Palm, C., Raghuram, N., Reis, S., Scholz, R.W., Sims, T., Westhoek, H., F.S., Z., with contributions from, Ayyappan, S., Bouwman, A.F., Bustamante, M., Fowler, D., Galloway, J.N., Gavito, M.E., Garnier, J., Greenwood, S., Hellums, D.T., Holland, M., Hoysall, C., Jaramillo, V.J., Klimont, Z., Ometto, J.P., Pathak, H., Plocq Fichelet, V., Powlson, D., Ramakrishna, K., Roy, A., Sanders, K., Sharma, C., Singh, B., Singh, U., Yan, X.Y., and Zhang, Y. (2013) Our Nutrient World: The challenge to produce more food and energy with less pollution. [Online]. Available: https://www.unenvironment.org/resources/report/our-nutrient-worldchallenge-produce-more-food-and-energy-less-pollution [Accessed 01/08/2018].
- 449. **FAO** (2017) *World fertilizer trends and outlook to 2020* [Online]. Rome, Italy. Available: <u>http://www.fao.org/3/a-i6895e.pdf</u> [Accessed 01/08/2018].
- 450. **FAO** (2018) *Nitrogen inputs to agricultural soils from livestock manure: New statistics* [Online]. Rome, Italy. Available: <u>http://www.fao.org/3/I8153EN/i8153en.pdf</u> [Accessed 01/08/2018].
- 451. **Dobbie, K.E., McTaggart, I.P., and Smith, K.A.** (1999) Nitrous oxide emissions from intensive agricultural systems: Variations between crops and seasons, key driving variables, and mean emission factors. *Journal of Geophysical Research: Atmospheres* **104,** 26891-9. doi: 10.1029/1999JD900378
- 452. **Bouwman, A.F., Boumans, L.J.M., and Batjes, N.H.** (2002) Modeling global annual N₂O and NO emissions from fertilized fields. *Global Biogeochemical Cycles* **16,** 28-1--9. doi: 10.1029/2001GB001812
- 453. **Erisman, J.W., Grennfelt, P., and Sutton, M.** (2003) The European perspective on nitrogen emission and deposition. *Environment International* **29**, 311-25. doi: 10.1016/S0160-4120(02)00162-9
- 454. Flynn, H.C., Smith, J., Smith, K.A., Wright, J., Smith, P., and Massheder, J. (2005) Climate- and crop-responsive emission factors significantly alter estimates of current and future nitrous oxide emissions from fertilizer use. *Global Change Biology* **11**, 1522-36. doi: 10.1111/j.1365-2486.2005.00998.x
- 455. **Flynn, H.C., Canals, L.M.i., Keller, E., King, H., Sim, S., Hastings, A., Wang, S., and Smith, P.** (2012) Quantifying global greenhouse gas

emissions from land-use change for crop production. *Global Change Biology* **18**, 1622-35. doi: 10.1111/j.1365-2486.2011.02618.x

- 456. Skiba, U., Jones, S.K., Dragosits, U., Drewer, J., Fowler, D., Rees, R.M., Pappa, V.A., Cardenas, L., Chadwick, D., Yamulki, S., and Manning, A.J. (2012) UK emissions of the greenhouse gas nitrous oxide. *Philosophical Transactions of the Royal Society B: Biological Sciences* 367, 1175-85. doi: 10.1098/rstb.2011.0356
- 457. Shcherbak, I., Millar, N., and Robertson, G.P. (2014) Global metaanalysis of the nonlinear response of soil nitrous oxide (N₂O) emissions to fertilizer nitrogen. *PNAS* **111**, 9199-204. doi: 10.1073/pnas.1322434111
- 458. Chadwick, D.R. (2005) Emissions of ammonia, nitrous oxide and methane from cattle manure heaps: effect of compaction and covering. *Atmospheric Environment* **39**, 787-99. doi: 10.1016/j.atmosenv.2004.10.012
- 459. Rees, R.M., Baddeley, J.A., Bhogal, A., Ball, B.C., Chadwick, D.R., Macleod, M., Lilly, A., Pappa, V.A., Thorman, R.E., Watson, C.A., and Williams, J.R. (2013) Nitrous oxide mitigation in UK agriculture. *Soil Science and Plant Nutrition* **59**, 3-15. doi: 10.1080/00380768.2012.733869
- 460. Bell, M.J., Hinton, N., Cloy, J.M., Topp, C.F.E., Rees, R.M., Cardenas, L., Scott, T., Webster, C., Ashton, R.W., Whitmore, A.P., Williams, J.R., Balshaw, H., Paine, F., Goulding, K.W.T., and Chadwick, D.R. (2015) Nitrous oxide emissions from fertilised UK arable soils: Fluxes, emission factors and mitigation. *Agriculture, Ecosystems & Environment* 212, 134-47. doi: 10.1016/j.agee.2015.07.003
- 461. Ryden, J.C., Whitehead, D.C., Lockyer, D.R., Thompson, R.B., Skinner, J.H., and Garwood, E.A. (1987) Ammonia emission from grassland and livestock production systems in the UK. *Environmental Pollution* 48, 173-84. doi: 10.1016/0269-7491(87)90032-7
- 462. **van der Weerden, T.J., and Jarvis, S.C.** (1997) Ammonia emission factors for N fertilizers applied to two contrasting grassland soils. *Environmental Pollution* **95,** 205-11. doi: 10.1016/S0269-7491(96)00099-1
- 463. Tilman, D., Balzer, C., Hill, J., and Befort, B.L. (2011) Global food demand and the sustainable intensification of agriculture. *PNAS* 108, 20260-4. doi: 10.1073/pnas.1116437108
- Derner, J.D., Hunt, L., Filho, K.E., Ritten, J., Capper, J., and Han, G. (2017) "Livestock Production Systems," in *Rangeland Systems: Processes, Management and Challenges,* ed. D.D. Briske. (Cham: Springer International Publishing), 347-72.
- 465. Sutton, M.A., Oenema, O., Erisman, J.W., Leip, A., van Grinsven, H., and Winiwarter, W. (2011) Too much of a good thing. *Nature* 472, 159-61. doi: 10.1038/472159a
- 466. **Beddington, J.** (2009) Food security: contributions from science to a new and greener revolution. *Philosophical Transactions of the Royal Society B: Biological Sciences* **365,** 61-71. doi: 10.1098/rstb.2009.0201
- 467. Crawford, N.M., and Glass, A.D.M. (1998) Molecular and physiological aspects of nitrate uptake in plants. *Trends Plant Sci* 3, 389-95. doi: 10.1016/S1360-1385(98)01311-9
- 468. **Fan, X., Naz, M., Fan, X., Xuan, W., Miller, A.J., and Xu, G.** (2017) Plant nitrate transporters: from gene function to application. *Journal of Experimental Botany* **68**, 2463-75. doi: 10.1093/jxb/erx011
- 469. Barraclough, D., Geens, E.L., Davies, G.P., and Maggs, J.M. (1985) Fate of fertilizer nitrogen. *Journal of Soil Science* 36, 593-603. doi: 10.1111/j.1365-2389.1985.tb00361.x

- 470. Weigelt, A., Bol, R., and Bardgett, R.D. (2005) Preferential Uptake of Soil Nitrogen Forms by Grassland Plant Species. *Oecologia* **142**, 627-35. doi: 10.1007/s00442-004-1765-2
- 471. Lark, R.M., Milne, A.E., Addiscott, T.M., Goulding, K.W.T., Webster, C.P., and O'Flaherty, S. (2004) Scale- and location-dependent correlation of nitrous oxide emissions with soil properties: an analysis using wavelets. *European Journal of Soil Science* **55**, 611-27. doi: 10.1111/j.1365-2389.2004.00620.x
- 472. Xu, L., Niu, H., Xu, J., and Wang, X. (2013) Nitrate-nitrogen leaching and modeling in intensive agriculture farmland in China. *The Scientific World Journal* **2013**, 353086-. doi: 10.1155/2013/353086
- 473. **Wang, H., Gao, J.-e., Li, X.-h., Zhang, S.-l., and Wang, H.-j.** (2015) Nitrate accumulation and leaching in surface and ground water based on simulated rainfall experiments. *PLoS One* **10**, e0136274. doi: 10.1371/journal.pone.0136274
- 474. **Marchi, E.C.S., Zotarelli, L., Delgado, J.A., Rowland, D.L., and Marchi, G.** (2016) Use of the Nitrogen Index to assess nitrate leaching and water drainage from plastic-mulched horticultural cropping systems of Florida. *International Soil and Water Conservation Research* **4**, 237-44. doi: 10.1016/j.iswcr.2016.12.001
- 475. **Huang, T., Ju, X., and Yang, H.** (2017) Nitrate leaching in a winter wheat-summer maize rotation on a calcareous soil as affected by nitrogen and straw management. *Scientific Reports* **7**, 42247. doi: 10.1038/srep42247
- 476. **Raun, W.R., and Johnson, G.V.** (1999) Improving Nitrogen Use Efficiency for Cereal Production. *Agronomy Journal* **91,** 357-63. doi: 10.2134/agronj1999.00021962009100030001x
- 477. **Kant, S., Bi, Y.-M., and Rothstein, S.J.** (2010) Understanding plant response to nitrogen limitation for the improvement of crop nitrogen use efficiency. *Journal of Experimental Botany* **62,** 1499-509. doi: 10.1093/jxb/erq297
- 478. **Marshall, A.H., Collins, R.P., Humphreys, M.W., and Scullion, J.** (2016) A new emphasis on root traits for perennial grass and legume varieties with environmental and ecological benefits. *Food and Energy Security* **5**, 26-39. doi: 10.1002/fes3.78
- 479. Warren, R.G., and Johnston, A.E. (1964) *The Park Grass experiment* [Online]. Rothamsted, UK. [Accessed 01/08/2018].
- 480. **Johnston, A.E., and Poulton, P.R.** (2018) The importance of long-term experiments in agriculture: their management to ensure continued crop production and soil fertility; the Rothamsted experience. *European Journal of Soil Science* **69**, 113-25. doi: 10.1111/ejss.12521
- 481. **Jenkinson, D.S., Poulton, P.R., Johnston, A.E., and Powlson, D.S.** (2004) Turnover of nitrogen-15-labeled fertilizer in old grassland. *Soil Science Society of America* **68**, 865-75. doi: 10.2136/sssaj2004.8650
- 482. **Postolache, C., and Botez, F.** (2014) Factors controlling key nitrogen fluxes in grassland ecosystems: Case studies Southern Romania. *Environmental engineering and management journal* **13,** 2201-10. doi: 10.30638/eemj.2014.245
- 483. **ADAS** (2007) "Diffuse nitrate pollution from agriculture strategies for reducing nitrate leaching", (ed.) ADAS Report to DEFRA. (London, UK).
- 484. Cardenas, L.M., Gooday, R., Brown, L., Scholefield, D., Cuttle, S., Gilhespy, S., Matthews, R., Misselbrook, T., Wang, J., Li, C., Hughes, G., and Lord, E. (2013) Towards an improved inventory of N₂O from agriculture: Model evaluation of N₂O emission factors and N fraction leached from different sources in UK agriculture. *Atmospheric Environment* **79**, 340-8. doi: 10.1016/j.atmosenv.2013.06.033

- 485. Beckwith, C.P., Lewis, P.J., Chalmers, A.G., Froment, M.A., and Smith, K.A. (2002) Successive annual applications of organic manures for cut grass: short-term observations on utilization of manure nitrogen. *Grass and Forage Science* **57**, 191-202. doi: 10.1046/j.1365-2494.2002.00317.x
- 486. **Humphrey, P.** (2011) "A Request for an Exemption regarding the N max rules, for Grass grown for Dehydration, from the British Association of Green Crop Driers (BAGCD) March 2012 ", in: *DEFRA Nitrate Regulations 2008 Consultation* NIAB TAG,).
- 487. Keeney, D., and Nelson, D. (1982) "Nitrogen-inorganic forms," in Methods of soil analysis - Part 2 - Chemical and microbiological properties, eds. A.L. Page, R.H. Miller & D.R. Keeney. (Madison, Wisconsin: Amercian Society of Agronomy).
- 488. **Smith, K.A., and Li, S.** (1993) Estimation of potentially mineralisable nitrogen in soil by KCl extraction. *Plant and Soil* **157,** 167-74. doi: 10.1007/BF00011045
- 489. Mulvaney, R.L. (1996) "Nitrogen-inorganic forms," in *Methods of soil analysis*, eds. D.L. Sparkes, A.L. Page, P.A. Helmke, R.H. Loeppert, P.N. Soltanpoor, M.A. Tabatabai, C.T. Johnston & M.E. Sumner. (Madison, Wisconsin: SSSA), 1123-84.
- 490. **Maynard, D.G., Klaira, Y.P., and Crumbaugh, J.A.** (2007) "Nitrate and exchangeable ammonium nitrogen," in *Soil sampling and methods of analysis,* eds. **M.R. Carter & E.G. Gregorich**. 2nd ed (Boca Raton Florida: CRC Press, Taylor and Francis Group), 71-80.
- 491. Setatou, H.B., and Simonis, A.D. (1996) Laboratory methods of measuring soil nitrogen status and correlation of measurements with crop responses. *Communications in Soil Science and Plant Analysis* **27**, 651-63. doi: 10.1080/00103629609369585
- 492. **Griess, P.** (1879) Bemerkungen zu der Abhandlung der HH. Weselsky und Benedikt "Ueber einige Azoverbindungen". *Berichte der deutschen chemischen Gesellschaft* **12**, 426-8. doi: 10.1002/cber.187901201117
- 493. **Pasquali, C.E.L., Gallego-Picó, A., Hernando, P.F., Velasco, M., and Alegría, J.S.D.** (2010) Two rapid and sensitive automated methods for the determination of nitrite and nitrate in soil samples. *Microchemical Journal* **94,** 79-82. doi: 10.1016/j.microc.2009.09.005
- 494. **Cameron, K.C., Di, H.J., and Moir, J.L.** (2013) Nitrogen losses from the soil/plant system: a review. *Annals of Applied Biology* **162**, 145-73. doi: 10.1111/aab.12014
- 495. Black, A.S., Sherlock, R.R., Cameron, K.C., Smith, N.P., and Goh, K.M. (1985) Comparison of three field methods for measuring ammonia volatilization from urea granules broadcast on to pasture. *Journal of Soil Science* **36**, 271-80. doi: 10.1111/j.1365-2389.1985.tb00331.x
- 496. **Islam, A., Chen, D., White, R.E., and Weatherley, A.J.** (2008) Chemical decomposition and fixation of nitrite in acidic pasture soils and implications for measurement of nitrification. *Soil Biology and Biochemistry* **40**, 262-5. doi: 10.1016/j.soilbio.2007.07.008
- 497. **Fenn, L.B., and Hossner, L.R.** (1985) "Ammonia volatilization from ammonium or ammonium-forming nitrogen fertilizers," in *Advances in Soil Science*, ed. **B.A. Stewart**. (New York, NY: Springer New York), 123-69.
- 498. **McGarry, S.J., O'Toole, P., and Morgan, M.A.** (1987) Effects of soil temperature and moisture content on ammonia volatilization from ureatreated pasture and tillage soils. *Irish Journal of Agricultural Research* **26,** 173-82.
- 499. Schmidhalter, U. (2005) Development of a quick on-farm test to determine nitrate levels in soil. *Journal of Plant Nutrition and Soil Science* 168, 432-8. doi: 10.1002/jpln.200520521

- 500. **Guntiñas, M.E., Leirós, M.C., Trasar-Cepeda, C., and Gil-Sotres, F.** (2012) Effects of moisture and temperature on net soil nitrogen mineralization: A laboratory study. *European Journal of Soil Biology* **48**, 73-80. doi: 10.1016/j.ejsobi.2011.07.015
- 501. **Al-Kanani, T., MacKenzie, A.F., and Barthakur, N.N.** (1991) Soil water and ammonia volatilization relationships with surface-applied nitrogen fertilizer solutions. *Soil Science Society of America* **55**, 1761-6. doi: 10.2136/sssaj1991.03615995005500060043x
- 502. **Turner, D.A., Edis, R.E., Chen, D., Freney, J.R., and Denmead, O.T.** (2012) Ammonia volatilization from nitrogen fertilizers applied to cereals in two cropping areas of southern Australia. *Nutrient Cycling in Agroecosystems* **93,** 113-26. doi: 10.1007/s10705-012-9504-2
- 503. **Daftardar, S.Y., and Shinde, S.A.** (1980) Kinetics of ammonia volatilization of anhydrous ammonia applied to a vertisol as influenced by farm yard manure, sorbed cations and cation exchange capacity. *Communications in Soil Science and Plant Analysis* **11,** 135-45. doi: 10.1080/00103628009367022
- 504. Whitehead, D.C., and Raistrick, N. (1993) The volatilization of ammonia from cattle urine applied to soils as influenced by soil properties. *Plant and Soil* **148**, 43-51. doi: 10.1007/BF02185383
- 505. Sutton, M.A., Nemitz, E., Erisman, J.W., Beier, C., Bahl, K.B., Cellier, P., de Vries, W., Cotrufo, F., Skiba, U., Di Marco, C., Jones, S., Laville, P., Soussana, J.F., Loubet, B., Twigg, M., Famulari, D., Whitehead, J., Gallagher, M.W., Neftel, A., Flechard, C.R., Herrmann, B., Calanca, P.L., Schjoerring, J.K., Daemmgen, U., Horvath, L., Tang, Y.S., Emmett, B.A., Tietema, A., Peñuelas, J., Kesik, M., Brueggemann, N., Pilegaard, K., Vesala, T., Campbell, C.L., Olesen, J.E., Dragosits, U., Theobald, M.R., Levy, P., Mobbs, D.C., Milne, R., Viovy, N., Vuichard, N., Smith, J.U., Smith, P., Bergamaschi, P., Fowler, D., and Reis, S. (2007) Challenges in quantifying biosphere-atmosphere exchange of nitrogen species. *Environmental Pollution* **150**, 125-39. doi: 10.1016/j.envpol.2007.04.014
- 506. Colman, B.P. (2010) Understanding and eliminating iron interference in colorimetric nitrate and nitrite analysis. *Environmental Monitoring and Assessment* 165, 633-41. doi: 10.1007/s10661-009-0974-x
- 507. **H. Yang, W., Herman, D., Liptzin, D., and Silver, W.** (2012) A new approach for removing iron interference from soil nitrate analysis. *Soil Biology and Biochemistry* **46**, 123–8. doi: 10.1016/j.soilbio.2011.12.003
- 508. **Sawicki, C.R., and Scaringelli, F.P.** (1971) Colorimetric determination of nitrate after hydrazine reduction to nitrite. *Microchemical Journal* **16**, 657-72. doi: 10.1016/0026-265X(71)90059-2
- 509. **Stevens, R.J., and Laughlin, R.J.** (1995) Nitrite transformations during soil extraction with potassium chloride. *Soil Science Society of America* **59**, 933-8. doi: 10.2136/sssaj1995.03615995005900030044x
- 510. **Ponette, Q., Andre, D., and Dufey, J.E.** (1996) Chemical significance of aluminium extracted from three horizons of an acid forest soil, using chloride salt solutions. *European Journal of Soil Science* **47**, 89-95. doi: 10.1111/j.1365-2389.1996.tb01375.x
- 511. **van Maarschalkerweerd, M., and Husted, S.** (2015) Recent developments in fast spectroscopy for plant mineral analysis. *Frontiers in Plant Science* **6**, 169-. doi: 10.3389/fpls.2015.00169
- 512. Delgado-Baquerizo, M., Reich, P.B., Khachane, A.N., Campbell, C.D., Thomas, N., Freitag, T.E., Abu Al-Soud, W., Sørensen, S., Bardgett, R.D., and Singh, B.K. (2017) It is elemental: soil nutrient stoichiometry drives bacterial diversity. *Environmental Microbiology* 19, 1176-88. doi: 10.1111/1462-2920.13642

- 513. Wang, Q., Cameron, K., Buchan, G., Zhao, L., Zhang, E.H., Smith, N., and Carrick, S. (2012) Comparison of lysimeters and porous ceramic cups for measuring nitrate leaching in different soil types. *New Zealand Journal of Agricultural Research* 55, 333-45. doi: 10.1080/00288233.2012.706224
- 514. **McBratney, A., Whelan, B., Ancev, T., and Bouma, J.** (2005) Future directions of precision agriculture. *Precision Agriculture* **6**, 7-23. doi: 10.1007/s11119-005-0681-8
- 515. **Shahandeh, H., Wright, A.L., Hons, F.M., and Lascano, R.J.** (2005) Spatial and temporal variation of soil nitrogen parameters related to soil texture and corn yield. *Agronomy Journal* **97,** 772-82. doi: 10.2134/agronj2004.0287
- 516. **Shanahan, J.F.** (2008) Responsive in-season nitrogen management for cereals. *Computers and electronics in agriculture* **v. 61,** pp. 51-0-2008 v.61 no.1. doi: 10.1016/j.compag.2007.06.006
- 517. Shaw, R., Lark, R.M., Williams, A.P., Chadwick, D.R., and Jones, D.L. (2016) Characterising the within-field scale spatial variation of nitrogen in a grassland soil to inform the efficient design of in-situ nitrogen sensor networks for precision agriculture. *Agriculture, Ecosystems & Environment* 230, 294-306. doi: 10.1016/j.agee.2016.06.004
- 518. **Miller, A.J., and Zhen, R.-G.** (1991) Measurement of intracellular nitrate concentrations in *Chara* using nitrate-selective microelectrodes. *Planta* **184,** 47-52. doi: 10.1007/BF00208235
- 519. **Miller, A.J.** (2013) Real time measurement of cytoplasmic ions with ionselective microelectrodes. *Methods in Molecular Biology* **953**, 243-54. doi: 10.1007/978-1-62703-152-3_16
- 520. Laird, D., Fleming, P., Wang, B., Horton, R., and Karlen, D. (2010) Biochar impact on nutrient leaching from a Midwestern agricultural soil. *Geoderma* **158**, 436-42. doi: 10.1016/j.geoderma.2010.05.012
- 521. Morris, E.C., Griffiths, M., Golebiowska, A., Mairhofer, S., Burr-Hersey, J., Goh, T., von Wangenheim, D., Atkinson, B., Sturrock, C.J., Lynch, J.P., Vissenberg, K., Ritz, K., Wells, D.M., Mooney, S.J., and Bennett, M.J. (2017) Shaping 3D root system architecture. *Current Biology* **27**, R919-R30. doi: 10.1016/j.cub.2017.06.043
- 522. Mairhofer, S., Johnson, J., Sturrock, C.J., Bennett, M.J., Mooney, S.J., and Pridmore, T.P. (2016) Visual tracking for the recovery of multiple interacting plant root systems from X-ray CT images. *Machine Vision and Applications* **27**, 721-34. doi: 10.1007/s00138-015-0733-7
- 523. **Bai, C., Liang, Y., and Hawkesford, M.J.** (2013) Identification of QTLs associated with seedling root traits and their correlation with plant height in wheat. *Journal of Experimental Botany* **64,** 1745-53. doi: 10.1093/jxb/ert041
- 524. **Wojciechowski, T., Gooding, M.J., Ramsay, L., and Gregory, P.J.** (2009) The effects of dwarfing genes on seedling root growth of wheat. *Journal of Experimental Botany* **60,** 2565-73. doi: 10.1093/jxb/erp107
- 525. Merbach, W., Mirus, E., Knof, G., Remus, R., Ruppel, S., Russow, R., Gransee, A., and Schulze, J. (1999) Release of carbon and nitrogen compounds by plant roots and their possible ecological importance. *Journal of Plant Nutrition and Soil Science* 162, 373-83. doi: 10.1002/(SICI)1522-2624(199908)162:4<373::AID-JPLN373>3.0.CO;2-#
- 526. **De Visser, R., Vianden, H., and Schnyder, H.** (1997) Kinetics and relative significance of remobilized and current C and N incorporation in leaf and root growth zones of *Lolium perenne* after defoliation: assessment by 13C and 15N steady-state labelling. *Plant, Cell & Environment* **20,** 37-46. doi: 10.1046/j.1365-3040.1997.d01-9.x

- Jackson, R.B., Mooney, H.A., and Schulze, E.-D. (1997) A global budget for fine root biomass, surface area, and nutrient contents. *PNAS* 94, 7362-6.
- 528. Scheurwater, I., Koren, M., Lambers, H., and Atkin, O.K. (2002) The contribution of roots and shoots to whole plant nitrate reduction in fastand slow-growing grass species. *Journal of Experimental Botany* **53**, 1635-42. doi: 10.1093/jxb/erf008
- 529. Roche, J., Turnbull, M.H., Guo, Q., Novák, O., Späth, J., Gieseg, S.P., Jameson, P.E., and Love, J. (2017) Coordinated nitrogen and carbon remobilization for nitrate assimilation in leaf, sheath and root and associated cytokinin signals during early regrowth of *Lolium perenne*. *Annals of Botany* **119**, 1353-64. doi: 10.1093/aob/mcx014
- 530. **Thornton, B., and Macduff, J.H.** (2002) Short-term changes in xylem N compounds in *Lolium perenne* following defoliation. *Annals of Botany* **89**, 715-22. doi: 10.1093/aob/mcf132
- 531. **Bigot, J., Lefevre, J., and Boucaud, J.** (1991) Changes in the amide and amino acid composition of xylem exudate from perennial ryegrass (*Lolium perenne* L.) during regrowth after defoliation. *Plant and Soil* **136**, 59-64. doi: 10.1007/BF02465220
- 532. Volenec, J.J., Ourry, A., and Joern, B.C. (1996) A role for nitrogen reserves in forage regrowth and stress tolerance. *Physiologia Plantarum* **97**, 185-93. doi: 10.1111/j.1399-3054.1996.tb00496.x
- 533. Clayton, S.J., Read, D.B., Murray, P.J., and Gregory, P.J. (2008) Exudation of alcohol and aldehyde sugars from roots of defoliated *Lolium perenne* L. grown under sterile conditions. *Journal of Chemical Ecology* 34, 1411-21. doi: 10.1007/s10886-008-9536-x
- 534. **Grayston, S.J., Vaughan, D., and Jones, D.** (1997) Rhizosphere carbon flow in trees, in comparison with annual plants: the importance of root exudation and its impact on microbial activity and nutrient availability. *Applied Soil Ecology* **5**, 29-56. doi: 10.1016/S0929-1393(96)00126-6
- 535. **Mawdsley, J.L., and Bardgett, R.D.** (1997) Continuous defoliation of perennial ryegrass (*Lolium perenne*) and white clover (*Trifolium repens*) and associated changes in the composition and activity of the microbial population of an upland grassland soil. *Biology and Fertility of Soils* **24**, 52-8. doi: 10.1007/BF01420220
- 536. **Chen, M., Chen, B., and Marschner, P.** (2008) Plant growth and soil microbial community structure of legumes and grasses grown in monoculture or mixture. *Journal of Environmental Sciences* **20**, 1231-7.
- 537. **Zhou, Y., Zhu, H., Fu, S., and Yao, Q.** (2017) Variation in soil microbial community structure associated with different legume species is greater than that associated with different grass species. *Frontiers in microbiology* **8**, 1007.
- 538. **Crookston, R.K., and Hicks, D.R.** (1988) Effect of early defoliation on maize growth and yield: an eleven-year perspective. *Crop Science* **28**, 371-3. doi: 10.2135/cropsci1988.0011183X002800020040x
- 539. **Wilkinson, S., Bacon, M.A., and Davies, W.J.** (2007) Nitrate signalling to stomata and growing leaves: interactions with soil drying, ABA, and xylem sap pH in maize. *Journal of Experimental Botany* **58**, 1705-16. doi: 10.1093/jxb/erm021
- 540. **Iqbal, N., Masood, A., and Khan, N.A.** (2012) Analyzing the significance of defoliation in growth, photosynthetic compensation and source-sink relations. *Photosynthetica* **50**, 161-70. doi: 10.1007/s11099-012-0029-3
- 541. **Jones, G.B., Alpuerto, J.B., Tracy, B.F., and Fukao, T.** (2017) Physiological effect of cutting height and high temperature on regrowth

vigor in orchardgrass. *Frontiers in Plant Science* **8**, 805-. doi: 10.3389/fpls.2017.00805

- 542. **Naku, M., Kambizi, L., and Matimati, I.** (2019) Functional roles of ammonium (NH₄⁺) and nitrate (NO₃⁻) in regulation of day- and night-time transpiration in *Phaseolus vulgaris*. *Functional Plant Biology*. doi: 10.1071/fp17179
- 543. **Pagès, L.** (2014) Branching patterns of root systems: quantitative analysis of the diversity among dicotyledonous species. *Annals of Botany* **114,** 591-8. doi: 10.1093/aob/mcu145
- 544. **Del Bianco, M., and Kepinski, S.** (2018) Building a future with root architecture. *Journal of Experimental Botany* **69,** 5319-23. doi: 10.1093/jxb/ery390
- 545. Silva, G.d.L.S., Carneiro, M.S.d.S., Furtado, F.M.V., Santos, F.J.d.S., Silva, M.S.d., Costa, N.d.L., and Magalhães, J.A. (2014) Some considerations on the root system of forage plants. *PUBVET* **8**, Art.1687.
- 546. **Paez-Garcia, A., Motes, C., Scheible, W.-R., Chen, R., Blancaflor, E., and Monteros, M.** (2015) Root Traits and Phenotyping Strategies for Plant Improvement. *Plants* **4**, 334. doi: 10.3390/plants4020334
- 547. **Federico, G.** (2005) *Feeding the world: An economic history of agriculture, 1800-2000.* Princeton University Press.
- 548. Schofield, D., Tyson, K.C., Garwood, E.A., Armstrong, A.C., Hawkins, J., and Stone, A.C. (1993) Nitrate leaching from grazed grassland lysimeters: effects of fertilizer input, field drainage, age of sward and patterns of weather. *Journal of Soil Science* **44**, 601-13. doi: 10.1111/j.1365-2389.1993.tb02325.x
- 549. **Edmeades**, **D.C.** (2003) The long-term effects of manures and fertilisers on soil productivity and quality: a review. *Nutrient Cycling in Agroecosystems* **66**, 165-80. doi: 10.1023/A:1023999816690
- 550. Velthof, G.L., Oudendag, D., Witzke, H.P., Asman, W.A., Klimont, Z., and Oenema, O. (2009) Integrated assessment of nitrogen losses from agriculture in EU-27 using MITERRA-EUROPE. *J Environ Qual* **38**, 402-17. doi: 10.2134/jeq2008.0108
- 551. García-Ruiz, J.M., Beguería, S., Nadal-Romero, E., González-Hidalgo, J.C., Lana-Renault, N., and Sanjuán, Y. (2015) A metaanalysis of soil erosion rates across the world. *Geomorphology* **239**, 160-73. doi: 10.1016/j.geomorph.2015.03.008
- 552. **Posthumus, H., Deeks, L.K., Rickson, R.J., and Quinton, J.N.** (2015) Costs and benefits of erosion control measures in the UK. *Soil Use and Management* **31**, 16-33. doi: 10.1111/sum.12057
- 553. **Townsend, T.J., Ramsden, S.J., and Wilson, P.** (2016) How do we cultivate in England? Tillage practices in crop production systems. *Soil Use and Management* **32,** 106-17. doi: 10.1111/sum.12241
- 554. **IFA, and FAO** (2001) "Global estimates of gaseous emissions of NH₃, NO and N₂O from agricultural land", (eds.) International Fertiliser Industry Association & Food and Agriculture Organization of the United Nations. (Rome, Italy: FAO).
- 555. **Bouwman, A.F., Boumans, L.J.M., and Batjes, N.H.** (2002) Emissions of N₂O and NO from fertilized fields: Summary of available measurement data. *Global Biogeochemical Cycles* **16,** 6-1-6-13. doi: 10.1029/2001GB001811
- 556. **Desjardins, R.L., Pattey, E., Smith, W.N., Worth, D., Grant, B., Srinivasan, R., MacPherson, J.I., and Mauder, M.** (2010) Multiscale estimates of N₂O emissions from agricultural lands. *Agricultural and Forest Meteorology* **150,** 817-24. doi: 10.1016/j.agrformet.2009.09.001
- 557. **Crown**© (2018) "Updated Energy and Emissions Projections 2017", (ed.) DEFRA. (London, UK).

- 558. **DEFRA** (2018) "Emissions of air pollutants in the UK, 1970 to 2016 ", (ed.) D.N. Statistics.).
- 559. Herrmann, D., Boller, B., Studer, B., Widmer, F., and Kölliker, R. (2006) QTL analysis of seed yield components in red clover (*Trifolium pratense* L.). *Theoretical Applied Genetics* **112**. doi: 10.1007/s00122-005-0158-1
- 560. **Herrmann, D., Boller, B., Studer, B., Widmer, F., and Kölliker, R.** (2008) Improving persistence in red clover: insights from QTL analysis and comparative phenotypic evaluation. *Crop Science* **48**. doi: 10.2135/cropsci2007.03.0143
- 561. **Molvig, L., Tabe, L.M., Eggum, B.O., Moore, A.E., Craig, S., Spencer, D., and Higgins, T.J.V.** (1997) Enhanced methionine levels and increased nutritive value of seeds of transgenic lupins (*Lupinus angustifolius* L.) expressing a sunflower seed albumin gene. *PNAS* **94**, 8393-8.
- 562. **Wang, Z.Y., and Ge, Y.** (2005) Agrobacterium-mediated high efficiency transformation of tall fescue (*Festuca arundinacea*). *Journal of Plant Physiology* **162,** 103-13. doi: 10.1016/j.jplph.2004.07.009
- 563. **Bajaj, S., Ran, Y., Phillips, J., Kularajathevan, G., Pal, S., Cohen, D., Elborough, K., and Puthigae, S.** (2006) A high throughput *Agrobacterium tumefaciens*-mediated transformation method for functional genomics of perennial ryegrass (*Lolium perenne* L.). *Plant Cell Reports* **25,** 651-9. doi: 10.1007/s00299-005-0099-9
- 564. Chen, T., Yang, Q., Gruber, M., Kang, J., Sun, Y., Ding, W., Zhang, T., and Zhang, X. (2012) Expression of an alfalfa (*Medicago sativa* L.) ethylene response factor gene *MsERF8* in tobacco plants enhances resistance to salinity. *Molecular Biology Reports* **39**, 6067-75. doi: 10.1007/s11033-011-1421-y
- 565. **Chen, T., Yang, Q., Zhang, X., Ding, W., and Gruber, M.** (2012) An alfalfa (*Medicago sativa* L.) ethylene response factor gene, *MsERF11*, enhances salt tolerance in transgenic Arabidopsis. *Plant Cell Reports* **31**, 1737-46. doi: 10.1007/s00299-012-1287-z
- 566. Lee, K.-W., Cha, J.-Y., Kim, K.-H., Kim, Y.-G., Lee, B.-H., and Lee, S.-H. (2012) Overexpression of alfalfa mitochondrial *HSP23* in prokaryotic and eukaryotic model systems confers enhanced tolerance to salinity and arsenic stress. *Biotechnology Letters* **34**, 167-74. doi: 10.1007/s10529-011-0750-1
- 567. Lee, K.-W., Choi, G.J., Kim, K.-Y., Ji, H.J., Park, H.S., Kim, Y.-G., Lee, B.H., and Lee, S.-H. (2012) Transgenic expression of *MsHsp23* confers enhanced tolerance to abiotic stresses in tall fescue. *Asian-Australasian Journal of Animal Sciences* 25, 818-23. doi: 10.5713/ajas.2012.12034
- 568. **Badenhorst, P., Smith, K., and Spangenberg, G.** (2016) Development of a molecular breeding strategy for the integration of transgenic traits in outcrossing perennial grasses. *Agronomy* **6**, 56. doi: 10.3390/agronomy6040056
- 569. Li, Z.-Y., Long, R.-C., Zhang, T.-J., Yang, Q.-C., and Kang, J.-M. (2016) Molecular cloning and characterization of the *MsHSP17.7* gene from *Medicago sativa* L. *Molecular Biology Reports* 43, 815-26. doi: 10.1007/s11033-016-4008-9
- 570. Li, Z., Long, R., Zhang, T., Wang, Z., Zhang, F., Yang, Q., Kang, J., and Sun, Y. (2017) Molecular cloning and functional analysis of the drought tolerance gene *MsHSP70* from alfalfa (*Medicago sativa* L.). *Journal of Plant Research* **130**, 387-96. doi: 10.1007/s10265-017-0905-9
- 571. **Panter, S., Mouradov, A., Badenhorst, P., Martelotto, L., Griffith, M., Smith, K., and Spangenberg, G.** (2017) Re-programming photosynthetic cells of perennial ryegrass (*Lolium perenne* L) for fructan

biosynthesis through transgenic expression of fructan biosynthetic genes under the control of photosynthetic promoters. *Agronomy* **7**, 36. doi: 10.3390/agronomy7020036

- 572. Badenhorst, P.E., Panter, S., Palanisamy, R., Georges, S., Smith, K.F., Mouradov, A., Mason, J., and Spangenberg, G.C. (2018) Molecular breeding of transgenic perennial ryegrass (*Lolium perenne* L.) with altered fructan biosynthesis through the expression of fructosyltransferases. *Molecular Breeding* **38**, 21. doi: 10.1007/s11032-018-0776-3
- 573. **Wang, Z.-Y., and Brummer, E.C.** (2012) Is genetic engineering ever going to take off in forage, turf and bioenergy crop breeding? *Annals of Botany* **110**, 1317-25. doi: 10.1093/aob/mcs027
- 574. Entz, M.H., Baron, V.S., Carr, P.M., Meyer, D.W., Smith, S.R., and McCaughey, W.P. (2002) Potential of forages to diversify cropping systems in the Northern Great Plains. *Agronomy Journal* **94**, 240-50. doi: 10.2134/agronj2002.2400
- 575. Karlen, D.L., Hurley, E.G., Andrews, S.S., Cambardella, C.A., Meek, D.W., Duffy, M.D., and Mallarino, A.P. (2006) Crop rotation effects on soil quality at three northern corn/soybean belt locations. *Agronomy Journal* **98**, 484-95. doi: 10.2134/agronj2005.0098
- 576. Maillard, É., Angers, D.A., Chantigny, M., Lafond, J., Pageau, D., Rochette, P., Lévesque, G., Leclerc, M.-L., and Parent, L.-É. (2016) Greater accumulation of soil organic carbon after liquid dairy manure application under cereal-forage rotation than cereal monoculture. *Agriculture, Ecosystems & Environment* 233, 171-8. doi: 10.1016/j.agee.2016.09.011
- 577. Reckling, M., Bergkvist, G., Watson, C.A., Stoddard, F.L., Zander, P.M., Walker, R.L., Pristeri, A., Toncea, I., and Bachinger, J. (2016) Trade-offs between economic and environmental impacts of introducing legumes into cropping systems. *Frontiers in Plant Science* **7**. doi: 10.3389/fpls.2016.00669
- 578. Reckling, M., Hecker, J.-M., Bergkvist, G., Watson, C.A., Zander, P., Schläfke, N., Stoddard, F.L., Eory, V., Topp, C.F.E., Maire, J., and Bachinger, J. (2016) A cropping system assessment framework— Evaluating effects of introducing legumes into crop rotations. *European Journal of Agronomy* **76**, 186-97. doi: 10.1016/j.eja.2015.11.005
- 579. Smith, L.G., Tarsitano, D., Topp, C.F.E., Jones, S.K., Gerrard, C.L., Pearce, B.D., Williams, A.G., and Watson, C.A. (2016) Predicting the effect of rotation design on N, P, K balances on organic farms using the NDICEA model. *Renewable Agriculture and Food Systems* **31**, 471-84. doi: 10.1017/S1742170515000381
- 580. Zander, P., Amjath-Babu, T.S., Preissel, S., Reckling, M., Bues, A., Schläfke, N., Kuhlman, T., Bachinger, J., Uthes, S., Stoddard, F., Murphy-Bokern, D., and Watson, C. (2016) Grain legume decline and potential recovery in European agriculture: a review. Agronomy for Sustainable Development 36, 26. doi: 10.1007/s13593-016-0365-y
- 581. Hermle, S., Anken, T., Leifeld, J., and Weisskopf, P. (2008) The effect of the tillage system on soil organic carbon content under moist, cold-temperate conditions. *Soil and Tillage Research* **98**, 94-105. doi: 10.1016/j.still.2007.10.010
- 582. Thomas, S.M., Beare, M.H., Francis, G.S., Barlow, H.E., and Hedderley, D.I. (2008) Effects of tillage, simulated cattle grazing and soil moisture on N₂O emissions from a winter forage crop. *Plant and Soil* 309, 131. doi: 10.1007/s11104-008-9586-4
- 583. Krauss, M., Berner, A., Burger, D., Wiemken, A., Niggli, U., and Mäder, P. (2010) Reduced tillage in temperate organic farming:

implications for crop management and forage production. *Soil Use and Management* **26,** 12-20. doi: 10.1111/j.1475-2743.2009.00253.x

- 584. Crotty, F.V., Fychan, R., Sanderson, R., Rhymes, J.R., Bourdin, F., Scullion, J., and Marley, C.L. (2016) Understanding the legacy effect of previous forage crop and tillage management on soil biology, after conversion to an arable crop rotation. *Soil Biology and Biochemistry* **103**, 241-52. doi: 10.1016/j.soilbio.2016.08.018
- 585. **Firbank, L.G., Petit, S., Smart, S., Blain, A., and Fuller, R.J.** (2008) Assessing the impacts of agricultural intensification on biodiversity: a British perspective. *Philosophical Transactions of the Royal Society B: Biological Sciences* **363,** 777-87. doi: 10.1098/rstb.2007.2183
- 586. Henle, K., Alard, D., Clitherow, J., Cobb, P., Firbank, L., Kull, T., McCracken, D., Moritz, R.F.A., Niemelä, J., Rebane, M., Wascher, D., Watt, A., and Young, J. (2008) Identifying and managing the conflicts between agriculture and biodiversity conservation in Europe–A review. Agriculture, Ecosystems & Environment **124**, 60-71. doi: 10.1016/j.agee.2007.09.005
- 587. Scherr, S.J., and McNeely, J.A. (2008) Biodiversity conservation and agricultural sustainability: towards a new paradigm of 'ecoagriculture' landscapes. *Philosophical Transactions of the Royal Society B: Biological Sciences* **363**, 477-94. doi: 10.1098/rstb.2007.2165
- 588. Sutherland, W.J., Bailey, M.J., Bainbridge, I.P., Brereton, T., Dick, J.T.A., Drewitt, J., Dulvy, N.K., Dusic, N.R., Freckleton, R.P., Gaston, K.J., Gilder, P.M., Green, R.E., Heathwaite, A.L., Johnson, S.M., Macdonald, D.W., Mitchell, R., Osborn, D., Owen, R.P., Pretty, J., Prior, S.V., Prosser, H., Pullin, A.S., Rose, P., Stott, A., Tew, T., Thomas, C.D., Thompson, D.B.A., Vickery, J.A., Walker, M., Walmsley, C., Warrington, S., Watkinson, A.R., Williams, R.J., Woodroffe, R., and Woodroof, H.J. (2008) Future novel threats and opportunities facing UK biodiversity identified by horizon scanning. *Journal of Applied Ecology* 45, 821-33. doi: 10.1111/j.1365-2664.2008.01474.x
- 589. **European Commission** (2011) *The EU Biodiversity Strategy to 2020* [Online]. Luxembourg. Available: doi: 10.2779/39229 [Accessed].
- 590. Harrison, P.A., Berry, P.M., Simpson, G., Haslett, J.R., Blicharska, M., Bucur, M., Dunford, R., Egoh, B., Garcia-Llorente, M., Geamănă, N., Geertsema, W., Lommelen, E., Meiresonne, L., and Turkelboom, F. (2014) Linkages between biodiversity attributes and ecosystem services: A systematic review. *Ecosystem Services* 9, 191-203. doi: 10.1016/j.ecoser.2014.05.006
- 591. Stockdale, E.A., Griffiths, B.S., Hargreaves, P.R., Bhogal, A., Crotty, F.V., and Watson, C.A. (2019) Conceptual framework underpinning management of soil health—supporting site-specific delivery of sustainable agro-ecosystems. *Food and Energy Security* **8**, e00158. doi: 10.1002/fes3.158
- 592. **Bianca, H., Jihad, M.A., Saul, B., Adnan, R., Shlomo, S., and Yaacov, O.** (2001) Effects of inoculation with *Azospirillum brasilense* on chickpeas (*Cicer arietinum*) and faba beans (*Vicia faba*) under different growth conditions. *Agronomie* **21,** 553-60. doi: 10.1051/agro:2001144
- 593. **Xunzhong, Z., Schmidt, R.E., Ervin, E.H., and Doak, S.** (2002) Creeping bentgrass physiological responses to natural plant growth regulators and iron under two regimes. *HortScience* **37**, 898-902. doi: 10.21273/HORTSCI.37.6.898
- 594. **Pereyra, M.A., Zalazar, C.A., and Barassi, C.A.** (2006) Root phospholipids in *Azospirillum*-inoculated wheat seedlings exposed to water stress. *Plant Physiology and Biochemistry* **44,** 873-9. doi: 10.1016/j.plaphy.2006.10.020

- 595. **Del Pilar Cordovilla, M., Berrido, S.I., Ligero, F., and Lluch, C.** (1999) *Rhizobium* strain effects on the growth and nitrogen assimilation in *Pisum sativum* and *Vicia faba* plant growth under salt stress. *Journal of Plant Physiology* **154,** 127-31. doi: 10.1016/S0176-1617(99)80328-9
- 596. **Egamberdiyeva, D., and Höflich, G.** (2003) Influence of growthpromoting bacteria on the growth of wheat in different soils and temperatures. *Soil Biology and Biochemistry* **35,** 973-8. doi: 10.1016/S0038-0717(03)00158-5
- 597. Rojas-Tapias, D., Moreno-Galván, A., Pardo-Díaz, S., Obando, M., Rivera, D., and Bonilla, R. (2012) Effect of inoculation with plant growth-promoting bacteria (PGPB) on amelioration of saline stress in maize (*Zea mays*). *Applied Soil Ecology* **61**, 264-72. doi: 10.1016/j.apsoil.2012.01.006
- 598. Su, A.Y., Niu, S.Q., Liu, Y.Z., He, A.L., Zhao, Q., Pare, P.W., Li, M.F., Han, Q.Q., Ali Khan, S., and Zhang, J.L. (2017) Synergistic effects of *Bacillus amyloliquefaciens* (GB03) and water retaining agent on drought tolerance of perennial ryegrass. *Int J Mol Sci* 18. doi: 10.3390/ijms18122651
- 599. He, A.-L., Niu, S.-Q., Zhao, Q., Li, Y.-S., Gou, J.-Y., Gao, H.-J., Suo, S.-Z., and Zhang, J.-L. (2018) Induced salt tolerance of perennial ryegrass by a novel bacterium strain from the rhizosphere of a desert shrub *Haloxylon ammodendron*. *Int J Mol Sci* **19**, 469. doi: 10.3390/ijms19020469
- 600. Nabati, D.A., Schmidt, R.E., and Parrish, D.J. (1994) Alleviation of salinity stress in kentucky bluegrass by plant growth regulators and iron. *Crop Science* 34, 198-202. doi: 10.2135/cropsci1994.0011183X003400010035x
- 601. Zhang, X., and E. Schmidt, R. (2000) Hormone-containing products' impact on antioxidant status of tall fescue and creeping bentgrass subjected to drought. *Crop Science* 40. doi: 10.2135/cropsci2000.4051344x
- 602. **Botta, A.** (Year) "Enhancing plant tolerance to temperature stress with amino acids: an approach to their mode of action": International Society for Horticultural Science (ISHS), Leuven, Belgium), 29-35.
- 603. Van Oosten, M.J., Pepe, O., De Pascale, S., Silletti, S., and Maggio, A. (2017) The role of biostimulants and bioeffectors as alleviators of abiotic stress in crop plants. *Chemical and Biological Technologies in Agriculture* 4, 5. doi: 10.1186/s40538-017-0089-5
- 604. Schirawski, J., and Perlin, M.H. (2018) Plant-microbe interaction 2017-The good, the bad and the diverse. *Int J Mol Sci* **19**, 1374. doi: 10.3390/ijms19051374
- 605. **Gaffney, J.S., Marley, N.A., and Clark, S.B.** (1996) "Humic and fulvic acids and organic colloidal materials in the environment," in *Humic and Fulvic Acids*. American Chemical Society), 2-16.
- 606. **Sutton, R., and Sposito, G.** (2005) Molecular structure in soil humic substances: The new view. *Environmental Science & Technology* **39**, 9009-15. doi: 10.1021/es050778q
- 607. **Canellas, L.P., Piccolo, A., Dobbss, L.B., Spaccini, R., Olivares, F.L., Zandonadi, D.B., and Façanha, A.R.** (2010) Chemical composition and bioactivity properties of size-fractions separated from a vermicompost humic acid. *Chemosphere* **78**, 457-66. doi: 10.1016/j.chemosphere.2009.10.018
- 608. **Trevisan, S., Francioso, O., Quaggiotti, S., and Nardi, S.** (2010) Humic substances biological activity at the plant-soil interface. *Plant Signaling & Behavior* **5**, 635-43. doi: 10.4161/psb.5.6.11211

- 609. Lamar, R., Olk, D.C., Mayhew, L., and Bloom, P.R. (2013) Evaluation of a proposed standardized analytical method for the determination of humic and fulvic acids in commercial products.
- 610. **Calvo, P., Nelson, L., and Kloepper, J.W.** (2014) Agricultural uses of plant biostimulants. *Plant and Soil* **383,** 3-41. doi: 10.1007/s11104-014-2131-8
- 611. Rose, M.T., Patti, A.F., Little, K.R., Brown, A.L., Jackson, W.R., and Cavagnaro, T.R. (2014) "Chapter two A meta-analysis and review of plant-growth response to humic substances: practical implications for agriculture," in *Advances in Agronomy*, ed. **D.L. Sparks**. Academic Press), 37-89.
- 612. Halpern, M., Bar-Tal, A., Ofek, M., Minz, D., Muller, T., and Yermiyahu, U. (2015) "Chapter two - The use of biostimulants for enhancing nutrient uptake," in *Advances in Agronomy*, ed. D.L. Sparks. Academic Press), 141-74.
- 613. Shah, Z.H., Rehman, H.M., Akhtar, T., Alsamadany, H., Hamooh, B.T., Mujtaba, T., Daur, I., Al Zahrani, Y., Alzahrani, H.A.S., Ali, S., Yang, S.H., and Chung, G. (2018) Humic substances: Determining potential molecular regulatory processes in plants. *Frontiers in Plant Science* **9**. doi: 10.3389/fpls.2018.00263
- 614. **Piccolo, A., Pietramellara, G., and Mbagwu, J.S.C.** (1997) Use of humic substances as soil conditioners to increase aggregate stability. *Geoderma* **75**, 267-77. doi: 10.1016/S0016-7061(96)00092-4
- 615. **Schulten, H.R., and Leinweber, P.** (2000) New insights into organicmineral particles: composition, properties and models of molecular structure. *Biology and Fertility of Soils* **30,** 399-432. doi: 10.1007/s003740050020
- 616. **Frimmel, F.H.** (2005) "Aquatic humic substances," in *Biopolymers Online*, ed. **A. Steinbüchel**.).
- 617. **Rodríguez, F.J., and Núñez, L.A.** (2011) Characterization of aquatic humic substances. *Water and Environment Journal* **25,** 163-70. doi: 10.1111/j.1747-6593.2009.00205.x
- 618. **Skogerboe, R.K., and Wilson, S.A.** (1981) Reduction of ionic species by fulvic acid. *Analytical chemistry* **53**, 228-32. doi: 10.1021/ac00225a023
- 619. Ayuso, M., Hernández, T., and García, C. (1996) Effect of humic fractions from urban wastes and other more evolved organic materials on seed germination. *Journal of the Science of Food and Agriculture* 72, 461-8. doi: 10.1002/(SICI)1097-0010(199612)72:4<461::AID-JSFA680>3.0.CO;2-#
- 620. **Halim, M., Conte, P., and Piccolo, A.** (2003) Potential availability of heavy metals to phytoextraction from contaminated soils induced by exogenous humic substances. *Chemosphere* **52**, 265-75. doi: 10.1016/S0045-6535(03)00185-1
- 621. **Iglesias, A., López, R., Gondar, D., Antelo, J., Fiol, S., and Arce, F.** (2009) Effect of pH and ionic strength on the binding of paraquat and MCPA by soil fulvic and humic acids. *Chemosphere* **76**, 107-13. doi: 10.1016/j.chemosphere.2009.02.012
- 622. García, A.C., de Souza, L.G.A., Pereira, M.G., Castro, R.N., García-Mina, J.M., Zonta, E., Lisboa, F.J.G., and Berbara, R.L.L. (2016) Structure-property-function relationship in humic substances to explain the biological activity in plants. *Scientific Reports* **6**, 20798. doi: 10.1038/srep20798
- 623. Aiken, G.R., McKnight, D.M., Thorn, K.A., and Thurman, E.M. (1992) Isolation of hydrophilic organic acids from water using nonionic macroporous resins. *Organic Geochemistry* **18**, 567-73. doi: 10.1016/0146-6380(92)90119-I

- 624. Swift, R.S. (1996) "Organic matter characterization," in *Methods of soil* analysis Part 3 Chemical methods, eds. D.L. Sparks, S.S.S.o.
 America & A.S.o. Agronomy. (Madison, Wisconsin, US: Soil Science Society of America).
- 625. **Peuravuori, J., and Pihlaja, K.** (1997) Isolation and characterization of natural organic matter from lake water: Comparison of isolation with solid adsorption and tangential membrane filtration. *Environment International* **23,** 441-51. doi: 10.1016/S0160-4120(97)00049-4
- 626. Fujitake, N., Strynar, M.J., Mishima, T., Tsukamoto, M., Yanagi, Y., Suzuki, T., Otsuka, H., Dec, J., and Ballag, J.-M. (2003) Optimization of a methylation procedure to obtain chloroform-soluble humic acids. *Soil Science and Plant Nutrition* **49**, 453-7. doi: 10.1080/00380768.2003.10410031
- 627. **Baglieri, A., Ioppolo, A., Nègre, M., and Gennari, M.** (2007) A method for isolating soil organic matter after the extraction of humic and fulvic acids. *Organic Geochemistry* **38**, 140-50. doi: 10.1016/j.orggeochem.2006.07.007
- 628. Brezonik, P.L., Bloom, P.R., Sleighter, R.L., Cory, R.M., Khwaja, A.R., and Hatcher, P.G. (2015) Chemical differences of aquatic humic substances extracted by XAD-8 and DEAE-cellulose. *Journal of Environmental Chemical Engineering* **3**, 2982-90. doi: 10.1016/j.jece.2015.03.004
- 629. Zherebker, A., Shirshin, E., Kharybin, O., Kostyukevich, Y., Kononikhin, A., Konstantinov, A.I., Volkov, D., Roznyatovsky, V.A., Grishin, Y.K., Perminova, I.V., and Nikolaev, E. (2018) Separation of benzoic and unconjugated acidic components of leonardite humic material using sequential solid-phase extraction at different pH values as revealed by fourier transform ion cyclotron resonance mass spectrometry and correlation nuclear magnetic resonance pectroscopy. *Journal of Agricultural and Food Chemistry* 66, 12179-87. doi: 10.1021/acs.jafc.8b04079
- 630. **Sabi, R., Vrey, P., and van Rensburg, C.E.J.** (2012) Carbohydratederived fulvic acid (CHD-FA) inhibits Carrageenan-induced inflammation and enhances wound healing: efficacy and toxicity study in rats. *Drug Development Research* **73**, 18-23. doi: 10.1002/ddr.20445
- 631. García-Mina, J.M., Antolín, M.C., and Sanchez-Diaz, M. (2004) Metal-humic complexes and plant micronutrient uptake: a study based on different plant species cultivated in diverse soil types. *Plant and Soil* **258**, 57-68. doi: 10.1023/B:PLSO.0000016509.56780.40
- 632. **Gondar, D., López, R., Fiol, S., Antelo, J.M., and Arce, F.** (2006) Cadmium, lead, and copper binding to humic acid and fulvic acid extracted from an ombrotrophic peat bog. *Geoderma* **135**, 196-203. doi: 10.1016/j.geoderma.2005.12.003
- 633. **Pandeya, S.B., Singh, A.K., and Dhar, P.** (1998) Influence of fulvic acid on transport of iron in soils and uptake by paddy seedlings. *Plant and Soil* **198,** 117-25. doi: 10.1023/A:1004256325090
- 634. Zhimang, G., Xiaorong, W., Xueyuan, G., Jing, C., Liansheng, W., Lemei, D., and Yijun, C. (2001) Effects of fulvic acid on the bioavailability of rare earth elements and GOT enzyme activity in wheat (*Triticum aestivum*). Chemosphere **44**, 545-51. doi: 10.1016/S0045-6535(00)00484-7
- 635. **Pinto, A.P., Mota, A.M., de Varennes, A., and Pinto, F.C.** (2004) Influence of organic matter on the uptake of cadmium, zinc, copper and iron by sorghum plants. *Science of The Total Environment* **326**, 239-47. doi: 10.1016/j.scitotenv.2004.01.004
- 636. Dobbss, L.B., Medici, L.O., Peres, L.E.P., Pino-Nunes, L.E., Rumjanek, V.M., Façanha, A.R., and Canellas, L.P. (2007) Changes

in root development of *Arabidopsis* promoted by organic matter from oxisols. *Annals of Applied Biology* **151**, 199-211. doi: 10.1111/j.1744-7348.2007.00166.x

- 637. García, A.C., Santos, L.A., Izquierdo, F.G., Sperandio, M.V.L., Castro, R.N., and Berbara, R.L.L. (2012) Vermicompost humic acids as an ecological pathway to protect rice plant against oxidative stress. *Ecological Engineering* **47**, 203-8. doi: 10.1016/j.ecoleng.2012.06.011
- 638. Vaccaro, S., Ertani, A., Nebbioso, A., Muscolo, A., Quaggiotti, S., Piccolo, A., and Nardi, S. (2015) Humic substances stimulate maize nitrogen assimilation and amino acid metabolism at physiological and molecular level. *Chemical and Biological Technologies in Agriculture* **2**, 5. doi: 10.1186/s40538-015-0033-5
- 639. **Zhang, X., Zhang, X., Liu, X., Shao, L., Sun, H., and Chen, S.** (2015) Improving winter wheat performance by foliar spray of ABA and FA under water deficit conditions. *Journal of Plant Growth Regulation* **35**, 83-96. doi: 10.1007/s00344-015-9509-6
- 640. **Bezuglova, O.S., Polienko, E.A., Gorovtsov, A.V., Lyhman, V.A., and Pavlov, P.D.** (2017) The effect of humic substances on winter wheat yield and fertility of ordinary chernozem. *Annals of Agrarian Science* **15**, 239-42. doi: 10.1016/j.aasci.2017.05.006
- 641. Nardi, S., Pizzeghello, D., Muscolo, A., and Vianello, A. (2002) Physiological effects of humic substances on higher plants. *Soil Biology and Biochemistry* **34**, 1527-36. doi: 10.1016/S0038-0717(02)00174-8
- 642. **Chen, Y., Clapp, C.E., and Magen, H.** (2004) Mechanisms of plant growth stimulation by humic substances: The role of organo-iron complexes. *Soil Science and Plant Nutrition* **50**, 1089-95. doi: 10.1080/00380768.2004.10408579
- 643. **Trevisan, S., Francioso, O., Quaggiotti, S., and Nardi, S.** (2010) Humic substances biological activity at the plant-soil interface: From environmental aspects to molecular factors. *Plant Signaling & Behavior* **5**, 635-43. doi: 10.4161/psb.5.6.11211
- 644. **Nardi, S., Pizzeghello, D., Schiavon, M., and Ertani, A.** (2016) Plant biostimulants: physiological responses induced by protein hydrolyzedbased products and humic substances in plant metabolism. *Scientia Agricola* **73**, 18-23. doi: 10.1590/0103-9016-2015-0006
- 645. **Tan, K.H., and Tantiwiramanond, D.** (1983) Effect of humic acids on nodulation and dry matter production of soybean, peanut, and clover. *Soil Science Society of America* **47**, 1121-4. doi: 10.2136/sssaj1983.03615995004700060012x
- 646. **Maji, D., Misra, P., Singh, S., and Kalra, A.** (2017) Humic acid rich vermicompost promotes plant growth by improving microbial community structure of soil as well as root nodulation and mycorrhizal colonization in the roots of *Pisum sativum. Applied Soil Ecology* **110,** 97-108. doi: 10.1016/j.apsoil.2016.10.008
- 647. Little, K., Rose, M., Patti, A., Cavagnaro, T., and Jackson, R. (Year) "Effect of application rate of commercial lignite coal-derived amendments on early-stage growth of *Medicago sativa* and soil health, in acidic soil conditions": Springer Netherlands), 1085-8.
- 648. Little, K.R., Rose, M.T., Jackson, W.R., Cavagnaro, T.R., and Patti, A.F. (2014) Do lignite-derived organic amendments improve early-stage pasture growth and key soil biological and physicochemical properties? *Crop and Pasture Science* **65**, 899-910. doi: 10.1071/CP13433
- 649. **Aydin, A., Kant, C., and Turan, M.** (2012) Humic acid application alleviate salinity stress of bean (*Phaseolus vulgaris* L.) plants decreasing membrane leakage. *African Journal of Agricultural Research* **7**, 1073-86. doi: 10.5897/AJAR10.274

- 650. **Traversa, A., Loffredo, E., Gattullo, C.E., Palazzo, A.J., Bashore, T.L., and Senesi, N.** (2014) Comparative evaluation of compost humic acids and their effects on the germination of switchgrass (*Panicum vigatum* L.). *Journal of Soils and Sediments* **14,** 432-40. doi: 10.1007/s11368-013-0653-y
- 651. Daneshvar Hakimi Maibodi, N., Kafi, M., Nikbakht, A., and Rejali, F. (2015) Effect of foliar applications of humic acid on growth, visual quality, nutrients content and root parameters of perennial ryegrass (*Lolium perenne* L.). *Journal of Plant Nutrition* **38**, 224-36. doi: 10.1080/01904167.2014.939759
- 652. **Chang, Z., Liu, Y., Dong, H., Teng, K., Han, L., and Zhang, X.** (2016) Effects of cytokinin and nitrogen on drought tolerance of creeping bentgrass. *PLoS One* **11**, e0154005-e. doi: 10.1371/journal.pone.0154005
- 653. **Nardi, S., Panuccio, M.R., Abenavoli, M.R., and Muscolo, A.** (1994) Auxin-like effect of humic substances extracted from faeces of *Allolobophora caliginosa* and *A. rosea*. *Soil Biology and Biochemistry* **26**, 1341-6. doi: 10.1016/0038-0717(94)90215-1
- 654. **Russell, L., Stokes, A.R., Macdonald, H., Muscolo, A., and Nardi, S.** (2006) Stomatal responses to humic substances and auxin are sensitive to inhibitors of phospholipase A2. *Plant and Soil* **283,** 175-85. doi: 10.1007/s11104-006-0011-6
- 655. **Trevisan, S., Pizzeghello, D., Ruperti, B., Francioso, O., Sassi, A., Palme, K., Quaggiotti, S., and Nardi, S.** (2010) Humic substances induce lateral root formation and expression of the early auxinresponsive *IAA19* gene and *DR5* synthetic element in *Arabidopsis. Plant Biology* **12,** 604-14. doi: 10.1111/j.1438-8677.2009.00248.x
- 656. **Piccolo, A., Celano, G., and Pietramellara, G.** (1993) Effects of fractions of coal-derived humic substances on seed germination and growth of seedlings (*Lactuga sativa* and *Lycopersicum esculentum*). *Biology and Fertility of Soils* **16**, 11-5. doi: 10.1007/BF00336508
- 657. **Asenjo, M.C.G., González, J.L., and Maldonado, J.M.** (2000) Influence of humic extracts on germination and growth of ryegrass. *Communications in Soil Science and Plant Analysis* **31,** 101-14. doi: 10.1080/00103620009370423
- 658. **Türkmen, Ö., Dursun, A., Turan, M., and Erdinç, Ç.** (2004) Calcium and humic acid affect seed germination, growth, and nutrient content of tomato (*Lycopersicon esculentum* L.) seedlings under saline soil conditions. *Acta Agriculturae Scandinavica, Section B — Soil & Plant Science* **54,** 168-74. doi: 10.1080/09064710310022014
- 659. **Allaway, W.H.** (1968) "Agronomic controls over the environmental cycling of trace elements," in *Advances in Agronomy*, ed. **A.G. Norman**. Academic Press), 235-74.
- 660. **Nagajyoti, P.C., Lee, K.D., and Sreekanth, T.V.M.** (2010) Heavy metals, occurrence and toxicity for plants: a review. *Environmental Chemistry Letters* **8**, 199-216. doi: 10.1007/s10311-010-0297-8
- 661. **Miyasaka, S.C., Buta, J.G., Howell, R.K., and Foy, C.D.** (1991) Mechanism of aluminum tolerance in snapbeans : root exudation of citric acid. *Plant Physiology* **96,** 737-43. doi: 10.1104/pp.96.3.737
- 662. **Lawlor, D.W.** (1970) Absorption of polyethylene glycols by plants and their effects on plant growth. *New Phytologist* **69**, 501-13. doi: 10.1111/j.1469-8137.1970.tb02446.x
- 663. **Williams, T.M.** (2006) "The mechanism of action of isothiazolone biocide", in: *Corrosion 2006.* (San Diego, California: NACE International).
- 664. **Barman, B.N., and Preston, H.G.** (1992) The effects of pH on the degradation of isothiazolone biocides. *Tribology International* **25**, 281-7. doi: 10.1016/0301-679X(92)90065-U

- 665. **Cavigelli, M.A., Robertson, G.P., and Klug, M.J.** (1995) Fatty acid methyl ester (FAME) profiles as measures of soil microbial community structure. *Plant and Soil* **170**, 99-113. doi: 10.1007/BF02183058
- 666. Griffiths, B.S., Ritz, K., Ebblewhite, N., and Dobson, G. (1998) Soil microbial community structure: Effects of substrate loading rates. *Soil Biology and Biochemistry* **31**, 145-53. doi: 10.1016/S0038-0717(98)00117-5
- 667. **Fierer, N., Schimel, J.P., and Holden, P.A.** (2003) Variations in microbial community composition through two soil depth profiles. *Soil Biology and Biochemistry* **35,** 167-76. doi: 10.1016/S0038-0717(02)00251-1
- 668. Grayston, S.J., Campbell, C.D., Bardgett, R.D., Mawdsley, J.L., Clegg, C.D., Ritz, K., Griffiths, B.S., Rodwell, J.S., Edwards, S.J., Davies, W.J., Elston, D.J., and Millard, P. (2004) Assessing shifts in microbial community structure across a range of grasslands of differing management intensity using CLPP, PLFA and community DNA techniques. *Applied Soil Ecology* **25**, 63-84. doi: 10.1016/S0929-1393(03)00098-2
- 669. **Ramsey, P.W., Rillig, M.C., Feris, K.P., Holben, W.E., and Gannon, J.E.** (2006) Choice of methods for soil microbial community analysis: PLFA maximizes power compared to CLPP and PCR-based approaches. *Pedobiologia* **50**, 275-80. doi: 10.1016/j.pedobi.2006.03.003
- 670. **Cardinali, A., Pizzeghello, D., and Zanin, G.** (2015) Fatty acid methyl ester (FAME) succession in different substrates as affected by the coapplication of three pesticides. *PLoS One* **10**, e0145501. doi: 10.1371/journal.pone.0145501
- 671. **Kim Thi Tran, C., Rose, M.T., Cavagnaro, T.R., and Patti, A.F.** (2015) Lignite amendment has limited impacts on soil microbial communities and mineral nitrogen availability. *Applied Soil Ecology* **95,** 140-50. doi: 10.1016/j.apsoil.2015.06.020
- 672. **Yao, H., Chapman, S.J., Thornton, B., and Paterson, E.** (2015) 13C PLFAs: a key to open the soil microbial black box? *Plant and Soil* **392,** 3-15. doi: 10.1007/s11104-014-2300-9
- 673. **Guo Gao, T., Yuan Xu, Y., Jiang, F., Zhen Li, B., Shui Yang, J., Tao Wang, E., and Li Yuan, H.** (2015) Nodulation characterization and proteomic profiling of *Bradyrhizobium liaoningense CCBAU05525* in response to water-soluble humic materials. *Scientific Reports* **5**, 10836. doi: 10.1038/srep10836
- 674. **Visser, S.A.** (1985) Physiological action of humic substances on microbial cells. *Soil Biology and Biochemistry* **17**, 457-62. doi: 10.1016/0038-0717(85)90009-4
- 675. **McLoughlin, A.J., and Küster, E.** (1972) The effect of humic substances on the respiration and growth of micro-organisms. *Plant and Soil* **37,** 17-25. doi: 10.1007/BF01578475
- 676. Lüscher, A., Mueller-Harvey, I., Soussana, J.F., Rees, R.M., and Peyraud, J.L. (2014) Potential of legume-based grassland-livestock systems in Europe: a review. *Grass and Forage Science* **69**, 206-28. doi: 10.1111/gfs.12124
- 677. **Preissel, S., Reckling, M., Schläfke, N., and Zander, P.** (2015) Magnitude and farm-economic value of grain legume pre-crop benefits in Europe: A review. *Field Crops Research* **175,** 64-79. doi: 10.1016/j.fcr.2015.01.012
- 678. Foyer, C.H., Lam, H.M., Nguyen, H.T., Siddique, K.H., Varshney, R.K., Colmer, T.D., Cowling, W., Bramley, H., Mori, T.A., Hodgson, J.M., Cooper, J.W., Miller, A.J., Kunert, K., Vorster, J., Cullis, C., Ozga, J.A., Wahlqvist, M.L., Liang, Y., Shou, H., Shi, K., Yu, J., Fodor, N., Kaiser, B.N., Wong, F.L., Valliyodan, B., and Considine, M.J. (2016) Neglecting legumes has compromised human health and

sustainable food production. *Nat Plants* **2**, 16112. doi: 10.1038/nplants.2016.112

- 679. Iannetta, P.P., Young, M., Bachinger, J., Bergkvist, G., Doltra, J., Lopez-Bellido, R.J., Monti, M., Pappa, V.A., Reckling, M., Topp, C.F., Walker, R.L., Rees, R.M., Watson, C.A., James, E.K., Squire, G.R., and Begg, G.S. (2016) A comparative nitrogen balance and productivity analysis of legume and non-legume supported cropping systems: The potential role of biological nitrogen fixation. *Frontiers in Plant Science* 7, 1700. doi: 10.3389/fpls.2016.01700
- 680. Ledgard, S.F., and Steele, K.W. (1992) Biological nitrogen fixation in mixed legume/grass pastures. *Plant and Soil* **141**, 137-53. doi: 10.1007/BF00011314
- 681. **Carlsson, G., and Huss-Danell, K.** (2003) Nitrogen fixation in perennial forage legumes in the field. *Plant and Soil* **253**, 353-72. doi: 10.1023/A:1024847017371
- 682. **Unkovich, M.J., and Pate, J.S.** (2000) An appraisal of recent field measurements of symbiotic N₂ fixation by annual legumes. *Field Crops Research* **65**, 211-28. doi: 10.1016/S0378-4290(99)00088-X
- 683. **Huss-Danell, K., Chaia, E., and Carlsson, G.** (2007) N₂ fixation and nitrogen allocation to above and below ground plant parts in red clovergrasslands. *Plant and Soil* **299**, 215-26. doi: 10.1007/s11104-007-9376-4
- 684. **Hardy, R.W.F., Holsten, R.D., Jackson, E.K., and Burns, R.C.** (1968) The acetylene-ethylene assay for N₂ fixation: Laboratory and field evaluation. *Plant Physiology* **43**, 1185. doi: 10.1104/pp.43.8.1185
- 685. **Neilson, A.H., and Sparell, L.** (1976) Acetylene reduction (nitrogen fixation) by *enterobacteriaceae* isolated from paper mill process waters. *Applied and environmental microbiology* **32**, 197-205.
- 686. Das, S., and De, T.K. (2018) Microbial assay of N₂ fixation rate, a simple alternate for acetylene reduction assay. *MethodsX* 5, 909-14. doi: 10.1016/j.mex.2017.11.010
- 687. Janda, J.M., and Abbott, S.L. (2007) 16S rRNA gene sequencing for bacterial identification in the diagnostic laboratory: Pluses, perils, and pitfalls. *Journal of Clinical Microbiology* 45, 2761. doi: 10.1128/JCM.01228-07
- 688. Limpens, J., Granath, G., Aerts, R., Heijmans, M.M.P.D., Sheppard, L.J., Bragazza, L., Williams, B.L., Rydin, H., Bubier, J., Moore, T., Rochefort, L., Mitchell, E.A.D., Buttler, A., van den Berg, L.J.L., Gunnarsson, U., Francez, A.J., Gerdol, R., Thormann, M., Grosvernier, P., Wiedermann, M.M., Nilsson, M.B., Hoosbeek, M.R., Bayley, S., Nordbakken, J.F., Paulissen, M.P.C.P., Hotes, S., Breeuwer, A., Ilomets, M., Tomassen, H.B.M., Leith, I., and Xu, B. (2012) Glasshouse vs field experiments: do they yield ecologically similar results for assessing N impacts on peat mosses? New Phytologist 195, 408-18. doi: 10.1111/j.1469-8137.2012.04157.x
- 689. Wang, Z., Gerstein, M., and Snyder, M. (2009) RNA-seq: a revolutionary tool for transcriptomics. *Nature Reviews Genetics* 10, 57-63. doi: 10.1038/nrg2484
- 690. **Marguerat, S., and Bähler, J.** (2010) RNA-seq: from technology to biology. *Cellular and Molecular Life Sciences* **67,** 569-79. doi: 10.1007/s00018-009-0180-6
- 691. **Costa-Silva, J., Domingues, D., and Lopes, F.M.** (2017) RNA-seq differential expression analysis: An extended review and a software tool. *PLoS One* **12**, e0190152. doi: 10.1371/journal.pone.0190152
- 692. **Mortazavi, A., Williams, B.A., McCue, K., Schaeffer, L., and Wold, B.** (2008) Mapping and quantifying mammalian transcriptomes by RNAseq. *Nature Methods* **5**, 621. doi: 10.1038/nmeth.1226

- 693. t Hoen, P.A., Ariyurek, Y., Thygesen, H.H., Vreugdenhil, E., Vossen, R.H., de Menezes, R.X., Boer, J.M., van Ommen, G.J., and den Dunnen, J.T. (2008) Deep sequencing-based expression analysis shows major advances in robustness, resolution and inter-lab portability over five microarray platforms. *Nucleic Acids Research* **36**, e141. doi: 10.1093/nar/gkn705
- 694. **Martin, L., Fei, Z., Giovannoni, J., and Rose, J.** (2013) Catalyzing plant science research with RNA-seq. *Frontiers in Plant Science* **4**, 66. doi: 10.3389/fpls.2013.00066
- 695. Conesa, A., Madrigal, P., Tarazona, S., Gomez-Cabrero, D., Cervera, A., McPherson, A., Szcześniak, M.W., Gaffney, D.J., Elo, L.L., Zhang, X., and Mortazavi, A. (2016) A survey of best practices for RNA-seq data analysis. *Genome biology* 17, 13-. doi: 10.1186/s13059-016-0881-8
- 696. **Stark, R., Grzelak, M., and Hadfield, J.** (2019) RNA sequencing: the teenage years. *Nature Reviews Genetics*. doi: 10.1038/s41576-019-0150-2
- 697. Bartlett, A., Lewis, J., and Williams, M.L. (2016) Generations of interdisciplinarity in bioinformatics. *New genetics and society* **35**, 186-209. doi: 10.1080/14636778.2016.1184965
- 698. Lewis, J., and Bartlett, A. (2013) Inscribing a discipline: tensions in the field of bioinformatics. *New genetics and society* **32**, 243-63. doi: 10.1080/14636778.2013.773172
- 699. **Garber, M., Grabherr, M.G., Guttman, M., and Trapnell, C.** (2011) Computational methods for transcriptome annotation and quantification using RNA-seq. *Nature Methods* **8**, 469. doi: 10.1038/nmeth.1613
- 700. El Yahyaoui, F., Küster, H., Ben Amor, B., Hohnjec, N., Pühler, A., Becker, A., Gouzy, J., Vernié, T., Gough, C., Niebel, A., Godiard, L., and Gamas, P. (2004) Expression profiling in *Medicago truncatula* identifies more than 750 genes differentially expressed during nodulation, including many potential regulators of the symbiotic program. *Plant Physiology* **136**, 3159-76. doi: 10.1104/pp.104.043612
- 701. Lohar, D.P., Sharopova, N., Endre, G., Peñuela, S., Samac, D., Town, C., Silverstein, K.A.T., and VandenBosch, K.A. (2006) Transcript analysis of early nodulation events in *Medicago truncatula*. *Plant Physiology* **140**, 221. doi: 10.1104/pp.105.070326
- 702. Kamphuis, L.G., Williams, A.H., D'Souza, N.K., Pfaff, T., Ellwood, S.R., Groves, E.J., Singh, K.B., Oliver, R.P., and Lichtenzveig, J. (2007) The *Medicago truncatula* reference accession A17 has an aberrant chromosomal configuration. *New Phytologist* **174**. doi: 10.1111/j.1469-8137.2007.02039.x
- 703. Moreau, S., Verdenaud, M., Ott, T., Letort, S., de Billy, F., Niebel, A., Gouzy, J., de Carvalho-Niebel, F., and Gamas, P. (2011) Transcription reprogramming during root nodule development in *Medicago truncatula. PLoS One* 6, e16463. doi: 10.1371/journal.pone.0016463
- 704. Boscari, A., Del Giudice, J., Ferrarini, A., Venturini, L., Zaffini, A.-L., Delledonne, M., and Puppo, A. (2013) Expression dynamics of the *Medicago truncatula* transcriptome during the symbiotic interaction with *Sinorhizobium meliloti*: which role for nitric oxide? *Plant Physiology* 161, 425-39. doi: 10.1104/pp.112.208538
- 705. Roux, B., Rodde, N., Jardinaud, M.-F., Timmers, T., Sauviac, L., Cottret, L., Carrère, S., Sallet, E., Courcelle, E., Moreau, S., Debellé, F., Capela, D., de Carvalho-Niebel, F., Gouzy, J., Bruand, C., and Gamas, P. (2014) An integrated analysis of plant and bacterial gene expression in symbiotic root nodules using laser-capture

microdissection coupled to RNA sequencing. *The Plant Journal* **77**, 817-37. doi: 10.1111/tpj.12442

- 706. Larrainzar, E., Riely, B.K., Kim, S.C., Carrasquilla-Garcia, N., Yu, H.-J., Hwang, H.-J., Oh, M., Kim, G.B., Surendrarao, A.K., Chasman, D., Siahpirani, A.F., Penmetsa, R.V., Lee, G.-S., Kim, N., Roy, S., Mun, J.-H., and Cook, D.R. (2015) Deep sequencing of the *Medicago truncatula* root transcriptome reveals a massive and early interaction between nodulation factor and ethylene signals. *Plant Physiology* **169**, 233. doi: 10.1104/pp.15.00350
- 707. Pecrix, Y., Staton, S.E., Sallet, E., Lelandais-Brière, C., Moreau, S., Carrère, S., Blein, T., Jardinaud, M.-F., Latrasse, D., Zouine, M., Zahm, M., Kreplak, J., Mayjonade, B., Satgé, C., Perez, M., Cauet, S., Marande, W., Chantry-Darmon, C., Lopez-Roques, C., Bouchez, O., Bérard, A., Debellé, F., Muños, S., Bendahmane, A., Bergès, H., Niebel, A., Buitink, J., Frugier, F., Benhamed, M., Crespi, M., Gouzy, J., and Gamas, P. (2018) Whole-genome landscape of *Medicago* truncatula symbiotic genes. Nat Plants 4, 1017-25. doi: 10.1038/s41477-018-0286-7
- 708. Sańko-Sawczenko, I., Łotocka, B., Mielecki, J., Rekosz-Burlaga, H., and Czarnocka, W. (2019) Transcriptomic changes in *Medicago truncatula* and *Lotus japonicus* root nodules during drought stress. *Int J Mol Sci* 20. doi: 10.3390/ijms20051204
- 709. Kouchi, H., Shimomura, K., Hata, S., Hirota, A., Wu, G.-J., Kumagai, H., Tajima, S., Suganuma, N., Suzuki, A., Aoki, T., Hayashi, M., Yokoyama, T., Ohyama, T., Asamizu, E., Kuwata, C., Shibata, D., and Tabata, S. (2004) Large-scale analysis of gene expression profiles during early stages of root nodule formation in a model legume, *Lotus japonicus*. *DNA Research* **11**, 263-74. doi: 10.1093/dnares/11.4.263
- 710. **Sugimura, Y., and Saito, K.** (2017) Comparative transcriptome analysis between *Solanum lycopersicum* L. and *Lotus japonicus* L. during arbuscular mycorrhizal development. *Soil Science and Plant Nutrition* **63**, 127-36. doi: 10.1080/00380768.2017.1280378
- 711. Kelly, S., Mun, T., Stougaard, J., Ben, C., and Andersen, S.U. (2018) Distinct *Lotus japonicus* transcriptomic responses to a spectrum of bacteria ranging from symbiotic to pathogenic. *Frontiers in Plant Science* 9, 1218-. doi: 10.3389/fpls.2018.01218
- 712. **Garg, R., Patel, R.K., Tyagi, A.K., and Jain, M.** (2011) De novo assembly of chickpea transcriptome using short reads for gene discovery and marker identification. *DNA Research* **18**. doi: 10.1093/dnares/dsq028
- Garg, R., and Jain, M. (2013) "RNA-seq for transcriptome analysis in non-model plants," in *Legume Genomics: Methods and Protocols*, ed. R.J. Rose. (Totowa, NJ: Humana Press), 43-58.
- 714. **Kant, C., Pradhan, S., and Bhatia, S.** (2016) Dissecting the root nodule transcriptome of chickpea (*Cicer arietinum* L.). *PLoS One* **11**, e0157908. doi: 10.1371/journal.pone.0157908
- 715. Duarte, J., Rivière, N., Baranger, A., Aubert, G., Burstin, J., Cornet, L., Lavaud, C., Lejeune-Hénaut, I., Martinant, J.-P., Pichon, J.-P., Pilet-Nayel, M.-L., and Boutet, G. (2014) Transcriptome sequencing for high throughput SNP development and genetic mapping in pea. *BMC Genomics* 15, 126. doi: 10.1186/1471-2164-15-126
- 716. Alves-Carvalho, S., Aubert, G., Carrère, S., Cruaud, C., Brochot, A.-L., Jacquin, F., Klein, A., Martin, C., Boucherot, K., Kreplak, J., da Silva, C., Moreau, S., Gamas, P., Wincker, P., Gouzy, J., and Burstin, J. (2015) Full-length de novo assembly of RNA-seq data in pea (*Pisum sativum* L.) provides a gene expression atlas and gives insights

into root nodulation in this species. *The Plant Journal* **84**, 1-19. doi: 10.1111/tpj.12967

- 717. Kulaeva, O.A., Afonin, A.M., Zhernakov, A.I., Tikhonovich, I.A., and Zhukov, V.A. (2017) "Transcriptomic studies in non-model plants: case of *Pisum sativum* L. and *Medicago lupulina* L.," in *Applications of RNA-seq and omics strategies - From microorganisms to human health,* eds. F.A. Marchi, P.D.R. Cirillo & E.C. Mateo. ([Online]: IntechOpen).
- 718. Sudheesh, S., Verma, P., Forster, J.W., Cogan, N.O.I., and Kaur, S. (2016) Generation and characterisation of a reference transcriptome for lentil (*Lens culinaris* medik.). *Int J Mol Sci* 17, 1887. doi: 10.3390/ijms17111887
- 719. Hayashi, S., Reid, D.E., Lorenc, M.T., Stiller, J., Edwards, D., Gresshoff, P.M., and Ferguson, B.J. (2012) Transient *Nod* factordependent gene expression in the nodulation-competent zone of soybean (*Glycine max* [L.] Merr.) roots. *Plant Biotechnol J* **10**, 995-1010. doi: 10.1111/j.1467-7652.2012.00729.x
- 720. O'Rourke, J.A., Yang, S.A., Miller, S.S., Bucciarelli, B., Liu, J., Rydeen, A., Bozsoki, Z., Uhde-Stone, C., Tu, Z.J., Allan, D., Gronwald, J.W., and Vance, C.P. (2013) An RNA-Seq transcriptome analysis of orthophosphate-deficient white lupin reveals novel insights into phosphorus acclimation in plants. *Plant Physiology* **161**. doi: 10.1104/pp.112.209254
- 721. Li, S.-W., Shi, R.-F., and Leng, Y. (2015) De novo characterization of the mung bean transcriptome and transcriptomic analysis of adventitious rooting in seedlings using RNA-seq. *PLoS One* **10**, e0132969. doi: 10.1371/journal.pone.0132969
- 722. Li, X., Acharya, A., Farmer, A.D., Crow, J.A., Bharti, A.K., Kramer, R.S., Wei, Y., Han, Y., Gou, J., May, G.D., Monteros, M.J., and Brummer, E.C. (2012) Prevalence of single nucleotide polymorphism among 27 diverse alfalfa genotypes as assessed by transcriptome sequencing. *BMC Genomics* **13**, 568. doi: 10.1186/1471-2164-13-568
- 723. Liu, Z., Chen, T., Ma, L., Zhao, Z., Zhao, P.X., Nan, Z., and Wang, Y. (2013) Global transcriptome sequencing using the illumina platform and the development of EST-SSR markers in autotetraploid alfalfa. *PLoS One* 8, e83549. doi: 10.1371/journal.pone.0083549
- 724. **Gao, R., S. Austin, R., Amyot, L., and Hannoufa, A.** (2016) Comparative transcriptome investigation of global gene expression changes caused by *miR156* overexpression in *Medicago sativa*. *BMC Genomics* **17**. doi 10.1186/s12864-016-3014-6
- 725. Zhang, S., Shi, Y., Cheng, N., Du, H., Fan, W., and Wang, C. (2015) De novo characterization of fall dormant and nondormant alfalfa (*Medicago sativa* L.) leaf transcriptome and identification of candidate genes related to fall dormancy. *PLoS One* **10**, e0122170. doi: 10.1371/journal.pone.0122170
- 726. Shu, Y., Liu, Y., Li, W., Song, L., Zhang, J., and Guo, C. (2016) Genome-wide investigation of microRNAs and their targets in response to freezing stress in *Medicago sativa* L., based on high-throughput sequencing. *G3: Genes Genomes Genetics* 6, 755. doi: 10.1534/g3.115.025981
- 727. Song, F., Li, J., Fan, X., Zhang, Q., Chang, W., Yang, F., and Geng, G. (2016) Transcriptome analysis of *Glomus mosseae/Medicago sativa* mycorrhiza on atrazine stress. *Scientific Reports* 6, 20245. doi: 10.1038/srep20245
- 728. **Song, L., Jiang, L., Chen, Y., Shu, Y., Bai, Y., and Guo, C.** (2016) Deep-sequencing transcriptome analysis of field-grown *Medicago sativa* L. crown buds acclimated to freezing stress. *Functional & Integrative Genomics* **16**, 495-511. doi: 10.1007/s10142-016-0500-5
- 729. Xu, B., Wang, Y., Zhang, S., Guo, Q., Jin, Y., Chen, J., Gao, Y., and Ma, H. (2017) Transcriptomic and physiological analyses of *Medicago sativa* L. roots in response to lead stress. *PLoS One* **12**, e0175307-e. doi: 10.1371/journal.pone.0175307
- 730. Lei, Y., Xu, Y., Hettenhausen, C., Lu, C., Shen, G., Zhang, C., Li, J., Song, J., Lin, H., and Wu, J. (2018) Comparative analysis of alfalfa (*Medicago sativa* L.) leaf transcriptomes reveals genotype-specific salt tolerance mechanisms. *BMC Plant Biology* **18**, 35. doi: 10.1186/s12870-018-1250-4
- Oldroyd, G.E., and Downie, J.A. (2008) Coordinating nodule morphogenesis with rhizobial infection in legumes. *Annual Review of Plant Biology* 59, 519-46. doi: 10.1146/annurev.arplant.59.032607.092839
- 732. Hartmann, A., Rothballer, M., and Schmid, M. (2008) Lorenz Hiltner, a pioneer in rhizosphere microbial ecology and soil bacteriology research. *Plant and Soil* **312**, 7-14. doi: 10.1007/s11104-007-9514-z
- 733. Bertin, C., Yang, X., and Weston, L.A. (2003) The role of root exudates and allelochemicals in the rhizosphere. *Plant and Soil* 256, 67-83. doi: 10.1023/A:1026290508166
- 734. **Uren, N.** (2007) "Types, amounts, and possible functions of compounds released into the rhizosphere by soil-grown plants." CRC Press), 1-21.
- 735. **Oldroyd, G.E.D.** (2013) Speak, friend, and enter: signalling systems that promote beneficial symbiotic associations in plants. *Nature Reviews Microbiology* **11**, 252. doi: 10.1038/nrmicro2990
- 736. **Raaijmakers, J.M., Paulitz, T.C., Steinberg, C., Alabouvette, C., and Moënne-Loccoz, Y.** (2009) The rhizosphere: a playground and battlefield for soilborne pathogens and beneficial microorganisms. *Plant and Soil* **321,** 341-61. doi: 10.1007/s11104-008-9568-6
- 737. **Mendes, R., Garbeva, P., and Raaijmakers, J.M.** (2013) The rhizosphere microbiome: significance of plant beneficial, plant pathogenic, and human pathogenic microorganisms. *FEMS Microbiology Reviews* **37**, 634-63. doi: 10.1111/1574-6976.12028
- 738. **Bianco, L.** (2014) Rhizobial infection in *Adesmia bicolor* (*Fabaceae*) roots. *Archives of Microbiology* **196,** 675-9. doi: 10.1007/s00203-014-1004-0
- 739. Udvardi, M.K., Kakar, K., Wandrey, M., Montanari, O., Murray, J., Andriankaja, A., Zhang, J.-Y., Benedito, V., Hofer, J.M.I., Chueng, F., and Town, C.D. (2007) Legume transcription factors: global regulators of plant development and response to the environment. *Plant Physiology* **144**, 538-49. doi: 10.1104/pp.107.098061
- 740. **Murray, J.D.** (2011) Invasion by invitation: rhizobial infection in legumes. *Molecular Plant-Microbe Interactions* **24,** 631-9. doi: 10.1094/mpmi-08-10-0181
- 741. Amor, B.B., Shaw, S.L., Oldroyd, G.E., Maillet, F., Penmetsa, R.V., Cook, D., Long, S.R., Denarie, J., and Gough, C. (2003) The NFP locus of *Medicago truncatula* controls an early step of Nod factor signal transduction upstream of a rapid calcium flux and root hair deformation. *The Plant Journal* **34**, 495-506. doi: 10.1046/j.1365-313X.2003.01743.x
- 742. Libault, M., Farmer, A., Joshi, T., Takahashi, K., Langley, R.J., Franklin, L.D., He, J., Xu, D., May, G., and Stacey, G. (2010) An integrated transcriptome atlas of the crop model Glycine max, and its use in comparative analyses in plants. *The Plant Journal* **63**, 86-99. doi: 10.1111/j.1365-313X.2010.04222.x
- 743. **Udvardi, M., and Poole, P.S.** (2013) Transport and metabolism in legume-rhizobia symbioses. *Annual Review of Plant Biology* **64,** 781-805. doi: 10.1146/annurev-arplant-050312-120235

- 744. **Bais, H.P., Weir, T.L., Perry, L.G., Gilroy, S., and Vivanco, J.M.** (2006) The role of root exudates in rhizosphere interactions with plants and other organisms. *Annual Review of Plant Biology* **57,** 233-66. doi: 10.1146/annurev.arplant.57.032905.105159
- 745. **Peters, N.K., Frost, J.W., and Long, S.R.** (1986) A plant flavone, luteolin, induces expression of *Rhizobium meliloti* nodulation genes. *Science* **233**, 977-80. doi: 10.1126/science.3738520
- 746. **Perret, X., Staehelin, C., and Broughton, W.J.** (2000) Molecular basis of symbiotic promiscuity. *Microbiology and Molecular Biology Reviews* **64**, 180-201. doi: 10.1128/mmbr.64.1.180-201.2000
- 747. Indrasumunar, A., Searle, I., Lin, M.H., Kereszt, A., Men, A., Carroll, B.J., and Gresshoff, P.M. (2011) Nodulation factor receptor kinase 1alpha controls nodule organ number in soybean (*Glycine max* L. Merr). *The Plant Journal* **65**, 39-50. doi: 10.1111/j.1365-313X.2010.04398.x
- 748. Loh, J., Garcia, M., and Stacey, G. (1997) NodV and NodW, a second flavonoid recognition system regulating nod gene expression in *Bradyrhizobium japonicum*. *Journal of Bacteriology* **179**, 3013-20. doi: 10.1128/jb.179.9.3013-3020.1997
- 749. Limpens, E., Franken, C., Smit, P., Willemse, J., Bisseling, T., and Geurts, R. (2003) LysM domain receptor kinases regulating rhizobial Nod factor-induced infection. *Science* **302**, 630-3. doi: 10.1126/science.1090074
- 750. Madsen, E.B., Madsen, L.H., Radutoiu, S., Olbryt, M., Rakwalska, M., Szczyglowski, K., Sato, S., Kaneko, T., Tabata, S., Sandal, N., and Stougaard, J. (2003) A receptor kinase gene of the *LysM* type is involved in legume perception of rhizobial signals. *Nature* **425**, 637-40. doi: 10.1038/nature02045
- 751. Radutoiu, S., Madsen, L.H., Madsen, E.B., Felle, H.H., Umehara, Y., Grønlund, M., Sato, S., Nakamura, Y., Tabata, S., Sandal, N., and Stougaard, J. (2003) Plant recognition of symbiotic bacteria requires two LysM receptor-like kinases. *Nature* **425**, 585-92. doi: 10.1038/nature02039
- 752. **Knogge, W., and Scheel, D.** (2006) LysM receptors recognize friend and foe. *PNAS* **103**, 10829. doi: 10.1073/pnas.0604601103
- 753. Nakagawa, T., Kaku, H., Shimoda, Y., Sugiyama, A., Shimamura, M., Takanashi, K., Yazaki, K., Aoki, T., Shibuya, N., and Kouchi, H. (2011) From defense to symbiosis: limited alterations in the kinase domain of LysM receptor-like kinases are crucial for evolution of legume-*Rhizobium* symbiosis. *The Plant Journal* 65, 169-80. doi: 10.1111/j.1365-313X.2010.04411.x
- 754. **Zipfel, C., and Oldroyd, G.E.D.** (2017) Plant signalling in symbiosis and immunity. *Nature* **543**, 328. doi: 10.1038/nature22009
- 755. **Maroti, G., and Kondorosi, E.** (2014) Nitrogen-fixing Rhizobiumlegume symbiosis: are polyploidy and host peptide-governed symbiont differentiation general principles of endosymbiosis? *Frontiers in microbiology* **5**, 326. doi: 10.3389/fmicb.2014.00326
- 756. **Spaink, H.P.** (2000) Root nodulation and infection factors produced by rhizobial bacteria. *Annu Rev Microbiol* **54**, 257-88. doi: 10.1146/annurev.micro.54.1.257
- Pfeilmeier, S., George, J., Morel, A., Roy, S., Smoker, M., Stransfeld, L., Downie, J.A., Peeters, N., Malone, J.G., and Zipfel, C. (2019) Expression of the *Arabidopsis thaliana* immune receptor EFR in *Medicago truncatula* reduces infection by a root pathogenic bacterium, but not nitrogen-fixing rhizobial symbiosis. *Plant Biotechnol J* 17, 569-79. doi: 10.1111/pbi.12999

- 758. **Ausubel, F.M.** (2005) Are innate immune signaling pathways in plants and animals conserved? *Nat Immunol* **6**, 973-9. doi: 10.1038/ni1253
- 759. **Zipfel, C.** (2014) Plant pattern-recognition receptors. *Trends Immunol* **35**, 345-51. doi: 10.1016/j.it.2014.05.004
- 760. Boutrot, F., and Zipfel, C. (2017) Function, discovery, and exploitation of plant pattern recognition receptors for broad-spectrum disease resistance. *Annual Review of Phytopathology* 55, 257-86. doi: 10.1146/annurev-phyto-080614-120106
- 761. Chiasson, D.M., Loughlin, P.C., Mazurkiewicz, D., Mohammadidehcheshmeh, M., Fedorova, E.E., Okamoto, M., McLean, E., Glass, A.D., Smith, S.E., Bisseling, T., Tyerman, S.D., Day, D.A., and Kaiser, B.N. (2014) Soybean SAT1 (Symbiotic Ammonium Transporter 1) encodes a bHLH transcription factor involved in nodule growth and NH4+ transport. PNAS 111, 4814-9. doi: 10.1073/pnas.1312801111
- 762. van Berkum, P., and Fuhrmann, J.J. (2000) Evolutionary relationships among the soybean *bradyrhizobia* reconstructed from 16S rRNA gene and internally transcribed spacer region sequence divergence. *International Journal of Systematic and Evolutionary Microbiology* 50, 2165-72. doi: 10.1099/00207713-50-6-2165
- 763. **Peters, G.A.** (1975) The *Azolla-Anabaena azollae* relationship. *Archives* of *Microbiology* **103**, 113-22. doi: 10.1007/BF00436337
- 764. **Fang, Y., and Hirsch, A.M.** (1998) Studying early nodulin gene *ENOD40* expression and induction by nodulation factor and cytokinin in transgenic alfalfa. *Plant Physiology* **116**, 53. doi: 10.1104/pp.116.1.53
- 765. Mathis, R., Grosjean, C., de Billy, F., Huguet, T., and Gamas, P. (1999) The early nodulin gene *MtN6* is a novel marker for events preceding infection of *Medicago truncatula* roots by *Sinorhizobium meliloti*. *Molecular Plant-Microbe Interactions* **12**, 544-55. doi: 10.1094/MPMI.1999.12.6.544
- 766. Galibert, F., Finan, T.M., Long, S.R., Pühler, A., Abola, P., Ampe, F., Barloy-Hubler, F., Barnett, M.J., Becker, A., Boistard, P., Bothe, G., Boutry, M., Bowser, L., Buhrmester, J., Cadieu, E., Capela, D., Chain, P., Cowie, A., Davis, R.W., Dréano, S., Federspiel, N.A., Fisher, R.F., Gloux, S., Godrie, T., Goffeau, A., Golding, B., Gouzy, J., Gurjal, M., Hernandez-Lucas, I., Hong, A., Huizar, L., Hyman, R.W., Jones, T., Kahn, D., Kahn, M.L., Kalman, S., Keating, D.H., Kiss, E., Komp, C., Lelaure, V., Masuy, D., Palm, C., Peck, M.C., Pohl, T.M., Portetelle, D., Purnelle, B., Ramsperger, U., Surzycki, R., Thébault, P., Vandenbol, M., Vorhölter, F.-J., Weidner, S., Wells, D.H., Wong, K., Yeh, K.-C., and Batut, J. (2001) The composite genome of the legume symbiont *Sinorhizobium meliloti. Science* 293, 668. doi: 10.1126/science.1060966
- 767. Checcucci, A., Azzarello, E., Bazzicalupo, M., Galardini, M., Lagomarsino, A., Mancuso, S., Marti, L., Marzano, M.C., Mocali, S., Squartini, A., Zanardo, M., and Mengoni, A. (2016) Mixed nodule infection in *Sinorhizobium meliloti–Medicago sativa* symbiosis suggest the presence of cheating behavior *Frontiers in Plant Science* 7 835. doi: 10.3389/fpls.2016.00835
- 768. Wang, X.L., Cui, W.J., Feng, X.Y., Zhong, Z.M., Li, Y., Chen, W.X., Chen, W.F., Shao, X.M., and Tian, C.F. (2018) Rhizobia inhabiting nodules and rhizosphere soils of alfalfa: A strong selection of facultative microsymbionts. *Soil Biology and Biochemistry* **116**, 340-50. doi: 10.1016/j.soilbio.2017.10.033
- 769. **Pueppke, S.G., and Broughton, W.J.** (1999) *Rhizobium* sp. strain NGR234 and *R. fredii* USDA257 share exceptionally broad, nested host

ranges. *Molecular Plant-Microbe Interactions* **12**, 293-318. doi: 10.1094/mpmi.1999.12.4.293

- 770. Schardl, C.L., Leuchtmann, A., and Spiering, M.J. (2004) Symbioses of grasses with seedborne fungal endophytes. *Annual Review of Plant Biology* **55**, 315-40. doi: 10.1146/annurev.arplant.55.031903.141735
- 771. **Bais, H.P., Fall, R., and Vivanco, J.M.** (2004) Biocontrol of *Bacillus subtilis* against infection of *Arabidopsis* roots by *Pseudomonas syringae* is facilitated by biofilm formation and surfactin production. *Plant Physiology* **134,** 307-19. doi: 10.1104/pp.103.028712
- 772. **Chen, T., Duan, L., Zhou, B., Yu, H., Zhu, H., Cao, Y., and Zhang, Z.** (2017) Interplay of pathogen-induced defense responses and symbiotic establishment in *Medicago truncatula*. *Frontiers in microbiology* **8**, 973-. doi: 10.3389/fmicb.2017.00973
- 773. Clúa, J., Roda, C., Zanetti, E.M., and Blanco, A.F. (2018) Compatibility between legumes and *Rhizobia* for the establishment of a successful nitrogen-fixing symbiosis. *Genes (Basel)* 9. doi: 10.3390/genes9030125
- 774. Kaschani, F., Nickel, S., Pandey, B., Cravatt, B.F., Kaiser, M., and van der Hoorn, R.A.L. (2012) Selective inhibition of plant serine hydrolases by agrochemicals revealed by competitive ABPP. *Bioorganic & medicinal chemistry* **20**, 597-600. doi: 10.1016/j.bmc.2011.06.040
- 775. Mindrebo, J.T., Nartey, C.M., Seto, Y., Burkart, M.D., and Noel, J.P. (2016) Unveiling the functional diversity of the alpha/beta hydrolase superfamily in the plant kingdom. *Current Opinion in Structural Biology* 41, 233-46. doi: 10.1016/j.sbi.2016.08.005
- 776. **Beintema, J.J.** (1994) Structural features of plant chitinases and chitinbinding proteins. *FEBS Lett* **350**, 159-63. doi: 10.1016/0014-5793(94)00753-5
- 777. Lannoo, N., and Van Damme, E.J.M. (2014) Lectin domains at the frontiers of plant defense. *Frontiers in Plant Science* **5**, 397. doi: 10.3389/fpls.2014.00397
- 778. **Tischner, R.** (2000) Nitrate uptake and reduction in higher and lower plants. *Plant, Cell & Environment* **23,** 1005-24. doi: 10.1046/j.1365-3040.2000.00595.x
- 779. **Zeijl, A.v., Guhl, K., Xiao, T.T., Shen, D., Geurts, R., and Kohlen, W.** (2018) Nitrate inhibition of nodule formation in *Medicago truncatula* is mediated by *ACC SYNTHASE 10. bioRxiv*, 434829. doi: 10.1101/434829
- 780. Barbulova, A., Rogato, A., D'Apuzzo, E., Omrane, S., and Chiurazzi, M. (2007) Differential effects of combined N sources on early steps of the Nod factor-dependent transduction pathway in *Lotus japonicus*. *Molecular Plant-Microbe Interactions* 20, 994-1003. doi: 10.1094/mpmi-20-8-0994
- 781. Liu, A., Contador, C.A., Fan, K., and Lam, H.-M. (2018) Interaction and regulation of carbon, nitrogen, and phosphorus metabolisms in root nodules of legumes. *Frontiers in Plant Science* 9. doi: 10.3389/fpls.2018.01860
- 782. **Popp, C., and Ott, T.** (2011) Regulation of signal transduction and bacterial infection during root nodule symbiosis. *Current Opinion in Plant Biology* **14**, 458-67. doi: 10.1016/j.pbi.2011.03.016
- 783. Kawaharada, Y., Kelly, S., Nielsen, M.W., Hjuler, C.T., Gysel, K., Muszyński, A., Carlson, R.W., Thygesen, M.B., Sandal, N., Asmussen, M.H., Vinther, M., Andersen, S.U., Krusell, L., Thirup, S., Jensen, K.J., Ronson, C.W., Blaise, M., Radutoiu, S., and Stougaard, J. (2015) Receptor-mediated exopolysaccharide perception controls bacterial infection. *Nature* **523**, 308. doi: 10.1038/nature14611
- 784. Krusell, L., Madsen, L.H., Sato, S., Aubert, G., Genua, A., Szczyglowski, K., Duc, G., Kaneko, T., Tabata, S., de Bruijn, F., Pajuelo, E., Sandal, N., and Stougaard, J. (2002) Shoot control of

root development and nodulation is mediated by a receptor-like kinase. *Nature* **420**, 422-6. doi: 10.1038/nature01207

- 785. Schnabel, E., Journet, E.P., de Carvalho-Niebel, F., Duc, G., and Frugoli, J. (2005) The *Medicago truncatula SUNN* gene encodes a CLV1like leucine-rich repeat receptor kinase that regulates nodule number and root length. *Plant Mol Biol* 58, 809-22. doi: 10.1007/s11103-005-8102-y
- 786. **Oka-Kira, E., and Kawaguchi, M.** (2006) Long-distance signaling to control root nodule number. *Current Opinion in Plant Biology* **9**, 496-502. doi: 10.1016/j.pbi.2006.07.012
- 787. Mortier, V., Den Herder, G., Whitford, R., Van de Velde, W., Rombauts, S., D'Haeseleer, K., Holsters, M., and Goormachtig, S. (2010) CLE peptides control *Medicago truncatula* nodulation locally and systemically. *Plant Physiology* **153**, 222-37. doi: 10.1104/pp.110.153718
- 788. Lim, C.W., Lee, Y.W., and Hwang, C.H. (2011) Soybean noduleenhanced CLE peptides in roots act as signals in *GmNARK*-mediated nodulation suppression. *Plant Cell Physiology* 52, 1613-27. doi: 10.1093/pcp/pcr091
- 789. Reid, D.E., Ferguson, B.J., and Gresshoff, P.M. (2011) Inoculationand nitrate-induced CLE peptides of soybean control NARK-dependent nodule formation. *Molecular Plant-Microbe Interactions* 24, 606-18. doi: 10.1094/mpmi-09-10-0207
- 790. Middleton, P.H., Jakab, J., Penmetsa, R.V., Starker, C.G., Doll, J., Kaló, P., Prabhu, R., Marsh, J.F., Mitra, R.M., Kereszt, A., Dudas, B., VandenBosch, K., Long, S.R., Cook, D.R., Kiss, G.B., and Oldroyd, G.E.D. (2007) An ERF transcription factor in *Medicago truncatula* that is essential for Nod factor signal transduction. *Plant Cell* 19, 1221. doi: 10.1105/tpc.106.048264
- 791. Vernie, T., Moreau, S., de Billy, F., Plet, J., Combier, J.P., Rogers, C., Oldroyd, G., Frugier, F., Niebel, A., and Gamas, P. (2008) *EFD* Is an *ERF* transcription factor involved in the control of nodule number and differentiation in *Medicago truncatula*. *Plant Cell* **20**, 2696-713. doi: 10.1105/tpc.108.059857
- 792. Wang, X., Chen, X., Wang, Z., Nikolay, D., Vladimir, C., and Gao, H. (2010) Isolation and characterization of *GoDREB* encoding an ERF-type protein in forage legume *Galegae orientalis*. *Genes & Genetic Systems* **85**, 157-66. doi: 10.1266/ggs.85.157
- 793. Peng, Z., Liu, F., Wang, L., Zhou, H., Paudel, D., Tan, L., Maku, J., Gallo, M., and Wang, J. (2017) Transcriptome profiles reveal gene regulation of peanut (*Arachis hypogaea* L.) nodulation. *Scientific Reports* 7, 40066. doi: 10.1038/srep40066
- 794. **Legocki, R.P., and Verma, D.P.S.** (1980) Identification of "nodulespecific" host proteins (nodulins) involved in the development of *Rhizobium*-Legume symbiosis. *Cell* **20,** 153-63. doi: 10.1016/0092-8674(80)90243-3
- 795. Scheres, B., van Engelen, F., van der Knaap, E., van de Wiel, C., van Kammen, A., and Bisseling, T. (1990) Sequential induction of *nodulin* gene expression in the developing pea nodule. *Plant Cell* **2**, 687. doi: 10.1105/tpc.2.8.687
- 796. **van de Wiel, C., Scheres, B., Franssen, H., van Lierop, M.J., van Lammeren, A., van Kammen, A., and Bisseling, T.** (1990) The early nodulin transcript *ENOD2* is located in the nodule parenchyma (inner cortex) of pea and soybean root nodules. *The EMBO journal* **9**, 1-7.
- 797. **Kouchi, H., and Hata, S.** (1993) Isolation and characterization of novel *nodulin* cDNAs representing genes expressed at early stages of soybean nodule development. *Molecular and General Genetics* **238**, 106-19. doi: 10.1007/BF00279537

- 798. Rivers, R.L., Dean, R.M., Chandy, G., Hall, J.E., Roberts, D.M., and Zeidel, M.L. (1997) Functional analysis of nodulin 26, an aquaporin in soybean root nodule symbiosomes. *J Biol Chem* 272, 16256-61. doi: 10.1074/jbc.272.26.16256
- 799. **Gamas, P., de Billy, F., and Truchet, G.** (1998) Symbiosis-specific expression of two *Medicago truncatula* nodulin genes, *MtN1* and *MtN13*, encoding products homologous to plant defense proteins. *Molecular Plant-Microbe Interactions* **11**, 393-403. doi: 10.1094/MPMI.1998.11.5.393
- 800. Becker, J.D., Moreira, L.M., Kapp, D., Frosch, S.C., Pühler, A., and Perlick, A.M. (2001) The nodulin *VfENOD18* is an ATP-binding protein in infected cells of *Vicia faba* L. nodules. *Plant Mol Biol* 47, 749-59. doi: 10.1023/A:1013664311052
- 801. Marsh, J.F., Rakocevic, A., Mitra, R.M., Brocard, L., Sun, J., Eschstruth, A., Long, S.R., Schultze, M., Ratet, P., and Oldroyd, G.E.D. (2007) *Medicago truncatula NIN* is essential for rhizobialindependent nodule organogenesis induced by autoactive calcium/calmodulin-dependent protein kinase. *Plant Physiology* **144**, 324. doi: 10.1104/pp.106.093021
- Roberts, D.M., and Routray, P. (2017) "The nodulin 26 intrinsic protein subfamily," in *Plant aquaporins: From transport to signaling*, eds. F. Chaumont & S.D. Tyerman. (Cham: Springer International Publishing), 267-96.
- 803. Liu, Y.-C., Lei, Y.-W., Liu, W., Weng, L., Lei, M.-J., Hu, X.-H., Dong, Z., Luo, D., and Yang, J. (2018) LjCOCH interplays with LjAPP1 to maintain the nodule development in *Lotus japonicus*. *Plant Growth Regulation* 85, 267-79. doi: 10.1007/s10725-018-0392-6
- 804. **Cheng, X.-G., Nomura, M., Takane, K., Kouchi, H., and Tajima, S.** (2000) Expression of two uricase (*Nodulin-35*) genes in a non-ureide type legume, *Medicago sativa*. *Plant and Cell Physiology* **41**, 104-9. doi: 10.1093/pcp/41.1.104
- 805. **Pringle, D., and Dickstein, R.** (2004) Purification of ENOD8 proteins from *Medicago sativa* root nodules and their characterization as esterases. *Plant Physiology and Biochemistry* **42,** 73-9. doi: 10.1016/j.plaphy.2003.10.004
- 806. Lafuente, A., Pajuelo, E., Caviedes, M.A., and Rodríguez-Llorente, I.D. (2010) Reduced nodulation in alfalfa induced by arsenic correlates with altered expression of early nodulins. *Journal of Plant Physiology* 167, 286-91. doi: 10.1016/j.jplph.2009.09.014
- 807. **Denarie, J., Debelle, F., and Prome, J.C.** (1996) *Rhizobium* lipochitooligosaccharide nodulation factors: signaling molecules mediating recognition and morphogenesis. *Annu Rev Biochem* **65,** 503-35. doi: 10.1146/annurev.bi.65.070196.002443
- 808. **Reddy, P.M., Ladha, J.K., Ramos, M.C., Maillet, F., Hernandez, R.J., Torrizo, L.B., Oliva, N.P., Datta, S.K., and Datta, K.** (1998) Rhizobial lipochitooligosaccharide nodulation factors activate expression of the legume early nodulin gene *ENOD12* in rice. *The Plant Journal* **14,** 693-702. doi: 10.1046/j.1365-313x.1998.00170.x
- 809. Liang, Y., Tóth, K., Cao, Y., Tanaka, K., Espinoza, C., and Stacey, G. (2014) Lipochitooligosaccharide recognition: an ancient story. *New Phytologist* 204, 289-96. doi: 10.1111/nph.12898
- 810. Muñoz, N., Soria-Díaz, M.E., Manyani, H., Sánchez-Matamoros, R.C., Serrano, A.G., Megías, M., and Lascano, R. (2014) Structure and biological activities of lipochitooligosaccharide nodulation signals produced by *Bradyrhizobium japonicum* USDA 138 under saline and osmotic stress. *Biology and Fertility of Soils* 50, 207-15. doi: 10.1007/s00374-013-0843-1

- 811. Bozsoki, Z., Cheng, J., Feng, F., Gysel, K., Vinther, M., Andersen, K.R., Oldroyd, G., Blaise, M., Radutoiu, S., and Stougaard, J. (2017) Receptor-mediated chitin perception in legume roots is functionally separable from Nod factor perception. *PNAS* **114**, E8118-e27. doi: 10.1073/pnas.1706795114
- 812. Berg, R.K., Jawson, M.D., Franzluebbers, A.J., and Kubik, K.K. (1989) *Bradyrhizobium japonicum* inoculation and seed priming for fluiddrilled soybean. *Soil Science Society of America* 53, 1712-7. doi: 10.2136/sssaj1989.03615995005300060017x
- 813. **Harris, D., Breese, W.A., and Rao, J.V.D.K.K.** (2005) The improvement of crop yield in marginal environments using on-farm seed priming: nodulation, nitrogen fixation, and disease resistance. *Australian journal of agricultural research* **56,** 1211-8. doi: 10.1071/AR05079
- 814. Alimadadi, A., Jahansouz, M.R., Besharati, H., and Tavakol Afshari, R. (2010) Evaluating the effects of phosphate solubilizing microorganisms, mycorrhizal fungi and seed priming on nodulation of chickpea. *Iranian Journal of Soil Research* 24, 44-53.
- 815. **Libault, M.** (2014) The carbon-nitrogen balance of the nodule and its regulation under elevated carbon dioxide concentration. *BioMed research international* **2014,** 507946-. doi: 10.1155/2014/507946
- 816. **Singleton, P.W., and Tavares, J.W.** (1986) Inoculation response of legumes in relation to the number and effectiveness of indigenous rhizobium populations. *Applied and environmental microbiology* **51**, 1013.
- 817. **Thies, J.E., Singleton, P.W., and Bohlool, B.B.** (1991) Influence of the size of indigenous rhizobial populations on establishment and symbiotic performance of introduced rhizobia on field-grown legumes. *Applied and environmental microbiology* **57**, 19.
- 818. **Beauchamp, C.J., Kloepper, J.W., Shaw, J.J., and Chalifour, F.P.** (2001) Root colonization of faba bean (*Vicia faba* L.) and pea (*Pisum sativum* L.) by *Rhizobium leguminosarum* bv. *viciae* in the presence of nitrate-nitrogen. *Can J Microbiol* **47,** 1068-74. doi: 10.1139/w01-113
- 819. **Matiru, V.N., and Dakora, F.D.** (2005) Xylem transport and shoot accumulation of lumichrome, a newly recognized rhizobial signal, alters root respiration, stomatal conductance, leaf transpiration and photosynthetic rates in legumes and cereals. *New Phytologist* **165**, 847-55. doi: 10.1111/j.1469-8137.2004.01254.x
- 820. Lobato, A.K.S., Gonçalves-Vidigal, M.C., Vidigal Filho, P.S., Andrade, C.A.B., Kvitschal, M.V., and Bonato, C.M. (2010) Relationships between leaf pigments and photosynthesis in common bean plants infected by anthracnose. New Zealand Journal of Crop and Horticultural Science 38, 29-37. doi: 10.1080/01140671003619308
- 821. Barón, M., Flexas, J., and Delucia, E. (2012) 22 Photosynthetic responses to biotic stress [Online]. Available: <u>https://www.life.illinois.edu/delucia/2014%20Publications/chapter%2022</u> %20web%20copy.pdf [Accessed 16/09/2019].
- 822. Vadez, V., Berger, J.D., Warkentin, T., Asseng, S., Ratnakumar, P., Rao, K.P.C., Gaur, P.M., Munier-Jolain, N., Larmure, A., Voisin, A.-S., Sharma, H.C., Pande, S., Sharma, M., Krishnamurthy, L., and Zaman, M.A. (2012) Adaptation of grain legumes to climate change: a review. Agronomy for Sustainable Development 32, 31-44. doi: 10.1007/s13593-011-0020-6
- 823. **Pierce, F.J., and Nowak, P.** (1999) "Aspects of Precision Agriculture," in *Advances in Agronomy*, ed. **D.L. Sparks**. Academic Press), 1-85.
- 824. **Niehaus, K., Lagares, A., and Pühler, A.** (1998) A *Sinorhizobium meliloti* lipopolysaccharide mutant induces effective nodules on the host plant *Medicago sativa* (alfalfa) but fails to establish a symbiosis with

Medicago truncatula. *Molecular Plant-Microbe Interactions* **11**, 906-14. doi: 10.1094/MPMI.1998.11.9.906

- 825. **Da, H.N., and Deng, S.P.** (2003) Survival and persistence of genetically modified *Sinorhizobium meliloti* in soil. *Applied Soil Ecology* **22**, 1-14. doi: 10.1016/S0929-1393(02)00127-0
- 826. Kuppusamy, K.T., Endre, G., Prabhu, R., Penmetsa, R.V., Veereshlingam, H., Cook, D.R., Dickstein, R., and Vandenbosch, K.A. (2004) LIN, a *Medicago truncatula* gene required for nodule differentiation and persistence of rhizobial infections. *Plant Physiology* 136, 3682-91. doi: 10.1104/pp.104.045575
- 827. **Mitra, R.M., and Long, S.R.** (2004) Plant and bacterial symbiotic mutants define three transcriptionally distinct stages in the development of the *Medicago truncatula/Sinorhizobium meliloti* symbiosis. *Plant Physiology* **134,** 595. doi: 10.1104/pp.103.031518
- 828. **Huhnke, R.L., Muck, R.E., and Payton, M.E.** (1997) Round Bale Silage Storage Losses of Ryegrass and Legume-Grass Forages. *Applied Engineering in Agriculture* **13**, 451. doi: 10.13031/2013.21625
- 829. **AHDB** (2018) *Factsheet 17: Grass silage analysis* [Online]. Available: <u>https://dairy.ahdb.org.uk/non_umbraco/download.aspx?media=12744</u> [Accessed 27/07/2018].
- 830. **Nutreco** (2018) *First cut grass silage 2018* [Online]. Available: <u>https://gb.trouwnutrition.co.uk/en/News/first-cut-grasss-silage-2018/1578686</u> [Accessed 27/07/2018].
- 831. IGCDC (1973) "Proceedings of the first international green crop drying congress", in: *First international green crop drying congress.* (eds.) C.L. Skidmore, Lord Netherthorpe, T. J., H.H. Rogers & R.J. Wilkins. (University of Oxford, UK.: E. & E. Plumridge, Linton, Cambridge).
- 832. **BAGCD** (1975) "Grass: The Journal of the British Association of Green Crop Driers". (UK).
- 833. **Vogt, G.** (2007) "The Origins of Organic Farming," in *Organic Farming: An International History,* ed. **W. Lockeretz**. CABI).
- 834. **BAGCD** (1992) "Memoranda and New Articles of Association: As adopted by Special Resolution passed on the 28th April 1992 Of The British Association of Green Crop Driers Limited.", (ed.) C. House. (London, UK: Her Majesty's Government).
- 835. **IGCDC** (1978) "Dri-crops 78: Proceedings of the second international green crop drying congress", in: *Second international green crop drying congress.* (eds.) H.R. Clark, H. R.E., A.E. Harvey, D.A. Christensen, E.M. Coxworth, G.M.J. Horton, J.A. Kernan & D.A. Pulkinen. (University of Saskatchewan, Canada: Extension Division, University of Saskatchewan Printing Services).
- 836. **European Commission** (2018) *The common agricultural policy at a glance* [Online]. Brussels, Belgium: European Commission,. Available: <u>https://ec.europa.eu/info/food-farming-fisheries/key-policies/common-agricultural-policy/cap-glance#legal-aspects</u> [Accessed 30/07/2018].
- 837. **Cunha, A., and Swinbank, A.** (2011) "The 1992 MacSharry Reform," in *An Inside View of the CAP Reform Process: Explaining the MacSharry, Agenda 2000, and Fischler Reforms*. (Oxford, UK: Oxford University Press).
- 838. **European Commission** (2003) Annual Report 2003 from the Commission to the Council and the European Parliament on the EC development policy and the implementation of external assistance in 2002 [Online]. Brussels, Belgium. Available: <u>https://eurlex.europa.eu/legal-content/en/TXT/?uri=CELEX:52003DC0527</u> [Accessed 07/08/2018].
- 839. **Crown**© (2005) "Secretary of State's Guidance for Vegetable Matter Drying Processes". (London, UK).

- 840. **European Commission** (2013) *Overview of CAP Reform 2014-2020* [Online]. Brussels, Belgium. Available: <u>https://ec.europa.eu/agriculture/sites/agriculture/files/policy-perspectives/policy-briefs/05_en.pdf</u> [Accessed 07/08/2018].
- 841. **Helminger, W., Martins, C., Henderyckx, E., and Scuvée, B.** (2016) *Agriculture, forestry and fishery statistics* [Online]. Luxembourg. Available: doi: 10.2785/917017 [Accessed 16/04/2018].
- 842. **Ramankutty, N., Evan, A.T., Monfreda, C., and Foley, J.A.** (2008) Farming the planet: 1. Geographic distribution of global agricultural lands in the year 2000. *Global Biogeochemical Cycles* **22**. doi: 10.1029/2007GB002952
- 843. **FAO, and FRA** (2015) *The Global Forest Resources Assessment 2015:* Desk reference [Online]. Rome, Italy. Available: <u>http://www.fao.org/3/a-i4808e.pdf</u> [Accessed 01/08/2018].
- 844. Kehoe, L., Romero-Muñoz, A., Polaina, E., Estes, L., Kreft, H., and Kuemmerle, T. (2017) Biodiversity at risk under future cropland expansion and intensification. *Nature Ecology & Evolution* 1, 1129-35. doi: 10.1038/s41559-017-0234-3
- 845. Foley, J.A., Ramankutty, N., Brauman, K.A., Cassidy, E.S., Gerber, J.S., Johnston, M., Mueller, N.D., O'Connell, C., Ray, D.K., West, P.C., Balzer, C., Bennett, E.M., Carpenter, S.R., Hill, J., Monfreda, C., Polasky, S., Rockström, J., Sheehan, J., Siebert, S., Tilman, D., and Zaks, D.P.M. (2011) Solutions for a cultivated planet. *Nature* 478, 337. doi: 10.1038/nature10452
- 846. Foley, J.A., Asner, G.P., Costa, M.H., Coe, M.T., DeFries, R., Gibbs, H.K., Howard, E.A., Olson, S., Patz, J., Ramankutty, N., and Snyder, P. (2007) Amazonia revealed: forest degradation and loss of ecosystem goods and services in the Amazon Basin. *Frontiers in Ecology* and the Environment 5, 25-32. doi: 10.1890/1540-9295(2007)5[25:ARFDAL]2.0.CO;2
- 847. Gibbs, H.K., Ruesch, A.S., Achard, F., Clayton, M.K., Holmgren, P., Ramankutty, N., and Foley, J.A. (2010) Tropical forests were the primary sources of new agricultural land in the 1980s and 1990s. *PNAS* **107**, 16732-7. doi: 10.1073/pnas.0910275107
- 848. Song, X.-P., Hansen, M.C., Stehman, S.V., Potapov, P.V., Tyukavina, A., Vermote, E.F., and Townshend, J.R. (2018) Global land change from 1982 to 2016. *Nature*. doi: 10.1038/s41586-018-0411-9
- 849. **Kaplan, J.O., Krumhardt, K.M., and Zimmermann, N.** (2009) The prehistoric and preindustrial deforestation of Europe. *Quaternary Science Reviews* **28**, 3016-34. doi: 10.1016/j.quascirev.2009.09.028
- 850. **European Commission** (2017) *REPORT FROM THE COMMISSION TO THE EUROPEAN PARLIAMENT AND THE COUNCIL on the implementation of the ecological focus area obligation under the green direct payment scheme* [Online]. Available: <u>https://eur-lex.europa.eu/legal-</u> <u>content/EN/ALL/?uri=COM:2017:152:FIN</u> [Accessed 10/08/2018].
- 851. **Crown**© (2018) *Basic Payment Scheme: rules for 2018* [Online]. Surrey, UK. Available: <u>https://www.gov.uk/guidance/bps-2018</u> [Accessed 10/08/2018].
- 852. **UN** (2004) *World population to 2300* [Online]. New York, USA. Available: <u>http://www.un.org/esa/population/publications/longrange2/WorldPop230</u> <u>Ofinal.pdf</u> [Accessed 01/08/2018].
- 853. **UN** (2017) *World Populations Prospects: The 2017 Revision* [Online]. Available: <u>https://esa.un.org/unpd/wpp/Download/Standard/Population/</u> [Accessed 01/08/2018].
- 854. Woods, J., Williams, A., Hughes, J.K., Black, M., and Murphy, R. (2010) Energy and the food system. *Philosophical Transactions of the*

Royal Society B: Biological Sciences **365**, 2991-3006. doi: 10.1098/rstb.2010.0172

- 855. **DEFRA** (2018) *Trends in UK sulphur dioxide, nitrogen oxides, nonmethane volatile organic compounds, ammonia and particulate matter (PM10, PM2.5) emissions* [Online]. London, UK. Available: <u>https://www.gov.uk/government/statistical-data-sets/env01-emissions-</u> <u>of-air-pollutants</u> [Accessed 07/08/2018].
- 856. **Rosas, F., Babcock, B.A., and Hayes, D.J.** (2015) Nitrous oxide emission reductions from cutting excessive nitrogen fertilizer applications. *Climatic Change* **132**, 353-67. doi: 10.1007/s10584-015-1426-y
- 857. **Crown**© (2007) "The Air Quality Strategy for England, Scotland, Wales and Northern Ireland", (eds.) DEFRA, Scottish Executive, Welsh Assembly Government & Department of the Environment Northern Ireland. (London, UK).
- 858. **UN** (2008) *Ambient Air Quality Directive (2008/50/EC)* [Online]. Available: <u>https://eur-lex.europa.eu/LexUriServ/LexUriServ.do?uri=OJ:L:2008:152:0001:0044:</u> <u>EN:PDF</u> [Accessed 01/08/2018].
- 859. **Crown**© (2010) "Air Quality Standards Regulations 2010". (London, UK).
- UNFCCC (1997) Kyoto Protocol to the United Nations Framework Convention on Climate Change adopted at COP3 [Online]. Kyoto, Japan. Available: <u>https://unfccc.int/resource/docs/cop3/07a01.pdf</u> [Accessed 12/08/18].
- 861. Laville, P., Lehuger, S., Loubet, B., Chaumartin, F., and Cellier, P. (2011) Effect of management, climate and soil conditions on N₂O and NO emissions from an arable crop rotation using high temporal resolution measurements. *Agricultural and Forest Meteorology* **151**, 228-40. doi: 10.1016/j.agrformet.2010.10.008
- 862. **Gómez-Baggethun, E., de Groot, R., Lomas, P.L., and Montes, C.** (2010) The history of ecosystem services in economic theory and practice: From early notions to markets and payment schemes. *Ecological Economics* **69**, 1209-18. doi: 10.1016/j.ecolecon.2009.11.007
- 863. **Ehrlich, P., and Ehrlich, A.** (1981) *Extinction : The Causes and Consequences of the Disappearance of Species.* New York: Random House.
- 864. **Braat, L.C., and de Groot, R.** (2012) The ecosystem services agenda:bridging the worlds of natural science and economics, conservation and development, and public and private policy. *Ecosystem Services* **1**, 4-15. doi: 10.1016/j.ecoser.2012.07.011
- 865. Crown© (2016) Natural Capital Committee: Terms of Reference [Online]. London, UK. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/517123/ncc-terms-of-reference.pdf</u> [Accessed 09/08/2018].
- 866. **Natural Capital Coalition** (2018) *Coalition organizations* [Online]. Available: <u>https://naturalcapitalcoalition.org/who/coalition-organizations/?mfilter=policy</u> [Accessed 09/08/2018].
- 867. Bünemann, E.K., Bongiorno, G., Bai, Z., Creamer, R.E., De Deyn, G., de Goede, R., Fleskens, L., Geissen, V., Kuyper, T.W., Mäder, P., Pulleman, M., Sukkel, W., van Groenigen, J.W., and Brussaard, L. (2018) Soil quality A critical review. *Soil Biology and Biochemistry* 120, 105-25. doi: 10.1016/j.soilbio.2018.01.030
- 868. **Kibblewhite, M.G., Ritz, K., and Swift, M.J.** (2008) Soil health in agricultural systems. *Philosophical Transactions of the Royal Society B: Biological Sciences* **363,** 685-701. doi: 10.1098/rstb.2007.2178

- 869. **Doran, J.W., and Zeiss, M.R.** (2000) Soil health and sustainability: managing the biotic component of soil quality. *Applied Soil Ecology* **15,** 3-11. doi: 10.1016/S0929-1393(00)00067-6
- 870. Welsh Assembly Government (2006) *Environment strategy for Wales* [Online]. Cardiff, UK. [Accessed 07/08/2018].
- 871. **European Commission** (2012) *The implementation of the soil thematic strategy and ongoing activities* [Online]. Brussels, Belgium. Available: <u>https://esdac.jrc.ec.europa.eu/Library/JRC_SOIL/Policy/DGENV/COM(201</u> <u>2)46_EN.pdf</u> [Accessed 07/08/2018].
- 872. Ola, A., Dodd, I.C., and Quinton, J.N. (2015) Can we manipulate root system architecture to control soil erosion? SOIL 1, 603-12. doi: 10.5194/soil-1-603-2015, 2015.
- 873. **Johnston, A.E., Poulton, P.R., and Coleman, K.** (2009) "Chapter 1 Soil Organic Matter: Its Importance in Sustainable Agriculture and Carbon Dioxide Fluxes," in *Advances in Agronomy*, ed. **D.L. Sparks**. Academic Press), 1-57.
- Kell, D.B. (2011) Breeding crop plants with deep roots: their role in sustainable carbon, nutrient and water sequestration. *Annals of Botany* 108, 407-18. doi: 10.1093/aob/mcr175
- 875. **Henryson, K., Sundberg, C., Kätterer, T., and Hansson, P.-A.** (2018) Accounting for long-term soil fertility effects when assessing the climate impact of crop cultivation. *Agricultural Systems* **164**, 185-92. doi: 10.1016/j.agsy.2018.03.001
- 876. **Ghimire, R., Norton, J.B., and Norton, U.** (2018) Soil organic matter dynamics under irrigated perennial forage–annual crop rotations. *Grass and Forage Science* **0**. doi: 10.1111/gfs.12378
- 877. Burkmar, R., Bell, C., and Field Studies Council (2015) Drivers of Biodiversity Loss: A research synthesis for the Tomorrow's Biodiversity Project [Online]. Shrewsbury, UK. Available: <u>http://www.fscbiodiversity.uk/sites/default/files/TomBioDrivers.pdf</u> [Accessed 09/08/2018].
- 878. Isaac, N., Outhwaite, C., August, T., Roy, D., and Powney, G. (2016) State of Nature 2016 [Online]. Available: <u>https://www.rspb.org.uk/globalassets/downloads/documents/conservatio</u> <u>n-projects/state-of-nature/state-of-nature-uk-report-2016.pdf</u> [Accessed 09/08/2018].
- 879. Crawley, M.J., Johnston, A.E., Silvertown, J., Dodd, M., de Mazancourt, C., Heard, M.S., Henman, D.F., and Edwards, G.R. (2005) Determinants of Species Richness in the Park Grass Experiment. *The American Naturalist* 165, 179-92. doi: 10.1086/427270
- 880. Craven, D., Isbell, F., Manning, P., Connolly, J., Bruelheide, H., Ebeling, A., Roscher, C., van Ruijven, J., Weigelt, A., Wilsey, B., Beierkuhnlein, C., de Luca, E., Griffin, J.N., Hautier, Y., Hector, A., Jentsch, A., Kreyling, J., Lanta, V., Loreau, M., Meyer, S.T., Mori, A.S., Naeem, S., Palmborg, C., Polley, H.W., Reich, P.B., Schmid, B., Siebenkäs, A., Seabloom, E., Thakur, M.P., Tilman, D., Vogel, A., and Eisenhauer, N. (2016) Plant diversity effects on grassland productivity are robust to both nutrient enrichment and drought. *Philosophical Transactions of the Royal Society B: Biological Sciences* 371, 20150277. doi: 10.1098/rstb.2015.0277
- 881. **Wall, D.H., Nielsen, U.N., and Six, J.** (2015) Soil biodiversity and human health. *Nature* **528**, 69. doi: 10.1038/nature15744
- 882. Luck, G.W., Harrington, R., Harrison, P.A., Kremen, C., Berry, P.M., Bugter, R., Dawson, T.P., de Bello, F., Díaz, S., Feld, C.K., Haslett, J.R., Hering, D., Kontogianni, A., Lavorel, S., Rounsevell, M., Samways, M.J., Sandin, L., Settele, J., Sykes, M.T., van den Hove, S., Vandewalle, M., and Zobel, M. (2009) Quantifying the Contribution

of Organisms to the Provision of Ecosystem Services. *BioScience* **59**, 223-35. doi: 10.1525/bio.2009.59.3.7

- 883. **Maddison, D., and Day, B.** (2015) *Improving Cost Benefit Analysis Guidance: A Report to the Natural Capital Committee* [Online]. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/517027/ncc-research-improving-cost-benefit-guidance-final-report.pdf</u> [Accessed 10/08/2018].
- 884. **Crown**© (2018) *Natural Capital Committee: Annual Report 2018: Fifth report to the Economic Affairs Committee* [Online]. Available: <u>https://assets.publishing.service.gov.uk/government/uploads/system/uploads/attachment_data/file/677873/ncc-annual-report-2018.pdf</u> [Accessed 10/08/2018].
- 885. Mézière, D., Colbach, N., Dessaint, F., and Granger, S. (2015) Which cropping systems to reconcile weed-related biodiversity and crop production in arable crops? An approach with simulation-based indicators. *European Journal of Agronomy* 68, 22-37. doi: 10.1016/j.eja.2015.04.004
- 886. Storkey, J., Doring, T., Baddeley, J.A., Marshall, A., Roderick, S., and Jones, H. (2011) Modelling the ability of legumes to suppress weeds. *Aspects of Applied Biology* **109**, 53-8.
- 887. Storkey, J., Döring, T., Baddeley, J., Collins, R., Roderick, S., Jones, H., and Watson, C. (2015) Engineering a plant community to deliver multiple ecosystem services. *Ecological Applications* 25, 1034-43. doi: 10.1890/14-1605.1
- 888. Storkey, J., Macdonald, A.J., Poulton, P.R., Scott, T., Köhler, I.H., Schnyder, H., Goulding, K.W.T., and Crawley, M.J. (2015) Grassland biodiversity bounces back from long-term nitrogen addition. *Nature* **528**, 401. doi: 10.1038/nature16444
- 889. JNCC, and DEFRA (2018) UK Biodiversity Indicators 2018 [Online]. York, UK. Available: <u>http://jncc.defra.gov.uk/page-4251</u> [Accessed 09/08/2018].
- 890. **European Commission** (2018) *Regulation (EU) No 1308/2013 of the European Parliament and of the Council of 17 December 2013 establishing a common organisation of the markets in agricultural products and repealing Council Regulations (EEC) No 922/72, (EEC) No 234/79, (EC) No 1037/2001 and (EC) No 1234/2007* [Online]. Available: <u>http://data.europa.eu/eli/reg/2013/1308/oj</u> [Accessed 08/08/2018].
- 891. **European Commission** (2018) *Agricultural markets* [Online]. European Commission, Available: <u>https://ec.europa.eu/agriculture/markets_en</u> [Accessed 08/08/2018].
- 892. **European Commission** (2018) *Proposal for a DIRECTIVE OF THE EUROPEAN PARLIAMENT AND OF THE COUNCIL on unfair trading practices in business-to-business relationships in the food supply chain* [Online]. Available: <u>https://eur-lex.europa.eu/legal-</u> <u>content/EN/ALL/?uri=CELEX:52018PC0173</u> [Accessed 08/08/2018].
- FAO (2018) GLEAM 2.0 Assessment of greenhouse gas emissions and mitigation potential [Online]. Rome, Italy. Available: http://www.fao.org/gleam/results/en/ [Accessed 01/08/2018].
- 894. IPCC (2014) Climate Change 2014: Synthesis Report. Contribution of Working Groups I, II and III to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change [Online]. Geneva, Switzerland. Available: <u>https://www.ipcc.ch/pdf/assessment-</u> report/ar5/syr/SYR AR5 FINAL full wcover.pdf [Accessed 08/08/2018].

This page is intentionally left blank